E COM VENTURES INC Form DEFM14A July 25, 2008 Table of Contents

No fee required.

Fee computed on table below per Exchange Act Rules 14a-6(i)(1) and 0-11.

UNITED STATES

SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

SCHEDULE 14A

Proxy Statement Pursuant to Section 14(a) of the

Securities Exchange Act of 1934

Filed by the Registrant x Filed by a Party other than the Registrant "	
Check the appropriate box: Preliminary Proxy Statement Definitive Proxy Statement Definitive Additional Materials Soliciting Material Pursuant to \$240.14a-12	Confidential, for Use of the Commission Only (as permitted by Rule 14a-6(e)(2))
	E Com Ventures, Inc.
	(Name of Registrant as Specified In Its Charter)
(Name of Per	rson(s) Filing Proxy Statement, if other than the Registrant)
Payment of Filing Fee (Check the appropriate box)):

	Aggregate number of securities to which transaction applies: 7,400,000
	Per unit price or other underlying value of transaction computed pursuant to Exchange Act Rule 0-11 (set forth the amount on which the filing fee is calculated and state how it was determined):
1	Transaction Value of \$155,770,000 calculated by multiplying \$21.05, the average of the high and low sales price of the registrant common stock on March 10, 2008, as reported on the NASDAQ Stock Market, by 7,400,000, the number of shares issuable in the transaction described in this proxy statement. In accordance with Section 14(g) of the Securities Exchange Act of 1934, as amende the filing fee was determined by multiplying 0.0000393 by the amount determined in the preceding sentence.
]	Proposed maximum aggregate value of transaction: \$155,770,000
,	Total fee paid: \$6,121.76
pa	aid previously with preliminary materials.
	box if any part of the fee is offset as provided by Exchange Act Rule 0-11(a)(2) and identify the filing for which the offsetting fee aid previously. Identify the previous filing by registration statement number, or the Form or Schedule and the date of its filing. Amount Previously Paid:

(3)	Filing Party:
(4)	Date Filed:

Table of Contents

E Com Ventures, Inc.

251 International Parkway

Sunrise, Florida 33325

July 25, 2008

To our Shareholders:

You are invited to attend a special meeting of E Com Ventures shareholders to be held at 8:30 a.m. on August 8, 2008, at the offices of Edwards Angell Palmer & Dodge LLP, 750 Lexington Avenue, New York, NY 10022. We have signed an agreement to merge our business with Model Reorg, Inc., a diversified U.S. wholesale and retail fragrance company that had unaffiliated net revenues of \$302.5 million for its fiscal year ended October 31, 2007 and \$156.9 million for the first six months of its fiscal 2008. The combination will create a large, independent, national, vertically integrated wholesale distributor and specialty retailer of perfumes and fragrances that we expect will be well-positioned to compete in the marketplace and drive growth, as well as to benefit from increased operating scale. The principal purpose of the special meeting is to take actions necessary to complete the merger, including approving the issuance of shares of our common stock and warrants in the merger and approving an amendment to our Amended and Restated Articles of Incorporation to increase the authorized number of shares of our common stock.

In the merger, we will issue 5,900,000 shares of our common stock and warrants (not exercisable for the first three years) to purchase 1,500,000 shares at an exercise price of \$23.94 per share in exchange for all the capital stock of Model Reorg. These would represent approximately 66% of our outstanding shares upon the closing of the merger, or approximately 71% assuming exercise of all warrants. Model Reorg s majority owners are Stephen L. Nussdorf and Glenn H. Nussdorf and their sister, Arlene Nussdorf. Stephen and Glenn Nussdorf currently beneficially own approximately 44.5% of our common stock and will beneficially own approximately 53.7% of our common stock after the merger, exclusive of shares issuable upon exercise of the warrants. Including shares received by Arlene Nussdorf in the merger, these Nussdorf family members would beneficially own approximately 72.3% of our capital stock in aggregate, or 75.7% assuming exercise of their warrants. Our other existing shareholders will continue to own the same number of shares as before the merger.

Our Board of Directors formed a special committee of independent directors to review the proposed merger. The special committee retained its own legal counsel, obtained valuation advice and a fairness opinion from an independent investment bank, and negotiated the terms of the merger with Model Reorg. On the special committee s recommendation, our Board determined that the terms of the merger are fair to our shareholders (other than those who are Model Reorg shareholders). The Board believes that combining Model Reorg with E Com Ventures will create valuable synergies and scale that will benefit all shareholders and recommends that you approve the proposals described in the accompanying Proxy Statement.

Our Board also approved changing our corporate name to Perfumania Holdings, Inc. to capitalize on the market recognition of our most important tradename, and we are also asking for your approval of this change at the special meeting.

The attached Proxy Statement describes the proposed merger and the actions to be taken at the special meeting. Please read the entire Proxy Statement carefully, including the Risk Factors section, before voting.

We are very excited about the opportunities this merger brings to our shareholders, and we thank you for your consideration and continued support.

Sincerely,

Michael W. Katz
President and Chief Executive Officer
YOUR VOTE IS IMPORTANT PLEASE SIGN, DATE AND RETURN THE ENCLOSED

Edgar Filing: E COM VENTURES INC - Form DEFM14A PROXY CARD OR VOTING INSTRUCTION FORM

E Com Ventures, Inc.

251 International Parkway

Sunrise, Florida 33325

NOTICE OF SPECIAL MEETING OF SHAREHOLDERS

To our Shareholders:

A Special Meeting of shareholders of E Com Ventures, Inc. will be held at 8:30 a.m. on Friday, August 8, 2008 at the offices of Edwards Angell Palmer & Dodge LLP, 750 Lexington Avenue, New York, NY 10022 for the purpose of considering and acting upon the following:

- Approving the issuance of shares of common stock and warrants under the Agreement and Plan of Merger dated as of December 21, 2007 by and among E Com Ventures, Inc., Model Reorg, Inc., the shareholders of Model Reorg, and Model Reorg Acquisition LLC, as amended, and the issuance of our common stock upon exercise of such warrants;
- 2. Approving an amendment to E Com Ventures Articles of Incorporation to increase the number of shares of common stock we are authorized to issue from 6.250,000 shares to 20,000,000 shares:
- 3. Approving an amendment to E Com Ventures Articles of Incorporation to change our corporate name to Perfumania Holdings, Inc.;
- 4. If necessary or appropriate, approving a proposal to postpone or adjourn the Special Meeting to enable us to solicit additional proxies in favor of the foregoing proposals; and
- 5. Such other business as may properly come before the Special Meeting.

Any action may be taken on the foregoing proposals at the Special Meeting on the date specified above or on any date or dates to which the Special Meeting may be postponed or adjourned.

Shareholders of record at the close of business on June 24, 2008 are entitled to notice of, and to vote at, the Special Meeting or at any postponements or adjournments of the Special Meeting.

By Order of the Board of Directors,

Donovan Chin Chief Financial Officer and Secretary

Sunrise, Florida

July 25, 2008

YOUR VOTE IS IMPORTANT

IT IS IMPORTANT THAT PROXIES BE RETURNED PROMPTLY. THEREFORE, WHETHER OR NOT YOU EXPECT TO ATTEND THE SPECIAL MEETING IN PERSON, PLEASE SIGN, DATE AND RETURN THE ENCLOSED PROXY CARD OR VOTING INSTRUCTION FORM AS SOON AS POSSIBLE IN THE ENCLOSED POSTAGE PRE-PAID ENVELOPE.

TABLE OF CONTENTS

<u>QUESTIONS AND ANSWERS ABOUT THE MERGER AND THE SPECIAL SHAREHOLDERS MEETING</u>	1
SUMMARY	6
The Companies	6
Recommendation of our Board of Directors and Special Committee; Reasons for the Merger	6
Opinion of Financo	7
Overview of the Merger and the Merger Agreement	7
Increase in Authorized Shares	7
Name Change	8
Date, Time and Place of the Special Meeting	8
Record Date and Voting Rights for the Special Meeting	8
Required Votes	8
Adjournment	8
No Appraisal or Preemptive Rights	9
Management Following the Merger	9
Interests of Certain Directors and Officers of E Com	9
Risk Factors	9
U.S. Federal Income Tax Consequences to E Com and its Shareholders	9
Anticipated Accounting Treatment	10
Regulatory Approvals	10
NASDAQ Capital Market Listing	10
Projections	10
	10
SELECTED HISTORICAL AND PRO FORMA COMBINED FINANCIAL DATA	12
Selected Historical Financial Data of E Com	12
Selected Historical Financial Data of Model Reorg	14
Selected Unaudited Pro Forma Condensed Combined Financial Data	16
Comparative Historical and Unaudited Pro Forma Combined Per Share Data	17
RISK FACTORS	19
FORWARD-LOOKING STATEMENTS	24
THE SPECIAL MEETING OF SHAREHOLDERS	24
Date, Time and Place	24
Purposes of the Special Meeting	24
Record Date; Shares Entitled to Vote; Quorum Requirement	25
Required Votes	25
Voting by Our Directors, Officers, and Affiliates	25
<u>Miscellaneous</u>	25
PROPOSAL 1 APPROVAL TO ISSUE SHARES AND WARRANTS	27
<u>General</u>	27
Background of the Merger	27
Reasons for the Merger	36
Opinion of Financo	37
Model Reorg Board and Shareholder Approval	47
Interests of Certain Persons in the Merger	47
Terms of the Merger Agreement	48
The Warrants	56
Registration Rights	56
Post-Merger Financing	57

Table of Contents	
Affiliate Debt	58
No Appraisal or Preemptive Rights	58
Management Following the Merger	58
Material U.S. Federal Income Tax Consequences of the Merger to E Com and Its Shareholders	59
Anticipated Accounting Treatment	59
NASDAQ Capital Market Listing	59
Regulatory Matters	60
Status of Shares Issued in Merger	60
Recommendations of the Special Committee and our Board of Directors	60
PROPOSAL 2 AUTHORIZATION OF ADDITIONAL SHARES	6
Description of the Amendment	6
Reasons for and Effect of the Amendment Description of any Common Steels	6
Description of our Common Stock Recommendation of our Board of Directors	62
PROPOSAL 3 NAME CHANGE Description of the Amendment	62
Reasons for the Change	62
Recommendation of our Board of Directors	62
PROPOSAL 4 ADJOURNMENT OF SPECIAL MEETING	62
Recommendation of our Board of Directors	62
CERTAIN RELATIONSHIPS AND RELATED PERSON TRANSACTIONS	6.
INFORMATION ABOUT E COM VENTURES	6′
Business of E Com No. 10 Pin in 1A 1 in FF: 11 Country I Pon 1 in FF:	6′
Management s Discussion and Analysis of Financial Condition and Results of Operations	7
INFORMATION ABOUT MODEL REORG	84
Business of Model Reorg	84
Management s Discussion and Analysis of Financial Condition and Results of Operations	88
UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL STATEMENTS	10
SELECTED FINANCIAL PROJECTIONS	103
MARKET PRICE AND DIVIDEND INFORMATION; RELATED SHAREHOLDER MATTERS	112
SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT	113
WHERE YOU CAN FIND MORE INFORMATION	110
SHAREHOLDER PROPOSALS FOR THE 2008 ANNUAL MEETING	110
INDEX TO FINANCIAL STATEMENTS	F-
FINANCIAL STATEMENTS	
Annex I Agreement and Plan of Merger; First Amendment	
Annex II Financo Opinion	
Annex III Certificate of Amendment to Articles of Incorporation to Increase Number of Authorized Shares	

(ii)

Annex IV Certificate of Amendment to Articles of Incorporation to Effect Name Change

E COM VENTURES, INC.

SPECIAL MEETING OF SHAREHOLDERS

PROXY STATEMENT

This Proxy Statement contains information about the Special Meeting of Shareholders of E Com Ventures, Inc. (E Com) to be held on Friday, August 8, 2008, beginning at 8:30 a.m., at the offices of Edwards Angell Palmer & Dodge LLP, 750 Lexington Avenue, New York, NY 10022. The approximate date that this Proxy Statement, the accompanying Notice of Special Meeting and the enclosed form of proxy are first being sent to shareholders is July 25, 2008.

QUESTIONS AND ANSWERS ABOUT THE MERGER

AND THE SPECIAL SHAREHOLDERS MEETING

Overview of the Merger

We entered into an Agreement and Plan of Merger on December 21, 2007 (the Original Merger Agreement) with Model Reorg, Inc. (Model Reorg), the shareholders of Model Reorg, and our subsidiary, Model Reorg Acquisition LLC (Merger Sub). On July 8, 2008, the parties executed a First Amendment to the Original Merger Agreement (the Amendment, which we refer to together as the Merger Agreement, appear in Annex I to this Proxy Statement. The Merger Agreement provides that, subject to the conditions set forth therein, at the closing:

Model Reorg will merge with and into Merger Sub (the Merger) and the merged entity will continue as our wholly-owned subsidiary; and

we will issue to the Model Reorg shareholders 5,900,000 shares of our common stock and warrants exercisable for the purchase of 1,500,000 shares of our common stock at an exercise price of \$23.94 per share (the Warrants), which we refer to collectively as the Merger Consideration.

The shares issued to the Model Reorg shareholders will represent approximately 66% of our outstanding shares immediately after the closing and approximately 71% assuming exercise of all the Warrants. Further details of the Merger and the Merger Consideration are provided under Approval to Issue Shares and Warrants Terms of the Merger Agreement and The Warrants.

When we use the term combined company in this Proxy Statement, we are referring to E Com together with Model Reorg, which, following the Merger, will be owned and operated by E Com as a wholly-owned subsidiary. The combined company will be called Perfumania Holdings, Inc. if shareholders approve the name change proposal described below.

This section provides brief answers to questions you may have about the Merger and the Special Meeting. For more complete responses to these questions and for additional information, please read this Proxy Statement in its entirety.

Q: Why am I receiving this Proxy Statement?

A: In order for us to complete the Merger, our shareholders must vote to approve (i) the issuance of our common stock and Warrants to the owners of Model Reorg, including the issuance of common stock upon exercise of the Warrants, and (ii) an amendment to our Amended and Restated Articles of Incorporation, as amended to date (which we refer to as our Articles of Incorporation), to increase the number of authorized shares of our common stock so that we have enough shares to issue in connection with the Merger and for future business purposes.

Table of Contents

We are also seeking shareholder approval to change our corporate name to Perfumania Holdings, Inc.

We are sending this Proxy Statement and the enclosed proxy card to our shareholders to solicit their vote on these proposals at the Special Meeting. This Proxy Statement contains important information about the Merger and the proposals to be approved.

Q: What is the total value of the Merger Consideration?

A: Because the Merger Consideration consists of our common stock and warrants, its value fluctuates with changes in the trading price of our common stock on NASDAQ. There is no trading market for the Warrants; accordingly, we calculate their value using the Black-Scholes pricing model, which involves making certain assumptions about the anticipated volatility of our stock price, future interest rates, and other matters. Since these assumptions are inherently variable and uncertain, they also contribute to fluctuations in the value of the Merger Consideration. Based on the closing market prices of our common stock and the assumptions we used to value the Warrants as of the respective dates, we determined that the total value of the Merger Consideration as of December 20, 2007 (the trading day before the date of the Original Merger Agreement) would have been \$161,586,000, and that as of June 12, 2008, it would have been \$138,969,000. The value of the Merger Consideration actually received by the Model Reorg shareholders at the closing will not be determined until the closing. You should note that the Special Committee s review of the fairness of the proposed Merger Consideration focused primarily on the estimated relative values of the two companies, rather than on the absolute value of the Merger Consideration as of any particular time.

Q: How will the share ownership of E Com s largest shareholders change in the Merger?

A: Stephen L. Nussdorf and Glenn H. Nussdorf will remain our largest shareholders. They currently own in the aggregate approximately 36.4% of our outstanding common stock, or approximately 44.5% assuming conversion into shares of a \$5.0 million subordinated convertible promissory note we issued for a loan they made to us in 2004, which we refer to as the Convertible Note. Stephen and Glenn Nussdorf and their sister, Arlene Nussdorf, also own an aggregate of 88.8% of the outstanding shares of Model Reorg. After the Merger, Stephen and Glenn Nussdorf would beneficially own an aggregate of approximately 53.7% of our outstanding common stock, assuming conversion of the Convertible Note, and they would hold Warrants that would, upon exercise, result in their beneficially owning an aggregate of approximately 57.7% of our shares. Including shares to be received by Arlene Nussdorf in the Merger, these Nussdorf family members would beneficially own approximately 72.3% of our capital stock in aggregate, or 75.7% assuming exercise of their Warrants.

Q: What will happen to the other E Com shareholders?

A: You will continue to own the same number of shares of our common stock that you owned immediately before the Merger. Because we will issue new shares to the current shareholders of Model Reorg, the percentage interest of E Com that your shares represent will be reduced. However, we will be a bigger company as a result of acquiring Model Reorg. See Selected Historical and Pro Forma Combined Financial Data Comparative Historical and Unaudited Pro Forma Combined Per Share Data for more information on the effect of the Merger on our shareholders ownership.

Q: Should I send in my share certificates?

A: No. Your share certificates will not be exchanged in the Merger.

Q: Do I have appraisal rights?

A: No. Our shareholders do not have appraisal rights under Florida law in connection with the Merger.

2

Table of Contents

- Q: Has the E Com Board of Directors made a recommendation on how I should vote?
- A: Yes. Our Board of Directors appointed a committee of independent directors, who have no interest in Model Reorg, to review and negotiate the terms of the Merger. We refer to them in this Proxy Statement as the Special Committee. Based on the Special Committee s recommendation, our Board of Directors has recommended that you vote FOR the issuance of the Merger Consideration and the increase in our authorized shares. The reasons for our Board s recommendations are discussed in detail in Approval to Issue Shares and Warrants Reasons for the Merger and Authorization of Additional Shares Reasons for and Effect of the Amendment. The Board also recommends approval of the name change as discussed in Name Change Reasons for the Change below.
- Q: Have the Model Reorg shareholders approved the Merger?
- A: Yes. In addition, those Model Reorg shareholders who also own shares of our common stock have agreed to vote their E Com shares in favor of the Merger.
- Q: When do you expect the Merger to be completed?
- A: We are working towards completing the Merger as quickly as possible. We hope to complete it by early in the third quarter of our 2008 fiscal year.
- Q: Where can I find more information about the companies?
- A: In addition to the descriptions under Information about E Com and Information about Model Reorg, you can find more information about E Com from the sources described under Where You Can Find More Information. Because Model Reorg is a private company that does not file reports with the Securities and Exchange Commission (SEC), there is limited information publicly available about Model Reorg other than what has been provided in this Proxy Statement.
- Q: Are there risks I should consider in deciding whether to vote to approve the issuance of the Merger Consideration pursuant to the Merger Agreement?
- A: Yes. In evaluating the issuance of the Merger Consideration under the Merger Agreement, you should carefully consider the information discussed in Risk Factors and Forward-Looking Statements and the other considerations described in this Proxy Statement.
- Q: Who will be the Board of Directors and management of the combined company after the Merger?
- A: We do not anticipate that our Board of Directors will change at the time of the Merger. Model Reorg officers will join our management team as the larger Model Reorg business is absorbed into ours. More information about our Board and our and Model Reorg s present management appears under Approval to Issue Shares and Warrants Management Following the Merger.
- Q: Why am I being asked to approve amendments to E Com s Articles of Incorporation?
- A: An amendment to our Articles of Incorporation increasing the number of shares we may issue is necessary for us to complete the Merger. We need our shareholders to approve Proposal 2 for the share increase so that we will have enough authorized shares to issue to Model Reorg s shareholders as required by the Merger Agreement, and for future business purposes. Therefore, the Merger depends on our shareholders approving both Proposals 1 and 2.

Proposal 2 is not conditioned on approval of any other proposal, since we are seeking approval of the increase whether or not the Merger issuances are approved.

Table of Contents

In order to change our name to Perfumania Holdings, Inc., we also need to amend our Articles of Incorporation as described in Proposal 3. This proposal also is not conditioned on approval of any other proposal.

Q: When and where is the Special Meeting and who is entitled to vote?

A: The Special Meeting will take place on Friday, August 8, 2008 at 8:30 a.m., at the offices of Edwards Angell Palmer & Dodge LLP, 750 Lexington Avenue, New York, NY 10022. Holders of record of our common stock as of the close of business on June 24, 2008 are entitled to vote at the Special Meeting or any postponement or adjournment of the Special Meeting.

Q What shareholder approvals are required to approve the proposals?

A: The affirmative vote of a majority of the votes cast is required to approve each proposal.

Q: What do I need to do now?

A: You should carefully read and consider the information contained in this Proxy Statement, including the Annexes. You should then complete, sign, and return the enclosed proxy card or voting instruction form in accordance with the accompanying instructions.

Q: If my E Com shares are held in street name by a broker or bank, will it vote my shares for me?

A: If you do not provide your broker or bank with instructions on how to vote your shares, it will not be permitted to vote them at the Special Meeting. You should be sure to return the enclosed voting instruction form with instructions on how to vote your shares.

Q: May I vote in person?

A: If your shares of common stock are registered directly in your name with our transfer agent, you are considered a shareholder of record and we have sent the Proxy Statement and proxy card directly to you. If you are a shareholder of record, you may attend the Special Meeting and vote your shares in person.

If your shares of common stock are held in the name of your broker or other financial institution, which is usually the case if you hold them in a brokerage or similar account, your shares are held in street name and your broker or other financial institution is the shareholder of record. It has sent the Proxy Statement to you with a voting instruction form. As the beneficial owner, you are also invited to attend the Special Meeting. However, only your broker or other financial institution is authorized to vote or grant a proxy for those shares. Accordingly, if you wish to vote those shares in person at the Special Meeting, you must contact your broker or other financial institution to obtain a proxy, which you must bring to the meeting.

Whether or not you intend to vote in person at the Special Meeting, please sign and return the enclosed proxy card or voting instruction form in the enclosed envelope in case you are unable to attend. This will not prevent you from voting in person if you do attend.

Q: Can I vote by telephone or electronically?

A. We have not established procedures to allow telephone or electronic voting by shareholders of record. Owners of shares held in street name may vote by telephone or the Internet if their bank or broker makes those methods available, in which case it will have enclosed the instructions with this Proxy Statement.

Q: What happens if I do not return a proxy card or otherwise provide voting instructions?

A: If you are in favor of the Merger, you should return your proxy card or voting instruction form. Your failure to return your proxy card or otherwise provide voting instructions with respect to any of the proposals will not have the effect of a vote for or against the proposal.

4

Table of Contents

Q: What do I do if I want to change my vote?

A: A shareholder giving a proxy has the power to revoke it at any time before the vote is taken at the Special Meeting by:

if you are a shareholder of record, submitting to our Corporate Secretary (at 251 International Parkway, Sunrise, Florida 33325) a written instrument revoking the proxy;

if you are a beneficial owner of shares held in street name, following the instructions you received;

submitting a duly executed proxy card or voting instruction form bearing a later date; or

voting in person at the Special Meeting.

- Q: Who should I call if I have questions about the Special Meetings or the Merger?
- A: You may call The Proxy Advisory Group at (212) 616-2180.
- Q: How can I find out the results of the voting at the Special Meeting?

A: Preliminary voting results will be announced at the Special Meeting. Final voting results will be published in our public filings with the SEC after the date of the Special Meeting.

5

SUMMARY

This summary highlights selected information from this Proxy Statement and may not contain all of the information that is important to you. For a more complete description of the terms of the Merger and the proposals to be approved at the Special Meeting, you should read this entire Proxy Statement carefully.

The Companies

E Com Ventures, Inc.

251 International Parkway

Sunrise, Florida 33325

(954) 335-9100

E Com is a retailer and wholesaler of fragrances and related products and other specialty items. It does business through two wholly-owned subsidiaries, Perfumania, Inc. (Perfumania) and perfumania.com, Inc. (perfumania.com) Perfumania is a specialty retailer and wholesale distributor of a wide range of brand name and designer fragrances. It operates a chain of retail stores selling fragrances at discounted prices, which are generally located in regional malls, manufacturers outlet malls, lifestyle centers, airports and suburban strip shopping centers. Perfumania s wholesale division distributes fragrances and related products, primarily to Model Reorg. Also, perfumania.com offers a selection of E Com s more popular products for sale over the Internet and serves as an alternative shopping experience for Perfumania retail customers.

E Com is a publicly-traded corporation. Its common stock is traded on the NASDAQ Capital Market under the symbol ECMV. Refer to Information About E Com and Security Ownership of Certain Beneficial Owners and Management for more information. You may also find more information about us and our subsidiaries at www.ecomv.com and www.perfumania.com. The information at these websites is not incorporated by reference into this Proxy Statement.

Model Reorg, Inc.

35 Sawgrass Drive, Suite 2

Bellport, New York 11713

(631) 866-4100

Model Reorg is a privately-owned, diversified fragrance company that manufactures and distributes nationally advertised designer fragrances through its subsidiaries. The wholesale subsidiary, which we refer to as Quality Fragrance Group, primarily distributes fragrances to mass merchandisers, drug store chains, specialty outlets, including Perfumania, and distributors. The retail subsidiary, which we refer to as Scents of Worth, sells products in retail stores on a consignment basis. The manufacturing subsidiary, which we refer to as Five Star, owns and licenses designer fragrance brands and contracts with third parties for the manufacture of owned and licensed brands. Refer to Information About Model Reorg and Security Ownership of Certain Beneficial Owners and Management for more information.

Recommendation of our Board of Directors and Special Committee; Reasons for the Merger

(see pp. 36 and 60)

The Special Committee of our Board of Directors unanimously approved the Merger, and our Board of Directors unanimously determined that the terms of the Merger are fair to our shareholders, other than those who also hold securities of Model Reorg. The Board has recommended that you vote FOR the share and Warrant issuance and the amendments to our Articles of Incorporation. The Board and the Special Committee believe that the Merger will improve our competitive position and ability to grow, as well as permit us to benefit from

6

Table of Contents

increased operating scale. The factors the Board and the Special Committee relied on to approve the Merger and related transactions and to recommend shareholder approval are described in more detail under Approval to Issue Shares and Warrants Reasons for the Merger and Recommendations of the Special Committee and our Board of Directors.

Opinion of Financo

(see p. 37)

In considering whether to recommend approval of the Merger, the Special Committee received a written opinion of Financo, Inc., which we refer to as Financo, that the Merger Consideration is fair, from a financial point of view, to our shareholders (other than those who own, or whose affiliates own, securities of Model Reorg), as of December 21, 2007, the date the Original Merger Agreement was signed. The full text of the Financo opinion is attached to this Proxy Statement as Annex II. You are encouraged to read the opinion carefully and in its entirety for a description of the assumptions made, matters considered and limitations on the review undertaken. Financo s opinion was delivered to the Special Committee and addresses only the fairness, from a financial point of view, of the Merger Consideration. The opinion does not constitute a recommendation to any shareholder as to how a shareholder should vote at the Special Meeting. See Approval to Issue Shares and Warrants Opinion of Financo.

Overview of the Merger and the Merger Agreement

(see pp. 48, 57 and 58)

At the closing under the Merger Agreement, Model Reorg will merge into Merger Sub, and the merged entity will continue Model Reorg s business as our wholly-owned subsidiary. At that time, all issued and outstanding shares of Model Reorg common stock will be exchanged for an aggregate of 5,900,000 shares of our common stock and Warrants to purchase 1,500,000 shares of our common stock at an exercise price of \$23.94 per share. The Warrants will not be exercisable until the third anniversary of the closing of the Merger (except upon a change of control of E Com), after which they will be exercisable until the tenth anniversary. The Merger Consideration will be allocated among the Model Reorg shareholders in accordance with their interests in Model Reorg. We will retain in escrow 295,000 shares from the Merger Consideration in order to secure our indemnification rights under the Merger Agreement. We currently expect the Merger to occur early in the third quarter of our 2008 fiscal year. At that time, we will enter into certain agreements and Model Reorg will restructure and refinance certain debt it currently owes to an affiliated company, as described below under Approval to Issue Shares and Warrants Terms of the Merger Agreement, Post-Merger Financing and Affiliate Debt.

Increase in Authorized Shares

(see p. 61)

Our Board of Directors has approved increasing our authorized number of shares of common stock to 20,000,000, subject to shareholder approval, in order to have enough shares to complete the Merger and for future business purposes. Our Articles of Incorporation currently authorize the issuance of up to 6,250,000 shares of common stock. On June 24, 2008, 3,059,041 shares of our common stock were outstanding, 147,696 shares were reserved for issuance upon exercise of stock options, 444,445 shares were reserved for issuance upon conversion of the Convertible Note and 661,523 shares were reserved for future awards under our stock option plan. We will issue 5,900,000 shares to the Model Reorg shareholders at the closing of the Merger, and will issue up to an additional 1,500,000 shares if all the Warrants are exercised. The proposed amendment to our Articles of Incorporation will authorize a sufficient number of shares for these purposes, as well as a reasonable additional

Table of Contents

number of shares for future issuances pursuant to benefit plans, for financings or acquisitions, or otherwise. This proposed amendment is not conditioned on approval of any other proposal. Except for issuances pursuant to our equity compensation plans for directors, employees and independent contractors, we have no current plans to issue any additional shares of common stock.

Name Change

(see p. 62)

Our Board of Directors has approved changing our corporate name from E Com Ventures, Inc. to Perfumania Holdings, Inc., subject to shareholder approval. Perfumania is the distinctive tradename by which we are known in the marketplace and will provide a strong corporate and commercial identification for the combined company. If this proposal is approved by the shareholders, we anticipate that our ticker symbol on the NASDAQ Capital Market will be changed to PERF.

Date, Time and Place of the Special Meeting

(see p. 24)

The Special Meeting will be held at the offices of Edwards Angell Palmer & Dodge LLP, 750 Lexington Avenue, New York, NY 10022, at 8:30 a.m. on Friday, August 8, 2008.

Record Date and Voting Rights for the Special Meeting

(see p. 25)

Only holders of record of our common stock at the close of business on June 24, 2008 (the Record Date) will be entitled to vote at the Special Meeting or any postponement or adjournment of the Special Meeting. On the Record Date, there were 3,059,041 shares of common stock outstanding and entitled to vote. Each share of our common stock is entitled to one vote on each matter to be voted upon at the Special Meeting. The presence at the Special Meeting, in person or by proxy, of holders of a majority of the issued and outstanding shares of our common stock is required for a quorum.

Required Votes

(see p. 25)

Approval of each proposal requires the affirmative vote of a majority of the total votes cast on the proposal. Accordingly, abstentions and broker non-votes will have no effect on the vote on any proposal. Glenn and Stephen Nussdorf, who held an aggregate of 36.4% of our outstanding common stock on the Record Date, have agreed to vote their shares in favor of all the proposals.

Adjournment

(see p. 62)

If necessary or appropriate, shareholders will be asked to approve a proposal to postpone or adjourn the Special Meeting to a later time in order for us to solicit additional proxies in favor of any of the proposals. In any event, the Special Meeting may be adjourned if a quorum is not present.

8

No Appraisal or Preemptive Rights

(see p. 58)

Our shareholders will not be entitled to demand appraisal of, or exercise dissenters—rights with respect to, their shares in connection with the matters to be approved at the Special Meeting. Our shareholders do not have preemptive rights to purchase any of the shares or Warrants to be issued in connection with the Merger.

Management Following the Merger

(see p. 58)

We do not anticipate that our Board of Directors will change at the time of the Merger. We expect that the Model Reorg management team will be integrated with our management team as the larger Model Reorg business is absorbed into ours.

Interests of Certain Directors and Officers of E Com

(see p. 47)

Some of our directors and officers have interests in the Merger that are in addition to those of our other shareholders and that may influence them to support the Merger without regard to your interests. See Approval to Issue Shares and Warrants Interests of Certain Persons in the Merger for more information.

Risk Factors

(see p. 19)

We and Model Reorg are subject to various risks associated with our respective businesses and industries. In addition, the Merger, including the possibility that it may not be completed, poses a number of risks to us and our shareholders, including the following:

The Merger may be completed despite material adverse changes resulting from economic or industry changes or other causes that could reduce the value of the Merger to our shareholders.

Even if the Merger is completed, the costs to do so may be higher than projected for any of a number of reasons, reducing the anticipated benefit to our shareholders.

If the combined company does not realize the anticipated benefits from the Merger, the market price of our common stock may decline, and our shareholders may not realize a benefit despite the ownership dilution they will experience in connection with the Merger.

These and other risks are discussed in greater detail under Risk Factors. In addition, risks related to our current business, which will continue to exist whether or not the Merger occurs, are described in our periodic reports filed with the SEC. You are encouraged to read and consider all of these risks carefully before you vote.

U.S. Federal Income Tax Consequences to E Com and its Shareholders

(see p. 59)

Because our shareholders will not be exchanging their shares in the Merger, the closing of the Merger and related transactions under the Merger Agreement will not have any U.S. federal income tax consequences to the holders of our common stock. As a tax-free reorganization under the Internal Revenue Code of 1986, as amended

9

Table of Contents

(the Internal Revenue Code), the Merger will not be taxable to us or Model Reorg for U.S. federal income tax purposes. The Merger will have the effect of limiting our ability to use our net operating losses from prior periods to reduce taxable income in the future.

Anticipated Accounting Treatment

(see p. 59)

We expect to treat the Merger as a reverse acquisition under the purchase method of accounting in accordance with accounting principles generally accepted in the United States of America (GAAP). For accounting purposes, Model Reorg is considered to be acquiring us in the Merger, so Model Reorg is financial statements, showing its acquisition of us, will be the combined company is financial statements from and after the Merger. However, the combined company will continue to use our fiscal year end, the Saturday closest to January 31, which is conventional for retailing businesses.

Regulatory Approvals

(see p. 60)

In order to complete the Merger we and Model Reorg were required to make certain filings with the Federal Trade Commission and wait until the applicable period for review had expired or been terminated. We and Model Reorg filed the required notification on January 24, 2008 and, on February 1, 2008, the Federal Trade Commission granted early termination of the waiting period.

NASDAQ Capital Market Listing

(see p. 59)

We have received a preliminary, oral notice that we will not be required to reapply for listing on the NASDAQ Capital Market under NASDAQ s reverse merger—rules. Accordingly, we expect that our common stock will continue to be listed on the NASDAQ Capital Market following the closing of the Merger. However, since it is a condition to closing the Merger that we continue to be listed, if the application is eventually required but is not accepted, the Merger will not occur unless we and Model Reorg waive that condition.

Projections

(see p. 108)

Neither we nor Model Reorg, as a matter of course, generally publish our respective business plans and strategies or make public projections as to future sales, earnings, or other results. However, we and Model Reorg prepared our respective prospective financial information set forth under Selected Financial Projections to present certain projections of financial performance that were provided to the Special Committee and Financo. This prospective financial information was not prepared with a view toward public disclosure or with a view toward complying with the guidelines established by the American Institute of Certified Public Accountants with respect to prospective financial information, but, in the view of our and Model Reorg s management, was prepared on a reasonable basis, reflects the best estimates and judgments available as of the date of their preparation, and presents, to the best of each management s knowledge and belief, the expected course of action and the expected future financial performance of the respective company. However, this information is not fact and should not be relied upon as being necessarily indicative of future results, and readers of this Proxy Statement are cautioned not to place undue reliance on the prospective financial information.

Table of Contents

Neither the Company s independent auditors, nor any other independent accountants, have compiled, examined, or performed any procedures with respect to the prospective financial information contained herein, nor have they expressed any opinion or any other form of assurance on such information or its achievability, and assume no responsibility for, and disclaim any association with, the prospective financial information.

The assumptions and estimates underlying the prospective financial information are inherently uncertain and, though considered reasonable by our and Model Reorg s management, respectively, as of the date of its preparation, are subject to a wide variety of significant business, economic, and competitive risks and uncertainties that could cause actual results to differ materially from those contained in the prospective financial information, including, among others, the risks and uncertainties described in Risk Factors and Forward-Looking Statements. Accordingly, there can be no assurance that the prospective results are indicative of our future performance or that of the combined company or that actual results will not differ materially from those presented in the prospective financial information. Inclusion of the prospective financial information in this Proxy Statement should not be regarded as a representation by any person that the results contained in the prospective financial information will be achieved.

Neither we nor Model Reorg intends to update or otherwise revise the prospective financial information to reflect circumstances existing or events occurring, including changes in general economic or industry conditions, since its preparation, even if any or all of the underlying assumptions are shown to be in error.

11

SELECTED HISTORICAL AND PRO FORMA COMBINED FINANCIAL DATA

The following tables present selected historical financial data, selected unaudited pro forma combined financial data, and comparative historical and unaudited pro forma per share data.

Selected Historical Financial Data of E Com

The selected historical financial data provided below for our fiscal years ended February 2, 2008, February 3, 2007 and January 28, 2006, and as of February 2, 2008 and February 3, 2007 have been taken or derived from our audited Consolidated Financial Statements included in this Proxy Statement beginning on page F-3. The selected historical financial data provided below for and as of the thirteen week periods ended May 3, 2008 and May 5, 2007 have been taken or derived from our unaudited condensed consolidated financial statements included in this Proxy Statement. Our financial statements for the reporting periods ending before 2008 have been restated as described in Note 3 to our Consolidated Financial Statements beginning on page F-14. These unaudited consolidated historical financial statements include all adjustments (consisting only of normal recurring adjustments) that in the opinion of our management are necessary for a fair presentation of our financial position and results of operations for these interim periods. Historical operating results of interim fiscal periods are not necessarily indicative of results that may be expected for the full year, particularly since our sales have been significantly higher in the fourth fiscal quarter than in the other quarters due primarily to increased purchases of fragrances as gift items during the holiday season. The selected financial data for the fiscal years ended January 29, 2005 and January 31, 2004, and as of January 28, 2006, January 29, 2005, and January 31, 2004 have been taken or derived from our audited consolidated financial statements for those years, which are not included in this Proxy Statement. This information is only a summary and you should read it together with our Management s Discussion and Analysis of Financial Condition and Results of Operations and historical financial statements and related notes contained elsewhere in this Proxy Statement. These historical results are not necessarily indicative of results to be expected in any future p

Our fiscal year results are based on a fifty-two or fifty-three week retail calendar ending on the Saturday closest to January 31. All references herein to fiscal years are to the calendar year in which the fiscal year begins; for example, fiscal year 2007 refers to the fiscal year that began on February 4, 2007 and ended on February 2, 2008. With the exception of fiscal year 2006, which contained fifty-three weeks, all fiscal years presented below contain fifty-two weeks. Our comparable store sales relate to retail stores that had been open for one year or more at the end of the period presented. For fiscal year 2006, this comparison has been adjusted to a fifty-two week basis by excluding the sales of the fifty-third week.

12

	Т	hirteen W	eeks l	Ended				F	iscal	Year Ended	d			
		Iay 3, 2008		May 5, 2007 (estated)	Fe	bruary 2, 2008		bruary 3, 2007 Restated)		nuary 28, 2006 (estated)		nuary 29, 2005 Restated)		nuary 31, 2004 Restated)
			`	,	iousa	nds, except	,	,	,	,	,	,	`	,
Statement of Operations Data:				,		•	Ü		•			,		
Net sales, retail division	\$	46,710	\$	42,488	\$	244,020	\$	229,783	\$	215,841	\$	201,425	\$	198,479
Net sales, wholesale division		9,819		5,651	\$	57,815		13,826		17,853		23,578		14,089
Total net sales		56,529		48,139		301,835		243,609		233,694		225,003		212,568
		21.075		10.000		100 (14		100.056		05.054		00.040		01.000
Gross profit, retail division		21,867		19,882		108,614		102,976		95,354		90,049		81,923
Gross profit, wholesale division		641		441		3,251		954		1,147		1,288		1,454
Total gross profit		22,508		20,323		111,865		103,930		96,501		91,337		83,377
Selling, general and														
administrative expenses		24,541		21,128		99,974		89,004		81,006		78,892		82,999
Depreciation and amortization		1,776		1,447		6,197		4,797		4,830		5,680		5,858
Expenses incurred in connection														
with change of control														4,931
Total operating expenses		26,317		22,575		106,171		93,801		85,836		84,572		93,788
1 2 1		,		,		,		,		,		,		ĺ
Income (loss) from operations		(3,809)		(2,252)		5,694		10,129		10,665		6,765		(10,411)
Other expenses:		(3,007)		(2,232)		3,074		10,127		10,005		0,703		(10,411)
Interest expense		(871)		(1,088)		(4,728)		(4,495)		(3,877)		(3,326)		(2,153)
Realized loss on investments		(0/1)		(1,000)		(4,720)		(4,423)		(3,077)		(3,320)		(2,133) (172)
Realized loss on investments														(172)
In														
Income (loss) before income		(4,680)		(3,340)	\$	966		5,634		6,788		3,439		(12,736)
taxes					Э									(12,730)
Income tax benefit (provision)		1,778		1,269		(960)		(1,244)		8,471		(150)		
	_		_		_		_		_		_		_	
Net income (loss)	\$	(2,902)	\$	(2,071)	\$	6	\$	4,390	\$	15,259	\$	3,289	\$	(12,736)
Weighted average shares														
outstanding:			_	0.50.450	_				_	0.40.4.46			_	151010
Basic		059,041		,058,170		3,058,797		,000,471		,949,146		2,832,107		,454,340
Diluted		059,041		,058,170		3,058,797		5,505,890		,463,480		3,001,844		,454,340
Basic net income (loss) per share	\$	(0.95)	\$	(0.68)	\$	0	\$	1.46	\$	5.17	\$	1.16	\$	(5.19)
Diluted net income (loss) per		(O. 7.7)		(0.75)	,		,				_		_	,=
share	\$	(0.95)	\$	(0.68)	\$	0	\$	1.38	\$	4.51	\$	1.11	\$	(5.19)

	May 3, 2008	May 5, 2007	February 2, 2008	February 3, 2007	January 28, 2006	January 29, 2005	January 31, 2004
Selected Operating Data as of:							
Number of stores open at end of							
period	313	272	303	267	239	223	232
Comparable store sales increase							
(decrease)	3.1%	(2.9)%	2.9%	2.3%	5.8%	1.8%	1.1%

	As of May 3, 2008	February 2, 2008 (\$ in the	February 3, 2007 (Restated) ousands)	As of January 28, 2006 (Restated)	January 29, 2005 (Restated)	January 31, 2004 (Restated)
Balance Sheet Data:						
Working capital (deficiency)	\$ (7,604)	\$ (2,722)	\$ 10,528	\$ 12,033	\$ 1,909	\$ (9,364)
Total assets	177,766	159,835	124,722	112,758	105,459	89,909
Long-term debt, excluding current						
portion	7,090	7,190	12,553	12,898	12,972	7,746
Total shareholders equity	32,959	35,861	35,742	30,543	12,370	7,394

Selected Historical Financial Data of Model Reorg

The selected historical statement of income data provided below for the fiscal years ended October 31, 2007, 2006 and 2005 and the balance sheet data as of October 31, 2007 and 2006 have been taken or derived from Model Reorg s audited consolidated financial statements included in this Proxy Statement beginning on page F-38. The statement of operations data for the six-month periods ended April 30, 2008 and 2007 and the balance sheet data as of April 30, 2008 have been taken or derived from Model Reorg s unaudited financial information included in this Proxy Statement beginning on page F-53. The selected historical statement of income data provided below for the fiscal years ended October 31, 2004 and 2003 and the balance sheet data as of October 31, 2005, 2004 and 2003 have been taken or derived from Model Reorg s audited consolidated financial statements for those years, which are not included in this Proxy Statement. This information is only a summary and you should read it together with Model Reorg s Management s Discussion and Analysis of Financial Condition and Results of Operations and historical financial statements and related notes contained elsewhere in this Proxy Statement. These historical results are not necessarily indicative of results to be expected in any future period.

Model Reorg s fiscal year ends on October 31. All references in discussions of Model Reorg s financial results to a particular year, with or without the word fiscal, are to the twelve-month period ending on October 31 of that year.

	Six Months Ended April 30				Fiscal Year Ended October 31								
								2006		2005		2004	2003
	2	008		2007		2007	(Restated)	(Restated)	(.	Restated)	(Restated)
				(\$ in tho	usai	nds, except w	veigh	ted average	ana	l per share a	moi	unts)	
Statement of Income Data:													
Net sales, Quality Fragrance Group		28,297	\$	120,140	\$	245,493	\$	257,826	\$	244,128	\$	169,831	\$ 97,183
Net sales, Scents of Worth		45,661		45,170		76,369		73,990		68,971		69,329	66,941
Net sales, Five Star		2,629		2,022		5,850		6,545		3,808		4,041	1,489
Total net sales	1	76,587		167,332		327,712		338,361		316,907		243,201	165,613
Gross profit, Quality Fragrance Group		32,022		29,764		57,584		59,886		51,292		36,888	25,559
Gross profit, Scents of Worth		17,642		17,341		30,040		29,945		27,941		28,838	35,446
Gross profit, Five Star		913		831		2,885		3,206		2,227		2,174	740
Total gross profit		50,577		47,936		90,509		93,037		81,460		67,900	61,745
Selling, warehouse, delivery and													
administrative expenses		32,950		29,509		60,113		57,548		54,694		43,908	38,411
Depreciation and amortization		713		822		1,411		1,721		2,464		2,346	1,833
(Recovery) provision on vendor advances						(2,367)		2,367					
Total operating expenses		33,663		30,331		59,157		61,636		57,158		46,254	40,244
				1= -0=		24 252		24.404		2.4.202			24 5 04
Income from operations		16,914		17,605		31,352		31,401		24,302		21,646	21,501
Interest expense		5,300		6,104		12,749		14,506		11,179		5,509	3,846
Income before income taxes		11,614		11,501		18,603		16,895		13,123		16,137	17,655
Income taxes		4,860		4,600		7,353		6,854		5,121		5,879	5,047
Net income	\$	6,754	\$	6,901	\$	11,250	\$	10,041	\$	8,002	\$	10,258	12,608
Weighted average shares outstanding:													
Basic and diluted(1)	- ,-	000,000		5,900,000		5,900,000		5,900,000		5,900,000		5,900,000	5,900,000
Basic and diluted net income per share(1)	\$	1.14	\$	1.17	\$	1.91	\$	1.70	\$	1.36	\$	1.74	\$ 2.14

			As	of October	31	
	As of April 30 2008	2007	2006 (Restated) (\$ in the	2005 (Restated) ousands)	2004 (Restated)	2003 (Restated)
Balance Sheet Data:						
Working capital	\$ 235,288	\$ 267,174	\$ 238,922	\$ 266,963	\$ 233,395	\$ 137,643
Total assets	306,537	344,988	306,844	318,931	311,388	167,021
Long-term debt, excluding current portion	179,644	215,227	197,521	231,995	204,757	110,194
Total shareholders equity	91,366	84,611	61,635	51,594	43,591	34,286

⁽¹⁾ See Footnotes (1) and (3) to the table beginning on page 17 for an explanation of the calculation of weighted average shares outstanding and basic and diluted net income per share.

Selected Unaudited Pro Forma Condensed Combined Financial Data

The following selected unaudited pro forma condensed combined financial data have been derived from and should be read with the Unaudited Pro Forma Condensed Combined Financial Statements and related notes beginning on page 101. This information is based on our and Model Reorg s historical statements of operations and balance sheets, giving effect to the Merger. The Merger will be accounted for under the purchase method of accounting. For accounting purposes, Model Reorg is considered to be acquiring us in this transaction. Accordingly, the purchase price is allocated among the fair values of our assets and liabilities, while the assets and liabilities of Model Reorg are reflected at historical cost in the combined pro forma balance sheet. The unaudited pro forma combined financial data are based on the estimates and assumptions set forth in the notes to such statements, which are preliminary and have been made solely for the purposes of developing such pro forma information. This information is for illustrative purposes only. The results might have been different had the companies always been combined. You should not rely on these data as being indicative of the historical results that would have been achieved had the companies always been combined or of the future results that the combined company will experience after the Merger.

The following unaudited pro forma combined statement of income combines the results of operations of E Com and Model Reorg, respectively, for the twelve months ended October 31, 2007, and the six months ended April 30, 2008, as if the transaction had occurred as of November 1, 2006. E Com s fiscal year end is the Saturday closest to January 31. The results of operations for E Com for the twelve months ended October 31, 2007 represent the 12-month period beginning October 29, 2006 and ending November 3, 2007, and those for the six months ended April 30, 2008 represent the six-month period beginning November 4, 2007 and ending May 3, 2008.

	Pro Forma Combined				
	Model Reorg and E Com				
	Six Months Ended	Twelve Months			
	April 30,	Ended			
	2008	October 31, 2007			
	(\$ in thousands, exc	ept weighted average			
	and per sho	are amounts)			
Statement of Income Data:					
Total net sales	\$ 302,565	\$ 538,619			
Cost of goods sold	187,712	339,337			
Gross profit	114,853	199,282			
Selling, warehouse, delivery and administrative expenses	88,718	156,096			
Depreciation and amortization	4,153	7,258			
Recovery on vendor advance		(2,367)			
•					
Income from operations	21,982	38,295			
Interest expense	7,198	17,661			
	,	,			
Income before income taxes	14,784	20,634			
Income tax expense	5,914	8,254			
	-,,	3,20 :			
Net income	\$ 8,870	\$ 12,380			
Net income	\$ 0,070	\$ 12,360			
W. L. L. Gl. O I'					
Weighted Average Shares Outstanding:					
Basic(1)	8,959,041	8,955,510			
Diluted(1)	9,478,899	9,469,706			
Basic net income per share(1)	\$ 0.99	\$ 1.38			
Diluted net income per share(1)	\$ 0.96	\$ 1.36			

	As of
	April 30, 2008
Balance Sheet Data:	
Working capital	\$ 268,655

Total assets	\$ 435,340
Long-term debt - excluding current portion	\$ 229,325
Total shareholders equity	\$ 117,489

(1) See Footnotes (1) and (3) to the table beginning on page 17 for an explanation of the calculation of weighted average shares outstanding and basic and diluted net income per share.

16

Comparative Historical and Unaudited Pro Forma Combined Per Share Data

The information below reflects the historical net income and book value per share of our common stock and the unaudited pro forma net income and book value per share of the combined company after giving effect to the Merger. Our historical net income for the twelve months ended October 31, 2007 combines the results of the fourth quarter of our 2006 fiscal year with those of the first nine months of our 2007 fiscal year, and our historical net income for the six months ended April 30, 2008 combines the results of the fourth quarter of our 2007 fiscal year with those of the first quarter of our 2008 fiscal year.

You should read the tables below in conjunction with our and Model Reorg s historical financial statements, and the respective related notes, beginning on page F-37 of this Proxy Statement, and the unaudited pro forma condensed combined financial information and notes related to such financial information beginning on page 101 of this Proxy Statement.

	Historical Twelve Months Ended								Pro Forma Combined Model Reorg and E Com			
	October 31, 2007 (Restated) Model		Six Months Ended April 30, 2008 Model				Twelve Months Ended		Six Months Ended			
	Е	Com	Reorg		E Com		Reorg		October 31, 2007		April 30, 2008	
Basic net income per share(1)(2)	\$	1.28	\$	1.91	\$	0.29	\$	1.14	\$	1.38	\$	0.99
Diluted net income per share(1)(2)	\$	1.23	\$	1.91	\$	0.29	\$	1.14	\$	1.36	\$	0.96
Shares used in the computation of basic net income per share(3)(4) Shares used in the computation of diluted net income per share(3)(4)	ĺ	055,510 569,706	- /-	900,000	ĺ)59,041)59,041	- /-	900,000	ĺ	955,510 469,706	ĺ	959,041 478,899

	Historical		
As of April 30, 2008	E Com	Model Reorg	
Book value per share	\$ 10.77	\$ 15.49	\$ 13.11
Shares used in the computation of book value per share(3)(4)	3,059,041	5,900,000	8,959,041

- (1) Basic net income per share is computed by dividing net income by the weighted average number of common shares outstanding during the period. In computing diluted net income per share, the weighted average number of common shares outstanding includes all common stock equivalents with exercise prices at or below the average market price for the respective period.
- (2) Pro forma combined net income per share reflects certain pro forma adjustments resulting from accounting for the Merger as a reverse acquisition that are set forth in the footnotes to the Unaudited Pro Forma Condensed Combined Statements of Operations, beginning on page 104. These adjustments, in total, decrease combined pro forma net income by approximately \$2.8 million, or \$0.31 basic income per share and \$0.29 diluted net income per share, for the twelve months ended October 31, 2007 and increase combined pro forma net income by approximately \$1.2 million, or \$0.14 basic and diluted net income per share, for the six months ended April 30, 2008.

17

Table of Contents

- (3) Model Reorg share numbers represent the 5.9 million shares issuable in the Merger in exchange for the common shares of Model Reorg. They do not include the 1.5 million shares issuable upon exercise of the Warrants issuable in the Merger since they would be antidilutive as E Com s average market price has historically been lower than the Warrant exercise price during each relevant period (including the fiscal periods shown in Selected Historical Financial Data of Model Reorg on page 14). Since Model Reorg has only 96.9 shares outstanding, this treatment permits a more realistic comparison of historical net income per share between E Com and Model Reorg.
- (4) Pro forma share numbers represent the weighted average shares outstanding of E Com for the respective periods, assuming in each case the issuance of the 5.9 million shares issuable in the Merger. Pro forma share numbers used in the computation of diluted net income per share also assume the issuance of 69,751 and 75,413 shares issuable upon exercise of stock options for the twelve months ended October 31, 2007 and the six months ended April 30, 2008, respectively, and 444,445 shares of our common stock upon conversion of the Convertible Note. The 1.5 million Warrant shares are not included in the pro forma share numbers since they would be antidilutive.

18

RISK FACTORS

Our shareholders should consider the following factors carefully in evaluating whether to approve the issuance of the Merger Consideration and the other proposals in this Proxy Statement. These factors should be considered in conjunction with the other information included in this Proxy Statement, including the forward-looking statements made herein. The following do not include all risks that the combined company will face as a result of the Merger. Additional risks related to our existing business and markets, which will continue to confront us whether or not the Merger occurs, are described in our public filings with the SEC, including our Forms 10-K and Forms 10-Q.

The Merger may be completed even though economic or industry changes or other causes may have a material adverse effect on the business of either company that could cause a decline in the combined company s earnings or stock price and reduce the value of the Merger to our shareholders.

In general, either we or Model Reorg may refuse to complete the Merger if there is an event or change having a material adverse effect on the other party between December 21, 2007, the date of the Original Merger Agreement, and the closing. However, certain types of events or changes do not permit either party to refuse to complete the Merger, even if they would have a material adverse effect on us or Model Reorg. These include the effects of (i) changes in conditions in the United States or global economy or capital or financial markets generally, including changes in interest or exchange rates, (ii) changes in general legal, regulatory, political, economic or business conditions or changes in GAAP, unless they disproportionately affect one of the parties, or (iii) any decline in the market price, or change in trading volume, of our common stock, or any failure to meet publicly announced revenue or earnings projections. If adverse changes occur but we must still complete the Merger, our stock price may suffer. This may in turn reduce the value of the Merger to our shareholders.

We may not be able to obtain the post-merger financing necessary to complete the Merger or such financing may be more expensive than we anticipate.

Consummation of the Merger is subject to a number of conditions, including the availability of a new third party \$250 million secured credit facility to replace our and Model Reorg s existing third party credit facilities. As described in Post-Merger Financing, we have received a commitment for such a facility. It expires on August 16, 2008 and is subject to certain conditions, including negotiation of acceptable loan documentation with the third party lender and documentation of the related party debt restructuring described in Approval to Issue Shares and Warrants Affiliate Debt with subordination terms acceptable to the third party lender. If the conditions are not met, the Merger could not happen without a waiver of the financing condition by the parties to the Merger Agreement. Such a waiver would be subject to the considerations described under Post-Merger Financing.

Certain of our directors, executive officers and shareholders have interests that are in addition to those of other shareholders, which may influence them to support the Merger.

Certain of our directors, executive officers and shareholders have interests in the Merger that are in addition to yours, which may influence them to support the Merger or seek to waive certain conditions in the Merger Agreement without regard to your interests. These are described under Approval to Issue Shares and Warrants Interests of Certain Persons in the Merger. You should consider whether these directors, executive officers and shareholders are more likely to support approval of the Merger, or the waiver of conditions that might benefit our shareholders, than if they did not hold these interests.

The combined company may not qualify for listing on NASDAQ after the Merger.

We currently expect that our shares will continue to be listed on the NASDAQ Capital Market following the Merger. However, we may be required to reapply for listing, and NASDAQ may not accept the listing. If it is not accepted, the Merger will not occur without a waiver from the parties to the Merger Agreement, since it is a condition to closing the Merger that we continue to be listed on the NASDAQ Capital Market.

Table of Contents

We have incurred and will incur significant costs in connection with the Merger, whether or not we complete it.

We have incurred significant costs related to the Merger and we expect to incur significant additional costs. These costs include financial advisory, legal and accounting fees and expenses and other charges. We may also incur additional unanticipated costs for any of a number of reasons. Such costs will reduce the assets that either company would have if the Merger is not consummated or that the combined company would have to operate its business after the Merger.

Completion of the Merger will result in substantial and immediate dilution to the voting power of our current shareholders.

Issuing shares of our common stock to the Model Reorg shareholders in the Merger will significantly dilute the voting power of our existing shareholders (from 63.3% of the outstanding shares held by our current shareholders other than Model Reorg shareholders before the Merger to 21.7% afterwards), and the issuance of additional shares upon exercise of the Warrants would cause further dilution (potentially reducing the ownership of our current shareholders other than the Nussdorfs to 18.6% of the outstanding shares). If the combined company does not realize the benefits from the Merger anticipated by our Board when they set the Merger Consideration, the market price of our common stock may decline as a result and our shareholders may not realize a benefit despite the ownership dilution they will experience.

Because the lack of a public market for Model Reorg s capital stock makes it difficult to evaluate the value of the Model Reorg stock, the Model Reorg shareholders may receive Merger Consideration that is greater than the fair market value of Model Reorg.

The outstanding capital stock of Model Reorg is privately held and is not traded in any public market. The lack of a public market makes it difficult to determine the fair value of Model Reorg. Since the percentage of our equity to be issued to Model Reorg s shareholders was determined based on negotiations between the parties, it is possible that the value of our common stock to be issued in the Merger will be greater than the fair value of Model Reorg.

The market price of our common stock may fluctuate significantly and the Merger Consideration will not be adjusted for such changes.

At the closing of the Merger, the outstanding Model Reorg common stock will be converted into the Merger Consideration. The number of shares of our common stock and Warrants issuable to the Model Reorg shareholders will not be adjusted for changes in the value of Model Reorg or E Com or in the market price of our common stock. We cannot abandon the Merger or resolicit the vote of our shareholders solely because of any such changes.

The market price and trading volume of our common stock has and may vary significantly between the date of the Original Merger Agreement and the completion of the Merger. These variations may be caused by, among other factors, responses to the risk factors described in this Proxy Statement, market expectations of the likelihood that the Merger will be completed and the timing of its completion, the market s perception of the merits of the Merger or the prospects for the combined company s post-Merger operations, or reasons unrelated to the combined company s performance, such as reports by industry analysts, investor perceptions, or announcements by our or Model Reorg s customers, competitors or suppliers regarding their own performance, as well as general economic and industry conditions. There can be no assurance that the market price or the trading volume of the combined company s common stock after the Merger will be at or above the market price and trading volume of our common stock on the dates of the Original Merger Agreement, this Proxy Statement, the Special Meeting, or the Merger.

20

Table of Contents

Our remedy for losses arising out of Model Reorg s breach of the Merger Agreement is limited.

Pursuant to the indemnification provisions in the Merger Agreement, we are entitled to indemnification for certain losses and expenses that are sustained in connection with any breach of Model Reorg s representations, warranties and covenants. The combined company must bear the amount of any such losses and expenses up to \$1 million. To compensate for losses and expenses above that amount, the combined company will retain in escrow 295,000 shares of the total Merger Consideration. Except with respect to claims for taxes, these escrowed shares of our common stock are the sole and exclusive remedy for any losses arising out of claims related to the Merger Agreement. These shares will be valued at a total of \$7,062,300 for this purpose, but the shares may actually be worth less at the time. If the combined company incurs losses as a result of any breach beyond the value of those shares, it will not be able to seek additional compensation from the former Model Reorg shareholders

Successful integration of the E Com and Model Reorg businesses may be difficult.

After the Merger, we and Model Reorg will integrate our respective operations. Delays and other unforeseen difficulties may impede the combined company in its efforts to integrate the E Com and Model Reorg operations successfully. Even if the two companies are able to integrate their operations, there can be no assurance that anticipated synergies will be achieved. (See Approval to Issue Shares and Warrants Reasons for the Merger.) Any failure or delay in integrating operations or achieving such synergies could have a material adverse effect on the business, results of operations, financial condition and cash flows of the combined company, and our shareholders may not realize a benefit despite the ownership dilution they will experience in the Merger.

After the Merger, control of our management and policies will remain with our principal shareholders, who could take actions that are not in the best interest of the other shareholders.

Stephen L. Nussdorf and Glenn H. Nussdorf currently beneficially own approximately 44.5% of our common stock, assuming conversion of the Convertible Note. They also own approximately 59% of Model Reorg s common stock. Immediately after the Merger, Stephen and Glenn Nussdorf will beneficially own an aggregate of approximately 53.7% of our outstanding common stock, assuming conversion of the Convertible Note, or 57.7%, assuming exercise of their Warrants. As a result, they will be able to direct our corporate policies, being able to act unilaterally to approve most actions requiring shareholder approval under law or our governing documents. In addition, Arlene Nussdorf, who is the sister of Stephen and Glenn Nussdorf, owns approximately 29% of Model Reorg s common stock. Including shares received by Arlene Nussdorf in the Merger, these Nussdorf family members (collectively, the Nussdorfs) would beneficially own approximately 72.3% of our capital stock in aggregate, or 75.7% assuming exercise of their Warrants. (See Security Ownership of Certain Beneficial Owners and Management.) The Nussdorfs collective stock ownership may have the effect of delaying or preventing policies or actions deemed desirable by our Board of Directors, such as a business combination that might be in the interests of our other shareholders, which in turn could materially and adversely affect the market price of our common stock. Conversely, such ownership may cause us to implement policies that are not in the best interests of our other shareholders.

Furthermore, after the Merger, the combined company will have approximately \$92.3 million of indebtedness to the Nussdorfs and their affiliates. (See Approval to Issue Shares and Warrants Affiliate Debt and Certain Relationships and Related Person Transactions Relationships and Transactions between Model Reorg or E Com and Other Affiliates.). As significant creditors, the Nussdorfs will have the ability to refuse consent to actions our Board may consider necessary or to demand repayment of the debt in the event of default.

After the Merger, we will be a controlled corporation under the NASDAQ rules. Accordingly, although there is no present intention to reduce the number of independent directors, we will no longer be required to have a board of directors with a majority of independent directors, to have executive compensation determined by an independent compensation committee, or to have our directors nominated by a majority of the independent directors. This could increase the extent to which the principal shareholders are able to control our operations. (See Approval to Issue Shares and Warrants Management Following the Merger.)

21

Table of Contents

We may need additional funds that, if available, could result in increased interest expenses or additional dilution to our shareholders. If additional funds are needed and are not available, the business of the combined company could be negatively impacted.

We will be a substantially larger company following the Merger, with a substantially increased debt burden. If the combined company is unable to generate sufficient cash flows from operations to service its obligations, it could face liquidity and working capital constraints, which could adversely impact its future operations and growth. If the combined company needs to raise additional funds to support its operations, it may not be able to do so on favorable terms, or at all. Without such funding, it may need to modify or abandon its growth strategy or eliminate product offerings, any of which could negatively impact its financial position. If such funds are raised through the issuance of equity securities, the percentage ownership of its shareholders will be reduced even further, and the holders of new equity securities may have rights, preferences or privileges senior to those of the holders of common stock.

We will incur substantial new indebtedness in order to close the Merger, which may adversely affect our cash flow and business operations.

It is a condition to closing the Merger that the combined company obtain a replacement senior secured credit facility of up to \$250 million to replace our and Model Reorg s existing facilities. (see Post-Merger Financing). Having this amount of indebtedness, in addition to the related party debt discussed above, may (i) make the combined company more vulnerable to adverse changes in general economic, industry and competitive conditions and (ii) place the combined company at a disadvantage compared to any of its competitors that may have less debt. Furthermore, during any period that the new credit facility bears interest at floating rates, the interest expense would increase if interest rates rise. The combined company may be required to dedicate a substantial portion of its cash flow to pay interest and principal on its debt. If the combined company does not have earnings to service the debt, it may need to refinance all or part of that debt, sell assets, borrow more money or sell securities, which it may not be able to do on commercially reasonable terms, or at all.

The terms of the new credit facility will include events of default and covenants that will limit the combined company from taking certain actions without obtaining the consent of the lenders. In addition, the credit facility will require the combined company to maintain certain financial ratios and will restrict its ability to incur additional indebtedness. These covenants and restrictions may limit the combined company s ability to respond to changing business and economic conditions and may prevent it from engaging in transactions that might otherwise be considered beneficial.

A breach of the provisions of the credit facility, including any inability to comply with the required financial ratios, could result in an event of default. If an event of default occurs (after any applicable notice and cure periods), the lenders will be entitled to terminate any commitment to make further extensions of credit under the credit facility and to accelerate the repayment of amounts outstanding (including accrued and unpaid interest and fees). Upon a default under the credit facility, the lenders could also foreclose against any collateral securing such obligations, which may be all or substantially all of the combined company s assets. If that occurred, the combined company may not be able to continue to operate as a going concern.

The Registration Rights Agreement may limit the combined company s ability to obtain financing.

As a condition to closing the Merger, we will agree, in certain circumstances, to register with the SEC the resale of the shares of our common stock that the Model Reorg shareholders receive as part of the Merger Consideration. (See Approval to Issue Shares and Warrants Registration Rights.) The Registration Rights Agreement will permit the Model Reorg shareholders to require that, in the event of any marketing limitation on the number of shares included in an applicable registration statement, their shares be registered on a pro rata basis with shares being registered for parties that have obtained registration rights in connection with providing financing to the combined company. This may limit our ability to obtain financing in the future.

22

Table of Contents

The absence of contracts with Model Reorg s retail store customers or suppliers could result in loss of key customers or suppliers, which would have a material adverse effect on the combined company s business.

Model Reorg has not had long-term or exclusive contracts with any of its retail store customers or with the suppliers of distributed brands. Suppliers of distributed brands generally can, at any time after the Merger, choose to reduce or eliminate the volume of their products distributed by the combined company, including by supplying products to Model Reorg s retail store customers directly or through another distributor. Retail store customers are generally able to cancel orders or delay the delivery of products on short notice. In addition, they may decide not to purchase products from Model Reorg for any reason, including because Model Reorg has become a subsidiary of E Com, their competitor. Moreover, our and Model Reorg s expense levels are based in part on expectations of future sales and, if expectations regarding future sales are inaccurate, the combined company may be unable to reduce costs in a timely manner to adjust for sales shortfalls. The loss of any key suppliers or customers, or a change in Model Reorg s relationship with any one of them, could have a material adverse effect on the combined company s business, prospects, and financial condition after the Merger.

If Model Reorg is unable to effectively manage the inventory it sells on consignment, the combined company will not achieve its expected results.

Model Reorg s business includes a significant portion of consigned sales, and its revenue recognition policy defers recognition of revenue for this type of sales. Consignment sales remain in inventory until the products are sold to end users and, if not sold, the inventory may be returned to Model Reorg upon termination of the consignment relationships. The turnover frequency of Model Reorg s inventory on consignment is critical to generating regular cash flow in amounts necessary to keep financing costs to targeted levels and to purchase additional inventory. If this turnover is not sufficiently frequent, the combined company s financing costs may exceed targeted levels and it may be unable to generate regular cash flow in amounts necessary to purchase additional inventory to meet the demand for other products. In addition, slow inventory turnover may force the combined company to reduce prices and accept lower margins to sell consigned products.

Any weakness in internal control over financial reporting or disclosure controls and procedures could result in a loss of investor confidence in our financial reports and lead to a stock price decline.

We will be required to evaluate our internal control over financial reporting under Section 404 of the Sarbanes-Oxley Act of 2002 and report the results in our annual reports on Form 10-K. We are also required to maintain effective disclosure controls and procedures. After the Merger, our Section 404 evaluation will need to include the internal controls of Model Reorg, a substantially larger business, and our disclosure controls and procedures will need to expand to encompass the activities of the Model Reorg business. Model Reorg has not previously had to evaluate and publicly report on its internal control over financial reporting. If management identifies any material weakness in Model Reorg s internal controls that are not remedied, we will be unable to assert that our internal controls are effective. Any failure to have effective internal control over financial reporting or disclosure controls and procedures covering the business of the combined company could cause investors to lose confidence in the accuracy and completeness of the combined company s financial reports, limit its ability to raise financing or lead to regulatory sanctions, any of which could result in a material adverse effect on the combined company s business or a decline in the market price of its common stock.

23

FORWARD-LOOKING STATEMENTS

Some of the statements in this Proxy Statement, including those that contain the words anticipate, believe, plan, estimate, expect, should, and other similar expressions, are forward-looking statements within the meaning of the Private Securities Litigation Reform Act of 1995, as amended. Those forward-looking statements involve known and unknown risks, uncertainties and other factors that may cause our actual results, performance or achievements or those of our industry to be materially different from any future results, performance or achievements expressed or implied by those forward-looking statements. Among the factors that could cause actual results, performance or achievement to differ materially from those described or implied in the forward-looking statements are:

our ability to service our obligations;
our ability to comply with the covenants in our credit facilities;
general economic conditions, including the level of discretionary spending by consumers;
competition;
the consummation of the Merger;
our ability to integrate the Model Reorg business and operations into ours;
our ability to maintain and expand supplier and customer relationships;
our ability to raise additional capital to finance expansion;
the risks inherent in new product introductions or our entry into new geographic markets; and
other factors described under Risk Factors in this Proxy Statement and in our filings with the SEC. Copies of our SEC filings are available from the SEC or may be obtained upon request from us. We do not undertake any obligation to update the information contained herein, which speaks only as of this date, other than as required by law.

THE SPECIAL MEETING OF SHAREHOLDERS

Date, Time and Place

This Proxy Statement and the accompanying form of proxy are being furnished on or about July 25, 2008 to holders of our common stock of record at the close of business on June 24, 2008 in connection with the solicitation of proxies by our Board of Directors for use at the Special Meeting of our shareholders to be held on Friday, August 8, 2008 at the offices of Edwards Angell Palmer & Dodge LLP, 750 Lexington Avenue, New York, NY 10022, commencing at 8:30 a.m., including at any postponement or adjournment of the Special Meeting.

Purposes of the Special Meeting

At the Special Meeting, we will ask our shareholders to approve the following:

- 1. The issuance of shares of our common stock and the Warrants under the Merger Agreement and the issuance of shares of our common stock upon exercise of the Warrants;
- 2. An amendment to our Articles of Incorporation to increase the number of shares of common stock we are authorized to issue from 6,250,000 shares to 20,000,000 shares;
- 3. An amendment to our Articles of Incorporation to change our corporate name to Perfumania Holdings, Inc.;

24

Table of Contents

- 4. If necessary or appropriate, a proposal to postpone or adjourn the Special Meeting to a later time to enable us to solicit additional proxies in favor of the foregoing proposals; and
- 5. Such other business as may properly come before the Special Meeting.

Implementation of Proposal 1 is conditioned on the approval of Proposal 2. The Merger cannot be completed unless our shareholders approve both proposals. Therefore, you should consider Proposals 1 and 2 together. However, implementation of Proposal 2 is not conditioned on approval of any other proposal, since we are seeking approval of the increase in authorized shares whether or not the Merger issuances are approved. Implementation of Proposal 3 also is not conditioned on approval of any other proposal.

Record Date; Shares Entitled to Vote; Quorum Requirement

Our Board of Directors has fixed the close of business on June 24, 2008 as the Record Date for the determination of shareholders entitled to notice of and to vote at the Special Meeting. At the Record Date, 3,059,041 shares of our common stock were outstanding and entitled to vote.

Each holder of shares of our common stock on the Record Date is entitled to cast one vote per share on each proposal submitted for a vote of shareholders at the Special Meeting. Votes may be cast either in person or by properly executed proxy.

The presence in person or by properly executed proxy of the holders of a majority of the outstanding shares of common stock on the Record Date is necessary to constitute a quorum for the transaction of business at the Special Meeting. If a quorum is not present at the Special Meeting, the meeting may be postponed or adjourned from time to time, without notice other than by announcement at the meeting, until a quorum is present or represented. Shares represented by proxies that are marked ABSTAIN, as well as broker non-votes, with respect to any matter, will be counted as present for the purpose of determining whether a quorum is present at the meeting.

Required Votes

Under the NASDAQ Capital Market listing criteria, approval of the issuance of the Merger Consideration and the shares issuable upon exercise of the Warrants requires the affirmative vote of a majority of the total votes cast on the proposal. Under the Florida Business Corporation Act, approval of each of the amendments to our Articles of Incorporation requires that the number of votes cast in favor of approval exceeds the number cast in opposition. These two standards are the same in operation. Accordingly, abstentions and broker non-votes will not be counted in the voting and will not have the effect of votes against any of the proposals.

Voting by Our Directors, Officers, and Affiliates

Glenn and Stephen Nussdorf, who held an aggregate of 1,113,144 shares (36.4%) of our outstanding common stock) on the Record Date, have agreed to vote their shares of our common stock in favor of all the proposals at the Special Meeting.

Miscellaneous

You may vote by proxy or in person at the Special Meeting.

Voting by Proxy

Shares of our common stock represented by properly executed proxies received at or before the Special Meeting and not revoked before being voted will be voted in the manner specified on such proxies. Properly executed proxies that do not contain voting instructions on a proposal will be voted FOR that proposal. If any

25

Table of Contents

matters not described above are properly brought before the Special Meeting, it is the intention of the persons named in the enclosed proxy card to vote the proxy in accordance with their best judgment. Properly executed proxies marked ABSTAIN, although counted for purposes of determining whether there is a quorum at the Special Meeting, will not be voted.

The voting instruction form enclosed with this Proxy Statement permits you to vote FOR, AGAINST or ABSTAIN with respect to each of the proposals. Our Board of Directors recommends that you vote FOR each of the proposals.

Voting in Person

If your shares of our common stock are registered directly in your name with our transfer agent, you are considered a shareholder of record and we have sent the Proxy Statement and proxy card directly to you. If you are a shareholder of record, you may attend the Special Meeting and vote your shares in person.

If your shares of common stock are held in the name of your broker or other financial institution, which is usually the case if you hold them in a brokerage or similar account, your shares are held in street name and your broker or other financial institution is the shareholder of record. It has sent the Proxy Statement to you with a voting instruction form. As the beneficial owner, you are also invited to attend the Special Meeting. However, only your broker or other financial institution is authorized to vote or grant a proxy for your shares. Accordingly, if you wish to vote your shares in person at the Special Meeting, you must contact your broker or other financial institution to obtain a proxy, which you must bring to the meeting.

Whether or not you intend to vote in person at the Special Meeting, we request that you sign and return the enclosed proxy card or voting instruction form in the enclosed envelope in case you are unable to attend. This would not prevent you from voting in person if you do attend.

Revocation of Proxies

A shareholder giving a proxy has the power to revoke it at any time before the vote is taken at the Special Meeting by:

if you are a stockholder of record, submitting to our Corporate Secretary (at 251 International Parkway, Sunrise, Florida 33325) a written instrument revoking the proxy;

if you are a beneficial owner of shares held in street name, following the instructions you received;

submitting a duly executed proxy card or voting instruction form bearing a later date; or

voting in person at the Special Meeting.

Solicitation of Proxies

Proxies are being solicited on behalf of our Board of Directors. We will pay the costs and expenses incurred in connection with the printing and mailing of this Proxy Statement and the solicitation of the enclosed proxy. In addition to solicitation by mail, our directors, officers and employees may solicit proxies in person or by other means of communication. Our directors, officers and employees will receive no additional compensation for such services, but we may reimburse them for reasonable out-of-pocket expenses in connection with such solicitation. Some of these directors and executive officers may have interests in the Merger that differ from yours, as described in Approval to Issue Shares and Warrants Interests of Certain Persons in the Merger. Brokers and other nominees will be requested to forward proxy solicitation materials to the beneficial owners of shares held of record by them, and we will reimburse them for the reasonable, out-of-pocket expenses they incur in doing so. We have engaged The Proxy Advisory Group, LLC, to assist in the solicitation of proxies and provide related advice and informational support for a services fee and the reimbursement of customary disbursements that are not expected to exceed \$10,000 in the aggregate.

26

Table of Contents

Assistance

If you need assistance in completing your proxy card or have questions regarding the Special Meeting, please contact:

The Proxy Advisory Group, LLC

18 East 41st Street, Suite 2000

New York, New York 10017

(212) 616-2180

PROPOSAL 1 APPROVAL TO ISSUE SHARES AND WARRANTS

We are asking you to approve the issuance of the shares of our common stock and the Warrants to the Model Reorg shareholders in accordance with the Merger Agreement, as well as the issuance of additional shares of our common stock upon exercise of the Warrants, so we may complete the Merger.

The following describes the Merger and the Merger Agreement. Although we believe that this description covers the material terms of the Merger and the Merger Agreement, it may not contain all of the information that is important to you. We encourage you to read carefully this entire Proxy Statement, including the Merger Agreement, which is attached as Annex I, for a more complete understanding of the Merger. The following description is subject to, and is qualified in its entirety by reference to, the Merger Agreement.

General

In the Merger, Model Reorg will merge into Merger Sub, and the surviving entity will continue Model Reorg s business as our wholly-owned subsidiary. As consideration for the Merger, we will issue to Model Reorg s shareholders an aggregate of 5,900,000 shares of our common stock and Warrants to purchase an aggregate of 1,500,000 shares of our common stock, exercisable beginning on the third anniversary of the Merger at an exercise price of \$23.94 per share. These shares will represent approximately 65.9% of our outstanding common stock immediately after the Merger, and 70.8% assuming exercise of all the Warrants.

Background of the Merger

Glenn H. Nussdorf and Stephen L. Nussdorf have owned a substantial interest in E Com since 2003, and we and Model Reorg and our respective affiliated persons and entities have engaged in significant commercial dealings. Those relationships are described in Certain Relationships and Related Person Transactions. The following describes the discussions and events specifically relating to the Merger and leading up to the signing of the Merger Agreement.

The Nussdorfs began considering the possibility of a business combination between us and one or more of their affiliated companies in 2003. After February 2004, when Stephen Nussdorf and Michael Katz were elected to our Board, the Board periodically discussed the possibility of a business combination with Model Reorg. However, the idea did not receive serious consideration until the summer of 2005, when Messrs. Nussdorf and Katz indicated in Board discussions that they believed that our growth potential and earnings prospects could be enhanced by such a combination.

In September 2005, Model Reorg engaged Wachovia Capital Markets LLC (Wachovia) as its financial advisor and to help arrange for financing to fund the operation and growth of the combined company. Later that month, Model Reorg held an organizational conference call including Wachovia, its counsel, Mr. Katz and representatives of Edwards Angell Palmer & Dodge LLP (EAPD), representing Model Reorg, regarding potential deal structuring and Florida legal considerations.

Table of Contents

Thereafter, one of our independent directors, Paul D. Garfinkle, on our behalf, and Mr. Katz, on behalf of Model Reorg, had several informal discussions regarding possible transaction structures and financing needs. Wachovia also commenced its due diligence review of our business and financial affairs, including holding a meeting on November 14, 2005 with Messrs. Katz and Garfinkle and other members of our and Model Reorg s management.

At a meeting of our Board on October 6, 2005, Mr. Katz discussed Model Reorg s intent to evaluate a possible combination with us. He outlined the possibility of obtaining improved terms from vendors, as well as the savings that might result from operating synergies in administrative functions and warehouse operations, and there was a general discussion. The Board authorized the execution of a confidentiality agreement with Model Reorg to permit Model Reorg to review and evaluate our business. The Board also authorized Mr. Garfinkle to investigate law and investment banking firms to help us evaluate the possible combination. Our general outside counsel, Akerman Senterfitt, advised us that Mr. Katz s role in any substantive consideration of such a combination should be solely in his capacity as an officer of Model Reorg and that communications by Model Reorg with respect to any such combination should be directed only to our independent directors.

Informal discussions to explore the benefits and risks of such a transaction, and Model Reorg s due diligence investigation, occurred during the remainder of 2005 and 2006. During this period, our independent directors periodically reviewed the business and financial impacts of combining the companies.

At its meeting on September 6, 2006, our Board reviewed the discussions to that date and determined to establish the Special Committee to consider the possible transaction. The Special Committee consists of Joseph Bouhadana, Carole Ann Taylor, and Mr. Garfinkle, the three members of our Board who are not our employees or consultants, who are independent under NASDAQ and SEC rules, and who are not shareholders of Model Reorg. None of the members of the Special Committee directly own any of our securities other than options to purchase shares of our common stock granted under our 2000 Directors Stock Option Plan (an aggregate of 9,500 shares between them). None has any interest in Model Reorg or otherwise in the proposed transaction. The Special Committee was authorized to consider and evaluate any proposed combination, assess and make a recommendation to our Board whether it would be in our best interests and those of our shareholders to effectuate the proposed combination, conduct discussions and negotiations on our behalf concerning the proposed combination, including negotiating a definitive acquisition agreement, and incur appropriate expenses in connection with its work, including retaining, at our expense, an independent investment banking firm, legal counsel and such other advisors as the Special Committee deemed necessary, useful or advisable to assist and advise it in its work. Members of the Special Committee received a fee of \$10,000 for their services in considering the Merger and Merger Agreement.

The Special Committee thereupon held its first meeting and appointed Mr. Garfinkle as its Chairman.

On November 10, 2006, Model Reorg proposed to our Board that we acquire Model Reorg in exchange for our issuing an aggregate of 6,396,649 shares of our common stock to the Model Reorg shareholders. As a result, we would remain a publicly-traded company with the Model Reorg shareholders owning approximately 81% of our issued and outstanding common stock, assuming conversion of the Convertible Note. The offer letter also contemplated that:

the combined company would obtain additional senior third party credit facilities to finance its ongoing operations and growth;

obligations due from Model Reorg to Quality King Distributors, Inc. (Quality King), a separate company that is wholly-owned by Stephen and Glenn Nussdorf and their sister, Arlene Nussdorf, would be repaid, refinanced and/or converted into preferred stock of the combined company in the transaction; and

the transaction would be subject to the approval of (a) an independent committee of our Board and (b) our shareholders other than shareholders of Model Reorg.

28

Table of Contents

Model Reorg also delivered a draft merger agreement,

On November 17, 2006, we issued a press release announcing Model Reorg s offer and the formation of the Special Committee to review and evaluate the proposal.

On December 5, 2006, the Special Committee interviewed Troutman Sanders LLP (Troutman Sanders) to serve as its legal advisor. The Troutman Sanders representative discussed with the members of the Special Committee their duties and responsibilities. The Special Committee then authorized Mr. Garfinkle to obtain proposals from investment banking firms to provide financial advice to the Special Committee. The Special Committee also authorized Mr. Garfinkle to explore the retention of an accounting firm to assist with its due diligence investigation of Model Reorg. Thereafter, the Special Committee retained Troutman Sanders as its special counsel.

Beginning December 6, 2006, the Special Committee obtained proposals from four investment banking firms, two contacted by Troutman Sanders and two recommended by members of the Special Committee. One of these firms was Financo, which had been recommended based on its specialized experience and reputation in providing investment banking, consulting and financial advice to retailers. None of these firms had previously been engaged by us, Model Reorg, or any of their affiliates in any capacity. The Special Committee considered the proposed engagement terms of the four firms and met with two of them.

The discussions between Mr. Garfinkle, on our behalf, and Mr. Katz, on behalf of Model Reorg, continued during this period, clarifying the terms of the offer and considering the timing and process to consummate a merger, due diligence, potential financing and other matters germane to the contemplated merger. In addition, on December 19, 2006, we entered into a confidentiality agreement with Model Reorg allowing us to perform a due diligence investigation on Model Reorg.

Thereafter, the Special Committee reviewed information about Model Reorg, including its financial statements, negotiated an engagement letter with Financo, and engaged in discussions with an accounting firm regarding providing due diligence services.

On April 12, 2007, there was a meeting of the Special Committee, at which the Committee:

approved the letter dated April 9, 2007 engaging Financo to provide an initial valuation analysis, to determine whether the consideration for the proposed Merger was fair from a financial point of view to our shareholders (other than those shareholders who own, or whose affiliates own, securities of Model Reorg) and to render an opinion to the Special Committee as to its conclusion, and

authorized the engagement of an accounting firm to provide tax and financial preacquisition due diligence analysis and inquiries, tax and financial structuring assistance with contract and financing issues, and assistance with post-transaction transition issues. The Special Committee then met with representatives of Troutman Sanders and Wachovia, Messrs. Stephen Nussdorf and Katz, and Model Reorg and E Com management. Model Reorg management, with Wachovia s assistance, made presentations about the structure of the combined company and the savings and synergies foreseen. In particular, Model Reorg management described the contemplated relocation of warehousing and certain other functions from Florida to New York and the consolidation of systems, which they expected would save payroll and other overhead costs. Representatives of Financo then joined the meeting and the representatives of Model Reorg management and Wachovia answered questions from the Special Committee members and Financo and discussed possible financing.

During April, May and June 2007, Wachovia and Financo exchanged information about us and Model Reorg, and Troutman Sanders, Akerman Senterfitt and EAPD held several conference calls in which they discussed the draft merger agreement and the terms of the proposed Merger. During this period, Troutman

29

Table of Contents

Sanders communicated with the members of the Special Committee about the terms of the proposed merger and the draft merger agreement, the members engaged in discussions among themselves, and Messrs. Garfinkle and Katz had periodic conversations.

On June 1, 2007, Mr. Garfinkle, Mr. Katz, and Troutman Sanders held a telephone conference to discuss substantive issues regarding the proposed Merger and draft merger agreement. Among other things, the Special Committee had requested that we not be required to make any representations and warranties in the merger agreement with respect to our operations and that the merger agreement require the Model Reorg shareholders to indemnify us against potential losses arising from misrepresentations and breaches of warranties made by Model Reorg in the merger agreement. During a later call, EAPD communicated to Troutman Sanders that Model Reorg was unwilling to agree that the Merger should be conditioned on the approval of a majority of our shareholders other than shareholders of Model Reorg, as had been proposed in Model Reorg s November 10, 2006 letter. They expressed the concern that, given the composition of our shareholder base and the likelihood that some shareholders would not vote at all, an unreasonably small number of shareholders might have the power to block the Merger, possibly for their own purposes, notwithstanding the Special Committee s approval and the desire of a majority of our shareholders to complete the proposed combination. In the following weeks, Troutman Sanders and EAPD continued to discuss the draft merger agreement.

On June 11, 2007, the Special Committee held a meeting by conference telephone call with Troutman Sanders in which it reviewed the substantive issues in the draft merger agreement, financial information about us and Model Reorg, and the accounting treatment of, and financial reporting requirements for the combined company following, the proposed Merger.

On June 25, 2007, the Special Committee held a meeting by conference telephone and received a report from Mr. Garfinkle and Troutman Sanders concerning the negotiations over the terms of the transaction, the refinancing of Model Reorg s obligations to Quality King, the status of the due diligence investigation and the initial valuation analysis being prepared by Financo.

On June 27, 2007, EAPD and Troutman Sanders held a conference telephone call concerning issues in the draft merger agreement of concern to the Special Committee, including the number of shares to be issued, the terms on which the Quality King obligations would be refinanced, a proposed escrow fund for indemnification, and the extent of our required representations. Model Reorg determined to accept the principal changes requested by the Special Committee on June 1 on condition that the transaction not be subject to the separate approval of our shareholders other than those who are also Model Reorg shareholders. EAPD provided a revised draft of the merger agreement reflecting the changes, which was distributed to the Special Committee. At that time, management of both companies provided certain projections to the Special Committee.

On July 5, 2007, the Special Committee held a meeting by conference telephone to review and discuss the revised draft of the merger agreement, the status of the due diligence investigation, and the progress of Financo s work on valuation. Among other things, the Special Committee discussed the lack of a requirement for separate approval of a majority of the shares held by shareholders other than Model Reorg shareholders. The Special Committee also discussed alternatives for the refinancing of the obligations of Model Reorg to Quality King and the effects of Model Reorg s operation of licensed stores.

On July 9, 2007, the Special Committee received the accounting firm s diligence report. The report was provided to Financo, and it was discussed by the members of the Special Committee.

On July 30, 2007, Troutman Sanders and Akerman Senterfitt held a conference call to review the draft merger agreement with EAPD. On August 3, 2007, Troutman Sanders held a conference call with EAPD to negotiate the draft merger agreement.

From April 2007 until late July 2007, Mr. Katz engaged in negotiations with a financial institution, seeking a commitment to provide senior credit facilities to the combined company. They reached a substantially final agreement in principle in July 2007 but, before execution of a commitment letter, the lenders withdrew from the

30

Table of Contents

discussions, reportedly due to what had become the generally unsettled state of the credit markets. Since the terms of any such financing would be an important factor in determining the feasibility of a combination, we and Model Reorg then pursued other financing options.

On August 9, 2007, Troutman Sanders, Mr. Katz, and EAPD held a telephone conference to discuss the issue of shareholder voting and the status of the proposed senior financing for the ongoing operations of the combined company.

On August 10, 2007, the Special Committee held a conference call with Mr. Katz and Troutman Sanders in which Mr. Katz discussed the status of the proposed financing, describing his discussions regarding alternative financing with another lender and the refinancing of the Quality King obligations, and explained Model Reorg s position with respect to the disinterested shareholder voting condition. After Mr. Katz left the call, and after discussions with Troutman Sanders, the Special Committee determined that, while counsel should continue to seek inclusion of such a provision, its absence alone should not be the reason for rejecting a transaction if the Special Committee determines that the transaction is in our best interests and is fair to our shareholders other than those who are also Model Reorg shareholders.

On August 29, 2007, the new lender issued a commitment letter to us providing for a fully-underwritten \$250 million senior secured revolving credit facility and a \$30 million senior secured first-in-last-out term loan facility, and contemplating the refinancing of Model Reorg s obligations to Quality King on a subordinated basis. That commitment letter provided that it would expire in November 2007.

EAPD and Troutman Sanders discussed the proposed terms of the transaction by telephone on a number of occasions throughout September and October.

On September 12, 2007, Financo submitted to the Special Committee its initial preliminary valuation materials regarding us and Model Reorg based upon, among other factors, the respective earnings before interest, taxes, depreciation and amortization (EBITDA) for the twelve months ended April 30, 2007, and the debt for borrowed monies, net of cash on hand (net debt) as of April 30, 2007, of each of the companies.

On September 17, 2007, the Special Committee held a meeting by conference telephone with Financo and Troutman Sanders in which Financo discussed with the Special Committee the various valuation methodologies used and comparable companies considered by Financo in preparing the valuation information concerning the two companies, including assumptions used, inherent limitations in, and results of the analysis. The Special Committee requested Financo to update the data used in these preliminary valuation materials.

On September 19, 2007, Mr. Garfinkle and Mr. Katz discussed the Special Committee s position on several aspects of the proposed Merger in a telephone call, in particular the requisite shareholder vote, the number of shares issuable to the Model Reorg shareholders and the anticipated timing of Model Reorg s year-end audit.

On September 23, 2007, the Special Committee held a meeting by conference telephone in which the members discussed Model Reorg s position with respect to separate voting by our disinterested shareholders and other considerations relating to the proposed transaction.

On September 24, 2007, Financo provided the Special Committee with updated preliminary valuation material setting forth information regarding the valuations of both companies based on forward-looking information furnished by each company for the twelve months ending December 31, 2007, in addition to the historical information previously considered.

On October 5, 2007, the Special Committee held a meeting by conference telephone with Financo and Troutman Sanders in which Financo discussed with the Special Committee its updated preliminary valuation materials and there was a general discussion of the information relied on and Financo s analytical approach.

31

Table of Contents

Following Financo s participation in the meeting, the Special Committee reviewed information prepared by Troutman Sanders with respect to the number of shares of our common stock that would be issued to the shareholders of Model Reorg under the different methodologies and at various relative ranges of valuations contained in Financo s preliminary valuation materials.

Also on October 5, 2007, Model Reorg acquired by merger 100% of Jacavi LLC (Jacavi), an independent fragrance distributor, as had been contemplated during the Merger negotiations. As a result, the former owners of Jacavi acquired an 11.25% interest in Model Reorg and members of the Nussdorf family continued to hold the remaining ownership interest.

On October 8, 2007, the Special Committee held a meeting by conference telephone with Troutman Sanders, in which it was agreed that Financo should further update the historical information it had used in its valuations to the most current information provided by both companies and update the forward-looking analysis through the end of our current fiscal year in order to reflect more comparable information and to conform to the conventional retail fiscal year end.

On October 16, 2007, EAPD distributed revised drafts of the proposed merger agreement and the related Model Reorg disclosure schedules.

On October 19, 2007, Financo provided the Special Committee revised preliminary valuation materials, including historical information based on the actual EBITDA of both companies for the twelve months ended August 31, 2007 and projected EBITDA for both companies and the combined company for the twelve months ending January 31, 2008. The projected figures for Model Reorg included results of Jacavi on a proforma basis for periods following Model Reorg s acquisition of Jacavi.

On October 23, 2007, the Special Committee held a meeting by conference telephone with Troutman Sanders to review and analyze Financos October 19, 2007 preliminary valuation materials. The Special Committee also reviewed information prepared by Troutman Sanders computing the number of shares of our common stock that would be issued to the shareholders of Model Reorg if issued in proportion to the respective valuations of the two companies contained in Financos preliminary valuation materials, with certain adjustments. The Special Committee also reviewed certain provisions of the draft merger agreement with Troutman Sanders and the proposed refinancing of Model Reorgs obligations to Quality King.

Based on these reviews and discussions, the Special Committee determined to make a counter-proposal, subject to final resolution of open issues in the draft merger agreement, acceptable terms for the financing of the obligations of Model Reorg to Quality King, completion of the proposed senior financing set forth in the August 9, 2007 commitment letter and completion of our due diligence investigation of Model Reorg.

Mr. Garfinkle and Mr. Katz had a telephone conversation on October 24, 2007 in which Mr. Garfinkle communicated the Special Committee s proposed revised terms for the combination, including:

decreasing the number of shares to be issued to the Model Reorg shareholders from 6,396,649 to 5,500,000;

the Model Reorg shareholders indemnifying us for any tax liability of Model Reorg for periods before the consummation of the Merger; and

the number and valuation of the shares of our common stock to be held in the escrow fund.

In a November 5, 2007 telephone conversation, EAPD advised Troutman Sanders of Model Reorg s counter-proposal that, due to perceived changes in the two companies businesses since the November 2006 merger proposal, we issue enough shares of our common stock for the Model Reorg shareholders to beneficially own an aggregate of 85% of our common stock (inclusive of shares presently owned by Stephen and Glenn Nussdorf and giving effect to the conversion of the Convertible Note), or approximately 9,500,000 shares.

Table of Contents 47

32

Table of Contents

Troutman Sanders thereupon communicated this counter-offer to each Special Committee member and, later that day, the Special Committee held a meeting by conference telephone in which it discussed the counter-offer and requested that Mr. Garfinkle confirm its accuracy and the reasons for it. On the next day, Mr. Garfinkle and Mr. Katz had a telephone conversation about the counter-offer.

On November 7, 2007, the Special Committee held a meeting by conference telephone in which Mr. Garfinkle reported that he had spoken with Mr. Katz, who had confirmed the accuracy of Model Reorg s counter-offer. The Special Committee determined that, based on the preliminary valuations materials it had previously received, this proposal was not fair to our shareholders. It also expressed concern that, under Florida Law, stock ownership above 80% would permit the Model Reorg shareholders to acquire us by a short form merger with a company they control and without the approval of our shareholders. The Special Committee requested a meeting with Stephen Nussdorf and Mr. Katz.

On November 8, 2007, the members of the Special Committee held a conference telephone call with Stephen Nussdorf and Mr. Katz. The members of the Special Committee requested that Model Reorg reconsider its position or provide the Special Committee with information that Model Reorg had received from its financial advisors that supported Model Reorg s counter-offer.

On November 20, 2007, the August 29, 2007 financing commitment described above expired in accordance with its terms.

On November 26, 2007, the Special Committee met with Glenn Nussdorf and Mr. Katz to negotiate the economic terms of the proposed Merger, including the terms of the refinancing of Model Reorg s obligations to Quality King. During this meeting, Mr. Garfinkle stated that the Special Committee would not agree to increase the number of our shares issuable to the Model Reorg shareholders as proposed by Model Reorg but that, in addition to the issuance of 5,500,000 shares, the Special Committee would consider the issuance of warrants to purchase 500,000 shares at a price above the current market price of our common stock.

EAPD and Troutman Sanders continued to discuss the negotiations between Model Reorg and the Special Committee in periodic telephone conversations during November and December.

On December 4, 2007, Model Reorg responded with a proposal for the issuance of 6,400,000 of our shares and Warrants to purchase 1,500,000 additional shares at an exercise price equal to the market price of our common stock on that date.

On December 5, 2007, the Special Committee held a meeting by conference telephone to consider Model Reorg s revised offer and the refinancing of its obligations to Quality King. The members determined to make a counter-offer, that we issue 5,900,000 shares of common stock and Warrants to purchase 1,500,000 shares, exercisable at a price of \$1.00 above the market price of our common stock on the date of the Original Merger Agreement. On that basis, the Model Reorg shareholders would beneficially own approximately 78.3% of our outstanding common stock immediately after the Merger, or 82.2% assuming the exercise of their Warrants (inclusive of shares presently owned by Stephen and Glenn Nussdorf and giving effect to the conversion of the Convertible Note). Troutman Sanders communicated this counterproposal to EAPD.

On December 6, 2007 Mr. Katz advised Mr. Garfinkle that the Model Reorg shareholders agreed in principle to the Special Committee s counter-offer, other than with respect to the exercise price of the Warrants.

Over the succeeding days, Mr. Katz and Mr. Garfinkle negotiated the exercise price of the Warrants, resulting in an understanding that it would be equal to the average closing price of our common stock on NASDAQ for the 50 trading days ending immediately before the date of execution of the Original Merger Agreement. Model Reorg also agreed with the Special Committee s position that the Warrants not be exercisable for three years after the date of issuance and not be transferable, with limited exceptions. In addition, the terms

33

Table of Contents

on which the existing obligations of Model Reorg to Quality King would be satisfied were negotiated, with the Special Committee taking the position that the terms should be at least as favorable to us as those in the original senior credit facilities proposed in July 2007. Such terms were agreed to by Model Reorg as described under Terms of the Merger Agreement Affiliated Debt.

On December 10, 2007, EAPD provided a revised draft of the merger agreement and drafts of the form of Warrant and Registration Rights Agreement.

On December 12, 2007, Troutman Sanders, Akerman Senterfitt and EAPD held a conference call to discuss the new drafts of the Merger documents, and Model Reorg agreed to make certain changes, including eliminating the provision for net exercise of the Warrants.

From December 12, 2007 through December 20, 2007, EAPD, Troutman Sanders, and Akerman Senterfitt continued their telephone discussions of comments to the Merger documents, and EAPD prepared and distributed revised drafts. Merger Sub was organized in Delaware on December 13, 2007 to be our wholly-owned subsidiary into which Model Reorg would merge.

On December 17 and 18, 2007, the Special Committee held meetings by conference telephone with Troutman Sanders to review the current drafts of the merger agreement, reflecting changes requested by the Special Committee and agreed to by Model Reorg, including the following:

the indemnification provisions of the draft merger agreement had been bifurcated so that claims, if any, relating to tax liabilities above those accrued on Model Reorg s financial statements would be paid in cash by the Model Reorg shareholders without limitation:

the Model Reorg shareholders would agree that, notwithstanding any ability to do so under Florida law, they would not effect a short form merger of us with a company they control while their combined ownership of our outstanding shares was less than 90% of our outstanding common stock; and

the Model Reorg shareholders would agree to vote the shares of our common stock they hold for the proposals in this Proxy Statement needed to implement the Merger.

The Special Committee also reviewed Model Reorg supdated disclosure schedules, the Warrants and the Registration Rights Agreement.

On December 21, 2007, Model Reorg s Board of Directors and shareholders each approved the Original Merger Agreement and the proposed Merger on behalf of Model Reorg.

On December 21, 2007, the Special Committee met in person at our headquarters. Financo, whose representatives participated by conference telephone, presented its final valuation report to the Special Committee and responded to questions from the members. Financo then rendered its oral opinion (later confirmed in writing) to the Special Committee that, as of December 21, 2007, the Merger Consideration was fair from a financial point of view to our shareholders (other than those of our shareholders who own, or whose affiliates own, securities of Model Reorg). A copy of Financo sopinion is included as Annex II to this Proxy Statement and a discussion of the bases for that opinion is described under Opinion of Financo in this Proxy Statement.

The Special Committee then discussed the proposal in depth, considering the Financo report and opinion, as well as the factors described under Reasons for the Merger, among others. After discussion, the Special Committee unanimously approved the Merger and the related transactions and recommended the adoption of the Original Merger Agreement to our Board of Directors, subject to the Special Committee s right to withdraw, modify or amend such recommendation if the Special Committee determines, in good faith, that failure to take such action would be reasonably likely to result in a breach of its fiduciary duties to our shareholders under applicable law.

Table of Contents

Following the Special Committee meeting on December 21, 2007, our Board of Directors held a meeting that was attended by all members of the Board, with Stephen Nussdorf attending by conference telephone. The interests of Messrs. Nussdorf and Katz in the proposed Merger were noted. After discussion, based in part on the recommendation of the Special Committee and the fairness opinion the Special Committee received from Financo, and for reasons including those described under Reasons for the Merger, the Board unanimously:

determined that the terms of the Merger are fair from a financial point of view to the holders of our common stock (other than shareholders who own, or whose affiliates own, securities of Model Reorg);

adopted the Original Merger Agreement and declared the Original Merger Agreement and the related transactions to be advisable;

approved amendments of our Articles of Incorporation, subject to shareholder approval, (a) increasing the number of authorized shares of common stock to 20,000,000 and (b) changing our name to Perfumania Holdings, Inc.; and

recommended that our shareholders approve (a) the issuance of the Merger Consideration and the issuance of our common stock upon exercise of the Warrants, subject to the Board s right to withdraw, modify or amend its recommendation to the extent that it determines, in good faith, that failure to take such action would be reasonably likely to result in a breach of its fiduciary duties to our shareholders under applicable law, and (b) the foregoing amendments of our Articles of Incorporation.

After the Board meeting on December 21, 2007, the Original Merger Agreement was executed by us, Model Reorg and Merger Sub, and by the shareholders of Model Reorg. Later in the day we issued a press release announcing its execution.

After the execution of the Original Merger Agreement, we restated our historical amortization of rent expense and leasehold improvements as described in Note 3 to our audited Consolidated Financial Statements on page F-14, which resulted in an increase in net income of \$1.0 million in fiscal year 2005, a decrease in net income of \$100,000 in fiscal year 2006, and a decrease in net income of \$270,000 for the twelve months ended September 30, 2007. In addition, Model Reorg restated amounts reported in its fiscal 2005 and 2006 years related to inventory reserves, deferred revenue (included in accrued expenses) and federal and state income taxes as described in Note 1 to Model Reorg s audited Consolidated Financial Statements on page F-42, which resulted in increases in Model Reorg s net income in fiscal 2006 and 2005 of \$290,000 and \$370,000, respectively, and a decrease in net income of \$715,000 for the twelve months ended September 30, 2007 from the projected amounts previously provided to Financo and the Special Committee.

During the Spring of 2008, as contemplated by the Original Merger Agreement, we and Model Reorg negotiated with institutional lenders for a senior secured credit facility. We obtained an updated commitment on May 16, 2008 for the Senior Credit Facility described in Post-Merger Financing from the same institutional lender that had provided the August 29, 2007 financing commitment.

On May 30, 2008, EAPD and Troutman Sanders discussed possible revisions to the Original Merger Agreement to reflect the terms of the proposed Senior Credit Facility, changes to the restructuring of Model Reorg s related party debt required by that facility, and an extended deadline for consummation of the Merger. On June 3, 2008, members of Model Reorg management discussed the terms of the proposed debt arrangements with Troutman Sanders and EAPD. After a telephone conversation on June 5, 2008 among EAPD, Troutman Sanders and Akerman Senterfitt, EAPD distributed a draft of the First Amendment on June 9, 2008. On June 13, 2008, Troutman Sanders responded with its comments. During the following three weeks, EAPD and Troutman Sanders, in consultation with their clients, finalized the language of the First Amendment.

On June 17, 2008, the Special Committee met by conference call to consider the First Amendment. We and Model Reorg provided the Special Committee with interest expense projections regarding the proposed new Senior Credit Facility and affiliate debt arrangements showing that the combined company s aggregate interest expense, based on anticipated debt levels, is expected to be less on an overall basis than was projected under the arrangements contemplated by the Original Merger Agreement.

35

Table of Contents

We and Model Reorg also provided the Special Committee with certain modified historical financial results and projections taking into account the restatements of our and Model Reorg s financial statements described above. Compared to the historical financial results previously considered by the Special Committee and Financo, the restatements resulted in a decrease in our EBITDA for the twelve-month period ending September 30, 2007 of \$267,000, or 1.85%, and a decrease in Model Reorg s EBITDA for the same period of \$1 million, or 2.57%. The cumulative effect of the restatements on EBITDA for the five years ending October 31, 2007 was a decrease of \$744,000, or 1.6%, for us and an increase of \$4.36 million, or 2.91%, for Model Reorg. The Special Committee was advised that our restatement will have an immaterial impact in future years, while Model Reorg s restatement will have no future impact.

In addition, we and Model Reorg provided the Special Committee with updated business forecasts for the twelve months ending October 31, 2008, reflecting the current economic environment. These and the related assumptions are described in Selected Financial Projections, beginning on page 108. Taking these factors into account, including the effect of the restatements described above, we and Model Reorg advised the Special Committee that our EBITDA for the twelve months ending October 31, 2008 is projected to decline by 32.8% compared with the projections we originally provided the Special Committee and Financo, while Model Reorg s EBITDA for the same period is projected to decline by only 15.4%.

A factor in Financo s financial analysis of the fairness of the Merger Consideration was the EBITDAs of E Com and Model Reorg for the twelve months ended September 30, 2007 and their projected EBITDAs for the twelve months ending October 31, 2008 (see Opinion of Financo on page 37). The Special Committee considered the effects of the restatements on each company s historical and projected EBITDA to be immaterial. The Special Committee s review of the fairness of the proposed Merger Consideration focused primarily on the estimated relative values of us and Model Reorg, rather than the absolute value of the Merger Consideration as of any particular time. The Special Committee noted that the percentage decline in our projected EBITDA for the twelve months ending October 31, 2008 was twice the projected decline of Model Reorg s EBITDA for the same period. The Special Committee also concluded that the reasons for the Merger described under Reasons for the Merger below remained valid. Accordingly, after considering the First Amendment, the restatements and the other information described above, the Special Committee determined for these reasons and those described in Reasons for the Merger not to seek an update of Financo s fairness opinion nor modify its recommendation.

On July 7, 2008, our Board of Directors held a meeting that was attended by all the directors except Ms. Taylor to discuss the First Amendment and the Merger. The Board held another meeting on July 8, 2008, by conference telephone call, that was attended by all the directors. The interests of Messrs. Stephen Nussdorf and Michael Katz in the proposed Merger were noted. After discussion, based in part on the recommendation of the Special Committee, and for the reasons described in Reasons for the Merger, the Board unanimously approved the execution of the First Amendment.

Reasons for the Merger

Our Board of Directors believes that the Merger will provide substantial benefits to our shareholders. The combination with Model Reorg will create a large, independent, national, vertically integrated wholesale distributor and specialty retailer of perfumes and fragrances that we expect will be well-positioned to compete in the marketplace and drive growth, as well as to benefit from increased operating scale.

In making its determination to recommend approval of the Original Merger Agreement, the Special Committee consulted with our and Model Reorg s officers regarding the strategic and operational aspects of the combination and received reports of our diligence review of Model Reorg. In addition, the Special Committee consulted with representatives of Financo regarding valuations of each company and with respect to the fairness of the transaction from a financial point of view, and with special counsel to the Special Committee regarding legal matters. In the course of reaching its determinations, the Special Committee and our Board considered a variety of factors, including the following:

the anticipated benefits of vertical integration and diversification of our business;

36

Table of Contents

the increased cash flow that is expected to be available to support an acceleration of the growth of E Com s retail operations;

the higher consolidated capitalization, revenues and earnings base of the combined company that should result in both improved bank and private financing capability and afford us greater access to public capital markets;

the greater scale and stronger financial position that is expected to result in additional bargaining power and improved terms with suppliers and landlords;

the availability to us of more extensive, modern warehouse facilities that will provide better and more direct access to vendors and the ability to make more timely deliveries;

the addition of experienced management and purchasing personnel;

the Special Committee s belief that risks inherent in distribution businesses, such as that of Model Reorg, are lower than those in retail operations, such as that of E Com;

the accretion in earnings per share for our existing shareholders expected to result from the Merger; and

synergies anticipated to result from the Merger, including those discussed below.

The anticipated synergies include the ability to combine both companies—warehouse, shipping, information technology and administration functions into a single, modern facility, reducing overhead and minimizing redundancies, and other efficiencies, partially offset by increased costs of auditing and internal control and freight costs. Based on our estimates, we anticipate that these synergies will reduce the projected aggregate annual operating expenses of the two companies by approximately \$3.5 to \$4.0 million on a pre-tax basis, with a nominal impact on the twelve months ending October 31, 2008. See also—Selected Financial Projections. No assurance can be given that we will achieve synergies of any particular magnitude.

Potential negative factors addressed by the Special Committee and the Board during the period that the transaction was under consideration included:

the Nussdorfs acquisition of majority ownership and, thus, the ability to exercise legal control over most matters to be determined by the combined company s shareholders;

concern over possible Model Reorg customer perceptions of competition from the combined company; and

the greater leverage to be incurred by the combined company.

However, in the judgment of the Special Committee and the Board, these concerns were outweighed by the anticipated benefits for E Com and its shareholders.

In considering the First Amendment, the Special Committee and the Board reexamined the principal factors supporting its earlier decision as described above and concluded that the reasons for the Merger remain valid.

Opinion of Financo

Pursuant to a letter agreement dated as of April 9, 2007, Financo was engaged to provide a fairness opinion to the Special Committee. The Special Committee selected Financo based on Financo s reputation and experience in the retail and consumer industries in particular. Financo focuses on providing investment banking services, including merger and acquisition advisory services, to retail and consumer companies. In this capacity, Financo is continually engaged in valuing these businesses and maintains an extensive database regarding mergers and acquisitions in these industries for comparative purposes. At the meeting of the Special Committee on December 21, 2007, Financo rendered its opinion that, as of such date, and based upon and subject to the various assumptions and limitations set forth in its written opinion, the Merger Consideration was fair, from a financial point of view, to holders of our common stock (other than those who own, or whose affiliates own, securities of Model Reorg).

Table of Contents

Financo s opinion, which describes the assumptions made, matters considered and limitations on the review undertaken by Financo, is attached as Annex II to this Proxy Statement. Holders of our common stock are urged to, and should, read the Financo opinion carefully and in its entirety. The Financo opinion is directed to the Special Committee and addresses only the fairness of the Merger Consideration from a financial point of view to holders of our common stock (other than those who own, or whose affiliates own, securities of Model Reorg) as of the date of the opinion. The Financo opinion does not address any other aspect of the Merger and does not constitute a recommendation to any holder of our common stock as to how to vote at the Special Meeting. The summary of the Financo opinion set forth in this Proxy Statement is qualified in its entirety by reference to the full text of such opinion.

Financo s opinion does not address the underlying business decision to enter into the Merger Agreement or the Merger, nor does it evaluate alternative transaction structures or other financial or strategic alternatives.

In reading the discussion of the fairness opinion set forth below, you should be aware that Financo reviewed and considered such financial and other matters as Financo deemed relevant, including, among other things:

drafts of the Original Merger Agreement, the Warrants and the Registration Rights Agreement, each dated December 20, 2007, which, for purposes of this opinion, Financo assumed to be in all material respects identical to the forms of such agreements to be executed;

certain publicly available financial and other information for E Com, including its Annual Report on Form 10-K for the fiscal year ended February 3, 2007 and Quarterly Reports on Form 10-Q for the quarters ended May 5 and August 4, 2007 and certain other relevant financial and operating data furnished to Financo by our management;

historical financial information of Model Reorg, prepared and furnished to Financo by management of Model Reorg;

certain internal financial analyses, financial forecasts, reports and other information concerning E Com prepared and furnished to Financo by our management and management and representatives of Model Reorg;

certain internal financial analyses, financial forecasts and other information concerning Model Reorg furnished to Financo by management and representatives of Model Reorg;

discussions Financo had with certain members of our management concerning the historical and current business operations and strategy, financial condition and prospects of E Com and such other matters as Financo deemed relevant;

discussions Financo had with certain members of Model Reorg management concerning the historical and current business operations and strategy, financial condition and prospects of Model Reorg and such other matters as Financo deemed relevant;

certain operating results, the reported prices and historical trading activity of our common stock, and the operating results, reported prices and historical trading activity of the securities of certain publicly traded companies Financo deemed relevant;

certain financial terms of the Merger as compared to the financial terms of certain selected business combinations Financo deemed relevant; and

such other information, financial studies, analyses and investigations and such other factors that Financo deemed relevant for the purposes of its opinion.

In conducting its review and arriving at its opinion, Financo assumed and relied, with the consent of the Special Committee and without independent investigation, upon the accuracy and completeness of the information publicly available about E Com and Model Reorg, and the financial and other information provided to Financo or otherwise made available to, discussed with, or reviewed by or for Financo. Financo assumed no

38

Table of Contents

responsibility for the accuracy or completeness, and did not independently verify, any such information. Financo relied on the assurance provided by our management that they are unaware of any facts that would make such information incomplete or misleading. With respect to the transaction process conducted on our behalf, Financo assumed and relied upon, without assuming any responsibility for independent verification of, the accuracy and completeness of the information supplied, summarized or otherwise made available to, discussed with, or reviewed by or for Financo, including as to the completeness of the process. Financo s analyses were based, among other things, on the financial projections of E Com (the E Com Financial Projections) prepared by our management and the financial projections of Model Reorg prepared by management of Model Reorg (the Model Reorg Financial Projections and collectively with the E Com Financial Projections, the Financial Projections). With respect to the Financial Projections, which were furnished to, discussed with or reviewed for Financo by management of E Com or Model Reorg, as the case may be, Financo assumed that they were reasonably prepared on bases reflecting the best currently available estimates and good faith judgments of management of E Com or Model Reorg, as the case may be, as to the future competitive, operating and regulatory environments and related financial performance of E Com or Model Reorg, as the case may be, and that the assumptions derived therefrom provided a reasonable basis for Financo s opinion. Financo expressed no view as to such Financial Projections, or the assumptions on which they were based.

For purposes of rendering its opinion, Financo assumed in all respects material to its analysis that the representations and warranties of each party contained in the Original Merger Agreement were true and correct, that each party would perform all of the covenants and agreements required to be performed by it under the Original Merger Agreement and that all conditions to the consummation of the Merger will be satisfied without waiver or modification thereof. Financo also assumed that in the course of obtaining the necessary regulatory and third party approvals, consents and releases for the Merger, no modification, delay, restriction or condition will be imposed that will have a material adverse effect on the Merger and that the Merger will be consummated in accordance with applicable laws and regulations and the terms of the Original Merger Agreement, without delay, waiver, amendment or modification of any material term, condition or agreement. Financo relied without independent verification upon the views of our management concerning the business, operational and strategic benefits and implications of the Merger. Financo did not express any view on, and its opinion did not address, the fairness of the amount or nature of any compensation to be paid or payable to any of the officers, directors or employees of E Com or Model Reorg, or any class of such person, whether relative to the Merger Consideration or otherwise.

Financo was not asked to pass upon, and expressed no opinion with respect to, any matters, including any agreements between E Com and Model Reorg or any of their respective affiliates, other than the fairness from a financial point of view of the Merger Consideration to holders of our common stock (other than those who own, or whose affiliates own, securities of Model Reorg). In connection with its engagement, Financo did not evaluate, nor did we or the Special Committee request Financo to evaluate, alternative transaction structures or financial alternatives other than the Merger, and Financo did not otherwise participate in the transaction process. Financo s opinion does not address our underlying business decision to effect the Merger.

Financo did not make or obtain any independent evaluations, valuations or appraisals of the assets or liabilities of E Com or Model Reorg nor was Financo furnished with such materials. In addition, Financo has not conducted, nor has Financo assumed any obligation to conduct, any physical inspection of the properties or facilities of E Com or Model Reorg. Financo assumed with the consent of the Special Committee that there are no legal issues with regard to E Com or Model Reorg that would affect its opinion, and Financo relied on this assumption without undertaking any independent investigation or inquiry. Financo s opinion was necessarily based upon economic and market conditions and other circumstances as they existed and could be evaluated by it on the date it rendered its opinion. It should be understood that although subsequent developments may affect Financo s opinion, Financo did not have any obligation to update, revise or reaffirm Financo s opinion and Financo expressly disclaimed any responsibility to do so. Financo s opinion did not address the relative merits of the Merger as compared to other business strategies that might be available to E Com, nor did it address our underlying business decision to proceed with the Merger. Financo expressed no view as to the federal, state or local tax consequences of the Merger. Financo s opinion was approved by a fairness committee of Financo.

39

Table of Contents

The following is a summary explanation of the various sources of information and valuation methodologies employed by Financo in rendering its opinion. These analyses were presented to the Special Committee at its meeting on December 21, 2007. This summary describes the financial analyses used by Financo and deemed to be material, but does not purport to be a complete description of analyses performed by Financo in arriving at its opinion. Financo did not explicitly assign any relative weights to the various factors or analyses considered. This summary of financial analyses includes information presented in tabular format. In order to fully understand the financial analyses used by Financo, the tables must be read together with the text of each summary. The tables alone do not constitute a complete description of the financial analyses.

For purposes of its analysis, Financo evaluated on a standalone basis each of E Com, Model Reorg and the combined company.

E Com Valuation

E Com Comparable Companies Analysis

Financo compared E Com to a group of comparable public specialty retailing companies referred to as the E Com Comparable Companies. The E Com Comparable Companies are public companies that Financo deemed comparable to E Com s limited universe. Financo selected public companies competing in the specialty retailing industry selling third party brands at value prices with net sales for the trailing twelve months publicly available at the time of the analysis of between \$478 million and \$672 million. Financo also included in this group public companies competing in the specialty retailing of personal care products of a threshold size, with net sales for the trailing twelve months publicly available at the time of the analysis of between \$870 million and \$10,996 million.

The E Com Comparable Companies are:

Buckle, Inc.
Casual Male Retail Group, Inc.
Limited Brands, Inc.
Sally Beauty Holdings, Inc.
Shoe Carnival, Inc.
Ulta Salon, Cosmetics & Fragrance, Inc. and

The Wet Seal, Inc.

Financo considered the ratio of equity market capitalization, adjusted for cash and debt when appropriate, to reflect enterprise value, which we refer to as EV, to earnings before interest, taxes, depreciation and amortization, which we refer to as EBITDA, and the ratio of price to earnings, which we refer to as P/E, to derive multiples that indicate the value public equity markets place on companies in a particular market segment. In order to perform this analysis, Financo compared financial information of E Com with publicly available information for the E Com Comparable Companies. Financo considered the entire range of multiples yielded by the E Com Comparable Companies in establishing a more limited reference range of multiples which it used for its analysis. For this analysis, as well as other analyses, Financo examined publicly available information, as well as a range of estimates based on equity research reports.

The following table presents, based on closing prices as of December 20, 2007, the reference range of multiples for the E Com Comparable Companies of EV divided by EBITDA, and P/E:

	Reference Range of Multiples
9/30/07 EV/ EBITDA	8.0x - 9.5x
9/30/07 P/E	10.0x - 16.0x
Projected 10/31/08 EV/EBITDA	6.5x - 8.0x
Projected 10/31/08 P/E	8.0x - 14.0x

The comparables imply the following reference ranges for E Com s per share value:

	Reference Range of
	Values
9/30/07 EV/ EBITDA	\$17.51 - \$23.43
9/30/07 P/E	\$12.32 - \$19.71
Projected 10/31/08 EV/ EBITDA	\$12.15 - \$18.23
Projected 10/31/08 P/E	\$7.93 - \$13.88

No company utilized in the public company comparables analysis as a comparison to E Com is identical to E Com. In evaluating the comparables, Financo made numerous assumptions with respect to the specialty retailing industry s performance and general economic conditions, many of which are beyond the control of E Com. Mathematical analysis, such as determining the median, average or range, is not in itself a meaningful method of using comparable company data.

E Com Comparable Transactions Analysis

Financo considered ratios of EV to sales and EBITDA to derive multiples that strategic and financial acquirers have been willing to pay for companies in this particular market segment. In order to perform this analysis, Financo reviewed a number of transactions, including those involving companies deemed by Financo to be comparable to E Com based on financial performance, market focus and business model. Financo reviewed eighteen comparable merger and acquisition transactions announced from July 31, 2003 through September 18, 2007 involving companies in the specialty retailing industry. For this analysis, as well as other analyses, Financo examined publicly available information, as well as information from Financo s proprietary database of published and confidential merger and acquisition transactions in the retail and consumer industries.

The E Com comparable transactions consisted of the acquisitions of:

Factory Card & Party Outlet by Amscan Holdings, Inc.

United Retail Group by Red Cats USA, Inc.

Deb Shops, Inc. by Lee Equity Partners

Cutter & Buck, Inc. by New Wave Group AB

American Apparel by Endeavor Acquisition Corp.

Golf Galaxy, Inc. by Dick s Sporting Goods, Inc.

Sally Beauty Holdings, Inc. by Clayton, Dubilier & Rice, Inc.

Yankee Candle Co. Inc. by Madison Dearborn Partners Inc.

The Body Shop International PLC by L Oreal SA

Pacific Sales Kitchen & Bath Centers, Inc. by Best Buy Co., Inc.

Party City Corporation by Berkshire Partners & Weston Presidio

OshKosh B Gosh Inc. by William Carter Co.

Brookstone, Inc. by J.W. Childs Associates, Osim International and Temasek Holdings

Maurices Inc. by Dress Barn Inc.

Galyan s Trading Company, Inc. by Dick s Sporting Goods, Inc.

Loehmann s Holdings Inc. by Crescent Capital Investments, Inc.

InterTAN Inc. by Circuit City Stores Inc. and

The White House, Inc. by Chico s FAS Inc.

41

Table of Contents

The following table presents, as of December 20, 2007, the reference range of multiples of adjusted price (defined as equity price plus total debt minus cash and cash equivalents) divided by the seller s EBITDA in the last reported twelve months prior to acquisition for the transactions listed above:

 $\begin{tabular}{lll} \bf Reference Range of \\ \hline \bf Multiples \\ EV/ \, EBITDA & 9.0x - 11.0x \\ \end{tabular}$

These comparables imply the following reference range for per share value:

Reference Range of Values
EV/ EBITDA \$21.46 - \$29.36

No transaction utilized as a comparable in the transaction comparables analysis is identical to the Merger. In evaluating the comparable transactions, Financo made numerous assumptions with respect to the specialty retailing industry s performance and general economic conditions, many of which are beyond the control of E Com. Mathematical analysis, such as determining the average, median or range, is not in itself a meaningful method of using comparable transaction data.

Consideration of the Discounted Cash Flow Methodology

While discounted cash flow is a commonly used valuation methodology, Financo did not employ such an analysis of E Com for the purposes of its opinion. Discounted cash flow analysis is most appropriate for companies that exhibit relatively steady or somewhat predictable streams of positive future cash flow. The E Com Financial Projections assume a heightened level of capital expenditures in the short term, significantly reducing free cash flow, and do not extend for a sufficient period to reflect potential benefits from such expenditures. Because of these factors, and the inherent uncertainty in forecasting operating performance, future cash flows and a sustainable long-term growth rate for E Com, Financo considered a discounted cash flow analysis inappropriate for valuing E Com.

Model Reorg Valuation

Model Reorg Comparable Companies Analysis

Financo compared Model Reorg to a group of companies focused on the beauty industry and companies that distribute consumer products that we refer to as the Model Reorg Companies. The Model Reorg Companies are public companies that Financo deemed comparable to Model Reorg. Financo selected companies competing in the beauty industries, as well as consumer products distributors with net sales for the trailing twelve months publicly available at the time of the analysis between \$360 million and \$7,154 million.

The Model Reorg Comparable Companies are:

Beauty Companies

Alberto-Culver Company

Elizabeth Arden, Inc.

The Estee Lauder Companies, Inc.

Inter Parfums, Inc.

Revlon, Inc. and

Sally Beauty Holdings, Inc.

42

Consumer Products Distributors

AMCON Distributing Company

Core-Mark Holding Company, Inc. and

Nash Finch Co.

Financo considered the ratio of EV to EBITDA of the Model Reorg Comparable Companies to derive multiples that indicate the value public equity markets place on companies in a particular market segment. In order to perform this analysis, Financo compared financial information of Model Reorg with publicly available information for the Model Reorg Comparable Companies. Financo considered the entire range of multiples yielded by the Model Reorg Comparable Companies in establishing a more limited reference range of multiples which it used for its analysis. For this analysis, as well as other analyses, Financo examined publicly available information, as well as a range of estimates based on equity research reports.

The following table presents, based on closing prices as of December 20, 2007, the range of multiples for the Model Reorg Comparable Companies of EV divided by EBITDA:

	Reference
	Range of
	Multiples
9/30/07 EV/ EBITDA	8.0x - 10.5x
Projected 10/31/08 EV/ EBITDA	6.5x - 9.0 x

The comparables imply the following reference ranges for equity value of Model Reorg:

	Reference Range of
	Values
	(\$ in millions)
9/30/07 EV/ EBITDA	\$94.1 - \$191.3
Projected 10/31/08 EV/ EBITDA	\$60.8 - \$167.5

No company utilized in the public company comparables analysis as a comparison to Model Reorg is identical to Model Reorg. In evaluating the comparables, Financo made numerous assumptions with respect to the beauty and consumer product distribution industries performance and general economic conditions, many of which are beyond the control of Model Reorg. Mathematical analysis, such as determining the median, average or range, is not in itself a meaningful method of using comparable company data.

Model Reorg Comparable Transactions Analysis

Financo considered ratios of EV to sales and EBITDA to derive multiples that strategic and financial acquirers have been willing to pay for companies in comparable transactions. In order to perform this analysis, Financo reviewed a number of transactions, including those involving companies deemed by Financo to be comparable to Model Reorg based on financial performance, market focus and business model. Financo reviewed four comparable merger and acquisition transactions announced from April 29, 2002 through June 19, 2006 involving companies in the beauty and consumer products distribution business. For this analysis, as well as other analyses, Financo examined publicly available information, as well as information from Financo s proprietary database of published and confidential merger and acquisition transactions in the retail and consumer industries.

The Model Reorg comparable transactions consisted of the acquisitions of:

Sally Beauty Holdings, Inc. by Clayton, Dubilier & Rice, Inc.

Unilever Cosmetics International by Coty, Inc.

Cosmetic Essence by Onex Corp. and

Core-Mark International, Inc. by Fleming Companies Inc.

43

The following table presents, as of December 20, 2007, the reference range of multiples of adjusted price (defined as equity price plus total debt minus cash and cash equivalents) divided by the seller s EBITDA in the last reported twelve months prior to acquisition for the transactions listed above:

						Reference Range of
						Multiples
EV/ EBITDA						8.0x - 11.0x

These comparables imply the following reference range for equity value of Model Reorg:

Reference Range of Values
(\$ in millions)
EV/ EBITDA
\$94.1 - \$210.7

No transaction utilized as a comparable in the transaction comparables analysis is identical to the Merger. In evaluating the comparable transactions, Financo made numerous assumptions with respect to the beauty and consumer products distribution industry s performance and general economic conditions, many of which are beyond the control of Model Reorg. Mathematical analysis, such as determining the average, median or range, is not in itself a meaningful method of using comparable transaction data.

Model Reorg Discounted Cash Flow Analysis

Financo examined the value of Model Reorg on a standalone basis, based on projected free cash flow estimates that were derived from its discussions with management of Model Reorg and the Model Reorg Financial Projections. The free cash flow estimates were generated by applying financial projections from October 31, 2007 through October 31, 2010.

Based on a range of terminal EBITDA exit multiples from 7.5x to 10.5x and a range of discount rates from 7.8% to 8.8%, Financo calculated for Model Reorg implied total EVs ranging from \$336.4 million to \$462.8 million, and implied equity values ranging from \$119.7 million to \$246.0 million.

Combined Company Valuation

Combined Company Comparable Companies Analysis

Financo compared the combined company to the E Com Comparable Companies and the Model Reorg Companies, taken as a group, which we refer to as the Combined Comparable Companies. For this purpose, Financo performed the same analyses for the Combined Comparable Companies as a group as it did for the E Com Companies and the Model Reorg Companies in connection with its standalone analyses of E Com and Model Reorg.

The following table presents, based on closing prices as of December 20, 2007, the reference range of multiples for the Combined Comparable Companies of EV divided by EBITDA, and P/E:

	Reference Range of
	Multiples
9/30/07 EV/ EBITDA	8.0x - 10.5x
9/30/07 P/E	12.0x - 18.0x
Projected 10/31/08 EV/ EBITDA	6.5x - 9.0x
Projected 10/31/08 P/E	10.0x - 11.0x

44

The comparables imply the following reference ranges for per share value for the combined company:

	Reference Range of
	Values
9/30/07 EV/ EBITDA	\$16.59 - \$30.54
9/30/07 P/E	\$22.36 - \$33.53
Projected 10/31/08 EV/ EBITDA	\$12.79 - \$28.50
Projected 10/31/08 P/E	\$21.77 - \$34.83

No company utilized in the public company comparables analysis as a comparison to the combined company is identical to the combined company. In evaluating the comparables, Financo made numerous assumptions with respect to the performance of the specialty retailing, beauty and consumer product distribution industries and general economic conditions, many of which will be beyond the control of the combined company. Mathematical analysis, such as determining the median, average or range, is not in itself a meaningful method of using comparable company data.

Combined Company Comparable Transactions Analysis

Financo performed the same analysis for purpose of its valuation of the combined company, and reviewed the same comparable transactions taken as a group, as it did in connection with its valuations of E Com and Model Reorg on a standalone basis.

The Combined Company comparable transactions consisted of the acquisitions of:

Specialty Retailers

Factory Card & Party Outlet by Amscan Holdings, Inc.

United Retail Group by Red Cats USA, Inc.

Deb Shops, Inc. by Lee Equity Partners

Cutter & Buck, Inc. by New Wave Group AB

American Apparel by Endeavor Acquisition Corp

Golf Galaxy, Inc. by Dick s Sporting Goods, Inc.

Yankee Candle Co. Inc. by Madison Dearborn Partners Inc.

Sally Beauty Holdings, Inc. by Clayton, Dubilier & Rice, Inc.

The Body Shop International PLC by L Oreal SA

Pacific Sales Kitchen & Bath Centers, Inc. by Best Buy Co., Inc.

Party City Corporation by Berkshire Partners & Weston Presidio

OshKosh B Gosh Inc. by William Carter Co.

Brookstone Inc. by J.W. Childs Associates, Osim International and Temasek Holdings

Maurices Inc. by Dress Barn Inc.

Galyan s Trading Company, Inc. by Dick s Sporting Goods, Inc.

Loehmann s Holdings Inc. by Crescent Capital Investments, Inc.

InterTAN Inc. by Circuit City Stores Inc. and

The White House, Inc. by Chico s FAS Inc.

45

Beauty/Distribution Companies

Sally Beauty Holdings, Inc. by Clayton, Dubilier & Rice; Inc.

Unilever Cosmetics International by Coty, Inc.

Cosmetic Essence by Onex Corp. and

Core-Mark International, Inc. by Fleming Companies Inc.

The following table presents, as of December 20, 2007, the reference range of multiples of adjusted price (defined as equity price plus total debt minus cash and cash equivalents) divided by the seller s EBITDA in the last reported twelve months prior to acquisition for the transactions listed above:

Reference Range of Multiples
EV/ EBITDA 9.0x - 11.0x

These comparables imply the following reference range for per share value for the combined company:

Reference Range of Values
EV/ EBITDA \$22.17 - \$33.33

No transaction utilized as a comparable in the transaction comparables analysis is identical to the Merger. In evaluating the comparable transactions, Financo made numerous assumptions with respect to the specialty retailing and distribution industries performance and general economic conditions, many of which are beyond the control of the combined company. Mathematical analysis, such as determining the average, median or range, is not in itself a meaningful method of using comparable transaction data.

Combined Company Discounted Cash Flow Analysis

Financo examined the value of the combined company on a standalone basis, based on projected free cash flow estimates that were derived from the Financial Projections. The free cash flow estimates were generated by applying financial projections from October 31, 2007 through October 31, 2010.

Based on a range of terminal EBITDA exit multiples from 7.5x to 10.5x and a range of discount rates from 9.7% to 10.7%, Financo calculated for the combined company implied total EVs ranging from \$448.1 million to \$623.6 million, and implied equity values ranging from \$178.6 million to \$354.1 million.

The preparation of a fairness opinion is a complex process involving determinations as to the most appropriate and relevant methods of financial analysis and the application of these methods to the particular circumstances and, therefore, is not necessarily susceptible to partial analysis or summary description. Selecting portions of the analysis or the summary set forth above, without considering the analysis as a whole, could create an incomplete view of the processes underlying the opinion of Financo. In arriving at its fairness determination, Financo considered the results of all these constituent analyses and did not attribute any particular weight to any particular factor or analysis considered by it; rather, Financo made its determination as to fairness on the basis of its experience and professional judgment after considering the results of all such analyses. Certain Financo analyses are based upon forecasts of future results and are not necessarily indicative of actual future results, which may be significantly more or less favorable than suggested by such analyses. The foregoing summary does not purport to be a complete description of the analyses performed by Financo. Additionally, analyses relating to the value of businesses or securities are not appraisals. Accordingly, such analyses and estimates are inherently subject to substantial uncertainty.

Table of Contents

In performing its analyses, Financo made numerous assumptions with respect to industry performance and general business and economic conditions and other matters, many of which are beyond our control. The analyses performed by Financo are not necessarily indicative of actual values or actual future results, which may be significantly more or less favorable than suggested by such analyses. The Merger Consideration and other terms of the Merger Agreement were determined through arm s length negotiations between us and Model Reorg, and were approved by our Board of Directors subsequent to the determinations and recommendation of the Special Committee. In addition, Financo s opinion and presentation to the Special Committee was one of many factors taken into consideration by the Special Committee in making its decision to approve the Merger. Consequently, the Financo analyses as described above should not be viewed as determinative of the opinion of the Special Committee with respect to the value of E Com, Model Reorg or the combined company or whether the Special Committee would have been willing to agree to different consideration.

Based upon and subject to the qualifications and limitations set forth in its written opinion, a copy of which is attached as Annex II, Financo was of the opinion that, as of December 21, 2007, the Merger Consideration was fair, from a financial point of view, to holders of our common stock (other than those who owned, or whose affiliates owned, securities of Model Reorg).

Following the delivery of Financo s opinion and the execution of the Original Merger Agreement, both E Com and Model Reorg restated their historical financial statements and modified their earlier projections for the Special Committee, as described above in Background of the Merger. The Special Committee did not engage Financo to update its fairness opinion as a result of the First Amendment and, accordingly, Financo did not consider the updated projections that were prepared for the Special Committee in connection with the First Amendment.

Financo was retained to represent the Special Committee to give an opinion as to the fairness of the Merger Consideration, from a financial point of view, to the holders of our common stock (other than those who own, or whose affiliates own, securities of Model Reorg). Financo received an initial fee of \$100,000 in connection with its services upon engagement, a fee of \$125,000 upon delivery of valuation metrics to the Special Committee, a fee of \$50,000 upon delivery of updated valuation metrics as requested by the Special Committee, and a fee of \$225,000 upon advising the Special Committee that Financo was prepared to deliver its opinion, for a total of \$500,000. No portion of Financo s fee is contingent upon the consummation of the Merger.

Model Reorg Board and Shareholder Approval

The Board of Directors of Model Reorg approved the Merger Agreement and the Merger. By their execution of the Merger Agreement, the Model Reorg shareholders unanimously consented to the Merger Agreement and the Merger. This consent effectively waived their rights under New York law to dissent from the Merger and seek appraisal of the shares of Model Reorg capital stock they own.

Interests of Certain Persons in the Merger

In considering the recommendation of our Board to approve the issuance of the Merger Consideration and the other matters described in this Proxy Statement, our shareholders should be aware that certain members of our Board and certain of our officers have interests in the transaction that are different from, or in addition to, the interests of our other shareholders generally.

In particular, Stephen Nussdorf, Chairman of our Board of Directors, is a principal shareholder, director and executive officer of Model Reorg and will receive a proportionate share of the Merger Consideration in the Merger. Together with his brother, Glenn Nussdorf, Stephen Nussdorf currently beneficially owns in the aggregate approximately 36.4% of our outstanding common stock, or approximately 44.5% assuming conversion of the Convertible Note. After the Merger, Stephen and Glenn Nussdorf would beneficially own an aggregate of approximately 53.7% of our outstanding common stock, assuming conversion of the Convertible Note, and they would hold Warrants that would, upon exercise, result in their beneficially owning up to an aggregate of approximately 57.7% of our shares. In addition, although they are not shareholders of Model Reorg, Michael W. Katz, our President and Chief Executive Officer and also one of our directors, is a director and the Executive Vice President of Model Reorg and Raymond Piergiorgi, Model Reorg s Chief Operating Officer, has been consulting with E Commanagement on operations matters. However, Messrs. Nussdorf, Katz and Piergiorgi were not members of the Special Committee of independent directors that reviewed and recommended approval of the Merger.

47

Table of Contents

The Merger Agreement requires Model Reorg to restructure and refinance debt owed by Model Reorg and its subsidiaries to Quality King, a company that is wholly-owned by Stephen and Glenn Nussdorf and their sister, Arlene Nussdorf. The balance of principal and accrued interest at April 30, 2008 was \$76.9 million. The restructuring and refinancing are to be accomplished through a term loan of \$52.3 million from Glenn, Stephen and/or Arlene Nussdorf or trusts for the benefit of their families, which will be used in part to pay the Quality King Debt down to \$35 million. Including the Convertible Note, the combined company will be obligated to the Nussdorfs and their affiliates in the amount of up to \$92.3 million upon the closing of the Merger. See Affiliate Debt below for more information about the terms of the proposed related party debt restructuring.

The relationships among the shareholders and officers of Model Reorg, their affiliated companies, and E Com are discussed in more detail under Certain Relationships and Related Person Transactions beginning on page 63 below. As described in that section, other transactions involving affiliated parties are also expected to occur in connection with the Merger. In particular, Model Reorg and Quality King currently share in various operational arrangements under which each may benefit from economies of scale and additional convenience, and we expect those arrangements to continue after the Merger. In addition, Model Reorg subleases office and warehouse space from Quality King in a leased facility in Bellport, NY, where we expect a significant portion of our operations to be consolidated with Model Reorg superations.

Terms of the Merger Agreement

The following is a summary of selected provisions of the Merger Agreement. While we believe this description covers the material terms of the Merger Agreement, it may not contain all of the information that is important to you and it is qualified in its entirety by reference to the Merger Agreement. The Merger Agreement is attached as Annex I to this Proxy Statement and is considered part of this document. The forms of Warrant and Registration Rights Agreement are included as exhibits to the Merger Agreement in Annex I. We urge you to read the Merger Agreement and its exhibits carefully in their entirety for a more complete understanding of the Merger.

General

At the closing under the Merger Agreement, Model Reorg will merge into Merger Sub, and the surviving entity (which will own Model Reorg s assets, assume its obligations, and conduct its operations) will continue as our wholly-owned subsidiary. The outstanding capital stock of Model Reorg will be canceled, and the Model Reorg shareholders will receive the Merger Consideration in exchange. The closing of the Merger is agreed to occur no later than two business days following the satisfaction or waiver of all of the conditions to the Merger, or at another such time as we and Model Reorg may agree in writing (the Closing Date). However, because the Merger is subject to a number of conditions, we cannot predict exactly when the closing will occur or if it will occur at all.

Merger Consideration

The Merger Consideration issuable to the Model Reorg shareholders in the Merger will consist of an aggregate of 5,900,000 shares of our common stock and Warrants to acquire an aggregate of 1,500,000 shares of our common stock at an exercise price of \$23.94 per share. The Warrants may not be exercised until the third anniversary of the Closing Date and will then remain exercisable until the tenth anniversary. They are not transferable with certain limited exceptions.

The Merger Consideration consists of our securities, so its value fluctuates with changes in the trading price of our common stock on NASDAQ. There is no trading market for the Warrants; accordingly, we calculate their value using the Black-Scholes pricing model, which involves making certain assumptions about the anticipated volatility of our stock price, future interest rates, and other matters. Since these assumptions are inherently uncertain, they also contribute to fluctuations in the value of the Merger Consideration. Based on the respective closing market prices of our common stock and those assumptions as of the respective dates, we determined that,

48

as of December 20, 2007 (the trading day before the date of the Original Merger Agreement), the value of the Warrants was \$20,340,000 and the total value of the Merger Consideration was \$161,586,000 and that, as of June 12, 2008, the value of the Warrants was \$15,600,000 and the total value of the Merger Consideration was \$138,969,000. These values fall within the reference ranges for equity value of Model Reorg provided by Financo, as described above in Opinion of Financo Model Reorg Valuation. The value of the Merger Consideration actually received by the Model Reorg shareholders in the Merger will not be determined until the Merger closes. You should note that the Special Committee s review of the fairness of the proposed Merger Consideration focused primarily on the estimated relative values of the two companies, rather than on the absolute value of the Merger Consideration as of any particular time.

Representations and Warranties

The Merger Agreement contains customary representations and warranties by us and by Model Reorg relating to, among other things, corporate organization and similar corporate matters, capitalization, authority to enter into the Merger Agreement and transactions contemplated thereby, and enforceability of the Merger Agreement. In addition, we represented and warranted to Model Reorg that the Special Committee had recommended the adoption of the Merger Agreement to our Board of Directors, that the Special Committee had received the opinion of Financo as to the fairness of the Merger Consideration, from a financial point of view, to our shareholders (other than those who own, or whose affiliates own, securities of Model Reorg), and that our Board of Directors had duly adopted resolutions:

determining that the terms of the Merger are fair from a financial point of view to the holders of our common stock (other than our shareholders who own, or whose affiliates own, securities of Model Reorg),

adopting the Merger Agreement and declaring advisable the Merger Agreement and the related transactions, and

recommending that our shareholders approve the issuance of the shares of our common stock and the Warrants pursuant to the Merger and the issuance of our common stock upon exercise of the Warrants, subject to the right of our Board to withdraw, modify or amend such recommendation to the extent that the Board determines, in good faith, that failure to take such action would be reasonably likely to result in a breach of its fiduciary duties to our shareholders under applicable law.

In addition, the Merger Agreement includes representations and warranties by Model Reorg to us as to:

approval by the board of directors and shareholders of Model Reorg;

whether the Merger or the execution of the Merger Agreement will result in any conflict with or violation of any material agreements of Model Reorg and its subsidiaries;

the preparation and accuracy of the financial statements of Model Reorg and its subsidiaries;

any undisclosed liabilities of Model Reorg and its subsidiaries;

the accounts receivables and inventories of Model Reorg and its subsidiaries;

title to the assets of Model Reorg and its subsidiaries;

material changes or events affecting Model Reorg and its subsidiaries;

litigation matters affecting Model Reorg and its subsidiaries;

compliance with legal requirements by Model Reorg and its subsidiaries;

filing of tax returns and payment of taxes by Model Reorg and its subsidiaries;

labor and employee benefits relating to the employees of Model Reorg and its subsidiaries;

environmental matters affecting Model Reorg and its subsidiaries;

49

Table of Contents

insurance maintained by Model Reorg and its subsidiaries;

brokerage, financial advisor s or finder s fees or commissions payable by Model Reorg and its subsidiaries in connection with the Merger;

material contracts to which Model Reorg and its subsidiaries are parties and any violation, default or breach of those contracts;

leased property of Model Reorg and its subsidiaries;

intellectual property of Model Reorg and its subsidiaries; and

transactions with affiliates of Model Reorg.

The representations and warranties of Model Reorg are, in many respects, qualified by materiality and limited to the knowledge of the person making the representation and warranty, but their accuracy forms the basis of one of the conditions to our obligation to complete the Merger. Please note, however, that these representations and warranties were made only for purposes of the Merger Agreement and as of specific dates, were solely for the benefit of the parties thereto, and are subject to limitations agreed to between the parties, including that they are qualified by disclosures between the parties that are not included with this Proxy Statement. Accordingly, investors and third parties should not rely on these representations and warranties as independent characterizations of the actual state of facts at the time they were made or otherwise but should consider them together with the other information in this Proxy Statement or that we have disclosed in other filings with the SEC.

The representations and warranties of Model Reorg will survive the Merger until thirty days following the filing of our first Annual Report on Form 10-K with the SEC following the Merger that includes audited financial statements that reflect at least six months of post-Merger operations of the combined company. In addition, Model Reorg s representations and warranties as to taxes will survive until the respective statutes of limitations as to taxes expire.

Covenants; Conduct of Business Pending the Merger

We and Model Reorg each agreed, until the closing of the Merger, to conduct our respective businesses in the ordinary course in accordance with past practices, to preserve current material relationships with customers, vendors, suppliers and other persons, and to avoid taking any action that would cause any of our respective representations and warranties to be untrue in any material respect. Subject to certain limited exceptions, Model Reorg agreed that it will not, during the period before the closing of the Merger:

change its authorized, issued or outstanding shares of capital stock or issue any options, warrants, convertible or exchangeable securities or other rights that may result in the issuance of shares of capital stock, voting securities, or equity interests of Model Reorg;

effect any event or change that would reasonably be expected to have a material adverse effect on its business;

sell, transfer, distribute, abandon or otherwise dispose of, or mortgage, pledge or impose any encumbrance on, any property (including real estate) or other of its assets and those of its subsidiaries that are material to Model Reorg and its subsidiaries taken as a whole, except for (i) sales of inventory in the ordinary course of business consistent with past practice or (ii) dispositions of obsolete or worthless assets in the ordinary course of business consistent with past practice;

declare, set aside or pay any dividend or other distribution (whether in cash, stock or property) with respect to any class of its capital stock or that of its subsidiaries (other than dividends or other distributions paid to Model Reorg by its subsidiaries), or repurchase, redeem or otherwise acquire any capital stock or other securities of, or other ownership interest in, Model Reorg or any of its subsidiaries;

50

Table of Contents

split, combine or reclassify any of its capital stock;

change its financial or tax accounting methods, principles or practices, except insofar as may have been required by a change in GAAP or applicable law;

make a tax election or revoke or change any material tax election inconsistent with past practices or the settle or compromise any material tax liability of Model Reorg or any of its subsidiaries;

grant to any director or employee any award or bonus or other compensation, except to the extent accrued on its balance sheet;

grant to any employee an increase in (or acceleration of vesting or payment of) severance or termination pay, except as required under any employment, severance or termination agreements as previously disclosed to us;

enter into any (or amend in any material respect of any existing) employment, severance or termination agreement with any employee;

enter into any consulting agreement;

accelerate the vesting of any option issued by Model Reorg or any of its subsidiaries to acquire its capital stock;

make any capital expenditure, taken as a whole, in excess of \$1.5 million individually or in the aggregate;

establish, amend or terminate any collective bargaining agreement;

accelerate any accounts receivable other than in the ordinary course of business consistent with past practice;

generally delay any payments by Model Reorg or any of its subsidiaries to vendors or others to whom Model Reorg owes payments (except for disputed payments);

make loans to any persons other than in the ordinary course of business consistent with past practice or loans to any officers or directors of us or any of our subsidiaries;

settle any action, complaint, claim, petition, investigation, suit or other proceeding, whether civil or criminal, in law or in equity, before any arbitrator or governmental entity, to which it or any of its subsidiaries are a party or any of their respective properties is subject;

accelerate the payment, right to payment or vesting under any benefit, retirement, profit sharing or deferred compensation plan or other compensation arrangement of Model Reorg or any of its subsidiaries;

incur any indebtedness in an amount in excess of \$500,000 in the aggregate;

grant any license or sublicense of any rights under or with respect to any material intellectual property;

establish, amend or contribute to any pension, retirement, profit sharing, stock bonus, multi-employer or other benefit plan covering any of the current or former employees, officers, directors or consultants, except as required by law or pursuant to such benefit plan;

establish, enter into, financially commit, contribute to, amend the terms of, or terminate, any joint venture, partnership agreement or similar arrangement or any contract involving a sharing of profits, losses, business or opportunities with any other person; or

enter into any agreement or commitment to do any of the foregoing.

51

No Solicitation

The Merger Agreement provides that, except as described below, we and our subsidiaries will not, and will not authorize any of our agents to, directly or indirectly, encourage, solicit, initiate, or take other action to facilitate any inquiries or the making of any proposal that constitutes or may reasonably be expected to lead to an acquisition proposal (as defined below), participate in any way in any discussions or negotiations relating to or in furtherance of an acquisition proposal, or accept any acquisition proposal, or enter into any agreement, arrangement or understanding to abandon, terminate or fail to consummate the Merger.

An acquisition proposal is any offer to acquire all or any substantial part of our business or properties, or the capital stock or membership interests of us or any of our subsidiaries, whether by merger, consolidation, sale of assets, tender offer, or similar transaction or series of transactions involving us or any of our subsidiaries.

However, in the exercise of and as required by its fiduciary duties, our Board of Directors, acting in good faith, may:

furnish information (including, without limitation, confidential information) concerning us to a third person who makes an unsolicited request for such information for the purpose of making an acquisition proposal, and

engage in discussions or negotiations with a third party in making an acquisition proposal that our Board believes is reasonably capable of being consummated;

provided that we notify Model Reorg in writing of such request for information or acquisition proposal, including the identity of the third party and the principal financial terms and conditions of such proposal. Our Board of Directors is entitled to withdraw or modify its recommendation of the Merger to our shareholders and approve and recommend to its shareholders an acquisition proposal that our Board of Directors reasonably believes is likely to be superior from a financial point of view to the Merger and is capable of being consummated in a reasonable period of time.

Other Agreements

Under the Merger Agreement, we and Model Reorg have each agreed to use our commercially reasonable efforts to:

file or otherwise submit all applications, notices, reports and other documents reasonably required to be filed with a governmental entity with respect to the Merger;

file listing applications with NASDAQ with respect to the shares to be issued pursuant to the Merger and upon exercise of the Warrants:

take all actions necessary to complete the Merger;

take all actions necessary to ensure that no state takeover or similar law is applicable to the Merger; and

obtain all consents, approvals or waivers reasonably required in connection with the transactions contemplated by the Merger Agreement.

We and Model Reorg also agreed in the Merger Agreement to consult and agree with each other about any public statement either will make concerning the Merger, subject to certain exceptions, and to provide each other with reasonable access to our respective officers, employees, representatives, books and records and facilities.

The Merger Agreement requires Merger Sub, as the surviving entity of the Merger, for a period of six years after the Merger, to indemnify each person, who at the time of the Merger, was a director, officer or employee of Model Reorg to the fullest extent permitted under its charter, the Business Corporation Law of the State of New York and the Delaware Limited Liability Company Act.

52

Table of Contents

Between the date of the execution of the Merger Agreement and the completion of the Merger, Model Reorg is required to prepare and deliver monthly and other financial statements to us.

In addition, the Model Reorg shareholders agreed with us that, even if permitted to do so under Florida Law, they would not effect a short form merger of us with a company they control while their combined ownership of our outstanding shares was less than 90% of our outstanding common stock.

The Model Reorg shareholders who currently own shares of our common stock agreed in the Merger Agreement to vote those shares in favor of the matters for which approval of our shareholders is sought in this Proxy Statement.

Conditions to the Completion of the Merger

Under the Merger Agreement, as amended by the First Amendment, each party s obligation to complete the Merger is subject to the satisfaction or waiver by each of the parties, at or before the Merger, of various conditions, including the following:

our shareholders must have approved the issuance of the Merger Consideration and the issuance of shares of common stock upon exercise of the Warrants and the amendment to our Articles of Incorporation increasing the authorized shares of common stock;

no law, injunction, judgment or ruling enacted, promulgated, issued, entered, amended or enforced by any governmental entity shall be in effect enjoining, restraining, preventing or prohibiting consummation of the Merger or making the consummation of the Merger illegal;

the effectiveness of any listing applications with the NASDAQ and under state securities laws;

asset based loans in amounts and on terms substantially similar to the Senior Credit Facility contemplated in the May 16, 2008 commitment described in Post-Merger Financing shall have been obtained by us and/or Merger Sub and funds shall have been advanced and/or be available thereunder; and

the restructuring and refinancing of the obligations of Model Reorg to Quality King and the subordinated loan from certain Nussdorf family members or family trusts, as described below in Affiliate Debt, shall have been put in place In addition, the obligation of Model Reorg to complete the Merger is further subject to the satisfaction or waiver by Model Reorg of the following conditions:

all our representations and warranties in the Merger Agreement shall be true and correct on the date of the Merger Agreement and on the Closing Date of the Merger with the same force and effect as if made on the date on which the Merger is to be completed or, if such representations and warranties address matters as of a particular date, then as of that particular date, except where the failure of these representations and warranties to be true and correct, disregarding any materiality qualifications, individually or in the aggregate, would not reasonably be expected to have a material adverse effect on us, and we shall have delivered an officer s certificate to this effect;

we shall have performed or complied in all material respects with all covenants and obligations required to be performed or complied with by us on or before the closing of the Merger, and we shall have delivered an officer s certificate to this effect;

we and Merger Sub shall have delivered all consents necessary or advisable to be obtained for the closing of the Merger except where the failure to do so is not reasonably expected to have a material adverse effect on the combined company;

we shall have executed and delivered a Registration Rights Agreement with the Model Reorg shareholders providing piggyback registration rights as further described below in Terms of the Merger Agreement Registration Rights;

53

Table of Contents

we shall have executed and delivered the Warrants; and

there shall have been no effect, event or change that, with certain exceptions, has a material adverse effect on our business, operations, assets, liabilities, results of operations, prospects, or financial condition, taken as a whole with our subsidiaries, since December 21, 2007, the date of the Original Merger Agreement.

In addition, our obligation to complete the Merger is further subject to the satisfaction or our waiver of the following conditions:

all representations and warranties of Model Reorg in the Merger Agreement being true and correct as of December 21, 2007 and as of the Closing Date of the Merger with the same force and effect as if made on the date on which the Merger is to be completed or, if such representations and warranties address matters as of a particular date, then as of that particular date, except where the failure of these representations and warranties to be true and correct, disregarding any materiality qualifications, individually or in the aggregate, would not reasonably be expected to have a material adverse effect on Model Reorg, and Model Reorg shall have delivered an officer s certificate to this effect:

Model Reorg shall have performed or complied in all material respects with all covenants and obligations required to be performed or complied with by Model Reorg on or before the closing of the Merger and Model Reorg shall have delivered an officer s certificate to this effect;

Model Reorg shall have delivered the consents required under the Merger Agreement for the closing of the Merger, including those from third parties, governmental entities, and landlords;

Model Reorg shall, and shall cause Quality King to, have executed and delivered a Services Agreement on the terms described in Certain Relationships and Related Person Transactions; and

Model Reorg shall have caused each of its shareholders to execute and deliver a certificate of representations as described in the Merger Agreement.

We and Model Reorg each have the right to waive compliance by the other with any of the agreements contained in the Merger Agreement or the other s conditions, and our Board may or may not waive any particular condition in its discretion. Our Board has made no determination about whether or not it would waive any particular condition. It would make such a decision, if necessary, based on the best interests of our shareholders in the circumstances existing at the time.

Termination

The Merger Agreement may be terminated at any time before the completion of the Merger, whether before or after the required approval of our shareholders has been obtained, as set forth below:

by mutual written consent of us and Model Reorg, duly authorized by our respective boards of directors;

by either us or Model Reorg, if:

the Merger has not been completed by August 31, 2008, but this right to terminate the Merger Agreement will not be available to any party whose action or failure to act has been the cause of, or resulted in, the failure of the Merger to be completed by August 31, 2008;

a law prohibiting the Merger or making the Merger illegal has been enacted or an injunction, judgment, order, decree or ruling or any other action, in each case, permanently enjoining, restraining, preventing or prohibiting the Merger has become final and non-appealable; provided, that neither party may terminate the Merger if the injunction, judgment, order, decree or ruling was primarily due to the failure of such party to perform any of its obligations under the Merger Agreement;

Table of Contents

we enter into an agreement for an acquisition proposal that our Board of Directors reasonably believes is likely to be superior from a financial point of view to the Merger and is capable of being consummated in a reasonable time; or

our shareholders fail to approve the issuance of the shares of our common stock and the Warrants and the amendment of our charter to increase our authorized shares of common stock at the Special Meeting;

provided, however, that neither party may terminate this Agreement if its failure (or that of any of its affiliates) to fulfill any obligation or condition under the Merger Agreement has been the cause of, or resulted in, the failure to obtain approval;

by us, if:

the representations and warranties of Model Reorg set forth in the Merger Agreement shall not be true and correct and as a result the condition described above at Terms of the Merger Conditions to the Completion of the Merger with respect to representations and warranties of Model Reorg will not be satisfied; or

before the Closing Date, Model Reorg has breached or failed in any material respect to perform or comply with the Merger Agreement and such breach has not been cured within twenty business days after Model Reorg has received written notice of such breach; or

by Model Reorg, if:

our Board or any committee thereof shall have withdrawn or modified its approval or recommendation of the Merger;

our representations and warranties of set forth in the Merger Agreement shall not be true and correct and, as a result, the condition described above at Terms of the Merger Conditions to the Completion of the Merger with respect to our representations and warranties will not be satisfied; or

before the Closing Date, we have breached or failed in any material respect to perform or comply with the Merger Agreement and such breach has not been cured within twenty business days after we have received written notice of such breach.

Escrow and Indemnification

Pursuant to the indemnification provisions in the Merger Agreement, we are entitled to indemnification for individual losses and expenses exceeding \$10,000 that exceed \$1 million in the aggregate that are sustained in connection with any breach of Model Reorg's representations, warranties and covenants. We will retain and deposit into escrow on the Closing Date 295,000 shares of the total Merger Consideration. Except with respect to claims for taxes, these escrowed shares of our common stock are the sole and exclusive remedy for any losses arising out of claims related to the Merger Agreement. For this purpose, the escrowed shares will be valued at \$23.94 per share, which is equal to the average closing price of our common stock on NASDAQ for the 50 trading days ending immediately before the date of execution of the Merger Agreement. On the thirtieth day following the filing of our first Annual Report on Form 10-K with the SEC that includes audited financial statements that reflect at least six months of the post-Merger operations of the combined company, we will release any remaining escrow shares, subject to any amounts then in dispute relating to indemnification, to the Model Reorg shareholders based on their allocation of the Merger Consideration.

The Model Reorg stockholders have agreed to indemnify us against any taxes for periods before the closing of the Merger not arising from the Merger transactions that are not paid before the Merger or were not accrued for in Model Reorg s September 30, 2007 balance sheet. Any such indemnification must be paid in cash.

Table of Contents

Amendment

The Merger Agreement may be amended by written agreement of the parties at any time before the effective time of the Merger, except that after the Merger Agreement has been adopted by our shareholders, no amendment that, by law, requires further approval by our shareholders may be made without such further approval.

The Warrants

At the closing of the Merger, we will issue Warrants to purchase an aggregate of 1,500,000 shares of our common stock to the Model Reorg shareholders as part of the Merger Consideration. Each Warrant will entitle the holder to purchase shares of our common stock at an exercise price of \$23.94 per share, payable in cash. The number of shares issuable upon exercise of the Warrants and the exercise price may be adjusted from time to time to reflect stock splits, stock dividends, combinations, reclassifications, mergers and other recapitalization transactions affecting our common stock.

The Warrants will not be exercisable until the third anniversary of the closing of the Merger, provided that the Warrants will become exercisable earlier in the event of any (a) merger, share exchange or consolidation of the combined company, (b) sale of all or substantially all of the combined company s assets, or (c) tender offer or exchange offer pursuant to which not less than 90% of the then outstanding shares of our common stock are tendered or exchanged for other securities, cash or property, other than any such transaction that does not result in a transfer of majority ownership of our combined voting power or assets. Once exercisable, the Warrants will remain exercisable until the tenth anniversary. The holders of Warrants may not sell or transfer them without our consent except upon death or, during the holder s lifetime, to members of the holder s family for estate planning purposes.

The foregoing description of the terms of the Warrants is qualified in its entirety by reference to the full text of the form of Warrant, which appears as Exhibit A to the Merger Agreement in Annex I to this Proxy Statement.

Registration Rights

At the closing of the Merger, we and the Model Reorg shareholders will execute a piggyback Registration Rights Agreement under which the former Model Reorg shareholders will have the right to require us, at our expense, to include the shares received in the Merger (but not those issuable upon exercise of the Warrants), and any shares of common stock issued in respect thereof, in certain registration statements that we may file with the SEC under the Securities Act of 1933, as amended. The shareholders registration rights do not apply to registration statements we file to sell securities (a) issued pursuant to bona fide compensatory arrangements or as consideration for certain acquisitions of businesses or assets or (b) that are solely debt securities or nonconvertible preferred stock, including those issuable upon conversion of such debt securities.

The Registration Rights Agreement contains customary provisions allocating rights and responsibilities in the event the registration involves an underwritten offering and obligating us and the selling shareholders to indemnify each other against certain liabilities arising from any registration of securities. The Registration Rights Agreement limits our ability to grant superior registration rights to others in the future, except that we may grant registration rights to persons providing financing and/or selling businesses or assets to us as long as any such rights are of equal priority and pro rata with the registration rights of the former Model Reorg shareholders.

The right of any former Model Reorg shareholder to request registration pursuant to the Registration Rights Agreement will terminate when that shareholder holds less than the greater of (a) 1% of the outstanding shares of our common stock or (b) the current weekly trading volume of our common stock and is not an affiliate (a control person) with respect to us.

The foregoing description of the Registration Rights Agreement is qualified in its entirety by reference to the full text of that agreement, which appears as Exhibit B to the Merger Agreement in Annex I to this Proxy Statement.

Table of Contents

Post-Merger Financing

In May 2008, we obtained a commitment (the Financing Commitment) from an institutional lender to provide senior credit facilities to us and our subsidiaries in an aggregate amount of up to \$250,000,000 (the Senior Credit Facility). The Financing Commitment is scheduled to expire on August 16, 2008. It is a condition to closing under the Merger Agreement that funds be available under asset-based loans in the amount and on terms substantially similar to the Financing Commitment. This condition may only be waived by agreement of all parties, including us and Model Reorg. As a further condition to the closing of the Senior Credit Facilities and the Merger, we will be required to refinance a portion of the Model Reorg indebtedness to Quality King as described in Affiliate Debt below. Proceeds of the Senior Credit Facility will be used to pay amounts incurred in connection with the Merger and our and Model Reorg s existing senior credit facilities, and for working capital and general corporate purposes.

The Senior Credit Facility will consist of a revolving credit facility of up to \$250 million, which will include revolving loans that may be drawn, repaid and reborrowed up to the amount available under a borrowing base calculated with reference to a specified percentage of the combined company s eligible accounts and a specified percentage of the combined company s eligible inventory from time to time. The proposed Senior Credit Facility would also include a sub-limit of \$25 million for letters of credit and a sub-limit of \$12.5 million for swingline loans (that is, same-day loans from the lead or agent bank). The Financing Commitment provides for the Senior Credit Facility to have a term of three years and for there to be a prepayment and early termination fee of 1.00% of the principal amount pre-paid during the first year after the closing of the Senior Credit Facility and 0.50% of the principal amount pre-paid during the second year.

Interest under the proposed Senior Credit Facility will, at our election unless an Event of Default exists, be at either (i) the higher of The Wall Street Journal corporate base rate or the federal funds rate plus 0.50% (the Base Rate) or (ii) the applicable London interbank offered rate (LIBOR), plus in each case, specified margins. Such margins are determined based upon our excess availability (that is, at any time, an amount equal to (a) the lesser of the aggregate revolving commitments and the borrowing base at such time minus (b) the revolving exposure of all lenders) from time to time. The Financing Commitment provides for the interest rate margins (i) to be initially set at 2.50% per annum for LIBOR borrowings and 1.25% for Base Rate borrowings and (ii) after the first fiscal quarter ending at least six months after the closing of the Senior Credit Facility, to range from 2.25% to 2.75% for LIBOR borrowings and from 1.00% to 1.50% for Base Rate borrowings. Pursuant to the Financing Commitment, the lender reserves the right to increase the interest rate margins by up to 0.75% per annum if necessary in order to achieve a successful syndication of the Senior Credit Facility. We will also be required to pay fees based on the unused amount of the Revolving Credit Facility and the outstanding amount of letters of credit under that facility.

All obligations of the combined company under the Senior Credit Facility and under any interest rate protection or other hedging arrangements will be secured by a first priority perfected security interests in all our existing and after-acquired personal property and owned real property, including, without limitation, 100% (or, in the case of excluded foreign subsidiaries, 66%) of the outstanding equity interests in their subsidiaries.

The Senior Credit Facility will be subject to customary conditions, will require that we have \$20 million in unused availability at closing, and will require us to make customary representations and warranties at the closing date and upon each borrowing. We will be required to comply with certain customary financial covenants, including maximum total leverage ratio, minimum fixed charge coverage ratio and minimum inventory turnover, in each case with levels to be determined. The facility will also entail reporting requirements and impose affirmative and negative covenants, including without limitation, restrictions on additional debt, liens, distributions, mergers, acquisitions and other investments.

The Senior Credit Facility will also be subject to standard events of default, including the following: failure to pay principal, interest or any other amount when due; breach of representations and warranties or covenants; cross-default to other indebtedness; failure to satisfy or stay execution of judgments; bankruptcy or insolvency;

57

and change of ownership or control. Upon an event of default under the Senior Credit Facility, the agent for the lenders will have the right to terminate any commitment to lend under the Senior Credit Facility, the right to accelerate the amounts outstanding under the Senior Credit Facility and all rights available to a secured creditor under applicable law, including the right to foreclose on the collateral.

Affiliate Debt

The Merger Agreement, as amended by the First Amendment, also requires Model Reorg to repay a portion of the debt owed by Model Reorg and its subsidiaries to Quality King, modify the terms of the remaining debt due to Quality King, and obtain a new subordinated loan from Glenn, Stephen and/or Arlene Nussdorf or trusts for the benefit of their families (collectively, the Nussdorf Lenders). The debt to Quality King relates to advances by Quality King to Model Reorg and its subsidiaries for working capital. The balance of principal and accrued interest at April 30, 2008 was \$76.9 million. At the closing of the Merger, the Nussdorf Lenders will loan approximately \$52.3 million to Model Reorg, which will use a portion of the loan proceeds to pay down the principal amount of the Quality King debt to \$35 million.

Both the new Nussdorf loan and the remaining Quality King debt will be subordinated to the new Senior Credit Facility described in Post-Merger Financing and neither will be secured by the combined company s assets. Subject to the requirements of the Senior Credit Facility, the maturity date of the new Nussdorf loan will be at least six months beyond the maturity date of the Senior Credit Facility and the remaining Quality King debt will amortize quarterly beginning in January 2009 at the rate of \$2.5 million per quarter, with the balance due six months following the scheduled maturity date of the Senior Credit Facility. The new Nussdorf loan will bear interest at a rate equal to 2% over the rate in effect from time to time under the revolving credit facility included in the Senior Credit Facility, and the remaining Quality King debt will bear interest at a rate equal to 1% over the rate in effect from time to time under that revolving credit facility. No interest payments will be required on either the Nussdorf loan or the Quality King debt until January 2009. The First Amendment eliminated the provisions of the Original Merger Agreement for satisfying a portion of the Quality King debt by the consignment of merchandise.

No Appraisal or Preemptive Rights

Our shareholders will not be entitled to demand appraisal of, or exercise any dissenters—rights under Florida law with respect to, their shares in connection with the actions to be taken at the Special Meeting. Neither Florida law nor our Articles of Incorporation provide preemptive rights to our shareholders with respect to the shares or Warrants we will issue to the Model Reorg shareholders in the Merger.

Management Following the Merger

Our Board of Directors is not expected to change at the time of the Merger. Most members of our senior management will continue in their current roles. In addition, Model Reorg officers will join our management team as we integrate the Model Reorg business into what will be a significantly larger company. We do not expect to enter into any executive employment agreements in connection with the Merger, nor that any of the executive officers of either company will terminate their employment as a result of the Merger. The following are expected to be our directors and executive officers immediately following the Merger:

Name

Stephen L. Nussdorf Michael W. Katz

Joseph Bouhadana(1)(2)(3)

Paul Garfinkle(1)

Carole Ann Taylor(1)(2)(3)

Donna Dellomo

Raymond Piergiorgi

Donovan Chin

Age Positio

- 57 Chairman of the Board of Directors
- 60 President, Chief Executive Officer and Director
- 37 Director
- 67 Director; Chairman of the Audit Committee
- 60 Director
- 43 Chief Financial Officer
- 50 Chief Operating Officer
- 41 Chief Financial Officer, Perfumania, Inc.

58

- Member of the Audit Committee
- (2) Member of the Compensation Committee

(3) Member of the Stock Option Committee

Messrs. Bouhadana and Garfinkle and Ms. Taylor are independent directors in accordance with the requirements of the NASDAQ Stock Market and (with respect to our Audit Committee) SEC regulations under the Sarbanes-Oxley Act of 2002. After the Merger, we will be a controlled corporation under the NASDAQ rules. Accordingly, although there is no present intention to reduce the number of independent directors, we will no longer be required to have a board of directors with a majority of independent directors, to have executive compensation determined by an independent compensation committee, or to have our directors nominated by a majority of the independent directors. This could increase the extent to which the principal shareholders are able to control our operations. We will, however, still be required to have an audit committee consisting of three members, all of whom are independent under NASDAQ and SEC requirements.

Mr. Garfinkle was a partner of BDO Seidman, LLP, Model Reorg s independent registered public accounting firm, until his retirement in 2000, and he receives retirement benefits from that firm. Our management has not yet engaged an accounting firm to audit the combined company s financial statements after fiscal 2007. However, under SEC rules, BDO Seidman, LLP would not be independent with respect to us unless Mr. Garfinkle resigns from our Board upon the closing of the Merger. No action has been taken with respect to any such resignation.

Material U.S. Federal Income Tax Consequences of the Merger to E Com and Its Shareholders

We expect the Merger to qualify as a tax-free reorganization under Section 368 of the Internal Revenue Code. As such, the Merger will not be taxable to us, Merger Sub or Model Reorg for U.S. federal income tax purposes. Because our shareholders will not be exchanging their shares in the Merger, the closing of the Merger and related transactions under the Merger Agreement will not have U.S. federal income tax consequences to our shareholders. The Merger will, when considered with other acquisitions of shares of our common stock occurring before the Merger, result in an ownership change of E Com, as that term is described under Section 382 of the Internal Revenue Code. As a result, our ability to use our net operating losses from pre-Merger periods to reduce taxable income in post-Merger periods would be subject to an annual limitation. In general, the annual limitation would equal our value immediately before the ownership change multiplied by the federal long-term tax-exempt interest rate.

Anticipated Accounting Treatment

We expect to treat the Merger as a reverse acquisition under the purchase method of accounting in accordance with GAAP. For accounting purposes, Model Reorg is considered to be acquiring us in the Merger. Accordingly, the Merger Consideration will be allocated among the fair values of our assets and liabilities, based upon a valuation that has not yet been determined. E Com s results will be reflected in the results of the combined company from the date of the Merger going forward. Historical financial condition and results of operations shown for comparative purposes in periodic filings subsequent to the completion of the transaction will reflect Model Reorg. However, the combined company will continue to use our fiscal year end, the Saturday closest to January 31, which is conventional for retail businesses.

NASDAQ Capital Market Listing

Our common stock is currently listed on the NASDAQ Capital Market and we expect it to continue to be so listed following the Merger. However, because the Merger will be deemed a reverse acquisition, we may be required to reapply for listing with NASDAQ to continue that listing. We have received a preliminary, oral notice that we will not be required to reapply. If we were in fact required to reapply, and if the application were accepted, which we expect, our common stock would continue to be traded on the NASDAQ Capital Market

Table of Contents

following the closing of the Merger. If it were not accepted, the Merger would not occur without a waiver from the parties to the Merger Agreement, since it is a condition to closing the Merger that E Com continue to be listed on the NASDAQ Capital Market. Our Board has made no determination as to whether it would waive that condition. It would make such a decision, if necessary, based on what it believes to be the best interests of our shareholders in the circumstances existing at the time. Even if our Board waived this condition, Model Reorg might decide not to waive it, in which case the Merger would not occur.

Regulatory Matters

Under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations promulgated thereunder (the HSR Act), the Merger could not be completed until we and Model Reorg had made certain filings with the Federal Trade Commission and the United States Department of Justice and the applicable waiting period under the HSR Act had expired or been terminated. We and Model Reorg filed the required notification on January 24, 2008 and, on February 1, 2008, the Federal Trade Commission granted early termination of the waiting period.

Status of Shares Issued in Merger

The issuance of shares of our common stock and Warrants to the shareholders of Model Reorg in the Merger, and the issuance of shares of our common stock upon exercise of the Warrants, will not be registered under the Securities Act of 1933, as amended, in reliance on an exemption for the issuance of shares not involving a public offering. Accordingly, SEC rules provide that the shares may not be resold in the public market for at least six months after the Merger or the Warrant exercise, as the case may be. Thereafter, (i) any sales of those shares in the public market by a shareholder who is a director, executive officer or controlling shareholder of E Com will be subject to public availability of certain financial and other information about us, and the volume and manner of sale limitations in SEC Rule 144, and (ii) sales by non-affiliates will be subject for six months to our having made certain financial and other information publicly available, but to no further restrictions thereafter.

Recommendations of the Special Committee and our Board of Directors

On December 21, 2007, the Special Committee unanimously approved the Merger and the related transactions and recommended the adoption of the Original Merger Agreement to our Board of Directors, subject to the Special Committee s right to withdraw, modify or amend such recommendation if the Special Committee determines, in good faith, that failure to take such action would be reasonably likely to result in a breach of its fiduciary duties to our shareholders under applicable law.

Having received the recommendation of the Special Committee, for reasons including those described above, on December 21, 2007, our Board unanimously:

determined that the terms of the Merger are fair from a financial point of view to the holders of our common stock (other than shareholders who own, or whose affiliates own, securities of Model Reorg);

adopted the Original Merger Agreement and declared it and the related transactions to be advisable; and

recommended that our shareholders approve the issuance of the Merger Consideration in the Merger, and the issuance of our common stock upon exercise of the Warrants, subject to the Board's right to withdraw, modify or amend such recommendation to the extent that the Board determines, in good faith, that failure to take such action would be reasonably likely to result in a breach of its fiduciary duties to our shareholders under applicable law.

On June 17, 2008, the Special Committee considered the First Amendment and determined not to modify its recommendation to our Board of Directors. The Board subsequently approved execution of the First Amendment on July 8, 2008.

Table of Contents

Our Board recommends that you vote FOR the issuance of the Merger Consideration in the Merger and the issuance of shares upon exercise of the Warrants.

PROPOSAL 2 AUTHORIZATION OF ADDITIONAL SHARES

Description of the Amendment

Our Board of Directors has approved, subject to shareholder approval, an amendment to our Articles of Incorporation to increase the number of authorized shares of our common stock from 6,250,000 to 20,000,000. A copy of the proposed certificate of amendment is attached as Annex III to this Proxy Statement. You should read the certificate of amendment carefully as it is the legal document that governs the amendment to the Articles of Incorporation.

Reasons for and Effect of the Amendment

Our Articles of Incorporation currently authorize the issuance of up to 6,250,000 shares of common stock and 1,000,000 shares of preferred stock, \$0.10 par value. On the Record Date, 3,059,041 shares of our common stock were outstanding, 147,696 shares were reserved for issuance upon exercise of outstanding stock options, 444,445 shares were reserved for issuance upon conversion of the Convertible Note, and 661,523 shares were reserved for future awards under our stock option plan. We will issue 5,900,000 shares to the Model Reorg shareholders in the Merger and an additional 1,500,000 shares if and when the Warrants are exercised. We are required by the Merger Agreement to have sufficient authorized but unissued shares available to complete the Merger. The proposed amendment to our Articles of Incorporation will authorize a sufficient number of shares for the Merger as well as a reasonable additional number of shares for future issuances by the combined company pursuant to benefit plans, for financings or acquisitions, or otherwise. If the amendment is not approved, we will not be able to complete the Merger and the other transactions contemplated by the Merger Agreement. Therefore, the Merger depends on our shareholders approving both Proposal 1 and this Proposal 2. However, this proposed amendment does not depend on approval of any other proposal as we are seeking approval of the increase whether or not the Merger issuance is approved.

The increased number of authorized and unissued shares could, in certain circumstances, have an anti-takeover effect. For example, it would permit issuances that could substantially dilute the stock ownership of a person seeking to effect a change in the composition of our Board of Directors or contemplating a tender offer or other business combination transaction with us. However, the share increase is not being proposed in response to any effort by a third party of which we are aware to accumulate shares of our common stock or obtain control of us, nor is it part of a plan to recommend to the stockholders a series of provisions to address takeover concerns. Other than the share increase proposal, our Board does not currently contemplate recommending the adoption of any other amendments to our Articles of Incorporation that could be construed to affect the ability of third parties to take over or change the control of E Com. Other than for issuances pursuant to our existing equity compensation arrangements for directors, employees and independent contractors, we have no current plans to issue any shares of our common stock.

Description of our Common Stock

The holders of outstanding shares of our common stock are entitled to receive dividends out of assets legally available therefor at such times and in such amounts as the Board may determine from time to time. The shares of common stock are not convertible into any other securities, and the holders thereof have no preemptive or subscription rights to purchase any of our securities. Upon liquidation, dissolution or winding up of E Com, the holders of common stock would be entitled to receive pro rata the assets that are legally available for distribution to them after payment of all debts and other liabilities. Each outstanding share of common stock is entitled to one vote on all matters submitted to a vote of the stockholders, including election of directors. There is no cumulative voting in the election of directors.

61

Recommendation of our Board of Directors

On December 21, 2007, our Board of Directors concluded unanimously that the Merger is in the best interests of our shareholders and that the number of authorized shares should be increased. The Board recommends that you vote FOR this amendment to our Articles of Incorporation.

PROPOSAL 3 NAME CHANGE

Description of the Amendment

Our Board of Directors has approved, subject to shareholder approval, an amendment to our Articles of Incorporation to change our corporate name to Perfumania Holdings, Inc. If the name change is approved by our shareholders, we anticipate that our ticker symbol on the NASDAQ Capital Market will be changed to PERF. A copy of the proposed certificate of amendment is attached as Annex IV to this Proxy Statement.

Reasons for the Change

The primary reason for the corporate name change is that we believe it will promote increased brand recognition of our products. As the name of our primary operating subsidiary, Perfumania is the distinctive tradename by which we are known in the marketplace. It will provide a strong corporate and commercial identification for the combined company, allowing us to capitalize on the existing market recognition.

Recommendation of our Board of Directors

Our Board of Directors unanimously recommends that you vote FOR this amendment to our Articles of Incorporation.

PROPOSAL 4 ADJOURNMENT OF SPECIAL MEETING

If necessary or appropriate in the circumstances at the Special Meeting, we may ask our shareholders to authorize the named proxies to approve one or more postponements or adjournments of the Special Meeting if sufficient votes have not yet been cast to approve Proposals 1, 2 or 3 at the time of the Special Meeting, in order to enable us to solicit additional proxies. This proposal relates only to a postponement or adjournment of the Special Meeting for purposes of soliciting additional proxies to obtain the requisite shareholder approval to approve Proposals 1, 2 or 3. Our Board of Directors retains full authority to postpone or adjourn the Special Meeting for any other purpose, including the absence of a quorum, or to postpone the Special Meeting before it is convened, without the consent of the shareholders.

If the Special Meeting is postponed or adjourned, we would not be required to give notice of the time and place of the adjourned meeting unless our Board of Directors fixed a new record date for the Special Meeting.

Recommendation of our Board of Directors

Our Board of Directors unanimously recommends that you vote FOR this Proposal.

62

CERTAIN RELATIONSHIPS AND RELATED PERSON TRANSACTIONS

The following describes relationships and transactions that exist now, or that are proposed to exist as a result of our acquisition of Model Reorg in the Merger, involving us and in which any of our principal shareholders, directors, executive officers and/or members of their immediate families has a direct or indirect material interest. Ordinary compensatory arrangements with our directors or officers that have been or are expected to be approved by the Compensation Committee of our Board of Directors are not included. Our Audit Committee reviews quarterly, on an aggregate basis, the related party transactions pursuant to which we purchase or sell merchandise from or to companies in which the Nussdorfs are affiliated to determine the reasonableness of the transactions as compared to third party transactions. We do not have any written policies or procedures specifically addressing related party transactions with those companies. In reviewing these transactions, our Audit Committee considers, among other factors, the gross margin of product sales, the payment and credit terms of the transactions, and the turnover of inventory of product purchases. In ratifying these transactions, our Audit Committee has considered the commercial reasonableness of their terms.

Shareholder Relationships

Stephen L. Nussdorf, the Chairman of our Board, and Glenn H. Nussdorf, his brother, currently own an aggregate of 1,113,144, or 36.4%, of the outstanding shares of our common stock. They also hold the Convertible Note, which is convertible into 444,445 shares. Upon conversion of the Convertible Note, they would own an aggregate of 1,557,589 shares, or approximately 44.5%. Stephen and Glenn Nussdorf also own an aggregate of 59.2% the outstanding shares of capital stock of Model Reorg. After the Merger, Stephen and Glenn Nussdorf would beneficially own an aggregate of approximately 53.7% of our outstanding common stock, assuming conversion of the Convertible Note, and they would hold Warrants that would, upon exercise, result in their beneficially owning up to an aggregate of approximately 57.7% of our shares. Their sister, Arlene Nussdorf, who is not a shareholder of E Com, also owns 29.6% of the Model Reorg shares. Stephen and Glenn Nussdorf have agreed to vote all of their shares of our common stock in favor of the proposals being presented at the Special Meeting. Trusts benefiting Stephen, Glenn and Arlene Nussdorf and members of their families also own 100% of the outstanding shares of Quality King.

At the closing of the Merger, we will execute an agreement providing certain piggyback registration rights to the former Model Reorg shareholders with respect to the shares of our common stock they receive at the closing. In addition, the former Model Reorg shareholders will have a continuing obligation to indemnify us against certain liabilities (subject to certain limitations) in connection with the Merger. (See Approval to Issue Shares and Warrants Terms of the Merger Escrow and Indemnification and Registration Rights for further details.)

Management Relationships

Stephen and Glenn Nussdorf serve as directors of Model Reorg and its subsidiaries. Stephen, Glenn and Arlene Nussdorf serve as directors and principal officers of Quality King and its subsidiaries. Our President and Chief Executive Officer, Michael W. Katz, is also a director of Model Reorg and a senior executive of both Model Reorg and Quality King. In addition, because we do not currently have a Chief Operating Officer, Raymond Piergiorgi, Model Reorg s Chief Operating Officer, has been consulting with E Com management on operations matters. With the following exceptions, any compensation for such services has been paid by the respective entity for which the services were performed.

Mr. Piergiorgi has received no compensation from us. Mr. Katz s compensation for his service with Model Reorg has been paid by Quality King. He is expected to continue after the Merger to serve as an officer of Quality King, which will pay any related compensation.

Under a Stockholder Agreement dated as of January 1, 2003, Model Reorg issued shares of its common stock to Mr. Katz, subject to vesting. Model Reorg repurchased the shares from Mr. Katz on July 1, 2004 pursuant to the agreement for a total of \$1,002,250, which was paid over thirty-three months. Pursuant to the

63

Table of Contents

Agreement, Model Reorg will pay Mr. Katz an additional amount in the event of termination of his employment in certain circumstances, or his death or disability of 3% of the increase in book value of Model Reorg, excluding certain amounts, over \$17.6 million, the book value of Model Reorg as of June 30, 2002. At October 31, 2007, Model Reorg had accrued \$1.5 million with respect to this liability. Mr. Katz has an obligation under this Agreement not to compete with Model Reorg, its subsidiaries or its affiliates or to solicit their employees for one year following the termination of his employment.

Rene A. Garcia, a shareholder of Model Reorg, will acquire beneficial ownership of more than 5% of our common stock as a result of the Merger. Mr. Garcia and trusts benefiting members of his family contributed certain fragrance businesses they owned to Jacavi LLC, which Model Reorg acquired by merger on October 5, 2007 in exchange for approximately 11.25% of its common stock. Mr. Garcia and the trusts have escrowed their Model Reorg shares until October 5, 2008 to secure their obligation to indemnify Jacavi against damages for certain breaches of their agreement with Jacavi. If the Merger is completed, the Merger Consideration issuable to Mr. Garcia and his family trusts will be held in escrow by EAPD until that date. Model Reorg also agreed to indemnify Mr. Garcia and his family trusts for losses in excess of \$200,000 in the event of a breach of certain representations and warranties made by Model Reorg in the Jacavi acquisition agreement. Model Reorg s indemnification obligation is unsecured and is limited to \$4.0 million.

In connection with the Jacavi merger agreement, Mr. Garcia entered into an employment agreement, dated as of October 5, 2007, with Jacavi. This agreement provides for Mr. Garcia s employment as president and chief executive officer of Jacavi and its subsidiary, Distribution Concepts, LLC, at an initial annual base salary of \$300,000. The term of his employment is until December 31, 2010, but may be terminated earlier by either party subject to payment to Mr. Garcia of an amount equal to one year s base salary if his employment is terminated by Jacavi without Cause or by Mr. Garcia for Good Reason, or his employment terminates due to his Disability, each as defined in the agreement. In the event of termination due to Disability, he also would continue to receive for one year those fringe benefits to which he was entitled at the time of termination. Mr. Garcia is bound by noncompetition obligations with respect to Jacavi and Distribution Concepts for twelve months following termination of his employment unless termination is without Cause or by him for Good Reason and he provides Jacavi with a general release of liability.

Relationships and Transactions between Model Reorg and E Com

The following are relationships and transactions between a subsidiary of ours, on one hand, and subsidiaries of Model Reorg, on the other, that currently exist or have existed during our past three fiscal years. We expect that these relationships and transactions will substantially continue after the Merger, although they will be between our wholly-owned subsidiaries and will be eliminated in consolidation, and therefore will not appear in our consolidated financial statements.

Product Purchases and Sales

Substantially all wholesale sales by Perfumania (our subsidiary) in recent years have been to Quality Fragrance Group (a Model Reorg subsidiary). The bulk of these are sales of product that Perfumania can obtain through a supplier relationship that is not available to Quality Fragrance Group. These sales stem also from Perfumania s ability to obtain certain merchandise at better prices and in greater quantities than Quality Fragrance Group is able to obtain. Sales by Perfumania to Quality Fragrance Group totaled approximately \$57.7 million, \$13.6 million, \$17.9 million during our 2007, 2006, and 2005 fiscal years.

Similarly, Perfumania purchases product from Quality Fragrance Group and Five Star Fragrance (another Model Reorg subsidiary). These include brands for which Five Star Fragrance is the owner or licensee and other product that Quality Fragrance Group can obtain at better prices and in greater quantities than Perfumania can obtain. They also include certain product for which Perfumania owns the brand and contracts with Five Star Fragrance for manufacturing. Over the past three fiscal years, such purchases ranged from 23% to 28% of E

64

Table of Contents

Com s total merchandise purchases. These purchases totaled approximately \$32.1 million, \$32.4 million, and \$30.5 million, during our 2007, 2006, and 2005 fiscal years. At the end of our 2007, 2006, and 2005 fiscal years, the net amounts due from us to Model Reorg s subsidiaries were approximately \$36.1 million, \$16.9 million, and \$17.2 million, respectively.

These purchase and sale transactions are undertaken on a purchase order basis and without long-term or exclusive contracts, similar to the way each company conducts business with unaffiliated third parties. However, the prices charged to us by the Model Reorg subsidiaries in such purchases are typically lower than it charges unaffiliated customers. The receivables are generally payable in 90 days but, due to the seasonality of our business, Model Reorg generally extends those terms. These amounts are non-interest bearing, and we are dependent upon these extended terms for much of our liquidity during the year.

Information Technology Services Agreement

Under an Information Technology Services Agreement dated August 2, 2007 with Model Reorg, we manage and monitor Model Reorg s IT systems and provide related services, including data center operations, network support, security, electronic data interchange processing, systems planning, and various development and support activities, for which Model Reorg pays us a monthly service fee of \$25,000 plus expenses. The agreement will terminate by its terms in connection with the Merger, or either party may terminate it on 30 days notice.

Relationships and Transactions between Model Reorg or E Com and Other Affiliates

The following relationships and transactions between either Model Reorg and its subsidiaries or us and our subsidiaries, on one hand, and other affiliates of either company, on the other, currently exist or have existed during our past three fiscal years. Except as discussed below, these will not be directly affected by the Merger.

Quality King Distributors, Inc.

Quality King holds a promissory note issued by Model Reorg on December 20, 2002 in order to supplement a revolving line of credit entered into jointly by Model Reorg and Quality King with an unaffiliated syndicate of lenders. The balance owed under this promissory note fluctuates based on Model Reorg s cash flow requirements. The Note bears interest based on the monthly blended borrowing rates selected by Quality King under the syndicate s revolving line of credit, which fluctuate with LIBOR and the prime rate. Average interest rates and Model Reorg s interest expense on the promissory note for Model Reorg s 2007 and 2006 fiscal years were 7.6% (\$4.5 million) and 7.4% (\$6.4 million), respectively, and for the six months ended April 30, 2008 were 5.4% (\$1.96 million). Interest expense charged to operations relating to the note payable was \$4,571 for fiscal year 2007 and \$1,961 for the six months ended April 30, 2008.

The balance of principal and accrued interest at April 30, 2008 was \$76.9 million. The Merger Agreement requires Model Reorg to repay a portion of this debt and restructure the balance as a condition to closing. (See Approval to Issue Shares and Warrants Affiliate Debt for further details of the debt restructuring and refinancing.)

Model Reorg and Quality King share in various operational arrangements under which each may benefit from economies of scale and additional convenience. In the case of joint participation in third party arrangements, e.g., 401(k), medical, dental, and flex spending plans, a portion of the costs and overhead is allocated to Model Reorg based on the specific services provided and the number of employees involved. Until June 2008, Model Reorg s subsidiaries obtained other payroll and human resources services from Quality King, for which Model Reorg was allocated a proportionate share of the costs and overhead, based, for example, on payroll dollars, number of employees involved or number of transactions processed. Model Reorg also obtains legal services, for which it pays an agreed hourly rate. Model Reorg s management has determined that the costs for these services are comparable to those it would have paid to unaffiliated suppliers. At the closing of the Merger Agreement, we will execute a Services Agreement with

Table of Contents

Quality King providing for the continuation of Model Reorg s participation in the third party arrangements at our respective share of Quality King s cost, including allocated overhead, plus a 2% administrative fee, and the provision of legal services, subject to termination with respect to any such service on 30 days notice by either party. Quality King and one of its affiliates also provide trucking services to Model Reorg, which we expect to continue to obtain after the Merger. Since May 2007, they have billed Model Reorg at the same rates they charge third parties for trucking services, rather than allocating expenses. Amounts allocated to or paid by Model Reorg under the foregoing arrangements during its 2007, 2006, and 2005 fiscal years were \$3.7 million, \$5.1 million, and \$4.7 million, respectively.

Quality King occupies a leased 560,000 square foot office and warehouse facility in Bellport, NY. Model Reorg has been occupying approximately half of this facility since December 2007 under a sublease that terminates on September 30, 2027. Model Reorg is making sublease payments of \$191,565 per month, increasing by 3% annually, which is equivalent to current market rates, plus a proportionate share of costs attributable to the premises. We expect that a significant portion of our operations will be consolidated in this facility after the Merger.

Parlux Fragrances, Inc.

Parlux owned more than 10% of the outstanding shares of our common stock for a number of years, disposing of its entire interest during fiscal 2006. Ilia Lekach, our Chairman and Chief Executive Officer until February 2004, was Chairman and Chief Executive Officer of Parlux until February 2007. He was also the beneficial owner of more than 10% of our common stock until disposing of his entire interest during fiscal 2006. Since February 2007, Glenn Nussdorf has owned approximately 12% of the outstanding shares of common stock of Parlux. At that time, Mr. Nussdorf entered into an agreement with Parlux under which three persons agreed upon between Mr. Nussdorf and Parlux comprise 50% of Parlux s board of directors. One of those persons now serves as Parlux s Chief Executive Officer.

Perfumania purchased product from Parlux amounting to approximately \$44.7 million, \$17.6 million and \$23.0 million, during our 2007, 2006, and 2005 fiscal years, respectively. These represented approximately 21%, 12%, and 18%, respectively, of our total inventory purchases during those periods. The amounts due to Parlux at February 2, 2008 and February 3, 2007 were approximately \$12.5 million and \$7.2 million, respectively. According to Parlux s Form 10-K, Perfumania is one of its largest customers. Similar to our purchases from Model Reorg described above, Parlux provides extended payment terms to us that, according to Parlux, take into consideration the opportunity to sell Parlux products in Perfumania s many retail outlets, the lengthy relationship between the companies, the volume of Perfumania s purchases, its policy of no returns, Parlux s reduced spending for advertising and promotion and reduced distribution costs applicable to Perfumania orders. The amounts owed to Parlux are non-interest bearing, and we have been dependent upon the extended terms for liquidity during the year.

For the three years ending July 31, 2006, Parlux and Five Star were parties to an agreement under which Parlux distributed product acquired from Five Star on terms comparable to those charged to third parties. Parlux s payments under this agreement totaled approximately \$1.0 million.

Principal Shareholders

We issued the Convertible Note in the principal amount of \$5,000,000 to Stephen and Glenn Nussdorf on December 9, 2004 in exchange for a \$5,000,000 subordinated secured demand loan they had made to us in March 2004. The Convertible Note matures in January 2009. It bears interest at the prime rate plus 1%, requires quarterly interest payments and is secured by a security interest in our assets. There are no prepayment penalties, and the Convertible Note is subordinate to all bank related indebtedness. The Nussdorfs may convert the amounts due under the Convertible Note into an aggregate of 444,445 shares of our common stock at a conversion price of \$11.25, the closing market price of our common stock on the date the Convertible Note was issued.

66

INFORMATION ABOUT E COM VENTURES

Business of E Com

General

We were incorporated under Florida law in 2000. We operate through two wholly-owned subsidiaries: Perfumania, which is a specialty retailer and wholesaler of fragrances and related products, and perfumania.com, which is an Internet retailer of fragrances and other specialty items.

Perfumania is a specialty retailer and wholesale distributor of a wide range of brand name and designer fragrances. Perfumania operates a chain of retail stores specializing in the sale of fragrances at discounted prices up to 75% below the manufacturers suggested retail prices. Perfumania s wholesale division distributes fragrances and related products primarily to Model Reorg s subsidiary, Quality Fragrance Group. Perfumania.com offers a selection of our more popular products for sale over the Internet and serves as an alternative shopping experience for Perfumania retail customers. Approximately 83% of our net sales and 97% of our gross profit during the quarter ended May 3, 2008 were provided by our retail division.

Perfumania operates its wholesale business directly. The retail business is principally operated through Magnifique Parfumes and Cosmetics, Inc. (Magnifique), a wholly-owned subsidiary of Perfumania, although the stores are generally operated under the name Perfumania. Perfumania s retail stores are generally located in regional malls, manufacturers outlet malls, lifestyle centers, airports and suburban strip shopping centers. The number of retail stores in operation at February 2, 2008, February 3, 2007 and January 28, 2006 were 303, 267 and 239, respectively.

Sales of perfumania.com are reported with those of our retail business. For ease of reference, our retail and wholesale businesses are referred to as segments. See further discussion of segment information in Notes 13 and 9 to our Consolidated Financial Statements for fiscal 2007 and first quarter of fiscal 2008, respectively.

Our fiscal year ends on the Saturday closest to January 31. We refer to a fiscal year by the calendar year in which it begins. Accordingly, fiscal year 2007 ended on February 2, 2008, fiscal year 2006 ended on February 3, 2007 and fiscal year 2005 ended on January 28, 2006. With the exception of fiscal year 2006, which contained fifty-three weeks, each of the fiscal years presented contains fifty-two weeks.

References to our Consolidated Financial Statements are to our audited consolidated financial statements for the fiscal years ended February 2, 2008, February 3, 2007, January 28, 2006, and as of February 2, 2008 and February 3, 2007, that are included in this Proxy Statement beginning on page F-3.

Retail Division

Strategy

Each of Perfumania s retail stores generally offers approximately 300 different fragrance brands for women and men at prices up to 75% below the manufacturer s suggested retail prices. These stores stock brand name and designer brands such as Estee Lauder, Yves Saint Laurent, Calvin Klein, Giorgio Armani, Hugo Boss, Ralph Lauren/Polo, Perry Ellis, Liz Claiborne, Giorgio, Halston, Escada, Chanel, Sean Jean, Lacoste, Burberry, Azzaro, Guess, Donna Karan, and Paris Hilton. Perfumania also carries private label lines of bath & body treatment products under the name Jerome Privee, and cosmetics products under the name Mattese.

The cornerstone of Perfumania s marketing philosophy is to develop customer awareness that its stores offer an extensive assortment of brand name and designer fragrances at discount prices. Perfumania posts highly visible price tags in its stores, listing both the manufacturers suggested retail prices and Perfumania s discounted prices to enable customers to make price comparisons. In addition, Perfumania utilizes sales promotions such as gift with purchase and purchase with purchase offers. From time to time, Perfumania test markets additional specialty gift items in its stores.

Table of Contents

Perfumania s stores are full-service stores. Accordingly, store personnel are trained to establish personal rapport with customers, to identify customer preferences with respect to both product and price range, and to successfully conclude a sale. Management believes that knowledgeable sales associates and attentive service are key factors to the success of Perfumania s retail stores. Perfumania s store associates are compensated on a salary plus commission basis. Perfumania has several incentive programs that motivate store associates to sell merchandise that have higher profit margins. In addition, to provide an incentive to increase sales and reduce expenses, regional, district and store managers are eligible to receive a bonus if store profitability and operational goals are met. Management believes that a key component of Perfumania s ability to increase profitability will be its ability to hire, train and retain store associates, district and regional managers. Perfumania conducts comprehensive training programs for store associates that are designed to achieve higher levels of customer satisfaction.

Perfumania relies on its distinctive store design and window displays to attract the attention of prospective customers. In addition, Perfumania distributes advertising flyers and brochures by mail in and around its stores and in the malls in which its stores are located. Radio and television advertising is done occasionally in certain geographic regions that have a cluster of stores. The amount of advertising varies due to the seasonality of the business with the greatest portion in the fourth fiscal quarter. (See further discussion in Note 14 to our Consolidated Financial Statements for fiscal 2007.)

Retail Stores

Perfumania s standard store design includes signs and merchandise displays that are designed to enhance customer recognition of Perfumania s stores. Perfumania s stores average approximately 1,500 square feet; however, stores located in manufacturers outlet malls tend to be larger than Perfumania s other stores. A store is typically managed by one manager and one assistant manager. The average number of employees in a Perfumania store is five, including part-time help. Regional and district managers visit stores on a regular basis in an effort to ensure knowledgeable and attentive customer service and compliance with operational policies and procedures.

Store Location and Expansion

Perfumania s current business strategy focuses on maximizing sales by raising the average dollar sale per transaction, increasing transactions per hour, reducing expenses at existing stores, opening new stores in proven geographic markets and selectively closing under-performing stores. When opening new stores, Perfumania seeks locations primarily in regional and manufacturers outlet malls, lifestyle centers and, selectively, on a stand-alone basis in suburban shopping centers in metropolitan areas. To achieve economies of scale with respect to advertising and management costs, Perfumania evaluates whether to open additional stores in markets where it already has a presence or whether to expand into additional markets that it believes have a population density and demographics to support a cluster of stores.

As of May 3, 2008, Perfumania operated 313 stores in 35 states and Puerto Rico, with the highest concentrations being 50 locations in Florida, 37 in Texas, 25 in California, 19 in New York and 18 in Puerto Rico. In fiscal years 2007, 2006 and 2005, Perfumania opened 39, 36 and 23 stores, respectively, with another 11 opened during the thirteen weeks ended May 3, 2008. Perfumania continuously monitors store performance and from time to time closes under-performing stores, which typically have been older stores in less trafficked locations. Perfumania has closed 19 stores since the beginning of fiscal 2005. During fiscal year 2008, Perfumania will continue to focus on improving the profitability of its existing stores, and management expects to open approximately 45 stores and close approximately 3 stores.

Information Systems

Perfumania has an integrated information system including retail outlet and corporate systems. Perfumania.com has a completely integrated e-commerce system. These systems encompass significant phases of our operations, providing information for planning, purchasing, pricing, distribution. finance and human resource

68

Table of Contents

decisions. E-mail and other information are communicated between the corporate office and store locations. Daily compilation of sales, gross margin, and inventory levels enables management to analyze profitability and sell-through by item and product line as well as monitor the success of sale promotions. Inventory is tracked through its entire life cycle. Perfumania s point of sale system is standard in all its stores. The system enables communication, pricing and promotion programs, time and attendance reporting, and inventory control.

Perfumania.com

Perfumania.com provides a number of advantages for retail fragrance sales. Internet fragrance sales are highly competitive and we compete on the basis of selling price, merchandise variety, ease of selection and cost of delivery. Our Internet site enables us to display a larger number of products than traditional store-based or catalog sellers. In addition, the ability to frequently adjust featured selections and edit content and pricing provides significant merchandising flexibility. Our Internet site benefits from the ability to reach a large group of customers from a central location. We can also obtain demographic and behavioral data of customers, increasing opportunities for direct marketing and personalized services. Because brand loyalty is a primary factor influencing a fragrance purchase, we believe the ability to physically sense the product is not critical to the purchasing decision. Perfumania.com s online store provides its customers with value, selection, pricing and convenience.

Seasonality

Our retail business follows a seasonal pattern, with sales peaking over approximately five weeks during the December holiday period. During fiscal 2007, the holiday period accounted for approximately 26% of our net retail sales.

Wholesale Division

Perfumania distributes fragrances on a wholesale basis to Model Reorg s subsidiary, Quality Fragrance Group, which accounted for substantially all of our net wholesale sales in fiscal years 2007, 2006 and 2005, and the first thirteen weeks of fiscal 2008. (For more information, see Notes 6 and 8 to our Consolidated Financial Statements for fiscal 2007 and the first quarter of fiscal 2008, respectively.)

Sources of Supply

In recent years, Perfumania has purchased fragrances from up to 144 suppliers, including national and international manufacturers, distributors, wholesalers and importers. Perfumania generally makes its purchases based on consideration of a combination of prices, credit terms, quantities and merchandise selection and, accordingly, the extent and nature of Perfumania s purchases from its various suppliers change constantly. Perfumania s purchases generally peak in the third quarter in anticipation of the December holiday season, which results in higher retail sales in the fourth quarter than in the first three quarters. As is customary in the fragrance industry, Perfumania has no long-term or exclusive contracts with suppliers.

Approximately 15%, 23% and 22% of Perfumania s total merchandise purchased in fiscal years 2007, 2006 and 2005, respectively, and approximately 17% in the first thirteen weeks of fiscal 2008, was from subsidiaries of Model Reorg. Approximately 21%, 12% and 18% of Perfumania s total merchandise purchased in fiscal years 2007, 2006 and 2005, respectively, and approximately 23% in the first thirteen weeks of fiscal 2008 was from another affiliate, Parlux, a manufacturer and distributor of prestige fragrances and related beauty products. (See Certain Relationships and Related Person Transactions Shareholder Relationships.) Besides Model Reorg and Parlux, no other supplier accounted for more than 10% of our merchandise purchases during fiscal years 2007, 2006 or 2005.

69

Table of Contents

A portion of Perfumania s merchandise is purchased from secondary sources such as distributors, wholesalers and importers. Merchandise purchased from secondary sources includes trademarked and copyrighted products that were manufactured in the United States, sold to foreign distributors and then re-imported into the United States, as well as trademarked and copyrighted products manufactured and intended for sale in foreign countries. From time to time, U.S. trademark and copyright owners and their licensees and trade associations have initiated litigation or administrative agency proceedings, based on U.S. Customs Service regulations or trademark or copyright laws, seeking to halt the importation into the United States of such gray market merchandise or to restrict its resale in the United States, and some of these actions have been successful. However, the U.S. courts remain divided on the extent to which trademark, copyright or other existing laws or regulations can be used to restrict the importation or sale of gray market merchandise. In addition, from time to time federal legislation to restrict the importation or sale of gray market merchandise has been proposed but, to our knowledge, no such legislation has been adopted. No litigation or administrative proceedings related to gray market merchandise have been brought against us during the past three fiscal years and no such matters, to our knowledge, are pending.

As is often the case in the fragrance and cosmetics business, some of the merchandise purchased by Perfumania may have been manufactured by entities, particularly foreign licensees, who are not the owners of the trademarks or copyrights for the merchandise. Perfumania s secondary market sources generally will not disclose the identity of their suppliers, which they consider to be proprietary trade information. As a result, Perfumania may not always be able to demonstrate that the manufacturer of specific merchandise had proper authority from the trademark or copyright owner to produce the merchandise or permit it to be resold in the United States. Accordingly, there is a risk that, if Perfumania were called upon or challenged by the owner of a particular trademark or copyright to demonstrate that specific merchandise was produced and sold with the proper authority and it was unable to do so, Perfumania could, among other things, be restricted from reselling the particular merchandise or be subjected to other liabilities.

Perfumania s business activities could become the subject of legal or administrative actions brought by manufacturers, distributors or others, any of which actions could have a material adverse effect on our business or financial condition. In addition, future judicial, legislative or administrative agency action, including possible import, export, tariff or other trade restrictions, could limit or eliminate some of Perfumania s secondary sources of supply or any of its business activities.

Distribution

Perfumania utilizes independent national trucking companies to deliver merchandise to its stores. Retail store deliveries generally are made weekly, with more frequent deliveries during the holiday season. Such deliveries permit the stores to minimize inventory storage space and increase the space available for display and sale of merchandise. To expedite delivery of merchandise to its wholesale customers, Perfumania sometimes instructs its suppliers to ship merchandise directly to its wholesale customers. Sales of perfumania.com are shipped through national carriers and are typically delivered within a few days of being ordered.

Competition

Retail and wholesale perfume businesses are highly competitive. Perfumania s retail competitors include department stores, regional and national retail chains, drug stores, supermarkets, duty-free shops and other specialty retail stores. We believe that Perfumania is the largest specialty retailer of discounted fragrances in the United States in terms of number of stores. Some of Perfumania s competitors sell fragrances at discount prices and some are part of large national or regional chains that have substantially greater resources and name recognition than Perfumania. Perfumania s stores compete on the basis of selling price, promotions, customer service, merchandise variety, store location and ambiance. Perfumania believes that its perfumery concept, full-service sales staff, discount prices, large and varied selection of brand name and designer fragrances and attractive shopping environment are important to its competitive position.

70

Table of Contents

Employees

At May 3, 2008, we had 1,756 employees, of whom 1,552 were employed in Perfumania s retail stores, 70 were employed in Perfumania s warehouse and distribution operations and 134 were employed in executive, administrative and other positions. Temporary and part-time employees are added between Thanksgiving and Christmas. None of our employees are covered by a collective bargaining agreement and we consider our relationship with our employees to be good.

Trade Names and Service Marks

Perfumania s stores use the trade name and service mark Perfumania Also operates under the trade names Also Perfumania, Class Perfumes, Perfumania Too and Perfumania Plus. Perfumania has common law rights to its trade names and service mark in those general areas in which its existing stores are located and has registered the service mark Perfumania[®] with the U.S. Patent and Trademark Office. The registration expires in 2009 and may be renewed for 10-year terms thereafter.

Properties

Our executive offices and distribution center are located in a 179,000 square foot facility in Sunrise, Florida. The facility is leased through December 2017 pursuant to a lease that currently provides for monthly rent of approximately \$89,000 with specified increases. We anticipate that a significant portion of our operations will be consolidated into Model Reorg s Bellport, NY facility following the merger, as described in Certain Relationships and Related Person Transactions.

All of Perfumania s retail stores are located in leased premises. Most of the store leases provide for the payment of a fixed amount of base rent plus a percentage of sales, ranging from 3% to 15%, over certain minimum sales levels. Store leases typically require Perfumania to pay its proportionate share of common area expenses, real estate taxes, utility charges, insurance premiums and certain other costs. Some of Perfumania s leases permit the termination of the lease if specified minimum sales levels are not met. (See Note 12 to our audited Consolidated Financial Statements for additional information with respect to our store leases.)

Legal Proceedings

We are involved in legal proceedings in the ordinary course of business. Management cannot presently predict the outcome of these matters, although management believes that the ultimate resolution of these matters will not have a materially adverse effect on our financial position, operations or cash flows.

Management s Discussion and Analysis of Financial Condition and Results of Operations

Restatement of Financial Statements

Since the issuance of our 2006 financial statements, we identified errors in our prior accounting for operating leases and the period of amortization of leasehold improvements and have restated our financial statements for fiscal years 2006 and 2005.

Under GAAP, rent expense is amortized on a straight-line basis over the term of the lease. In prior periods, we had determined that the term of the lease begins on the commencement date of the lease, which generally coincides with the store opening date. We have re-evaluated FASB Technical Bulletin No. 85-3, Accounting for Operating Leases with Scheduled Rent Increases and other guidance provided by the SEC and determined that the lease term for amortization purposes should commence on the date we take physical possession of the leased space to commence construction of leasehold improvements, which is generally two months prior to a store opening date. We have restated our previously reported financial statements in connection with our accounting for rent expense. Rent expense is included in selling, general and administrative expenses on our consolidated statements of operations.

Table of Contents

We have also reviewed our leasehold improvements to ensure amortization over the shorter of their economic lives or their lease term. In prior periods, in determining the lease terms, in addition to the initial term, we had included one anticipated renewal term even if the lease did not have a stated renewal option. We have adjusted our lease terms for purposes of calculating amortization of leasehold improvements to include a renewal term only if that renewal option is specified in the lease agreement and we have concluded that exiting the lease after the initial term would result in economic loss and therefore normally we would anticipate exercising the respective renewal option.

The restatement resulted in a decrease in net income of \$0.1 million in fiscal year 2006, an increase in net income of \$1.0 million in fiscal year 2005 and an aggregate decrease in net income of \$2.7 million for all years prior to fiscal year 2005, with a corresponding increase to the accumulated deficit as of January 29, 2005. The majority of the adjustments relate to periods prior to fiscal year 2005.

See Notes 3 and 2 to our Consolidated Financial Statements for fiscal 2007 and the first quarter of fiscal 2008, respectively, for a summary of the effects of the restatement. This Management s Discussion and Analysis of Financial Condition and Results of Operation gives effect to the restatement.

General

Perfumania s retail division accounts for most of our net sales and gross profit. Perfumania s overall profitability depends principally on our ability to purchase a wide assortment of merchandise at favorable prices, attract customers and successfully conclude retail sales. Other factors affecting our profitability include general economic conditions, competition, availability of volume discounts, number of stores in operation, timing of store openings and closings and the effect of special promotions offered by Perfumania.

The following table sets forth selected items from our Consolidated Statements of Operations expressed as a percentage of total net sales for the periods indicated:

	13 Weeks	Fiscal Year			
	May 3, 2008	May 5, 2007	2007	2006	2005
Total net sales	100.00%	100.0%	100.0%	100.0%	100.0%
Total gross profit	39.8	42.2	37.1	42.7	41.3
Selling, general and administrative expenses	43.4	43.9	33.1	36.5	34.6
Depreciation and amortization	3.1	3.0	2.1	2.0	2.1
Total operating expenses	46.6	46.9	35.2	38.5	36.7
Income (loss) from operations	(6.7)	(4.7)	1.9	4.2	4.6
Other expense:					
Interest expense	(1.5)	(2.3)	(1.6)	(1.9)	(1.7)
Income (loss) before income taxes	(8.3)	(6.9)	0.3	2.3	2.9
Income tax benefit (provision)	3.1	2.6	(0.3)	(0.5)	3.6
Net (loss) income	(5.1)%	(4.3)%	0.0%	1.8%	6.5%
4 /			` /	` /	

Critical Accounting Estimates

Our Consolidated Financial Statements have been prepared in accordance with GAAP. Preparation of these statements requires management to make judgments and estimates. As such, some accounting policies have a significant impact on amounts reported in these financial statements. The judgments and estimates made can significantly affect results. Materially different amounts would be reported under different conditions or by using different assumptions. A summary of significant accounting policies can be found in Note 2 to our Consolidated Financial Statements for the year ending February 2, 2008.

72

Table of Contents

We consider an accounting policy to be critical if it is both important to the portrayal of our financial condition and results and requires significant judgment and estimates by management in its application. We have identified certain accounting policies that we consider critical to our business and results of operations and have provided below additional information on those policies.

Inventory Adjustments and Writeoffs

Inventories are stated at the lower of cost or market, with cost being determined on a weighted average cost basis. We review our inventory on a regular basis for excess and potentially slow moving inventory based on prior sales, forecasted demand, historical experience and through specific identification of obsolete or damaged merchandise and, we record adjustments to reduce the carrying value of inventory to the lower of cost or market in accordance with our assessment. If there are material changes to these estimates, additional writeoffs could be necessary. Inventory shrinkage is estimated and accrued between physical inventory counts.

Impairment of Long-Lived Assets

When events or changes in circumstances indicate that the carrying values of long-lived assets may be impaired, an evaluation of recoverability is performed by comparing the carrying value of the assets to projected future cash flows in addition to other quantitative and qualitative analyses. Inherent in this process is significant management judgment as to the projected cash flows. Upon indication that the carrying value of such assets may not be recoverable, we recognize an impairment loss as a charge against current operations. Property and equipment assets are grouped at the lowest level for which there are identifiable cash flows when assessing impairment. Cash flows for retail assets are identified at the individual store level. Factors that could trigger an impairment review include a significant underperformance relative to expected historical or projected future operating results, or a significant negative industry or economic trend. Judgments are also made as to whether under-performing stores should be closed. Even if a decision has been made not to close an under-performing store, the assets at that store may be impaired. If there are material changes to these judgments or estimates, additional charges could be necessary.

Valuation of Deferred Tax Assets

Statement of Financial Accounting Standard (SFAS) No. 109, Accounting for Income Taxes, requires that deferred tax assets be evaluated for future realization and reduced by a valuation allowance to the extent we believe it is more likely than not that a portion of these assets will not be realized. We consider many factors when assessing the likelihood of future realization of our deferred tax assets, including our recent cumulative earnings experience by taxing jurisdiction, expectations of future taxable income, the carry-forward periods available to us for tax reporting purposes, and other relevant factors. The range of possible judgments relating to the valuation of our deferred tax assets is very wide. Significant judgment is required in making these assessments, and it is very difficult to predict when, if ever, our assessment may conclude that the remaining portion of our deferred tax assets is realizable.

Fiscal Year 2007 Compared to Fiscal Year 2006

Management Overview

During fiscal year 2007 we realized our fourth consecutive year of net income. We have achieved these positive operating results due to increases in retail and wholesale sales, an improvement in the product mix and selection at our retail stores, opening new stores and closing underperforming stores. In addition, at the retail level, in order to increase sales we have focused our efforts on increasing the average dollar sale per transaction and the number of units sold per transaction. We had an accumulated deficit of approximately \$34.8 million at February 2, 2008.

73

Our goal is to continue to increase the number of our stores, either by opening new ones in markets that present opportunities for growth or by acquiring smaller fragrance retailers. We believe there are numerous opportunities for retail locations domestically and are focused on expansion in markets where we either have a presence or in new geographic regions where the population density and the demographics support a cluster of stores.

Revenues:

		For the year ended (\$ in thousands)					
	February 2, 2008	Percentage of Revenues	Febr	uary 3, 2007	Percentage of Revenues	Percentage Increase	
Retail	\$ 244,020	80.8%	\$	229,783	94.3%	6.2%	
Wholesale	57,815	19.2%		13,826	5.7%	318.2%	
Total Revenues	\$ 301,835	100.0%	\$	243,609	100.0%	23.9%	

In fiscal year 2007, our total revenues increased by \$58.2 million or 23.9% compared with the prior year. Retail sales increased by \$14.2 million or 6.2%, primarily as a result of the increase in the number of retail stores in operation. Fiscal year 2007 had fifty-two weeks compared with fifty-three weeks in fiscal year 2006. Retail sales numbers for fiscal year 2006 include this additional week. The average number of stores operated increased from 250 during fiscal year 2006 to 283 in fiscal year 2007. Comparable store sales increased by 2.9%. The comparable store sales calculation measures sales results from stores that have been open for one year or more. We exclude stores that are closed for renovation from comparable store sales from the month during which renovation commences until the first full month after reopening. Comparable store sales growth percentages were calculated excluding the extra week of fiscal year 2006. While the average retail price per unit sold during fiscal year 2007 increased 6.3% versus 2006, the total number of units sold remained virtually unchanged. We attribute the increase in the average retail price per unit sold to changes in our product mix and increases in the retail selling prices on selected merchandise. The number of units sold was affected by softness in the United States economy and resulting weak mall traffic in 2007.

Wholesale sales increased by \$44.0 million or 318.2% from \$13.8 million in fiscal year 2006 to \$57.8 million in fiscal year 2007. Virtually all wholesale sales were made to Quality Fragrance Group, a subsidiary of Model Reorg. The increase in wholesale sales is due to significantly more merchandise availability at better prices than had been available from our suppliers during fiscal year 2006 and an increase in the demand from Model Reorg s subsidiaries. Through our supplier relationships, we are able to obtain certain merchandise at better prices and quantities than Quality Fragrance Group can. Wholesale sales are expected to fluctuate due to market conditions, the availability of desirable merchandise and the demand of our wholesale customers, including Model Reorg. The volume of wholesale sales in fiscal year 2007 was significantly higher than any of our most recent five years, and should not necessarily be considered indicative of wholesale sales volume to be expected in future quarters or years. See further discussion at Note 6 to our Consolidated Financial Statements for fiscal 2007.

Cost of Goods Sold:

Cost of goods sold include the cost of merchandise sold, inventory valuation adjustments, inventory shortages, damages and freight charges.

		For the year ended (\$ in thousands)			
	February 2, 2008	Febr	ruary 3, 2007	Percentage Increase	
Retail	\$ 135,405	\$	126,808	6.8%	
Wholesale	54,564		12,871	323.9%	
Total cost of goods sold	\$ 189,969	\$	139,679	36.0%	

74

Gross Profit:

		For the year ended (\$ in thousands)		
	February 2, 2008	Febr	ruary 3, 2007	Increase
Retail	\$ 108,614	\$	102,975	5.5%
Wholesale	3,251		954	240.7%
Total gross profit	\$ 111,865	\$	103,929	7.6%

Total gross profit increased principally as a result of higher sales in both the wholesale and retail divisions. In calculating gross profit, we exclude the costs related to our distribution center. These costs are included in selling, general and administrative expenses. Our gross profit and gross profit percentages may not be comparable to those of other retailers, which may include the costs related to their distribution network in costs of goods sold, rather than selling, general and administrative expenses.

Gross Profit Margin Percentages:

	For t	For the year ended		
	February 2, 2008	February 3, 2007		
Retail	44.5%	44.8%		
Wholesale	5.6%	6.9%		
Gross profit margin	37.1%	42.7%		

Wholesale sales yield a lower gross margin than retail sales. The decrease in gross margin on retail sales resulted principally from increased promotional activities during the fourth quarter of fiscal year 2007, offset to a lesser extent by increases in retail selling prices on selected merchandise. Promotional activities were increased during the fourth quarter due to an anticipated decrease in mall traffic as economic conditions slowed.

Operating Expenses:

Selling, general and administrative expenses include payroll and related benefits for our store operations, field management, distribution center and corporate office; rent, common area maintenance, real estate taxes and utilities for our stores, distribution center and corporate office; advertising, insurance, supplies, professional fees and other administrative expenses. Distribution center costs included in selling, general and administrative expenses in fiscal years 2007 and 2006 were approximately \$2.8 million and \$2.6 million, respectively.

		For the year ended (\$ in thousands)		
	February 2, 2008	Febru	uary 3, 2007	Percentage Increase (Decrease)
Selling, general and administrative	\$ 99,974	\$	89,004	12.3%
Depreciation and amortization	6,197		4,797	29.2%
Total operating expenses	106,171		93,801	13.2%
Income from operations	\$ 5,694	\$	10,129	(43.8)%

The majority of our selling, general and administrative expenses relate to the retail division. The increase in selling, general and administrative expenses is attributable primarily to the increase in the number of new stores and the additional payroll, occupancy and store opening expenses

needed to operate these stores. Of the total \$11.0 million increase in selling, general and administrative expenses in fiscal year 2007, \$4.4 million related to stores that opened during 2007 and \$3.6 is attributable to stores that opened during 2006 were open for a full year

75

in 2007. In addition, \$1.2 million relates to Merger expenses incurred in fiscal year 2007. These expenses consist primarily of financial advisory, legal and due diligence fees. As a percentage of retail sales, selling, general and administrative expenses increased from 38.7% to 41.0%, largely due to the opening of these new stores and these Merger expenses. In fiscal 2007, the 75 new stores that were opened during fiscal years 2007 and 2006 in the aggregate negatively impacted income from operations, in part as a result of lower than expected sales.

Depreciation and amortization expenses in fiscal year 2007 increased by \$1.4 million from fiscal year 2006 primarily as a result of property and equipment additions associated with the opening of new retail stores.

Other Expenses

		roru	ie year enueu	
		(\$ in	thousands)	
				Percentage
	February 2, 2008	Februa	ary 3, 2007	Increase
Interest expense	\$ 4.728	\$	4.495	5.2%

The increase in interest expense is due to higher interest rates on average in 2007 compared with 2006. The interest rates on the majority of our borrowings are at variable rates. Also, the average balance on our credit facility increased to \$40.3 million in 2007 compared to \$34.3 million in 2006, while the balance on the Convertible Note was unchanged.

Income Tax Provision:

	For the ye		
	(\$ 111 11101		Percentage
February 2, 2008	February 3,	, 2007	Decrease
\$ (960)	\$ (1	1.244)	(22.8)%

The effective tax rate in fiscal year 2007 differs from the federal statutory rate due to non-deductible permanent items, primarily expenses incurred related to the Merger.

Fiscal Year 2006 Compared to Fiscal Year 2005

Revenues:

For the year ended (\$ in thousands)

For the year ended

						Percentage
		Percentage of			Percentage of	Increase
	February 3, 2007	Revenues	Janu	ary 28, 2006	Revenues	(Decrease)
Retail	\$ 229,783	94.3%	\$	215,841	92.4%	6.5%
Wholesale	13,826	5.7%		17,853	7.6%	(22.6)%
Total Revenues	\$ 243,609	100.0%	\$	233,694	100.0%	4.2%

In fiscal year 2006 total revenue increased by \$9.9 million or 4.2%, compared with the prior year. Total retail sales increased by \$13.9 million or 6.5%, of which 1.3% related to the fifty-third week of fiscal year 2006. Excluding the effect of the fifty-third week, total retail sales increased 5.2%, which included comparable store sales increases of 2.3%. The average number of stores operated increased from 234 during fiscal year 2005 to 250 in fiscal year 2006. The improvements in retail sales during fiscal year 2006 were due to the greater availability of merchandise brands, the quantity of products available for sale, and the offer to our customers of more competitive retail prices. The result was an increase in the number of customer transactions, the average dollar sale per transaction and the retail selling prices on selected merchandise.

Wholesale sales decreased by \$4.0 million or 22.6% as demand for certain wholesale products were less than anticipated and less than the prior year. Virtually all wholesale sales were made to Quality Fragrance Group. Through our supplier relationships, we were able to obtain certain merchandise at better prices and quantities than Quality Fragrance Group.

Cost of Goods Sold:

Cost of goods sold include the cost of merchandise sold, inventory valuation adjustments, inventory shortages, damages, and freight charges.

			he year ended n thousands)	
				Percentage
				Increase
	February 3, 2007	Janu	ary 28, 2006	(Decrease)
Retail	\$ 126,808	\$	120,487	5.2%
Wholesale	12,872		16,706	(23.0)%
Total cost of goods sold	\$ 139,680	\$	137,193	1.8%
Gross Profit:				

			ne year ended thousands)	
				Percentage
				Increase
	February 3, 2007	Janua	ary 28, 2006	(Decrease)
Retail	\$ 102,975	\$	95,354	8.0%
Wholesale	954		1,147	(16.8)%
Total gross profit	\$ 103,929	\$	96,501	7.7%

Total gross profit increased principally as a result of higher sales and gross profit in the retail division, offset by the lower sales and lower gross profit in the wholesale division. In calculating gross profit, we exclude the costs related to our distribution center. These costs are included in selling, general and administrative expenses.

Gross Profit Margin Percentages:

	For	For the year ended		
	February 3, 2007	January 28, 2006		
Retail	44.8%	44.2%		
Wholesale	6.9%	6.4%		
Gross profit margin	42.7%	41.3%		

The increase in gross margin on retail sales resulted principally from increases in retail selling prices on selected merchandise as well as lower cost of merchandise realized over the past twelve months.

Operating Expenses:

Selling, general and administrative expenses include payroll and related benefits for our store operations, field management, distribution center and corporate office; rent, common area maintenance, real estate taxes and utilities for our stores, distribution center and corporate office; advertising, insurance, supplies, professional fees and other administrative expenses. Distribution center costs included in selling, general and administrative expenses in fiscal years 2006 and 2005 were approximately \$2.6 million and \$2.4 million, respectively.

			he year ended 1 thousands)	
				Percentage
	February 3, 2007	Janua	ary 28, 2006	Increase (Decrease)
Selling, general and administrative	\$ 89,004	\$	81,006	9.9%
Depreciation and amortization	4,797		4,830	(0.7)%
Total operating expenses	93,801		85,836	9.3%
Income from operations	\$ 10,129	\$	10,665	(5.0)%

The majority of our selling, general and administrative expenses relate to the retail division. The increase in selling, general and administrative expenses is attributable primarily to the increase in the number of new stores and the additional payroll, occupancy and store opening expenses needed to operate these stores. As a percentage of retail sales, selling, general and administrative expenses increased from 37.5% to 38.7%, largely due to the opening of these new stores. In fiscal 2006, the 36 new stores that were opened in the aggregate negatively impacted income from operations, in part as a result of lower than expected sales, whereas in fiscal 2005, the 23 new stores opened in that year in the aggregate contributed positive income from operations. Increases in wholesale sales volume generally result in nominal increases in selling, general and administrative expenses. Also, in fiscal 2006 we accrued \$0.8 million for the settlement of a legal complaint filed against us.

Depreciation and amortization expenses in fiscal year 2006 were reduced by \$0.3 million from fiscal year 2005 primarily as a result of property and equipment associated with older retail stores and software costs associated with year 2000 upgrades that have become fully amortized.

Other Expenses:

		(\$ in thousands)	
			Percentage
	February 3, 2007	January 28, 2006	Increase
terest expense	\$ 4,495	\$ 3.877	15.9%

For the year ended

The increase in interest expense is due to higher interest rates in 2006 compared with 2005, as the average balance on our credit facility decreased and the balance on the Convertible Note was unchanged. The interest rates on the majority of our borrowings are at variable rates.

Income Tax (Provision) Benefit:

| For the year ended (\$ in thousands) | February 3, 2007 | January 28, 2006 | Income tax (provision) benefit | \$ (1,244) | \$ 8,471 |

In fiscal year 2006, management determined that it is more likely than not that certain deferred tax assets of our Puerto Rican subsidiary, which previously had a full valuation allowance, will be realized. Accordingly, the reversal of the valuation allowance on these assets of approximately \$2.7 million has been reflected as a benefit in the tax provision for fiscal year 2006. Also, in fiscal year 2006, we recorded an adjustment to the net operating losses of our Puerto Rican subsidiary of approximately \$1.8 million as a result of an ongoing Puerto Rican tax audit, which resulted in a decrease to the gross deferred tax asset for the Puerto Rican net operating loss but also resulted in a corresponding decrease to the valuation allowance for this item.

Income tax benefits recorded in fiscal year 2005 resulted from management s assessment that it was more likely than not that we would realize the benefit of certain deferred tax assets. The prior year s valuation allowance was reduced by approximately \$11.9 million during the year ending January 28, 2006 due to

Table of Contents

management s determination that approximately \$10.3 million of our deferred tax assets would be utilized. As a result of the reversal of the valuation allowance from the prior fiscal year, a net tax benefit of \$8.5 million was reflected in our financial statements for fiscal year 2005.

Comparison of the Thirteen Weeks Ended May 3, 2008 with the Thirteen Weeks Ended May 5, 2007.

Net sales increased 17.4% from \$48.1 million in the thirteen weeks ended May 5, 2007 to \$56.5 million in the thirteen weeks ended May 3, 2008. The increase in sales was due to increase of \$4.2 million in each of wholesale and retail sales.

Retail sales increased 9.9% from \$42.5 million in the thirteen weeks ended May 5, 2007 to \$46.7 million in the thirteen weeks ended May 3, 2008. The average number of stores operated increased to 308 in the first quarter of fiscal 2008 from 271 in the prior year period, which resulted in a portion of the increase in retail sales. Perfumania s comparable store sales increased by 3%. The average retail price per unit sold during the thirteen weeks ended May 3, 2008 increased 9.1% compared with the prior year period and the total number of units sold increased by 1.1%. We attribute the increase in the average retail price per unit sold to increases in the retail selling prices on selected merchandise as well as changes in our product mix. The number of units sold was affected by increasing softness in the United States economy and the resulting weak mall traffic.

Wholesale sales increased 73.8% from \$5.6 million in the thirteen weeks ended May 5, 2007 to \$9.8 million in the thirteen weeks ended May 3, 2008. Substantially all wholesale sales during the first quarter of fiscal years 2008 and 2007 were made to Model Reorg s subsidiaries. The increase in wholesale sales is due to significantly more merchandise availability at better prices than had been available from our suppliers in the first quarter of 2007, and an increase in the demand of Model Reorg and its subsidiaries product needs. Quarterly wholesale sales are expected to fluctuate due to market conditions, the availability of desirable merchandise and the demand of our wholesale customers, including Model Reorg. The volume of wholesale sales in the thirteen weeks ended May 3, 2008 should not necessarily be considered indicative of wholesale sales volume to be expected in future quarters. See further discussion at Note 8 to our Consolidated Financial Statements for the first quarter of fiscal 2008.

Gross profit increased 10.8% from \$20.3 million in the thirteen weeks ended May 5, 2007 (42.2% of total net sales) to \$22.5 million in the thirteen weeks ended May 3, 2008 (39.8% of total net sales). Of the \$2.2 million increase in gross profit, \$2.0 related to retail sales and \$0.2 million related to wholesale sales. The increase in gross profit was due to increases in both retail and wholesale sales volume as discussed above. As a percentage of net sales, total gross profit in the thirteen weeks ended May 3, 2008 decreased as compared to the thirteen weeks ended May 5, 2007 due to the higher ratio of wholesale sales as a percentage of net sales in the thirteen weeks ended May 3, 2008 versus the comparable period last year. Retail gross profit as a percentage of retail sales was 46.8% for both thirteen week periods ended May 3, 2008 and May 5, 2007. Wholesale gross profit as a percentage of wholesale sales was 6.5% and 7.8% for the thirteen weeks ended May 3, 2008 and May 5, 2007, respectively. Gross margin on wholesale sales decreased due to the change in the product assortment sold in the thirteen weeks ended May 3, 2008 compared to the comparative thirteen week period in fiscal 2007.

Selling, general and administrative expenses increased 16.2% from \$21.1 million in the thirteen weeks ended May 5, 2007 to \$24.5 million in the thirteen weeks ended May 3, 2008. The increase was largely attributable to the increase in new stores and the additional payroll, occupancy and store opening expenses needed to operate these stores. Included in selling, general and administrative expenses for the thirteen weeks ended May 3, 2008 are \$0.2 million of expenses related to the Merger. Depreciation and amortization was approximately \$1.8 million in the thirteen weeks ended May 3, 2008 compared to \$1.4 million for the thirteen weeks ended May 5, 2007. The increase is attributable to depreciation incurred on new stores that have opened over the past twelve months offset by lower depreciation on certain stores that became fully depreciated during the same time period.

79

Interest expense was approximately \$0.9 million for the thirteen weeks ended May 3, 2008, compared with approximately \$1.1 million in the comparable period of 2007. The decrease in interest expense was due to lower interest rates on our borrowings offset by a higher average balance on our bank line of credit during the thirteen weeks ended May 3, 2008 compared with the comparable period of 2007. The majority of our borrowings incur interest based on either prime or Libor and these interest rates have decreased over the past year.

An income tax benefit of \$1.8 million was recorded as a result of the Company s net loss during the thirteen weeks ended May 3, 2008, compared with a tax benefit of \$1.3 million during the comparable period of 2007. The Company s effective tax rate for each of the thirteen-week periods ended May 3, 2008 and May 5, 2007 was a tax benefit of 38%.

As a result of the foregoing, our net loss increased to approximately (\$2.9) million in the thirteen weeks ended May 3, 2008, compared to a net loss of (\$2.1) million in the thirteen weeks ended May 5, 2007. Net loss per share for the first fiscal quarter of 2008 and 2007 was (\$0.95) and (\$0.68), respectively.

Liquidity and Capital Resources

Our principal funding requirements are for inventory purchases, opening new stores and renovation of existing stores. During fiscal years 2007 and 2006, we financed these requirements primarily through cash flows from operations, borrowings under our line of credit and credit terms from our vendors, including extended terms from our affiliates. During the first thirteen weeks of fiscal 2008, these capital requirements generally were satisfied through borrowings under our credit facility.

A summary of our cash flows for the year ended February 2, 2008 and the thirteen-week period ended May 3, 2008 is as follows:

Thirteen Weeks Ended		Year Ended		
May 3, 2008	May 5, 2007	February 2, 2008	Febru	ary 3, 2007
	(\$ i	n thousands)		
\$ (6,388)	\$ (6,271)	\$ 8,734	\$	2,625
(3,024)	(2,563)	(14,593)		(9,797)
9,657	9,398	5,611		7,194
245	564	(248)		22
1,035	1,283	1,283		1,261
\$ 1,280	\$ 1,847	\$ 1,035	\$	1,283
	\$ (6,388) (3,024) 9,657 245 1,035	May 3, 2008 May 5, 2007 (\$ i. \$ (6,388) \$ (6,271) (3,024) (2,563) 9,657 9,398 245 564 1,035 1,283	May 3, 2008 May 5, 2007 (\$ in thousands) \$ (6,388) \$ (6,271) \$ 8,734 (3,024) (2,563) (14,593) 9,657 9,398 5,611 245 564 (248) 1,035 1,283 1,283	May 3, 2008 May 5, 2007 (\$ in thousands) February 2, 2008 (\$ in thousands) February 2, 2008 (\$ in thousands) \$ (6,388) \$ (6,271) \$ 8,734 (\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$

Perfumania s senior secured credit facility provides for borrowings of up to \$60 million, depending on our levels of eligible inventories. The credit facility expired on May 12, 2008. In May 2008, we obtained a ninety day extension of our current facility to August 11, 2008, and the lender also made available to us a commitment to extend the facility through May 2009 that we did not execute. Advances under the line of credit are based on a formula of eligible inventories and bear interest at a floating rate ranging from (a) prime to prime plus 1.25% or (b) LIBOR plus 2.5% to 3.75% depending on a financial ratio test. As of May 3, 2008, \$42.6 million was outstanding under the line of credit and \$12.0 million was available to support normal working capital requirements and other general corporate purposes based upon our eligible borrowing base, net of the restrictions

Table of Contents

concerning minimum undrawn availability discussed below. Advances are secured by a first lien on all assets of Perfumania. The credit facility contains limitations on additional borrowings, capital expenditures and other items, and contains various covenants including a fixed charge coverage ratio, a leverage ratio and capital expenditure limits as defined. The credit facility also prohibits us from paying cash dividends. We are required to maintain, at all times, a minimum undrawn availability of not less than \$5,000,000 and a monthly average undrawn availability, as defined, of not less than \$7,500,000. In May 2008, the fixed charge covenant and the covenant concerning the limitation on capital expenditures were amended with an effective date of February 1, 2008. Perfumania was in compliance with all covenant requirements as of May 3, 2008 and expects to be in compliance through the extension period discussed above.

The Merger is subject to a number of terms, one of which includes the requirement that the combined company obtain an asset-based credit facility substantially as described in Post-Merger Financing. Failure of the Merger to close prior to the expiration of our current credit facility on August 11, 2008 will result in our having to obtain an additional extension to the term of this facility or an alternative source of financing. Management believes that we would be able to obtain either; however no assurance can be given that our efforts would be successful.

We also have outstanding the Convertible Note in the principal amount of \$5.0 million, payable to Stephen and Glenn Nussdorf, as described in Certain Relationships and Related Person Transactions. The Convertible Note matures in January 2009. If we are unable to extend its due date beyond January 2009, without considering the impact of the Merger, we may be unable to finance our business.

At May 3, 2008, we had a working capital deficit of approximately \$7.6 million, compared to working capital deficit of approximately \$2.7 million at February 2, 2008. The change since February 2, 2008 was primarily due to the net loss during the current period and increased spending on store construction in the first thirteen weeks of fiscal 2008. Our business follows a seasonal pattern as purchases of fragrances as gift items increases during the holiday season, resulting in significantly higher fourth quarter retail sales. The seasonality of our operations leads to significant fluctuations in certain asset and liability accounts, primarily inventories, accounts payable, affiliates and non-affiliates and bank line of credit, between fiscal year end and subsequent interim periods.

Net cash used in operating activities during the thirteen weeks ended May 3, 2008 was approximately \$6.4 million, compared with approximately \$6.3 million used in operating activities during the same period of the prior year. Our accounts payable to affiliates and non-affiliates increased due to the timing of payments to our merchandise vendors and because of increased inventory purchases during the thirteen weeks ended May 3, 2008 compared with the same period last year. The increase in inventory is due to planned purchases for store growth and selective opportunistic purchases of desirable merchandise. Our purchases from related parties, including Model Reorg s subsidiaries, are generally payable in 90 days; however due to the seasonality of our business these terms are generally extended. Related party accounts are historically brought closer to terms at the end of the holiday season; however we are dependant upon these extended terms for much of our liquidity during the year.

In fiscal year 2007, net cash provided by operating activities was approximately \$8.7 million and was generated primarily by net earnings adjusted for non-cash depreciation and amortization expenses, an increase in accounts payable, affiliates, due to the timing of payments to affiliates, offset by an increase in our inventory levels. Our inventory levels increased primarily due to the opening of 39 new stores in fiscal year 2007 as well as new vendor relationships developed with direct manufacturers in 2007 that have resulted in increased availability of additional desirable fragrance brands.

Investing activities generally represent spending for new store openings and the renovation of existing stores. Net cash used in investing activities was approximately \$3.0 million in the first thirteen weeks ended May 3, 2008, compared to \$2.6 million in the thirteen weeks ended May 5, 2008. The current period s investing

81

activities primarily represented spending for renovation of existing stores and new stores that either opened during the thirteen weeks ended May 3, 2008 or that are scheduled for completion during fiscal year 2008. During the thirteen weeks ended May 3, 2008, Perfumania opened 11 new stores and relocated two existing stores, compared to five new store openings and four store relocations during the comparable period in fiscal 2007. Net cash used in investing activities in fiscal year 2007 was approximately \$14.6 million. During fiscal year 2007 we opened 39 new stores and remodeled 52 stores. At May 3, 2008, Perfumania operated 313 stores, compared to 272 stores as of May 5, 2007. We currently plan to open approximately 40 new stores and close one store during the remainder of fiscal year 2008. We also intend to continue improving the appearance of our existing stores and opening new stores in markets that present opportunities for growth. We expect capital expenditures (without giving effect to the Merger) to be in the range of \$14 to \$16 million in fiscal year 2008 and we believe that these capital requirements will be satisfied by borrowings under any new credit facility and from cash flows from operations.

Net cash provided by financing activities during the first thirteen weeks of fiscal 2008 was approximately \$9.7 million, primarily from borrowings under our line of credit, compared with approximately \$9.4 million for the same period in the prior year. In fiscal year 2007, net cash provided by financing activities was approximately \$5.6 million, primarily due to the increase in our net bank borrowings. The increase was primarily due to the increase in vendor payments at year-end 2007 compared with 2006. Payments to vendors are funded by bank borrowings, which increases our cash provided by financing activities.

Based on past performance and current expectations, we believe that our cash balances and the available borrowing capacity under either the replacement Senior Credit Facility to be established upon the closing of the Merger or, should the Merger not occur, our existing credit facility (providing an additional maturity date extension is received), continued extended terms from our affiliates and our projected future operating results will generate sufficient liquidity to support our working capital needs, capital expenditures and debt service for the next twelve months. However, there can be no assurance that management s plans and expectations will be successful.

The following table provides a summary of our future contractual obligations as of February 2, 2008. Certain of these obligations are reflected in our consolidated balance sheet as of February 2, 2008, while others are disclosed as future obligations. See also Approval to Issue Shares and Warrants - Post-Merger Financing for a discussion of the Senior Credit Facility we expect that the combined company will enter into upon the closing of the Merger.

	Payments due by period				
	(\$ in thousands)				
		less than			more than
Contractual Obligations	Total	1 year	1-3 years	3-5 years	5 years
Bank line of credit	\$ 32,841	\$ 32,841			
Subordinated convertible note payable affiliate(1)	5,323	5,323			
Capital lease obligations(1)	12,858	1,210	2,472	2,653	6,523
Operating lease obligations(2)	110,371	19,129	33,156	25,172	32,914
Total	\$ 161,393	\$ 58,503	\$ 35,628	\$ 27,825	\$ 39,437

- (1) Debt amounts include principal maturities and expected interest payments. Rates utilized to determine interest payments for variable rate debt are based on an estimate of future interest rates.
- (2) Maintenance, real estate taxes and contingent rent obligations are excluded. *Off-Balance Sheet Arrangements*

We have no off-balance sheet arrangements as defined by Item 303(a)(4) of SEC Regulation S-K.

Table of Contents

Recent Accounting Standards

In March 2008, the FASB issued SFAS 161, Disclosures about Derivative Instruments and Hedging Activities an amendment of FASB Statement No. 133 (SFAS 161). SFAS 161 requires enhanced disclosures regarding derivatives and hedging activities, including: (a) the manner in which an entity uses derivative instruments; (b) the manner in which derivative instruments and related hedged items are accounted for under Statement of Financial Accounting Standards No. 133, Accounting for Derivative Instruments and Hedging Activities; and (c) the effect of derivative instruments and related hedged items on an entity s financial position, financial performance, and cash flows. SFAS 161 is effective for financial statements issued for fiscal years and interim periods beginning after November 15, 2008. As SFAS 161 relates specifically to disclosures, it will have no impact on our results of operations, financial position or cash flows.

In December 2007, the FASB issued SFAS No. 141(R), *Business Combinations* (SFAS 141(R)). SFAS 141(R) establishes principles and requirements for how an acquirer recognizes and measures in its financial statements the identifiable assets acquired, the liabilities assumed, the goodwill acquired, and any noncontrolling interest in the acquiree. This statement also establishes disclosure requirements to enable the evaluation of the nature and financial effect of the business combination. SFAS 141(R) is effective for fiscal years beginning after December 15, 2008. We are in the process of evaluating the effect that the adoption of SFAS 141(R) will have on our results of operations, financial position, and cash flows.

In December 2007, the FASB issued SFAS No. 160, *Noncontrolling Interests in Consolidated Financial Statements an amendment of ARB No. 51* (SFAS 160). SFAS 160 amends ARB 51 to establish accounting and reporting standards for the noncontrolling interest in a subsidiary and for the deconsolidation of a subsidiary. It clarifies that a noncontrolling interest in a subsidiary is an ownership interest in the consolidated entity that should be reported as equity in the consolidated financial statements. SFAS 160 is effective for fiscal years beginning on or after December 15, 2008. We are in the process of evaluating the effect that the adoption of SFAS 160 will have on our results of operations, financial position, and cash flows.

In February 2007, the FASB issued SFAS 159, *The Fair Value Option for Financial Assets and Financial Liabilities Including an amendment of FASB Statement No. 115* (SFAS 159). SFAS 159 permits entities to choose to measure eligible items at fair value at specified election dates and report unrealized gains and losses on items for which the fair value option has been elected in earnings at each subsequent reporting date. SFAS 159 is effective for fiscal 2008. The adoption of SFAS 159 did not have any effect on our results of operations, financial position or cash flows.

In September 2006, the FASB issued SFAS No. 157, Fair Value Measurements (SFAS 157). SFAS 157 defines fair value, establishes a framework for measuring fair value and expands disclosure of fair value measurements. SFAS 157 applies under other accounting pronouncements that require or permit fair value measurements and, accordingly, does not require any new fair value measurements. SFAS 157 is effective for the first interim period beginning in fiscal 2008 for financial assets and liabilities, as well as for any other assets and liabilities that are carried at fair value on a recurring basis in financial statements. In November 2007, the FASB provided a one year deferral for the implementation of SFAS 157 for other nonfinancial assets and liabilities. The adoption of SFAS 157 did not have a material effect on our results of operations, financial position or cash flows.

Changes in Foreign Exchange Rates; Inflation

Although large fluctuations in foreign exchange rates could have a material effect on the prices we pay for products purchased from outside the United States, such fluctuations have not been material to our results of operations to date. Transactions with foreign suppliers are denominated in United States dollars. We believe inflation has not had a material impact on our results of operations and we are generally able to pass through cost increases by increasing sales prices.

83

Table of Contents

Quantitative and Qualitative Disclosures about Market Risk

We conduct business primarily in the United States, and the functional currency for our overseas transactions is the United States dollar. As a result, we are not at risk of any foreign exchange translation exposure on a prospective basis.

Our exposure to market risk for changes in interest rates relates primarily to our bank line of credit. The bank line of credit bears interest at a variable rate, as discussed above under Liquidity and Capital Resources. We mitigate interest rate risk by continuously monitoring the interest rates and reacting to changes in LIBOR and prime rates. As a result of borrowings associated with our operating and investing activities, we are exposed to interest rate risk. As of February 2, 2008 and February 3, 2007, our primary source of funds for working capital and other needs was a line of credit that provides for borrowings up to \$60 million.

Of the \$45.4 million and \$39.8 million of short-term and long-term borrowings on our balance sheet as of February 2, 2008 and February 3, 2007, respectively, approximately 16.7% and 19.9%, respectively, represented fixed rate instruments. The line of credit bears interest at a floating rate ranging from (a) prime to prime plus 1.25%, or (b) LIBOR plus 2.5% to 3.75% depending on financial ratio tests. For fiscal year 2007, the credit facility accrued interest at an average rate of 8.1%. A hypothetical 10% adverse move in interest rates would have increased fiscal year 2007 interest expense by approximately \$0.3 million.

During the quarter ended May 3, 2008, there were no material changes in our market risks as described above.

INFORMATION ABOUT MODEL REORG

Business of Model Reorg

General

Model Reorg, Inc., a New York corporation, was incorporated in 1997. Model Reorg operates as a wholesaler, retailer and manufacturer of designer fragrances through wholly-owned subsidiaries that include Quality King Fragrance, Inc. (Quality Fragrance Group), Scents of Worth, Inc., Five Star Fragrance Company, Inc., and Jacavi, a fragrance distributor that Model Reorg acquired in October 2007. The wholesale division primarily distributes nationally advertised designer fragrances to mass merchandisers, drug store chains, specialty outlets, distributors and Perfumania. The retail division sells products in retail stores on a consignment basis. The manufacturing division owns and licenses designer fragrance brands and contracts to third parties the manufacture of owned and licensed brands.

Information for each of the three industry segments in which Model Reorg operates (wholesale distribution, retail and manufacturing) is provided in Note 16 and Note 5 to Model Reorg s Consolidated Financial Statements for October 31, 2007 and April 30, 2008, respectively. Each of the segments has its own assets, liabilities, revenues, and operating expenses. All intersegment transactions are eliminated in consolidation.

In recent years Model Reorg has purchased an increasing portion of its product directly from manufacturers rather than from distributors. This has increased product offerings and opened up more sales channels because of competitive pricing and because certain retailers deal only with companies that purchase product direct from the manufacturer.

Wholesale Business

The wholesale division, which operates through Quality Fragrance Group, distributes designer fragrances to mass market retailers, drug and other chain stores, retail wholesale clubs, traditional wholesalers, and other distributors throughout the United States. It buys designer fragrances principally from the brand owner /

84

Table of Contents

manufacturer. Quality Fragrance Group strives to increase its selection of brands, sizes and price points in order to be a one stop shop for its customers. Its scope was expanded with the acquisitions of Northern Group, Inc. (Northern) in 2004 and Jacavi in 2007, which provided new relationships with manufacturers of several major fragrance brands.

Quality Fragrance Group s sales are principally to retailers such as Wal-Mart, Walgreens, Kohl s, Sears, Marshalls, JCPenney, Target, TJ Maxx, and E Com s subsidiary, Perfumania. Quality Fragrance Group also operates a direct sales department that services over 10,000 pharmacies and specialty stores, such as AmerisourceBergen and Cardinal Health, throughout the United States.

Mass market retail sales were approximately 40% and 42% of Model Reorg s total sales in fiscal 2007 and 2006, respectively, while wholesale sales were approximately 35% in both years. Approximately 20% and 17% of Quality Fragrance Group s net revenues in fiscal 2007 and fiscal 2006, respectively, were derived from sales to other Model Reorg subsidiaries, and approximately 7% and 9%, respectively, from sales to Perfumania. One Quality Fragrance Group customer, Wal-Mart, accounted for approximately 16% of Model Reorg s total unaffiliated revenues in fiscal 2007. Quality Fragrance Group provides extended payment terms to Perfumania, which has had a material impact on Model Reorg s working capital. Inventory practices do not significantly affect working capital, and Quality Fragrance Group invests working capital in inventory on an opportunistic basis to meet anticipated customer needs or to capture advantageous prices.

Retail Business

Scents of Worth has operated Model Reorg s retail business since 1997. It is the largest national designer fragrance consignment program, with contractual relationships to sell products on a consignment basis in approximately 3,000 stores, including nearly 1,400 Kmart locations, nationwide. Its other retail customers include Burlington Coat Factory, Filene s Basement, Loehmann s, Fred Meyer, Daffy s and Duane Reade. Scents of Worth was designated a critical vendor to Kmart following Kmart s reorganization in 2003. Scents of Worth determines the pricing and the products displayed in each of its retail consignment locations and pays a percentage of the sales proceeds to the retailer for its profit and overhead applicable to these sales. Overhead includes sales associate payroll and benefits, rental of fragrance space and, in some instances, an inventory shrink allowance. Consignment fees vary depending in part on whether Scents of Worth or the retailer absorbs inventory shrinkage. Scents of Worth s total sales represented approximately 23% and 22% of Model Reorg s total sales for fiscal 2007 and fiscal 2006, respectively.

Manufacturing Business

The manufacturing division is operated by Five Star, which owns and licenses designer and other fragrance brands, paying royalties to the licensors based on a percentage of sales. Certain licenses contain minimum royalty provisions. Five Star manufactures, on behalf of Perfumania, Perfumania s Jerome Privee product line, which includes bath and body products and scented candles. Five Star s manufacturing is sub-contracted out to third party manufacturers. Its owned and licensed brands are sold principally through Model Reorg s wholesale and retail divisions. Approximately 68% and 66% of Five Star s fiscal 2007 and fiscal 2006 revenues, respectively, were derived from sales to other Model Reorg subsidiaries, and approximately 29% and 30%, respectively, were derived from sales to Perfumania. The majority of the unrelated party sales are to international customers. The rising cost of oil in recent years has had a direct impact on the cost of raw materials, as oil is used in the production of the plastic packaging materials Five Star purchases. In reaction to this, Five Star has increased its sourcing of components from the Far East in order to keep the cost of goods as low as possible.

Seasonality and Quarterly Results

Model Reorg s business is highly seasonal, with the most significant activity occurring from September through December each year. Quality Fragrance Group s sales to retailers are stronger during the months of September through November, since retailers need to receive merchandise well before the holiday season begins,

85

Table of Contents

with 35% of its total revenues being generated during these three months. Scents of Worth s revenues are the greatest in December, with approximately 25% of its revenues being generated this month, as is typical for a retail operation. Historically, Model Reorg has typically generated income from operations during all 12 months of the fiscal year. However, depending on outstanding debt levels, interest on debt financing may exceed income from operations in a particular month. This may happen in one of the lower revenue months for Model Reorg, such as February.

Strategy

Model Reorg s business strategy is to use its experience in the fragrance industry, knowledge of the fragrance market, and business relationships to procure products on favorable terms, enabling it to distribute its products to customers or its consignment retail locations at competitive prices. In addition, Model Reorg supplements its offerings with owned or licensed prestige designer fragrances brands. It seeks to increase the portfolio of brands for both wholesale distribution and retail sale by presenting a diverse sales opportunity for a designer s brand, thereby enhancing its purchasing opportunities.

Model Reorg emphasizes future growth by broadening its product offering to wholesale customers while also growing the Scents of Worth retail consignment business. It also expects to take advantage of opportunities for the Five Star manufacturing business to license or purchase mature designer brands that do not require significant additional expenditure to create retail market demand.

Suppliers

In recent years, Model Reorg has purchased approximately 70% of its fragrances directly from brand owners / manufacturers and 30% from distributors. Its suppliers include most of the dozen largest fragrance manufacturers in the United States. The distributors represent, for the most part, long-standing relationships and many are also customers of Model Reorg. Model Reorg maintains a regular dialogue with all designer fragrance brand manufacturers directed toward broadening its product offering to both wholesale and consignment retail customers. Model Reorg believes that having both wholesale and retail customers is desirable to most designer fragrance brand manufacturers and enhances its opportunities to further expand these relationships. In addition, Model Reorg s success in obtaining product at favorable prices has been enhanced by its ability to commit significant capital quickly to any single purchase and purchase in bulk quantities or, in the right circumstances, to prepay vendors for products purchased in exchange for a substantial discount. In addition, to take advantage of market opportunities Model Reorg often purchases products in excess of anticipated short term demand, which it holds in inventory for a longer-than-average time.

Marketing and Sales

Model Reorg s marketing activities largely involve advertising in trade publications that focus on the proprietary brands. Quality Fragrance Group and Scents of Worth work with retailers to develop in store promotions employing signage, displays or unique packaging to merchandise and promote products. Both companies also work with retailers to develop ad campaigns for specific events as required by the retailers, e.g., mailers, inserts and national print advertising.

Sales representatives maintain regular dialogue with customers to generate selling opportunities and to assist them in finding scarce products at low prices. All sales personnel have access to current inventory information that is generally updated with each order, allowing immediate order confirmation to customers and ensuring that ordered products are in stock for prompt shipment. Model Reorg s proprietary management information system affords customers access to current information on price and product availability to enable them to better plan inventory investments. Customers may elect to receive this information in a variety of electronic formats that are updated daily, and can transmit orders electronically directly to Model Reorg s data processing system. The reports generated for customers by the management information system also provide information on

86

Table of Contents

manufacturers special price offerings and promotional programs to demonstrate the savings that can be generated so that customers can validate the value of purchasing from Model Reorg compared to other sources. Model Reorg believes that this gives its salespeople a competitive advantage in customer interactions.

Intellectual Property Rights

Model Reorg s portfolio of fragrance brands is of great importance to its business. Five Star owns the Lutece, Norell, Pavlova, Realm, Raffinee and Royal Secret brands, among others. It licenses designer and other fragrance brands, such as Bijan, Gale Hayman, Michael Jordan, Pierre Cardin, Royal Copenhagen, and Vicky Tiel, often acquiring exclusive worldwide distribution rights. Some of these licenses are renewable on a year-to-year basis, while the rest have terms that typically run from two to five years.

Model Reorg primarily relies on trademark law to protect its intellectual property rights. In addition to using registered trademarks covering licensed brands, Model Reorg has a proprietary portfolio of more than 40 U.S. registered trademarks and applications and more than 150 foreign registered trademarks and applications. All of the brands listed in the preceding paragraph are registered trademarks that are owned by Model Reorg or its licensors. U.S. trademark rights are generally renewable for as long as the mark is used. Model Reorg occasionally registers the copyright to packaging materials, and it also relies on trade secret and other contractual restrictions. From time to time, Model Reorg brings litigation against those who, in its opinion, infringe its proprietary rights, but there can be no assurance that either such efforts, or any contractual restrictions used, will be adequate or effective. Also, owners of other brands may, from time to time, allege that Model Reorg has violated their intellectual property rights, which may lead to litigation and material expense.

Competition

Competition varies among the markets in which Model Reorg competes. In selling to retailers, Model Reorg competes with a wide range of chains and large and small stores, as well as manufacturers, including some of Model Reorg s suppliers. In the wholesale business, Model Reorg competes with many distributors, of which Elizabeth Arden is the largest. Generally, the basis of competition is brand recognition, quality and price. Model Reorg believes that the most important reasons for its competitive success include its established relationships with manufacturers and large customers, popular recognition of its proprietary and licensed brands, and its efficient, low-cost sourcing strategy and ability to deliver products to consumers at competitive prices. Some of its competitors may enjoy competitive advantages, including greater financial resources that can be devoted to competitive activities, such as sales and marketing, brand development and strategic acquisitions.

Legal Proceedings

Model Reorg is subject to the same risks of litigation and administrative proceedings as described in Information about E Com Business of E Com Sources of Supply. In addition, Model Reorg is currently involved in other legal proceedings in the ordinary course of business. Management cannot presently predict the outcome of any these matters, although management believes that the ultimate resolution of these matters will not have a materially adverse effect on its financial condition, operations or cash flows.

Employees and Properties

At April 30, 2008, Model Reorg had 283 employees, of which 210 were involved in warehousing, 50 in marketing, sales and operations, and 23 in finance and administration. Substantially all the warehouse employees are represented by a union. Model Reorg has never experienced a work stoppage, strike or other interruption in business as a result of a labor dispute.

Model Reorg s principal executive offices are located at 35 Sawgrass Drive, Suite 2, Bellport, New York 11713, and its telephone number is (631) 866-4100. Model Reorg subleases 280,000 square feet of this new, 560,000 square foot facility from Quality King (see Certain Relationships and Related Person Transactions) and began using this space in December 2007.

87

Management s Discussion and Analysis of Financial Condition and Results of Operations

General

In October 2007, Model Reorg acquired by merger all of the equity of Jacavi in exchange for the issuance of 10.9 shares of Model Reorg common stock (11.25% of the outstanding shares after the transaction), valued at approximately \$10.8 million, to the former shareholders of Jacavi. The acquisition was accounted for as a purchase and the results of operations of Jacavi for the month of October 2007 are consolidated with Quality Fragrance Group and are reflected in Model Reorg s financial statements for the year ended October 31, 2007. The impact on overall results was immaterial.

Model Reorg continues to expand its manufacturer vendor relationships, which is expected to increase revenues by improving product mix for existing customers as well as attracting new customers. Model Reorg has implemented cost saving measures by sourcing components for the manufacturing process from overseas vendors and decreasing the use of outside consultants. It has realized synergies from the consolidation of Northern s and Jacavi s operations into Five Star and Quality Fragrance Group.

Model Reorg s revenues for the three and six-month periods ending April 30, 2008 decreased by 4.3% and increased by 5.5% respectively, compared with the same periods in fiscal 2007. The decrease in sales for the second quarter of fiscal 2008 was the result of the tightening of credit resources available to wholesale customers and a reduction in consumer retail spending. The increase in sales for the six months ended April 30, 2008 as compared to the same period in the prior year was related to sales to Perfumania of new product offering resulting from Model Reorg s additional manufacturer relationships established in fiscal 2008.

The following table sets forth selected items from Model Reorg s consolidated statements of income expressed as a percentage of total net sales for the periods indicated:

	Six Months End 2008	led April 30, 2007	FISCAL YEA 2007 2006		2005
Total net sales	100.0%	100.0%	100.0%	100.0%	100.0%
Total gross profit	28.6	28.6	27.6	27.5	25.7
Selling, warehouse, delivery and administrative					
expenses	18.6	17.6	18.4	17.0	17.3
Depreciation and amortization	0.4	0.5	0.4	0.5	0.8
(Recovery) provision on vendor advances			(0.7)	0.7	
Total operating expenses	19.0	18.1	18.1	18.2	18.1
Income from operations	9.6	10.5	9.5	9.3	7.6
Interest expense	3.0	3.6	3.9	4.3	3.5
Income before income taxes	6.6	6.9	5.6	5.0	4.1
Income taxes	2.8	2.8	2.2	2.0	1.6
Net income	3.8%	4.1%	3.4%	3.0%	2.5%

Critical Accounting Policies and Estimates

Model Reorg s consolidated financial statements have been prepared in accordance with GAAP. Preparation of these statements requires management to make judgments and estimates that affect the reported amounts of assets, liabilities, revenues and expenses. On an on-going basis, management evaluates its estimates, including those related to bad debts, inventories, asset impairments, sales returns and allowances, and other contingent asset and liabilities. As such, some accounting policies have a significant impact on amounts reported in these financial statements. The judgments and estimates can significantly affect financial results. Materially different amounts would be reported under different conditions or by using different assumptions. Model Reorg considers

Table of Contents

an accounting policy to be critical if it is both important to the portrayal of Model Reorg s financial condition and results of operations, and requires significant judgment and estimates by management in its application. Model Reorg has identified certain critical accounting policies that affect the significant estimates and judgments used in the preparation of its financial statements.

Revenue Recognition

Sales are recognized when title passes, which occurs either upon shipment of products, delivery to the customer or sale to the ultimate customers from consignment inventories. Allowances for estimated returns, rebate incentives and pricing adjustments are provided when sales are recognized and are recorded as either a reduction of sales or as selling expenses. Accruals and allowances are estimated based on available information from third party data.

Allowance for Doubtful Accounts

Each Model Reorg subsidiary maintains an allowance for doubtful accounts for the estimated losses resulting from the inability of its customers to make the required payments. Management reviews its receivables on a daily basis and monitors its receivable balances to ensure they are within the credit limits assigned to customers. Quality Fragrance Group insures its unrelated party receivables under a credit insurance policy. Management works very closely with the insurance company to ensure that all customers—credit is assigned within the guidelines of the policy limiting maximum annual loss exposure to the policy deductible. The policy is subject to a \$150,000 annual deductible. There are certain customers that are specifically insured and are assigned an insurable credit limit. All other accounts are insured up to 1.5 times their highest trailing 12 month credit, not to exceed \$700,000. Management reviews receivables to ensure they are collectible. If any accounts are deemed uncollectible, they are fully reserved. During fiscal 2007, uncollectible accounts totaled less than \$100,000.

Inventory and Reserves

Inventories are valued at the lower of cost or market with cost being determined using the weighted average cost method, which approximates first-in, first-out (FIFO). The cost of inventory includes product costs only. Model Reorg reviews its inventory on a regular basis for excess or potentially slow moving inventory based on prior sales, forecasted demands, historical experience and the identification of obsolete or damaged merchandise. It records adjustments to the carrying value of inventory to the lower of cost or market in accordance with management s assessment. If there are material changes to these estimates, additional write-offs could be necessary. Inventory shrinkage is estimated and reserved for at the end of each accounting period by reviewing past adjustments for inventory shrinkage as a percentage of total inventory. Inventory reserves as a percentage of total inventory for fiscal 2007 represented less than 1.5% of total inventory, which is a decrease of approximately 1% from fiscal 2006. This is a result of more detailed reviews of inventory obsolescence, increased cycle counts, and better management of inventory purchases. If reserves are not accounted for properly, inventories and cost of goods sold could be understated or overstated.

Impairment of Long-Lived Assets

When events or changes in circumstances indicate that the carrying values of long-lived assets may be impaired, an evaluation of recoverability is performed by comparing the carrying value of the assets to the projected future cash flows in addition to other quantitative and qualitative analysis. Inherent in this process is significant management judgment as to the projected cash flows and fair value of the assets and liabilities. Upon indication that the carrying value of such assets may not be recoverable, management will recognize an impairment loss as a charge against current operations. Factors that could trigger an impairment review include a significant underperformance relative to expected historical or projected future operating results or a negative industry or economic trend. Intangibles related to brand acquisitions are tested annually for impairment. The

89

Table of Contents

projected income from operations by brand over the next ten years have indicated that there is no impairment adjustment necessary. If any of these brands are either sold or discontinued prior to the intangible being fully amortized, it would require the write-off of the net book value of the intangible to the statement of income.

Goodwill

Statement of Financial Accounting Standard (SFAS) No. 142, Goodwill and Other Intangible Assets specifies that management must test goodwill for impairment whenever changes in circumstances suggest that the carrying amount may not be recoverable, but at least annually. There is a two step process for impairment testing of goodwill. The first step compares the fair value of the reporting unit with its carrying amount including goodwill. The second step, if necessary, measures the amount of the impairment. As of October 31, 2007, Model Reorg s balance sheet reflects two goodwill assets, one from the acquisition of Northern in October 2004 and the other from the acquisition of Jacavi in October 2007.

Sales Returns and Allowance

Quality Fragrance Group sells seasonal merchandise to specific mass market retailers on a guaranteed sales basis. Management estimates the potential returns liability for these sales and records a returns accrual as the sales are being made. Once management believes that all returns have been processed, any remaining accrual balance will be adjusted to the financials in the current period as a change resulting from the change in management s estimate. The sales returns and allowance reserves requires management to make significant estimates and judgments regarding the sell through of merchandise. These estimates are done by evaluating returns history and sales of similar products sold to the retailers during the prior year. If the potential returns are not accounted for in the period of the original sale, the sales may be overstated or understated in the period of reporting.

Many mass market retailers are given allowances for co-op advertising, markdowns, defective merchandise, new store opening and freight. Model Reorg requires all allowances to be documented for the period to which they relate and the basis of calculation, including proof of performance. When authorizing allowances, the profitability of each customer is reviewed to ensure that the allowance will not affect net profitability below approved guidelines. If promotional allowances are not accounted for in the proper period, this could cause a misstatement of sales and selling expenses between accounting periods.

Fiscal Year 2007 Compared to Fiscal Year 2006

Revenues

For the year ended October 31, (\$ in thousands)

					Percentage
		Percentage of		Percentage of	(Decrease)
	2007	Revenue	2006	Revenue	Increase
Quality Fragrance Group	\$ 245,493	74.9%	\$ 257,826	76.2%	(4.8)%
Scents of Worth	76,369	23.3%	73,990	21.9%	3.2%
Five Star	5,850	1.8%	6,545	1.9%	(10.6)%
Total revenues	\$ 327,712	100.0%	\$ 338,361	100.0%	(3.1)%

Sales to unaffiliated purchasers represented 92.3% of total revenues in fiscal 2007, compared with 90.4% in fiscal 2006. Quality Fragrance Group s annual decrease in revenues was attributable to decreases in sales to Perfumania of \$6.7 million, a decrease of 25.0% compared to the prior year, and to wholesale customers of \$5.6 million, a decline of 11.2%. Perfumania has continued to increase its sourcing of products direct from manufacturers, and wholesale distributors were affected by the slowdown in the economy, which has caused lenders to tighten their available credit resources.

Five Star also saw a decrease of \$0.5 million, or 9.6%, in sales to Perfumania as there were fewer new product launches in 2007. The decrease in wholesale sales relates to a tightening of the credit resources available for some of the wholesale customers, which decreases the amounts they are able to purchase. Related party sales to Perfumania were \$25.2 million and \$32.4 million in fiscal 2007 and 2006, respectively.

Direct Cost of Goods Sold

Direct cost of goods sold includes the cost of merchandise, inventory reserve adjustments and freight charges. Individual components of cost of goods sold were relatively consistent from 2006 to 2007, with the exception of a \$250,000 reduction in Scents of Worth s inventory reserve in 2006.

	For tl	For the year ended October 31,			
		(\$ in thousands))		
			Percentage		
	2007	2006	(Decrease) Increase		
Quality Fragrance Group	\$ 187,909	\$ 197,940	(5.1)%		
Scents of Worth	46,330	44,045	5.2%		
Five Star	2,964	3,339	(11.2)%		
Total cost of goods sold	\$ 237,203	\$ 245,324	(3.3)%		

Gross Profit

		(\$ in thousands	·)
			Percentage
			(Decrease)
	2007	2006	Increase
Quality Fragrance Group	\$ 57,584	\$ 59,886	(3.8)%
Scents of Worth	30,040	29,945	0.3%
Five Star	2,885	3,206	(10.0)%
Total gross profit	\$ 90,509	\$ 93,037	(2.7)%

For the year ended October 31,

The change in total gross profit was attributable to the decrease in total sales for 2007. In addition, Scents of Worth s gross profit dollars for fiscal 2007 reflect a greater amount of markdown dollars and promotions during fiscal 2007 than in fiscal 2006.

Gross Profit Percentages

	For the year ended	For the year ended October 31,		
	2007	2006		
Quality Fragrance Group	23.5%	23.2%		
Scents of Worth	39.3%	40.5%		
Five Star	49.3%	49.0%		
Gross profit percentage	27.6%	27.5%		

Total gross profit percentage for Model Reorg remained consistent in fiscal 2007 as compared to fiscal 2006.

Operating Expenses

Selling, warehouse, delivery and administrative expenses include payroll and benefits for warehouse, sales, purchasing and other administrative personnel, rent, utilities, shipping supplies, freight out, consignment fees, advertising and sales promotion, sales commissions, insurance and other administrative expenses.

		For the year ended October 31, (\$ in thousands)			
	2007	2006	Percentage Increase (Decrease)		
Selling, warehouse, delivery and administrative	\$ 60,113	\$ 57,548	4.5%		
Depreciation and amortization	1,411	1,721	(18.0)%		
(Recovery) provision on vendor advances	(2,367)	2,367	(200.0)%		
Total operating expense	\$ 59,157	\$ 61,636	(4.0)%		
Income from operations	\$ 31,352	\$ 31,401	(0.2)%		

The increase in selling, warehouse, delivery and administrative expenses for fiscal 2007 relates to the increase in consignment fees as related to the increase in Scents of Worth sales and the increase in rental and warehousing costs due to delays in the relocation and consolidation of Model Reorg s warehouses to the new Bellport facility.

In fiscal 2007, a provision on vendor advances of \$2.4 million to entities controlled by the shareholders of Jacavi that had been recorded in fiscal 2006 was reversed as the advance became realizable.

Model Reorg receives services from Quality King pursuant to a service agreement. The agreement provides for the allocation of expenses, which are calculated based on various assumptions and methods. (See Certain Relationships and Related Person Transactions.) The methods employed utilize various allocation bases including the number of transactions processed, estimated delivery miles, warehouse square footage, payroll dollars and sales and inventory ratios. Allocated operating expenses for fiscal 2007 were \$4.1 million, or 6.9% of total operating expenses, as compared to \$5.4 million or 8.7% for 2006.

In May 2007, Model Reorg began receiving invoices for its freight services from Quality King, rather than an allocation. The freight charges are comparable to those charged by unrelated third party carriers.

In August 2007, Model Reorg entered into a service agreement with E Com whereby E Com provides information technology management support services for all of Model Reorg s computer systems at a rate of \$25,000 per month, plus out-of-pocket expenses. July 2007 was the last month Model Reorg was allocated any IT service charges from Quality King.

Depreciation and amortization expenses in fiscal year 2007 decreased \$0.3 million from fiscal 2006, primarily as a result of property and equipment becoming fully depreciated and intangible costs for certain licenses becoming fully amortized.

Interest Expense

For the year ended October 31, (\$ in thousands)

			,,	Percentage
	2	2007	2006	Decrease
Interest expense	\$ 1	2,749	\$ 14,506	(12.1)%

Table of Contents

The decrease in interest expense relates to a decrease in average outstanding borrowings of approximately \$21 million for fiscal 2007 as compared to fiscal 2006. Average interest rates on borrowings remained relatively unchanged year over year.

Income Tax Provision

For the year ended October 31, (\$ in thousands)

				Percentage
	2	007	2006	Increase
Income taxes	\$ 7	,353	\$ 6,854	7.3%

The increase in the tax provision is a result of the increase in income before taxes of \$1.7 million for fiscal 2007 as compared to fiscal 2006. Model Reorg is a C corporation and has an effective tax rate of approximately 40% of pretax income each year. See Recent Accounting Standards below for a discussion of the anticipated impact of the adoption of FASB Interpretation No. (FIN) 48, Accounting for Uncertainty in Income Taxes- an interpretation of FASB Statement No. 109.

Fiscal Year 2006 Compared to Fiscal Year 2005

Revenues

For the year ended October 31, (\$ in thousands)

		· · · · · · · · · · · · · · · · · · ·							
		Percentage of			Percentage				
	2006	Revenue	2005	Revenue	Increase				
Quality Fragrance Group	\$ 257,826	76.2%	\$ 244,128	77.0%	5.6%				
Scents of Worth	73,990	21.9	68,971	21.8	7.3%				
Five Star	6,545	1.9	3,808	1.2	71.9%				
Total Revenues	\$ 338.361	100.0%	\$ 316,907	100.0%	6.8%				

In fiscal 2006, Quality Fragrance Group s increase in revenues was attributable to a wider variety of product mix and availability, which resulted in an increase in sales for existing and new customers. The increase for Scents of Worth was also a result of improved product mix, product availability and new consignment opportunities with retailers. Five Star s increase was attributable to new product launches.

Related party sales to Perfumania were \$32.2 million and \$38.2 million in fiscal 2006 and 2005, respectively, including a decrease of \$9.0 million, or 25.8%, for Quality Fragrance Group and an increase of \$3.0 million, or 129.0%, for Five Star. The decrease in Quality Fragrance Group sales resulted from Perfumania s increased sourcing of certain products direct from manufacturers. The increase in Five Star sales was the result of the launch of a Mattese makeup line and the introduction of Jerome Privee Moments Bath and Body products and candles. These products are being manufactured for and sold exclusively to Perfumania.

Direct Cost of Goods Sold

Direct cost of goods sold includes the cost of merchandise, inventory reserve adjustments and freight in charges. Individual components of cost of goods sold were consistent from 2005 to 2006, except that a \$1.4 million increase in Quality Fragrance Group s inventory reserve for slow-moving inventory in 2005 was not required in 2006.

For the	year	ended	October	31,
	(\$ in	thousa	nde)	

	(\$ in invusanus)			
	2006	2005	Percentage Increase	
Quality Fragrance Group	\$ 197,940	\$ 192,836	2.6%	
Scents of Worth	44,045	41,030	7.3%	
Five Star	3,339	1,581	111.2%	
Total cost of goods sold	\$ 245,324	\$ 235,447	4.2%	

Gross Profit

For the year ended October 31, (\$ in thousands)

	(\$\psi in inousumus)			
			Percentage	
	2006	2005	Increase	
Quality Fragrance Group	\$ 59,886	\$ 51,292	16.8%	
Scents of Worth	29,945	27,941	7.2%	
Five Star	3,206	2,227	44.0%	
Total gross profit	\$ 93,037	\$ 81,460	14.2%	

Gross profit increased for Quality Fragrance Group, Scents of Worth and Five Star principally as a result of increased sales.

Gross Profit Percentages

	For the year ended	October 31,
	2006	2005
Quality Fragrance Group	23.2%	21.0%
Scents of Worth	40.5%	40.5%
Five Star	49.0%	58.5%
Gross profit percentage	27.5%	25.7%

The increase in Quality Fragrance Group gross profit percentage relates to an increase in the amount of inventory being purchased from manufacturers, which lowers the cost of merchandise, a decrease in related party sales to Perfumania, which are at a lower margin than other unaffiliated sales, and the 2005 Quality Fragrance Group inventory reserve of \$1.4 million described above relating to a lower of cost or market adjustment. The decrease in gross profit percentage for Five Star related to the increase in sales to Perfumania, which are at a lower margin than sales to international customers.

Operating Expenses

For the year ended October 31, (\$ in thousands)

		(\$ in thousands)
			Percentage Increase
	2006	2005	(Decrease)
Selling, warehouse, delivery and administrative	\$ 57,548	\$ 54,694	5.2%
Depreciation and amortization	1,721	2,464	(30.2)%
Provision on vendor advances	2,367		100.0%
Total operating expense	\$ 61,636	\$ 57,158	7.8%

Income from operations \$31,401 \$24,302 29.2%

94

The increase in selling, warehouse, delivery and administrative expenses is a result of the increase in sales. The expenses that are directly attributable to sales are shipping supplies, freight out, sales commissions, consignment fees, advertising and sales promotion. These expenses increased 3.6% over the prior year.

During fiscal 2006, Model Reorg recorded an estimated provision for losses of \$2.4 million, or approximately 50% of total advances made to entities controlled by the shareholders of Jacavi, as management was uncertain that 100% of these advances were collectible.

Allocated operating expenses under the Quality King service agreement for fiscal 2006 were \$5.4 million, or 8.7% of total operating expenses, compared with \$4.8 million, or 8.3% of operating expenses, for fiscal 2005.

Depreciation and amortization expenses in fiscal year 2006 decreased \$0.7 million from fiscal 2005, primarily as a result of property and equipment becoming fully depreciated and intangible costs for certain licenses becoming fully amortized.

Interest Expense

For the year ended October 31,

The increase in interest expense is due to higher interest rates in 2006 as compared to 2005, as average debt year over year remained relatively unchanged.

Income Tax Provision

The increase in the tax provision is a direct result of the increase in income before income taxes. Model Reorg is a C Corporation and has an effective tax rate of approximately 40% of pretax income each year.

Three And Six-Month Periods Ended April 30, 2008 Compared To Three And Six-Month Periods Ended April 30, 2007

Net Revenues

	For the three months ended April 30, (\$ in thousands)			For the six months ended April 30, (\$ in thousands)		
			Percentage (Decrease)			Percentage
	2008	2007	Increase	2008	2007	Increase
Quality Fragrance Group	\$ 45,730	\$ 49,141	(6.9)%	\$ 128,297	\$ 120,140	6.8%
Scents of Worth	15,299	14,705	4.0%	45,661	45,170	1.1%
Five Star	1,004	1,005	(0.1)%	2,629	2,022	30.0%
Total net revenues	\$ 62,033	\$ 64,851	(4.3)%	\$ 176,587	\$ 167,332	5.5%

Included in net revenues are related party sales to Perfumania of \$6.4 million and \$5.7 million for the three months ended April 30, 2008 and 2007, respectively, or 10.3% and 8.8% of total revenues, an increase of \$0.7 million or 12.3%. Sales to Perfmania were \$19.7 million and \$11.2 million for the six months ended April 30, 2008 and 2007, respectively, or 11.2% and 6.7% of total revenues, an increase of \$8.5 million or 75.9%. In the second quarter of fiscal 2008, Quality Fragrance Group s decrease in net revenues of approximately \$4.0 million was attributable to declines in both wholesale and retail sales. The decrease in wholesale revenues is the result of tightening credit resources generally, which decreases customers ability to purchase. The retail accounts were adversely affected by the national reduction in consumer spending as the economy continues to soften. The increase in Scents of Worth sales is attributable to an increase of approximately 152 store locations, an increase of 5% from the second quarter of fiscal 2007 to the second quarter of fiscal 2008.

The increase in sales to Perfumania accounted for substantially all of the increase in revenues for Model Reorg for the six months ended April 30, 2008 as compared to the same period in 2007. This was the result of additional manufacturer relationships established during fiscal 2008, an increase in product offerings with existing manufacturers and new product launches by Five Star.

Direct Cost of Goods Sold

Direct cost of goods sold includes the cost of merchandise, inventory reserve adjustments and freight in charges. Individual components of cost of goods sold were consistent for the three and six months ended April 30, 2008 as compared to the same periods in the prior year.

	For the three months ended April 30, (\$ in thousands)			For the six months ended April 30, (\$ in thousands)		
	2008	2007	Percentage Increase (Decrease)	2008	2007	Percentage Increase
Quality Fragrance Group	\$ 34,814	\$ 34,500	0.9%	\$ 96,275	\$ 90,376	6.5%
Scents of Worth	9,415	8,894	5.9%	28,019	27,829	0.7%
Five Star	669	712	(6.0)%	1,716	1,191	44.1%
Total cost of goods sold	\$ 44,898	\$ 44,106	1.8%	\$ 126,010	\$ 119,396	5.5%

Gross Profit

	For the three months ended April 30, (\$ in thousands)			For the six months ended April 30, (\$ in thousands)			
	2008	2007	Percentage (Decrease) Increase	2008	2007	Percentage Increase	
Quality Fragrance Group	\$ 10,916	\$ 14,640	(25.4)%	\$ 32,022	\$ 29,764	7.6%	
Scents of Worth	5,884	5,811	1.3%	17,642	17,341	1.7%	
Five Star	335	294	13.9%	913	831	9.9%	
Total gross profit	\$ 17,135	\$ 20,745	(17.4)%	\$ 50,577	\$ 47,936	5.5%	

The change in total gross profit dollars for the second quarter of fiscal 2008 as compared to the second quarter of fiscal 2007 was primarily attributable to the decrease in Quality Fragrance Group sales. Also contributing to the change were lower margins resulting from increases in manufacturer pricing and from sales of discontinued brands at lower prices than while the respective distribution arrangements were in effect.

Gross Profit Percentages

	For the three me	onths ended April 30,	For the six m	For the six months ended April 30,		
	2008	2007	2008	2007		
Quality Fragrance Group	23.9%	29.8%	25.0%	24.8%		
Scents of Worth	38.5%	39.5%	38.6%	38.4%		
Five Star	33.4%	29.3%	34.7%	41.1%		
Gross profit percentage	27.6%	32.0%	28.6%	28.6%		

Gross profit percentages for Model Reorg remained consistent for the six months ended April 30, 2008, as compared to the same period in the prior year, but they decreased during the three months ended April 30, 2008 compared to the same period in the prior year due to Quality Fragrance Group selling a larger portion of lower margin products and Scents of Worth offering greater discounts in 2008 to sell through holiday merchandise remaining in the stores. Five Star s increase in margin relates to product launches during the second quarter of fiscal 2008, which typically are sold at higher margins than existing brands.

Operating Expenses

	For the three months ended April 30, (\$ in thousands)			For the six months ended April 30, (\$ in thousands)		
	2008	2007	Percentage Increase (Decrease)	2008	2007	Percentage Increase (Decrease)
Selling, warehouse, delivery and administrative	\$ 13,417	\$ 12,334	8.8%	\$ 32,950	\$ 29,509	11.7%
Depreciation and amortization	373	407	(8.4)%	713	822	(13.3)%
Total operating expense	\$ 13,790	\$ 12,741	8.2%	\$ 33,663	\$ 30,331	11.0%
Income from operations	\$ 3,345	\$ 8,004	(58.2)%	\$ 16,914	\$ 17,605	(3.9)%

The increase in selling, warehouse, delivery and administrative expenses for the three months ended April 30, 2008 as compared to the three months ended April 30, 2007 relates to costs to relocate multiple warehouses into the new Bellport facility, as well as increased accounting, auditing and litigation costs. In addition to these costs, the increase for the six months ended April 30, 2008 as compared to the same period in the prior year includes an increase in shipping and warehousing costs related to the increase in sales for Quality Fragrance Group.

Included in operating expenses are allocated operating expenses under the Quality King services arrangements for the three months ended April 30, 2008, which were \$1.8 million or 13.3% of total operating expenses as compared to \$3.0 million or 23.8% of total operating expenses for the three months ended April 30, 2007. Allocated operating expenses for the six months ended April 30, 2008 were \$4.9 million or 14.5% of total operating expenses, as compared to \$6.0 million or 19.7% of total operating expenses for the six months ended April 30, 2007.

Interest Expense

	For the th	For the three months ended April 30, (\$ in thousands)			six months ende (\$ in thousand	. ,
		Percentage				Percentage
	2008	2007	Decrease	2008	2007	Decrease
Interest expense	\$ 2,135	\$ 2,770	22.9%	\$ 5,300	\$ 6,104	13.2%

97

The decrease in interest expense relates to a decrease in the interest rates on the revolving credit facility and intercompany payable to Quality King of approximately 2.5% during the second quarter 2008 and 1.9% during the six months ended April 30, 2008, as compared to the same periods in the prior year.

Income Taxes

	For the	For the three months ended April 30, (\$\frac{s}{in thousands}\$)			For the six months ended April 30,		
					(\$ in thousand	(s)	
			Percentage			Percentage	
	2008	2007	Decrease	2008	2007	Increase	
Income taxes	\$ 505	\$ 2.094	75.9%	\$ 4.860	\$ 4.600	5.7%	

The decrease in income taxes for the three months ended April 30, 2008 as compared to the three months ended April 30, 2007 is principally the result of the decrease in income before taxes of \$4.0 million. In addition, income taxes for both the three and six months ended April 30, 2008 reflect additional taxes paid for state income tax audits.

Liquidity and Capital Resources

Model Reorg s principal capital requirements for operating purposes are to fund inventory purchases, finance extended terms on accounts receivable, specifically related party receivables from E Com, and pay down accounts payable and debt. During fiscal 2007, Model Reorg financed these requirements through net cash flows from financing activities, specifically the revolving credit borrowings, while during the first half of fiscal 2008, it relied primarily on reductions in inventory and collections of unaffiliated receivables.

A summary of Model Reorg s cash flows for the years ended October 31, 2007 and 2006, and the six months ended April 30, 2008 and 2007 are as follows:

	Six Months Ended April 30,		Year Ended	October 31,	
	2008	2007	2007	2006	
		(\$ in thou	sands)		
Summary Cash Flow Information:					
Cash provided by (used in) operating activities	\$ 41,502	\$ 37,886	\$ (12,307)	\$ 39,234	
Cash used in investing activities	(3,771)	(1,241)	(2,115)	(4,610)	
Cash (used in) provided by financing activities	(35,583)	(37,175)	14,673	(34,408)	
Increase (decrease) in cash	2,148	(530)	251	216	
Cash at beginning of period	2,988	2,737	2,737	2,521	
Cash at end of period	\$ 5,136	\$ 2,207	\$ 2,988	\$ 2,737	

Management expects to continue to finance fiscal 2008 requirements primarily through operating activities.

Model Reorg, along with Quality King, is part of a \$500 million revolving line of credit with a syndicate of lenders that expires on April 30, 2010. As of April 30, 2008 and October 31, 2007, \$102.2 million and \$138.9 million, respectively, was outstanding for Model Reorg under this line of credit. The credit agreement provides for borrowing based on eligible accounts receivable and inventories and is secured by Model Reorg s accounts receivable and inventories. The interest rate applicable to borrowings under the credit agreement is based on the monthly blended borrowing rates, which are based on, at Quality King s option, the LIBOR rate or the prime rate plus an applicable spread based on availability. At April 30, 2008 and October 31, 2007, substantially all of the revolving credit borrowings were at LIBOR plus the applicable spread, the effective rate of which approximated 4.4% and 6.6%, respectively. The borrowings under the credit agreement are classified as long term based on the expiration date of the facility.

Model Reorg has a note payable to Quality King which is classified as long-term since Quality King will not require payment of the loan within one year. This note supplements Model Reorg s borrowings under the line of credit described above. Model Reorg is charged interest by Quality King based on the monthly blended borrowing rates, which are based on the LIBOR and prime rates, plus a spread based on excess availability. Interest rates on the note for the six months ended April 30, 2008 and fiscal year ended October 31, 2007 averaged 5.4% and 7.6%, respectively. Interest expense charged to operations relating to the note payable was \$4,571 for fiscal year 2007 and \$1,961 for the six months ended April 30, 2008.

For the six months ended April 30, 2008, net cash provided by operating activities was approximately \$41.5 million, resulting from a decrease in accounts receivable and inventory after the holiday season. Net cash used in investing activities for the six months ended April 30, 2008 was approximately \$3.8 million. These were primarily purchases of property, plant and equipment for the relocation to the Bellport warehouse and capitalized Merger-related expenses. Net cash used in financing activities of \$35.6 million principally relates to the decrease in the revolving credit borrowings and pay down of affiliated note payable from the funds provided by operating activities.

In fiscal 2007, net cash used in operating activities was approximately \$12.3 million, compared to \$39.2 million provided by operating activities in fiscal 2006. The variance of approximately \$51.5 million between the two years principally resulted from increases of \$7.5 million in accounts receivable relating to the timing of sales, \$12.6 million in extended receivable terms to Perfumania, and \$18.8 million in inventory purchases resulting from additional manufacturer relationships and product availability, as well as an increase in the payments on accounts payable of \$12.0 million related to the build of inventory and the terms on the purchases from manufacturers. Model Reorg funded these increased uses of cash in operating activities through increased borrowings under its revolving credit facility and from Quality King, as a result of which cash provided by financing activities increased approximately \$49.1 million in fiscal 2007 as compared to fiscal 2006. Net cash used in investing activities in fiscal 2007 was approximately \$2.1 million. Investing activities generally represent deposits made for property, plant and equipment for the relocation to the Bellport facility and capitalized transaction costs.

Management believes that Model Reorg s borrowing capacity under its current facility, projected cash flows from operations, affiliated borrowings and credit terms from vendors will provide sufficient liquidity to support its capital expenditures and debt service for the foreseeable future. There can be no assurance that management s plans and expectations will be successful.

Critical Accounting Policies and Estimates

There have been no material changes to our critical accounting policies and estimates from our fiscal year ended October 2007.

Contractual Obligations

Following is a summary of Model Reorg s contractual obligations as of October 31, 2007. See also Approval to Issue Shares and Warrants Post-Merger Financing for a discussion of the Senior Credit Facility we expect that the combined company will enter into upon the closing of the Merger.

	Payments due by period					
	(\$ in thousands)					
		Less than			More than	
Contractual Obligations (1)	Total	1 year	1-3 years	3-5 years	5 years	
Revolving line of credit (2)	\$ 138,278	\$	\$ 138,278	\$	\$	
Operating lease obligations	64,066	2,735	8,760	7,959	44,612	
Minimum royalty obligations	5,604	2,085	3,119	400		
Other	845	209	543	93		
Total	\$ 208,793	\$ 5,029	\$ 150,700	\$ 8,452	\$ 44,612	

(1) Affiliated borrowings from Quality King in the amount of \$76.3 million at October 31, 2007 are not included in this schedule as there are no specified repayment terms on these borrowings.

(2) This balance represents principal only as the interest rate is variable and accrues on outstanding balances, which vary significantly throughout the year.

99

Table of Contents

Off-Balance Sheet Arrangements

Model Reorg has no off-balance sheet arrangements as defined in Item 303(a)(4) of SEC Regulation S-K.

Recent Accounting Standards

In June 2006, the FASB issued FIN 48, which clarifies the accounting for the uncertainty in income taxes recognized under SFAS No. 109, Accounting for Income Taxes. FIN 48 prescribes a recognition threshold and measurement attribute for financial statement recognition and measurement of a tax position taken or expected to be taken in a tax return and also provides guidance on various related matters such as derecognition, interest and penalties and disclosures. The effective date of FIN 48 for non-public enterprises is for annual periods beginning after December 15, 2007. Model Reorg has begun evaluating the financial impact of applying the provisions of FIN 48 to all tax provisions and it does not believe there will be a material financial impact upon the initial adoption of FIN 48.

100

UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL STATEMENTS

The following unaudited pro forma condensed combined financial statements give effect to the proposed Merger. Pursuant to the Merger Agreement, we will acquire all of the outstanding share capital of Model Reorg in exchange for the Merger Consideration. Two Model Reorg shareholders currently own 44.5% of our common stock on a fully-diluted basis. Following the Merger, the former shareholders of Model Reorg will hold an aggregate of approximately 79.3% of our outstanding common stock on a fully diluted basis. Due to a number of factors, including that the shares issued to Model Reorg s shareholders in the Merger will constitute a majority of the outstanding shares of our common stock after the transaction, Model Reorg is deemed to be the acquiring company for accounting purposes. Accordingly, the purchase price is allocated among the fair values of our assets and liabilities, while the assets and liabilities of Model Reorg are reflected at historical cost in the combined condensed pro forma balance sheet. The transaction will be accounted for under the purchase method of accounting in accordance with SFAS 141. Under the purchase method of accounting, since Model Reorg is deemed to be the acquiror for accounting purposes, the total estimated purchase price, calculated as described below in Note 1 to the unaudited pro forma condensed combined balance sheet, is allocated to the net tangible and intangible assets acquired and liabilities assumed of E Com in connection with the transaction, based on their estimated fair values as of the completion of the Merger.

The unaudited pro forma condensed combined financial statements presented below are based upon the historical financial statements of both companies, reflecting the restatement of our financial statements for reporting periods ending before 2008 as described in Note 3 to our Consolidated Financial Statements beginning on page F-14, adjusted to give effect to our acquisition by Model Reorg. The pro forma adjustments are described in the accompanying notes presented on the following pages. Because Model Reorg is considered to be the acquiring company, the statement of operations information is presented below as of October 31, 2007 and April 30, 2008, the last day of Model Reorg s most recently completed fiscal year and interim period, respectively, and the balance sheet information is presented as of April 30, 2008. However, going forward, the combined company will continue to use our fiscal year end, the Saturday closest to January 31, which is conventional for retail businesses, as well as the corresponding Saturdays as the end of each fiscal quarter.

The amounts in the unaudited pro forma condensed combined financial statements are subject to a final determination of the fair market value of our assets acquired and liabilities assumed. The final purchase price allocation may be materially different from the allocation used in calculating goodwill for the pro forma adjustments shown below. As a result, actual income from operations may differ significantly from the pro forma amounts included below. There can be no assurance that such adjustments will not be material to the combined financial statements.

All intercompany balances and transactions between Model Reorg and E Com, consisting primarily of sales, purchases, profits in ending inventory, receivables and payables, as of the dates and for the periods of these unaudited pro forma combined financial statements, have been eliminated in the unaudited pro forma combined financial statements. Certain reclassification adjustments have been made to conform our historical reported balances to Model Reorg s financial statement basis of presentation.

The unaudited pro forma condensed combined financial statements are provided for illustrative purposes only and do not purport to represent the financial condition or results of operations had the Merger been completed as of the dates indicated, nor are they necessarily indicative of future consolidated results of operations or financial position.

The unaudited pro forma condensed combined financial statements do not include the realization of cost savings from operating efficiencies, synergies, or other restructurings effects resulting from the Merger.

The unaudited pro forma condensed combined financial statements should be read in conjunction with our historical financial statements and accompanying notes and the historical financial statements and accompanying notes of Model Reorg, which are included elsewhere in this Proxy Statement.

101

Unaudited Pro Forma Condensed Combined Balance Sheet

	As of April 30 Historical		As of April 30, 200	08	Pro Forma
	E Com	Model Reorg	(\$ in thousands)	Pro Forma Adjustments	Combined Model Reorg and E Com
Assets			(+		
Current:					
Cash	\$ 1,280	\$ 5,136		\$	\$ 6,416
Accounts receivable net	1,216	28,407			29,623
Receivables from an affiliate	235	35,844	(a)	(36,079)	
Inventories, net	118,857	193,620	(h)	(4,923)	306,172
			(i)	(1,382)	
Advances to suppliers for future purchases		2,085			2,085
Prepaids and other	8,527	5,724	(1)	(1,366)	12,885
Total current assets	130,115	270,816		(43,750)	357,181
Property and equipment, net	38,476	3,141		, , ,	41,617
Goodwill	1,904	20,434	(g) (m)	2,172 (1,904)	22,606
Other assets, net	7,273	12,147	(f)	(5,484)	13,936
	\$ 177,768	\$ 306,538		\$ (48,966)	\$ 435,340
Liabilities and Shareholders Equity					
Current:					
Accounts payable, non affiliates	22,899	22,965			45,864
Accounts payable, affiliates	55,891	144	(a)	(36,079)	19,956
Accrued expenses	10,974	12,211	(j)	(2,522)	17,134
			(1)	(3,529)	
Bank line of credit	42,590		(b)	(42,590)	
Current maturities of long-term debt	5,364	208			5,572
Total current liabilities	137,718	35,528		(84,720)	88,526
Long-term debt and other	7,090	102,733	(b)	32,502	142,325
Payable to affiliates	7,090	76,912	(b)	10,088	87,000
Total liabilities	144,808	215,173		(42,130)	317,851
Commitments and Contingencies					
Shareholders equity					
Common stock	40	1	(e)	59	99
			(d)	(1)	
Additional paid in capital	79,183	13,905	(e)	(59)	50,128
			(k)	(37,686)	
			(d)	1	
			(f)	(5,484)	
			(g)	2,172	
			(m)	(1,904)	
Retained earnings (deficit)	(37,686)	78,934	(k)	37,686	75,839
			(c)	(1,475)	
			(h)(i)(j)	(3,783)	
			(1)	2,163	

Edgar Filing: E COM VENTURES INC - Form DEFM14A

Treasury Stock	(8,577)	(1,475)	(c)	1,475	(8,577)
Total shareholders equity	32,960	91,365		(6,836)	117,489
	\$ 177,768	\$ 306,538	:	\$ (48,966)	\$ 435,340

Table of Contents

(a)	Adjustment to eliminate intercompany payable and receivables between E Com and Model Reorg
(b)	Adjustment to reflect the post-merger financing, which is expected to include a new a \$250 million revolving credit facility and the pay down of related party debt. (See Approval to Issue Shares and Warrants Post-Merger Financing and Affiliate Debt.)
(c)	Adjustment to record retirement of Model Reorg treasury stock
(d)	Adjustment to record the cancellation of Model Reorg common stock
(e)	Adjustment to record the issuance of additional 5,900,000 shares of E Com common stock
(f)	Adjustment to record estimated transaction fees, which are reflected as a component of purchase price. (See Note 1 below.)
(g)	Adjustment to record goodwill related to the Merger. Goodwill is calculated as the excess of purchase price over the fair value of the assets acquired and liabilities assumed. See Note 1 below.
(h)	Adjustment to remove any affiliated profit in E Com inventory as of April 2008
(i)	Adjustment to remove any affiliated profit in Model Reorg inventory as of April 2008
(j)	Tax effect of the pro forma adjustments at 40%, which approximates Model Reorg s effective tax rate
(k)	Adjustment to eliminate E Com s historical accumulated deficit
(1)	Adjustment to reverse E Com s historical accrued straight line rent and deferred rent, which does not represent a legal obligation of the combined company, and will have no value, upon completion of the Merger, and its related Deferred Tax Asset, and to recognize post-Merger rent expense on a straight-line basis
Note deter	Adjustment to eliminate E Com s pre-merger goodwill of \$1.9 million <i>I</i> : Since Model Reorg is deemed to be the acquirer for accounting purposes, in accordance with GAAP, the total purchase price will be mined based on the fair value of the outstanding shares of E Com prior to the Merger, plus the Model Reorg transaction costs. Preliminary ated transaction costs and purchase price are as follows (the actual purchase price will be calculated based on the fair value of the E Com s at the date the Merger is consummated and the actual transaction costs):

Table of Contents 145

Fair value of E Com common stock(x)

(*\$ in thousands*) \$ 62,037

Estimated transaction costs	5,484
	\$ 67 521

(x) 3,059,041 E Com common shares outstanding as of April 30, 2008 at a closing market price of \$20.28 per share.

Under the purchase method of accounting, for purposes of the table above, the total preliminary estimated purchase price is allocated to the E Com net tangible and identifiable intangible assets acquired and liabilities assumed based on their estimated fair values as of the completion date of the Merger. The preliminary estimated purchase price has been allocated based on estimates taking into account various factors as described in the introduction to these unaudited pro forma condensed combined financial statements. The allocation of the preliminary estimated purchase price is as follows:

	(\$ in	thousands)
Current assets	\$	123,159
Property and equipment, net		38,476
Other assets		7,273
Goodwill		2,172
Liabilities assumed		(103,559)
	\$	67 521

Unaudited Pro Forma Condensed Combined Statements Of Operations

		Hist	orical	Γwelve Mont	hs Ended O	ctober	31, 2007		o Forma
	1	E Com*		odel Reorg (\$ in thousan	ds, except we	Ad	o Forma justments d average		del Reorg d E Com
				and p	er share am	ounts)		
Net sales, unaffiliated	\$	236,089	\$	302,530		\$		\$	538,619
Net sales, affiliated		48,910		25,182	(a)		(74,092)		
Total net sales		284,999		327,712			(74,092)		538,619
Cost of goods sold		174,426		237,203	(a)		(74,092)		339,337
					(b)(c)		1,800		
Gross profit		110,573		90,509			(1,800)		199,282
Selling, warehouse, delivery and administrative expenses		95,812		60,113	(f)		171		156,096
Depreciation and amortization		5,847		1,411					7,258
Recovery on vendor advances				(2,367)					(2,367)
Income from operations		8,914		31,352			(1,971)		38,295
Interest expense		4,912		12,749					17,661
Income before income taxes		4,002		18,603			(1,971)		20,634
Income taxes		79		7,353	(d)(e)		822		8,254
Net income	\$	3,923	\$	11,250		\$	(2,793)	\$	12,380
Weighted average shares									
Basic(h)	3	3,055,510		5,900,000				8	3,955,510
Diluted(h)	3	3,569,706		5,900,000				ç	9,469,706
Net income (loss) per share									
Basic(h)	\$	1.28	\$	1.91		\$	(0.31)	\$	1.38
Diluted(h)	\$	1.23	\$	1.91		\$	(0.29)	\$	1.36

^{*} E Com s historical statement of operations for the twelve months ended October 31, 2007 combines the results of the fourth quarter of E Com s 2006 fiscal year with those of the first nine months of its 2007 fiscal year.

- (a) Adjustment to eliminate affiliated sales and purchases
- (b) Adjustment to remove the increase in affiliated profit in E Com inventory in the amount of \$800 as of October 31, 2007
- (c) Adjustment to remove the increase in affiliated profit in Model Reorg inventory in the amount of \$1,000 as of October 31, 2007
- (d) Tax effect of the pro forma adjustments at 40%, which approximates Model Reorg s effective tax rate (See Note 1 below)

/	`	TD 1'	1 1 1	•	cc .: .	c · . 1	100	(C) T (111
1 e	•)	To adjust tax expens	se to the combined (romnanies	effective tax rate o	t annroximatel	V 40%	See Note	I below
(-	',	10 dajust tun enpens	se to the combined t	companics	criccure tax rate o	i approximatei	y 1070	DCC I TOTC	I DCIOW

(f) Adjustment to reverse E Com s historical accrued straight line rent, which does not represent a legal obligation of the combined company, and will have no value, upon completion of the Merger, and to recognize post-Merger rent expense on a straight-line basis

104

(g) Basic net income per share is computed by dividing net income by the weighted average number of common shares outstanding during the period. In computing diluted net income per share, the weighted average number of common shares outstanding includes all common stock equivalents with exercise prices at or below the average market price for the respective period. The calculations for the twelve months ended October 31, 2007 are as follows:

						P	ro Forma
						Com	bined Model
	E	Com (\$ in the]	Model Reorg ercent weigh	Pro Forma Adjustments ted average and per sh		eorg and E Com
Numerator:		(φ ιπ ιπο	usurus,	except weight	ca average and per sn	ure umoi	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
Net income (loss) Basic	\$	3,923	\$	11,250	\$(2,793)	\$	12,380
Add: Interest on Convertible Note		473					473
Net income (loss) Diluted	\$	4,396	\$	11,250	\$ (2,793)	\$	12,853
Denominator:							
Weighted average number of shares for basic net							
income (loss) per share	3,	055,510	5.	,900,000	8,955,510		8,955,510
Shares issuable upon exercise of stock options		69,751			69,751		69,751
Convertible Note	4	444,445			444,445		444,445
Denominator for dilutive net income (loss) per							
share	3,	569,706	5,	,900,000	9,469,706		9,469,706
Basic net income (loss) per common share	\$	1.28	\$	1.91	\$ (0.31)	\$	1.38
Diluted net income (loss) per common share	\$	1.23	\$	1.91	\$ (0.29)	\$	1.36

Pro forma share numbers represent the weighted average shares outstanding of E Com for the twelve months ended October 31, 2007, assuming the issuance at the beginning of the period of the 5.9 million shares issuable in the Merger. Pro forma share numbers used in the computation of diluted net income per share also assume the issuance of 69,751 shares issuable upon exercise of stock options for the twelve months ended October 31, 2007 and 444,445 shares of our common stock upon conversion of the Convertible Note. The 1.5 million Warrant shares are not included in the pro forma share numbers since they would be antidilutive.

Note 1: The adjustment of \$0.822 million to increase income tax expense consists of a provision for \$1.542 million and a benefit of \$0.72 million. The provision for \$1.542 million results from the requirement to reflect the effective E Com income tax expense in this pro forma statement of operations for a period other than that of E Com s normal fiscal year and the \$0.171 million to reverse E Com s historical accrued straight line rent, which will have no value upon completion of the Merger, and to recognize post-Merger rent expense on a straight-line basis. The benefit results from the reduction of the pro forma income tax expense as a result of the elimination of an aggregate of \$1.8 million of affiliated profit in inventory.

105

		His	storical	Six M	onths ended A	pril 30), 2008	P	ro Forma
]	E Com*		Model Reorg (\$ in thou	sands, except	Ad	o Forma justments ed average		bined Model g and E Com
				a	nd per share a	mount	g)		
Net sales, unaffiliated	\$	145,663	\$		na per snare a	\$	-,	\$	302,565
Net sales, affiliated		21,287		19,685	(a)		(40,972)	·	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
Total net sales		166,950		176,587			(40,972)		302,565
Cost of goods sold		102,225		126,010	(a)		(40,972)		187,712
ŭ					(b)(c)		449		
Gross profit		64,725		50,577			(449)		114,853
Selling, warehouse, delivery and administrative				ĺ					
expenses		56,990		32,950	(f)		(212)		88,718
•					(h)		(1,434)		
Depreciation and amortization		3,440		713					4,153
Recovery on vendor advances									
Income from operations		4,295		16,914			773		21,982
Interest expense		1,898		5,300					7,198
Income before income taxes		2,397		11,614			773		14,784
Income tax expense		1,502		4,860	(d)(e)		(448)		5,914
Net income	\$	895	\$	6,754		\$	1,221	\$	8,870
				,			,		,
Weighted Average shares									
Basic(g)	3	3,059,041		5,900,000					8,959,041
Diluted(g)		3,059,041		5,900,000					9,478,899
Net Income (loss) per share									
Basic(g)	\$	0.29	\$	1.14		\$	0.14	\$	0.99
Diluted(g)	\$	0.29	\$	1.14		\$	0.14	\$	0.96

^{*} E Com s historical statement of operations for the six months ended April 30, 2008 combines the results of the last quarter of E Com s 2007 fiscal year with those of the first quarter of its 2008 fiscal year.

- (a) Adjustment to eliminate affiliated sales and purchases
- (b) Adjustment to remove the increase in affiliated profit in E Com Ventures inventory in the amount of \$0.805 million as of April 30, 2008
- (c) Adjustment to add the decrease in affiliated profit in Model Reorg inventory in the amount of \$0.356 million as of April 30, 2008
- (d) Tax effect of the pro forma adjustments at 40%, which approximates Model Reorg s effective tax rate

- (e) To adjust tax expense to the combined company s effective tax rate of approximately 40% (See Note 1 below)
- (f) Adjustment to reverse E Com s historical accrued straight line rent and deferred rent, which does not represent a legal obligation of the combined company, and will have no value, upon completion of the Merger, and to recognize post-Merger rent expense on a straight-line basis.

106

(g) Basic net income per share is computed by dividing net income by the weighted average number of common shares outstanding during the period. In computing diluted net income per share, the weighted average number of common shares outstanding includes all common stock equivalents with exercise prices at or below the average market price for the respective period. The calculations for the six months ended April 30, 2008 are as follows:

							Pro For	ma Combined
	F	Com		Model Reorg		Forma ustments		el Reorg and E Com
	L			s, except weig	•			
Numerator:		(,		, , ,		,		,
Net income Basic	\$	895	\$	6,754	\$	1,221	\$	8,870
Add: Interest on Convertible Note								187
Net income Diluted	\$	895	\$	6,754	\$	1,221	\$	9,057
	т		-	-,	-	-,	*	7,02.
Denominator:								
Weighted average number of shares for basic net								
income per share	3,0	059,041	5.	,900,000	8,	959,041		8,959,041
Shares issuable upon exercise of stock options								75,413
Convertible Note								444,445
Denominator for dilutive net income per share	3,0	059,041	5.	,900,000	8,	959,041		9,478,899
·								
Basic net income per common share	\$	0.29	\$	1.14	\$	0.14	\$	0.99
r r	Ψ	2.3/	4		Ψ		Ψ	0.77
Diluted net income per common share	\$	0.29	\$	1.14	\$	0.14	\$	0.96

Pro forma share numbers represent the weighted average shares outstanding of E Com for the six months ended April 30, 2008, assuming the issuance at the beginning of the period of the 5.9 million shares issuable in the Merger. Pro forma share numbers used in the computation of diluted net income per share also assume the issuance of 75,413 shares issuable upon exercise of stock options for the six months ended April 30, 2008 and 444,445 shares of our common stock upon conversion of the Convertible Note. The 1.5 million Warrant shares are not included in the pro forma share numbers since they would be antidilutive.

(h) Adjustment to reverse the writeoff of E Com transaction costs

Note 1: The adjustment of \$0.448 million to decrease income tax expense consists of a provision for \$0.574 million and a benefit of \$1.022 million. The provision for \$0.574 million results from the reversal of the writeoff of \$1.431 million of E Com s transaction costs. The benefit results from the reduction of the pro forma tax expense as a result of the elimination of an aggregate of \$0.449 million of affiliated profit in inventory; \$0.212 million to reverse E Com s historical accrued straight-line rent and to recognize post-Merger rent expense on a straight-line basis; and an adjustment to realign tax expense to Model Reorg s estimated effective rate of 40%.

SELECTED FINANCIAL PROJECTIONS

Neither we nor Model Reorg, as a matter of course, make public forecasts or projections as to future financial performance. However, during 2007, we and Model Reorg s management prepared prospective financial information to present certain projections of financial performance for the respective companies, and these projections were provided to the Special Committee, Financo and our Board in connection with their financial analyses of the proposed Merger. In June 2008, we and Model Reorg s management updated our projections for EBITDA for the twelve months ended October 31, 2008, which we provided to the Special Committee and our Board for use in their consideration of the First Amendment, together with projections of interest costs through October 31, 2010.

None of these projections were prepared with a view toward public disclosure, nor with a view toward complying with the guidelines of the American Institute of Certified Public Accountants with respect to prospective financial information, but, in the view of our management and Model Reorg management, they were prepared on a reasonable basis, reflect the best estimates and judgments available as of the date of their preparation, and present, to the best of each management sknowledge and belief as of the date of their preparation, the expected course of action and the expected future financial performance of the respective company. However, this information is not fact and should not be relied upon as being necessarily indicative of future results, and readers of this Proxy Statement are cautioned not to place undue reliance on this information. The projections are forward-looking statements inherently subject to the general risks that we and Model Reorg face in our businesses, including those discussed under Risk Factors in this Proxy Statement and the factors described under Forward-Looking Statements in this Proxy Statement.

Neither our nor Model Reorg s independent auditors, nor any other independent accountants, have compiled, examined, or performed any procedures with respect to the prospective financial information contained herein, nor have they expressed any opinion or any other form of assurance on such information or its achievability, and they assume no responsibility for, and disclaim any association with, this prospective financial information.

Set forth below are material projections of EBITDA for the respective companies on a standalone basis and for the combined company that we and Model Reorg prepared. All these projections other than the ones labelled Updated were provided to the Special Committee, Financo and our Board in 2007. Those labelled Updated were provided to the Special Committee and our Board in June 2008. These financial forecasts and the assumptions underlying them have not been updated since their respective dates of preparation.

Twelve Months Ended

		October 31, (\$ in thousands)				
	20	08	•			
	Original	Updated	2009	2010		
EBITDA						
E Com	\$ 14,743	\$ 9,902	\$ 16,424	\$ 19,466		
Model Reorg	42,694	36,118	47,066	49,081		
Combined Company	57,437	46,020	63,490	68,547		

108

EBITDA consists of net income excluding income tax provision, interest expense, depreciation and amortization. Projected EBITDA for E Com and Model Reorg for the respective twelve-month periods ending October 31 is reconciled to net income as follows:

	20	08		
	Original	Updated	2009	2010
E Com				
EBITDA	\$ 14,743	\$ 9,902	\$ 16,424	\$ 19,466
Interest expense	4,119	3,764	4,771	5,399
Income taxes	2,021	138	2,009	2,411
Depreciation and amortization expense	5,312	5,784	6,363	7,721
Net Income	\$ 3,291	\$ 216	\$ 3,281	\$ 3,935
			,	. ,
Model Reorg				
EBITDA	\$ 42,694	\$ 36,118	\$ 47,066	\$ 49,081
Interest expense	11,744	11,318	15,527	15,151
Income taxes	12,106	9,364	12,190	13,134
Depreciation and amortization expense	1,319	1,389	1,364	1,396
Net Income	\$ 17,525	\$ 14,047	\$ 17,985	\$ 19,400
Combined Company				
EBITDA	\$ 57,437	\$ 46,020	\$ 63,490	\$ 68,547
Interest expense	15,863	15,082	20,298	20,550
Income taxes	14,127	9,502	14,199	15,545
Depreciation and amortization expense	6,631	7,173	7,727	9,117
•				
Net Income	\$ 20,816	\$ 14,263	\$ 21,266	\$ 23,335

While presented with numeric specificity, the foregoing financial projections reflect numerous important assumptions, many of which are highly subjective, made by our and Model Reorg management in light of business, industry and market conditions at the time of their respective preparation. These assumptions include the following:

With respect to E Com:

Opening of 50 (for the Updated 2008 EBITDA projections above, an increase from the 45 used for the Original projection), 45 and 50 new stores in 2008, 2009, 2010, respectively, and closing of an average of five stores per year.

A calculated rate of average retail sales growth over the three years that is higher than our experience in recent years due to the greater number of new stores opened during the period, partially offset by the projected lower average sales volume derived from new stores generally.

An additional amount of retail sales growth due to: increased marketing efforts leading to greater recognition of the Perfumania brand name, continuing improvement in merchandise assortment, a recent national trend of consumers favoring retail stores over department stores, a reduction in the number of department stores due to industry consolidation, and continuing remodeling of Perfumania stores to standardize and improve store design and appearance.

A small increase in wholesale sales in 2008.

Capital expenditures of \$15.1 million (for the Updated 2008 EBITDA projections above, an increase from the \$12.9 million used for the Original projection), \$14.9 million and \$17.7 million in 2008, 2009 and 2010, respectively, relating primarily to opening of new stores, renovation of existing stores and replacement/upgrade of store point of sale register hardware and software.

109

Table of Contents

Moderate increase in store operating expenses as a percentage of retail sales due to the projected increase in new store openings. (New stores tend to incur higher operating expenses as a percentage of sales in the first few years after opening compared with more mature stores due to new store lower initial sales contributions.)

A gradual decrease in corporate office expenses as a percentage of retail sales.

A moderate increase in interest expense due to assumed increased utilization of credit facility to support new store openings. With respect to Model Reorg:

The Original 2008 EBITDA projection above was based on the assumption of an increase in revenues of approximately 10% for fiscal 2008 relating to the establishment of two significant manufacturer relationships not previously available to Model Reorg. The assumed increase was reduced to 3.5% for purposes of the Updated 2008 EBITDA projection.

While an increase of approximately 5% was projected for sales to existing retail customers during fiscal 2008 for purposes of the Original 2008 projection, the Updated 2008 EBITDA projection assumes no such increase.

No net increase in the number of customers in fiscal 2009 and 2010, but a 4.5% annual increase in sales in those years due to more extensive merchandise offerings.

Capital expenditures of \$0.8 million, \$1.7 million and \$1.3 million in 2008, 2009 and 2010, respectively.

Warehouse consolidation savings from combining seven warehouse locations into the single Bellport facility, such as rent, supplies, payroll and related benefits, were budgeted at \$0.1 million per month beginning February 2008.

Salaries were budgeted with a 4% cost of living increase each fiscal year.

We and Model Reorg also provided the Special Committee and our Board with interest expense projections regarding the proposed new Senior Credit Facility and Affiliate Debt arrangements showing that the combined company s aggregate interest expense, based on anticipated debt levels, is expected to be less than was projected under the arrangements contemplated by the Original Merger Agreement.

There can be no assurance that the assumptions made in preparing the financial projections will prove accurate. In addition, the financial projections do not take into account any of the costs of the transactions contemplated by the Merger Agreement, including the costs of the Merger, the related financing transactions, and any integration costs or savings from the Merger. Management cannot predict the actual integration costs or the savings, if any, that will be achieved when E Com and Model Reorg are combined.

EBITDA has limitations, including that it:

is not necessarily comparable to other similarly titled financial measures of other companies due to the potential inconsistencies in the method of calculation;

does not reflect actual cash expenditures, future requirements for capital expenditures or contractual commitments;

does not reflect the significant interest expense or cash requirements necessary to service interest or principal payments on our, Model Reorg s or the combined company s debt;

does not reflect any cash income taxes that we, Model Reorg or the combined company may be required to pay; and

although depreciation and amortization are non-cash charges, the assets being depreciated and amortized will often have to be replaced in the future and these measures do not reflect any cash requirements for such replacements.

110

Table of Contents

Therefore, EBITDA should not be considered either in isolation or as a substitute for analysis of our results as reported under GAAP. Furthermore, EBITDA should not be considered as an alternative to operating income (loss) or net income (loss) as a measure of operating performance or to net cash provided by operating, investing or financing activities, or as a measure of the ability of any of us, Model Reorg or the combined company to meet cash needs.

The assumptions and estimates underlying the prospective financial information set forth above are inherently uncertain and, though considered reasonable by our and Model Reorg management, respectively, as of the date of its preparation, are subject to a wide variety of significant business, economic, and competitive risks and uncertainties that could cause actual results to differ materially from those contained in the prospective financial information, including, among others, the risks and uncertainties described in Risk Factors and Forward-Looking Statements. Accordingly, there can be no assurance that the prospective results are indicative of our future performance or that of the combined company or that actual results will not differ materially from those presented in the prospective financial information. Inclusion of the prospective financial information in this Proxy Statement should not be regarded as a representation by any person that the results contained in the prospective financial information will be achieved.

Neither we nor Model Reorg intends to update or otherwise revise the prospective financial information to reflect circumstances existing or events occurring, including changes in general economic or industry conditions, since its preparation, even if any or all of the underlying assumptions are shown to be in error.

111

MARKET PRICE AND DIVIDEND INFORMATION; RELATED SHAREHOLDER MATTERS

Our common stock is traded on the NASDAQ Capital Market under the symbol ECMV. The following table sets forth the high and low closing sales prices per share of our common stock for the periods indicated, as reported on the NASDAQ Capital Market.

	High	Low
Fiscal 2008		
First Quarter	\$ 26.92	\$ 14.02
Second Quarter (through July 24)	\$ 23.25	\$ 12.62
Fiscal 2007		
First Quarter	\$ 31.53	\$ 23.07
Second Quarter	27.87	21.95
Third Quarter	26.41	19.01
Fourth Quarter	27.31	18.99
Fiscal 2006		
First Quarter	\$ 21.92	\$ 16.00
Second Quarter	23.20	10.60
Third Quarter	16.65	8.83
Fourth Quarter	24.17	13.94

As of June 24, 2008, we had 46 shareholders of record, which excludes shareholders holding shares in street name. On November 16, 2006 and December 20, 2007, the last full trading days before the days on which we issued press releases announcing the proposed Merger and the execution of the Original Merger Agreement, respectively, the closing prices per share of our common stock as reported on the NASDAQ Capital Market was \$18.08 and \$23.94, respectively.

We have not declared or paid any dividends on our common stock and do not currently intend to declare or pay cash dividends in the foreseeable future. Payment of dividends, if any, will be at the discretion of the Board of Directors after taking into account various factors, including our financial condition, results of operations, current and anticipated cash needs and plans for expansion. In addition, any third party senior credit facility that we obtain from time to time will restrict our payment of dividends.

There is currently no public market for Model Reorg common stock. Upon completion of the Merger, all outstanding shares of Model Reorg common stock will convert automatically into shares of our common stock in accordance with the Merger Agreement. There are seven holders of the outstanding shares of Model Reorg common stock.

112

SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT

The following tables set forth the beneficial ownership by the persons listed below of shares of our common stock and Model Reorg common stock, respectively, at the Record Date, June 24, 2008, and giving effect to the Merger. At the Record Date, we had 3,059,041 shares outstanding and Model Reorg had 96.9 shares outstanding. Beneficial ownership is determined under the rules of the SEC and includes voting or dispositive power over the securities. Shares of our common stock subject to the Convertible Note and the Warrants, and shares subject to options that are exercisable currently or within 60 days of the Record Date are considered beneficially owned by the person holding the Convertible Note, Warrants, and/or options, respectively, for the purpose of computing the percentage ownership of that person, but are not treated as outstanding for the purpose of computing the percentage ownership of any other person. Unless otherwise indicated, each person has sole voting and investment power over the shares shown or shares such power with his or her spouse.

Ownership of E Com Before the Merger

The following table sets forth the beneficial ownership of our common stock at the Record Date by each person that is the beneficial owner of more than five percent of our common stock, each of our current director and officers, and all such directors and executive officers as a group.

	Shares Owned	Percent of Class
Principal Shareholders		
Stephen L. Nussdorf and Glenn H. Nussdorf	1,557,589(1)	44.5%
35 Sawgrass Drive		
Bellport, New York 11713		
Jeffrey L. Feinberg	600,703(2)	19.6%
JLF Asset Management, L.L.C.		
2775 Via de la Valle Suite 204		
Del Mar, CA 92014		
Steven L. Martin	213,900(3)	7.0%
Slater Capital Management, L.L.C.		
825 Third Avenue 33rd Floor		
New York, NY 10022		
Other Directors and Executive Officers		
Michael W. Katz	100,000(4)	3.2%
Alan Grobman	0	
Joseph Bouhadana	5,000(4)	*
Paul Garfinkle	4,500(4)	*
Carole A. Taylor	3,000(4)	*
Donovan Chin	27,250(4)	*
All current directors and executive officers as a group (7 persons)	1,697,339(4)	46.6%

^{*} Indicates less than 1%.

(1)

Includes 444,445 shares of common stock issuable upon conversion of the Convertible Note as to which Stephen and Glenn Nussdorf would share voting and investment power.

(2) As reported in a Schedule 13G filed with the SEC on February 13, 2008, Mr. Feinberg and JLF Asset Management, L.L.C. share voting and investment power over these shares. Mr. Feinberg is the Managing Member of JLF Asset Management, L.L.C., which controls JLF Partners I, L.P. and JLF Offshore Fund, Ltd., which share voting and investment power over their respective shares. As reported in a Form 4 filed on March 21, 2008, JLF Offshore Fund, Ltd. holds 334,558 of these shares (10.9% of the class) and the remaining 266,145 shares are held in accounts of private investment vehicles.

113

- (3) As reported in a Schedule 13G filed with the SEC on February 15, 2008, Mr. Martin is the manager of Slater Capital Management, L.L.C., with which he shares voting and investment power over these shares, which are held in private investment funds for the benefit of clients. Additional information was provided in a Form 13F filed with the SEC by Slater Capital Management, L.L.C. on May 13, 2008.
- (4) Includes shares of common stock issuable upon the exercise of stock options exercisable currently or within 60 days of the Record Date in the following amounts: Michael W. Katz (100,000); Joseph Bouhadana (5,000); Paul Garfinkle (4,500); Carole A. Taylor (3,000); Donovan Chin (27,250); and all current directors and executive officers as a group (139,750).

Ownership of Model Reorg Before the Merger

The following table sets forth the beneficial ownership of Model Reorg common stock at the Record Date by each person that is the beneficial owner of more than five percent of Model Reorg s common stock, each current director and executive officer of Model Reorg, and all such directors and executive officers as a group.

	Shares Owned	Percent of Class
Stephen L. Nussdorf	28 ² /3	29.5838
Glenn H. Nussdorf	28 ² /3	29.5838
Arlene Nussdorf	28 ² /3	29.5838
Rene A. Garcia	10.9(1)	11.2486
Michael W. Katz	0	
All current directors and executive officers as a group (4 persons)	57 1/3	59.167

(1) Includes a total of 5.232 shares (5.4% of the class) held by trusts for the benefit of Mr. Garcia s children, for which Mr. Garcia and Rafael Villoldo serve as trustees and as to which they disclaim any pecuniary interest.

Pro Forma Ownership of Combined Company

The following table provides information, after giving pro forma effect to the Merger, with respect to the anticipated beneficial ownership of our common stock by each person expected to be the beneficial owner of more than five percent of our common stock anticipated to be outstanding after the Merger, each person expected to be a director or executive officer following the Merger, and all such directors and executive officers as a group (based on each such person scurrent beneficial ownership of our and/or Model Reorg shares). The amounts shown for the Model Reorg shareholders include the aggregate 295,000 shares that we will hold in escrow following the Merger.

	CI O I		Shares Owned	
	Shares Owned Excluding Warrants	Percent of Class	Assuming Exercise of Warrants(1)	Percent of Class
Principal Shareholders				
Stephen L. Nussdorf and Glenn H. Nussdorf	5,048,477(2)	53.7%	5,935,991(3)	57.7%
Arlene Nussdorf	1,745,444	19.5%	2,189,201(4)	23.3%
Rene A. Garcia	663,668(5)	7.4%	832,397(6)	9.1%
Jeffrey L. Feinberg	600,703(7)	6.7%	600,703(7)	6.7%
Other Directors and Executive Officers				
Michael W. Katz	100,000(8)	1.1%	100,000(8)	*
Joseph Bouhadana	5,000(8)	*	5,000(8)	*
Paul Garfinkle	4,500(8)	*	4,500(8)	*
Carole A. Taylor	3,000(8)	*	3,000(8)	*
Donna Dellomo		*		*
Raymond Piergiorgi		*		*
Donovan Chin	27,250(8)	*		*
All directors and executive officers as a group (8 persons)	5,188,227(8)	54.4%	6,075,741(8)(9)	58.2%

Table of Contents

*	Indicates less than 1%.
(1)	Warrants are not exercisable until the third anniversary of the effective date of the Merger, subject to acceleration in certain circumstance (see Approval to Issue Shares and Warrants The Warrants).
(2)	Includes 444,445 shares of common stock issuable upon conversion of the Convertible Note as to which Stephen and Glenn Nussdorf would share voting and investment power.
(3)	Includes 443,757 shares issuable upon exercise of Warrants issuable to each of Stephen and Glenn Nussdorf.
(4)	Includes 443,757 shares issuable upon exercise of Warrants.
(5)	Includes a total of 318,561 shares held by trusts for the benefit of Mr. Garcia s children, for which Mr. Garcia and Rafael Villoldo serve trustees and as to which they disclaim any pecuniary interest.
(6)	Includes a total of 168,729 shares issuable upon exercise of Warrants issuable to Mr. Garcia and the trusts for the benefit of Mr. Garcia schildren.
(7)	As reported in a Schedule 13G filed with the SEC on February 13, 2008, Mr. Feinberg and JLF Asset Management, L.L.C. share voting and investment power over these shares. Mr. Feinberg is the Managing Member of JLF Asset Management, L.L.C., which controls JLF Partners I, L.P. and JLF Offshore Fund, Ltd., which share voting and investment power over their respective shares.
(8)	Includes shares of common stock issuable upon the exercise of stock options exercisable currently or within 60 days of the Record Date in the following amounts: Michael W. Katz (100,000); Joseph Bouhadana (5,000); Paul Garfinkle (4,500); Carole A. Taylor (3,000); Donovan Chin (27,250); and all current directors and executive officers as a group (139,750).
(9)	Includes an aggregate of 887,514 shares issuable upon exercise of Warrants beneficially owned by Stephen Nussdorf.

Table of Contents 164

115

WHERE YOU CAN FIND MORE INFORMATION

We file annual, quarterly and current reports, proxy statements and other information with the SEC. You may read and copy any document we file at the SEC s Public Reference Room at 100 F Street, N.E., Room 1580, Washington, D.C. 20549, and you may also obtain copies by mail, at prescribed rates, from the Public Reference Section of the SEC at the same address. You may call the SEC at 1-800-732-0330 for information on the Public Reference Room. The SEC maintains a website that contains annual, quarterly and current reports, proxy statements and other information that issuers, including us, file electronically with the SEC. The SEC s website is located at www.sec.gov.

We have supplied all information contained in this Proxy Statement relating to us, and Model Reorg has supplied all such information relating to Model Reorg.

You should rely only on the information contained in this Proxy Statement to vote on the proposals at the Special Meeting. We have not authorized anyone to provide you with information that is different from what is contained in this Proxy Statement. This Proxy Statement is dated July 25, 2008. You should not assume that the information contained in the Proxy Statement is accurate as of any date other than such date, and neither the mailing of this Proxy Statement to shareholders nor the issuance of our common stock in the Merger shall create any implication to the contrary.

SHAREHOLDER PROPOSALS FOR THE 2008 ANNUAL MEETING

Shareholder proposals intended to be presented at our 2008 Annual Meeting of Shareholders pursuant to the provisions of Rule 14a-8 of the Securities and Exchange Commission promulgated under the Securities Exchange Act of 1934, as amended, must be received by our Corporate Secretary at the address below by September 12, 2008 for inclusion in our Proxy Statement and form of proxy relating to such Annual Meeting. Any shareholder proposal submitted other than for inclusion in our proxy materials for that meeting must be delivered to us no later than November 26, 2008, or such proposal will be considered untimely. If a shareholder proposal is received after November 26, 2008, we may vote in our discretion as to the proposal all of the shares for which we have received proxies for the 2008 Annual Meeting of Shareholders.

Send all proposals or nominations to Donovan Chin, Secretary, E Com Ventures, Inc., 251 International Parkway, Sunrise, Florida 33325.

116

INDEX TO FINANCIAL STATEMENTS

	Page
E Com Ventures, Inc. and Subsidiaries	
Audited Financial Statements	
Report of Independent Registered Public Accounting Firm	F-2
Consolidated Balance Sheets as of February 2, 2008 and February 3, 2007 (as restated)	F-3
Consolidated Statements of Operations for the Fiscal Years Ended February 2, 2008.	
February 3, 2007 (as restated), and January 28, 2006 (as restated)	F-4
Consolidated Statements of Shareholders Equity for the Fiscal Years Ended	
February 2, 2008, February 3, 2007 (as restated), and January 28, 2006 (as restated)	F-5
Consolidated Statements of Cash Flows for the Fiscal Years Ended February 2, 2008,	
February 3, 2007 (as restated), and January 28, 2006 (as restated)	F-6
Notes to Consolidated Financial Statements	F-7
Unaudited Financial Statements	
Condensed Consolidated Balance Sheets as of May 3, 2008 and February 2, 2008	F-26
Condensed Consolidated Statements of Operations for each of the thirteen-week periods	
ended May 3, 2008 and May 5, 2007 (as restated)	F-27
Condensed Consolidated Statements of Cash Flows for each of the thirteen-week periods	
ended May 3, 2008 and May 5, 2007 (as restated)	F-28
Notes to Condensed Consolidated Financial Statements	F-29
Model Reorg, Inc. and Subsidiaries	
Audited Financial Statements	
Independent Auditors Report	F-37
Consolidated Balance Sheets as of October 31, 2007 and 2006 (as restated)	F-38
Consolidated Statements of Income for the Fiscal Years Ended	
October 31, 2007, 2006 (as restated) and 2005 (as restated)	F-39
Consolidated Statements of Stockholders Equity for the Fiscal Years Ended	
October 31, 2007, 2006 (as restated) and 2005 (as restated)	F-40
Consolidated Statements of Cash Flows for the Fiscal Years Ended	
October 31, 2007, 2006 and 2005 (each as restated)	F-41
Notes to Consolidated Financial Statements	F-42
Unaudited Financial Statements	
Consolidated Balance Sheets as of April 30, 2008 and October 31, 2007	F-53
Consolidated Statement of Income for each of the three and six-month periods	
ended April 30, 2008 and April 30, 2007	F-54
Consolidated Statements of Cash Flows for the six-month periods	
ended April 30, 2008 and April 30, 2007	F-55
Notes to Consolidated Financial Statements	F-56

F-1

To the Board of Directors and Shareholders of

E Com Ventures, Inc. and Subsidiaries

REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

E Com Ventures, Inc.
Sunrise, Florida
We have audited the accompanying consolidated balance sheets of E Com Ventures, Inc. and subsidiaries (the Company) as of February 2, 2008 and February 3, 2007, and the related consolidated statements of operations, shareholders equity, and cash flows for each of the three years in the period ended February 2, 2008. These financial statements are the responsibility of the Company s management. Our responsibility is to express an opinion on the financial statements based on our audits.
We conducted our audits in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. The Company is not required to have, nor were we engaged to perform, an audit of its internal control over financial reporting. Our audits included consideration of internal control over financial reporting as a basis for designing audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control over financial reporting.

Accordingly, we express no such opinion. An audit also includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, such consolidated financial statements present fairly, in all material respects, the financial position of E Com Ventures, Inc. and

subsidiaries as of February 2, 2008 and February 3, 2007, and the results of their operations and their cash flows for each of the three years in the

period ended February 2, 2008, in conformity with accounting principles generally accepted in the United States of America.

As discussed in Note 3, the accompanying consolidated financial statements for the years ended February 3, 2007 and January 28, 2006 have

been restated.

/S/ DELOITTE & TOUCHE LLP

Deloitte & Touche LLP

Certified Public Accountants

Fort Lauderdale, Florida

June 5, 2008

E COM VENTURES, INC. AND SUBSIDIARIES

CONSOLIDATED BALANCE SHEETS

	FEB	BRUARY 2, 2008	FEBRUARY 3, 2007 (AS RESTATED, SEE NOTE 3)		
ASSETS:					
Current assets:					
Cash	\$	1,035,073	\$	1,282,546	
Trade receivables, no allowance required		1,057,499		954,664	
Deferred tax asset-current		2,261,856		2,821,584	
Inventories, net		107,479,019		78,427,029	
Prepaid expenses and other current assets		2,228,013		3,469,201	
Total current assets		114,061,460		86,955,024	
Property and equipment, net		36,587,935		28,246,433	
Goodwill		1,904,448		1,904,448	
Deferred tax asset: non-current		6,980,518		7,228,179	
Other assets, net		300,250		388,099	
Total assets	\$	159,834,611	\$	124,722,183	
LIABILITIES AND SHAREHOLDERS EQUITY:					
Current liabilities:					
Accounts payable non affiliates	\$	19,609,065	\$	16,748,142	
Accounts payable affiliates		48,650,294		24,110,130	
Accrued expenses and other liabilities		10,327,794		8,304,225	
Bank line of credit		32,840,872		26,919,115	
Subordinated convertible note payable affiliate		5,000,000			
Current portion of obligations under capital leases		355,376		345,424	
Total current liabilities		116,783,401		76,427,036	
Subordinated convertible note payable affiliate				5,000,000	
Long-term portion of obligations under capital leases		7,190,268		7,552,915	
Total liabilities		123,973,669		88,979,951	
Commitments and contingencies (See Note 12)					
Shareholders equity: Preferred stock, \$0.10 par value, 1,000,000 shares authorized, none					
issued					
Common stock, \$.01 par value, 6,250,000 shares authorized;					
3,957,290 and 3,950,664 shares issued at fiscal year-end 2007 and					
2006, respectively		39,573		39,507	
Additional paid-in capital		79,182,694		79,069,780	
Accumulated deficit		(34,784,381)		(34,790,111)	
Treasury stock, at cost, 898,249 shares		(8,576,944)		(8,576,944)	
Total shareholders equity		35,860,942		35,742,232	
Total liabilities and shareholders equity	\$	159,834,611	\$	124,722,183	

See accompanying notes to consolidated financial statements.

E COM VENTURES, INC. AND SUBSIDIARIES

CONSOLIDATED STATEMENTS OF OPERATIONS

	February 2, 2008	FOR THE FISCAL YEA February 3, 2007 February 2, 2008 (as restated, see Note 3		January 28, 2006		
Net sales to:	1 columny 2, 2000	(us resture	u, see 1 (dee e)	(451	istatea, see 1 (ote 5)	
Unrelated customers	\$ 244,119,470	\$ 2	229,985,494	\$	215,849,482	
Affiliates	57,715,160		13,623,604		17,844,599	
	- 1,1 - 1, - 1		-,,-		,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	
	301,834,630	2	243,609,098		233,694,081	
Cost of goods sold to:	301,031,030		213,007,070		233,071,001	
Unrelated customers	135,507,782	1	26,983,531		120,490,161	
Affiliates	54,461,686		12,695,946		16,702,761	
	2 1, 10 2,000		,0,0,0,0		,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	
	189,969,468	1	139,679,477		137,192,922	
	107,707,400		137,077,777		137,172,722	
Gross profit	111 965 160	1	102 020 621		96,501,159	
Gross profit	111,865,162	J	103,929,621		90,301,139	
Operating expenses:	00.074.054		00 002 077		01.007.602	
Selling, general and administrative expenses	99,974,054		89,003,867		81,005,603	
Depreciation and amortization	6,196,880		4,796,947		4,830,371	
Total operating expenses	106,170,934		93,800,814		85,835,974	
Income from operations	5,694,228		10,128,807		10,665,185	
Other expenses:						
Interest expense						
Affiliates	(451,354)		(465,798)		(371,458)	
Other	(4,277,077)	(4,028,943)			(3,506,018)	
	(4,728,431)		(4,494,741)		(3,877,476)	
Income before income taxes	965,797		5,634,066		6,787,709	
Income tax (provision) benefit	(960,067)		(1,244,404)		8,471,308	
meone un (provision) conent	(500,007)		(1,2 : 1, 10 1)		0,171,000	
Net income	\$ 5,730	\$	4,389,662	\$	15,259,017	
Net income	\$ 5,730	Φ	4,369,002	φ	13,239,017	
	ф	ф	1.46	ф	5 17	
Basic net income per common share	\$	\$	1.46	\$	5.17	
Diluted net income per common share	\$	\$	1.38	\$	4.51	
Weighted average number of shares outstanding:						
Basic	3,058,797		3,000,471		2,949,146	
Diluted	3,058,797		3,505,890		3,463,480	

See accompanying notes to consolidated financial statements.

F-4

E COM VENTURES, INC. AND SUBSIDIARIES

CONSOLIDATED STATEMENTS OF SHAREHOLDERS EQUITY

FOR THE FISCAL YEARS ENDED FEBRUARY 2, 2008, FEBRUARY 3, 2007 AND JANUARY 28, 2006

	Common Stock		Additional	ditional Treasury Stock			
			Paid-In			Accumulated	
	Shares	Amount	Capital	Shares	Amount	Deficit	Total
Balance at January 29, 2005 as originally reported	3,834,684	\$ 38,347	\$ 75,347,588	898,249	\$ (8,576,944)	\$ (51,748,982)	\$ 15,060,009
Prior period adjustment, see Note 3						(2,689,808)	(2,689,808)
Balance at January 29, 2005 (as restated, see Note							
3)	3,834,684	38,347	75,347,588	898,249	(8,576,944)	(54,438,790)	12,370,201
Net income (as restated, see Note 3)						15,259,017	15,259,017
Exercise of stock options	22,532	225	83,507				83,732
Receipt of profits under Section 16(b) short swing							
profit liability of the Exchange Act			181,591				181,591
Excess tax benefit from exercise of stock options			2,242,000				2,242,000
Executive compensation contributed to capital			406,000				406,000
Balance at January 28, 2006 (as restated, see Note							
3)	3,857,216	38,572	78,260,686	898,249	(8,576,944)	(39,179,773)	30,542,541
Net income (as restated, see Note 3)						4,389,662	4,389,662
Exercise of stock options	93,448	935	681,209				682,144
Share based compensation expense			64,256				64,256
Excess tax benefit from exercise of stock options			63,629				63,629
Balance at February 3, 2007 (as restated, see Note							
3)	3,950,664	39,507	79,069,780	898,249	(8,576,944)	(34,790,111)	35,742,232
Net income						5,730	5,730
Exercise of stock options	6,626	66	36,170				36,236
Share based compensation expense			70,719				70,719
Excess tax benefit from exercise of stock options			6,025				6,025
Balance at February 2, 2008	3,957,290	\$ 39,573	\$ 79,182,694	898,249	\$ (8,576,944)	\$ (34,784,381)	\$ 35,860,942

See accompanying notes to consolidated financial statements.

E COM VENTURES, INC. AND SUBSIDIARIES

CONSOLIDATED STATEMENTS OF CASH FLOWS

				E FISCAL YEAR EN	Jai	January 28, 2006		
C1 fl f	Februar	y 2, 2008	(as res	tated, see Note 3)	(as res	stated, see Note 3)		
Cash flows from operating activities: Net income	¢	5 720	\$	1 290 662	\$	15 250 017		
Net income	\$	5,730	Ф	4,389,662	Þ	15,259,017		
Adjustments to reconcile net income to net cash provided								
by operating activities:								
Deferred income taxes	8	307,389		1,063,404		(8,871,305)		
Tax benefit from exercise of stock options		34,428		63,629				
Excess tax benefit from stock option exercises		(6,025)		(63,629)				
Provision for impairment of assets and store closings	3	33,550		165,847		162,370		
Depreciation and amortization	6,1	96,880		4,796,947		4,830,371		
Share-based compensation		70,719		64,256				
Change in operating assets and liabilities:								
Trade receivables	(1	.02,835)		(135,592)		(123,260)		
Inventories	(29,0	51,990)		(5,450,185)		5,952,794		
Prepaid expenses and other current assets		241,188		(2,519,055)		199,577		
Other assets		87,171		68,850		218,281		
Accounts payable, non-affiliate		361,818		3,142,103		(4,640,529)		
Accounts payable, affiliates		540,164		(2,795,303)		3,677,108		
Accrued expenses and other liabilities		15,872		(166,143)		1,859,501		
riceraed expenses and other habitates	1,7	13,072		(100,115)		1,000,001		
Net cash provided by operating activities	8,7	734,059		2,624,791		18,523,925		
Cash flows from investing activities:								
Additions to property and equipment	(14,5	592,855)		(9,797,300)		(7,143,201)		
Net cash used in investing activities	(14,5	592,855)		(9,797,300)		(7,143,201)		
Cash flows from financing activities:								
Net borrowings (repayments) under bank line of credit	5,9	21,757		6,771,137		(11,380,234)		
Principal payments under capital lease obligations		352,695)		(322,299)		(254,912)		
Proceeds from exercise of stock options		36,236		682,144		83,732		
Excess tax benefit from stock option exercises		6,025		63,629		,		
Receipt of profits under								
Section 16(b)- short swing profit liability of the Exchange Act						181,591		
Net cash provided by (used in) financing activities	5,6	511,323		7,194,611		(11,369,823)		
(Decrease) increase in cash and cash equivalents		247,473)		22,102		10,901		
Cash and cash equivalents at beginning of period	1,2	282,546		1,260,444		1,249,543		
Cash and cash equivalents at end of period	\$ 1,0	035,073	\$	1,282,546	\$	1,260,444		
Cash paid during the period for:								
Interest		43,408	\$	4,379,131	\$	3,612,573		
Income taxes	\$ 2	231,444	\$	306,963	\$	200,000		

See accompanying notes to consolidated financial statements.

F-6

E COM VENTURES, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

FOR THE FISCAL YEARS ENDED FEBRUARY 2, 2008, FEBRUARY 3, 2007 AND JANUARY 28, 2006

NOTE 1 NATURE OF BUSINESS

E Com Ventures, Inc., a Florida corporation (ECOMV or the Company), performs all of its operations through two wholly-owned subsidiaries, Perfumania, Inc. (Perfumania), a Florida corporation, which is a specialty retailer and wholesaler of fragrances and related products and perfumania.com, inc., a Florida corporation which is an Internet retailer of fragrances and other specialty items.

Perfumania s retail stores are located in regional malls, manufacturers outlet malls, life style centers, airports and on a stand-alone basis in suburban strip shopping centers, throughout the United States and Puerto Rico. The number of retail stores in operation at February 2, 2008, February 3, 2007, and January 28, 2006 were 303, 267 and 239, respectively.

NOTE 2 SIGNIFICANT ACCOUNTING POLICIES

Significant accounting principles and practices used by the Company in the preparation of the accompanying consolidated financial statements are as follows:

FISCAL YEAR END

The Company s fiscal year ends on the Saturday closest to January 31 to enable the Company s operations to be reported in a manner which more closely coincides with general retail reporting practices and the financial reporting needs of the Company. In the accompanying notes, fiscal year 2007, 2006 and 2005 refer to the years ended February 2, 2008, February 3, 2007 and January 28, 2006, respectively. Fiscal years 2007 and 2005 each contain fifty-two weeks, while fiscal year 2006 contains fifty-three weeks, with the fifty-third week falling in the fiscal fourth quarter.

MANAGEMENT ESTIMATES

The preparation of financial statements in conformity with accounting principles generally accepted in the United States of America (GAAP) requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities as of the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. The most significant estimates made by management in the accompanying consolidated financial statements relate to the valuation of inventory balances, self-insured health care accruals, long-lived asset impairments and estimated useful lives of property and equipment, deferred tax assets and goodwill. Actual results could differ from those estimates.

RECLASSIFICATIONS

Certain amounts from the prior periods have been reclassified to conform to the current period presentation. Net sales and cost of goods sold on our consolidated statements of operations for fiscal years 2006 and 2005 were reclassified to disclose transactions with both unrelated customers and affiliates. These reclassifications had no effect on net income, as previously reported.

PRINCIPLES OF CONSOLIDATION

The consolidated financial statements include accounts of E Com Ventures, Inc. and its wholly owned subsidiaries. All intercompany balances and transactions have been eliminated in consolidation.

Table of Contents

CASH AND CASH EQUIVALENTS

The Company considers all highly liquid investments purchased with an original maturity of three months or less to be cash equivalents.

TRADE RECEIVABLES

The Company s trade receivables consist primarily of credit card receivables relating to the Company s retail stores. Generally, there are three to four days of retail sales transactions outstanding with third-party credit card vendors at any point in time.

INVENTORIES

Inventories, consisting of finished goods, are stated at the lower of cost or market with cost being determined on a weighted average method. The cost of inventory includes product cost and freight charges. Writeoffs of potentially slow moving or damaged inventory are recorded based on management s analysis of inventory levels, future sales forecasts and through specific identification of obsolete or damaged merchandise. Inventory shrinkage is estimated and accrued between physical inventory counts.

PROPERTY AND EQUIPMENT

Property and equipment is carried at cost, less accumulated depreciation and amortization. Depreciation for property and equipment, which includes assets under capital leases, is calculated using the straight-line method over the estimated useful lives of the related assets. Leasehold improvements are amortized over the shorter of the term of the lease including one stated renewal period that is reasonably assured, or the estimated useful lives of the improvements, generally ten years. Costs of major additions and improvements are capitalized and expenditures for maintenance and repairs which do not extend the useful life of the asset are expensed when incurred. Gains or losses arising from sales or retirements are reflected in operations.

GOODWILL

Goodwill represents the excess purchase price paid over net assets of businesses acquired resulting from the application of the purchase method of accounting. Goodwill is not amortized but is tested annually for impairment at the end of the Company s fiscal year or more frequently if events or changes in circumstances indicate the carrying value of goodwill may not fully be recoverable. No impairment was identified as a result of the annual tests in fiscal years 2007, 2006, and 2005.

OTHER INTANGIBLE ASSETS

Other intangible assets include store design, real estate leases and non-compete agreements which were recorded based upon their relative fair values at the date of acquisition as determined by management. Other intangible assets do not include goodwill. The amortization of intangible assets totaled approximately \$140,000 in fiscal year 2005. There was no amortization of intangible assets during fiscal years 2007 and 2006 as all intangible assets with finite lives were fully amortized as of the beginning of the fiscal year 2006.

GIFT CARDS

Upon the purchase of a gift card by a retail customer, a liability is established for the cash value of the gift card. The liability is included in accrued expenses and other liabilities. The liability is relieved and revenue is recognized at the time of the redemption of the gift card. Over time, some portion of gift cards issued is not redeemed. This amount is recorded as a reduction of selling, general and administrative expenses, when it can be determined that the likelihood of the gift card being redeemed is remote and there is no legal obligation to remit

F-8

Table of Contents

the unredeemed gift cards to relevant jurisdictions (often referred to as gift card breakage). Gift cards issued by the Company do not have expiration dates. In fiscal year 2007, we recorded gift card breakage of approximately \$140,000. There was no gift card breakage recorded in prior years.

REVENUE RECOGNITION

Revenue from wholesale transactions is recorded upon shipment of inventory when risk of ownership and title transfers to the buyer. Revenue from store sales is recorded, net of discounts, at the point of sale. Revenue from Internet sales is recognized at the time products are delivered to customers. Shipping and handling revenue from our Internet sales is included as a component of net sales and amounted to approximately \$298,000, \$289,000 and \$367,000 in fiscal years 2007, 2006 and 2005, respectively. Revenue from gift cards is recognized at the time of redemption. Returns of store and Internet sales are allowed within 30 days of purchase.

SALES RETURN ALLOWANCE

Allowances for retail sales returns are estimated and accrued based on our historical return patterns. Historical sales returns have not been significant. In fiscal year 2007, the Company accrued a sales return allowance of approximately \$150,000. No allowance was recorded in prior years.

COST OF GOODS SOLD

Cost of goods sold include the cost of merchandise sold, inventory valuation, inventory shortages, damages and freight charges. Costs relating to our distribution center are included in selling, general and administrative expenses and were approximately \$2.8 million, \$2.6 million and \$2.4 million in fiscal years 2007, 2006 and 2005, respectively.

SELLING, GENERAL AND ADMINISTRATIVE EXPENSES

Selling, general and administrative expenses include payroll and related benefits for our store operations, field management, distribution center and corporate office; rent, common area maintenance, real estate taxes and utilities for our stores, distribution center and corporate office; advertising, insurance, supplies, professional fees and other administrative expenses. Included in selling, general and administrative expenses in fiscal year 2007 are \$1.2 million of expenses related to a planned merger with Model Reorg, Inc. These expenses consist primarily of financial advisory, legal and due diligence fees. See further discussion at Note 6.

INCOME TAXES

Income tax expense is based principally on pre-tax financial income. Deferred tax assets and liabilities are recognized for the differences between the financial reporting carrying values and the tax bases of assets and liabilities and are measured using the enacted tax rates and laws that will be in effect when the differences are expected to reverse. A valuation allowance is recognized to reduce net deferred tax assets to amounts that management believes are more likely than not expected to be realized.

The Company adopted the provisions of FIN 48, Accounting for Uncertainty in Income Taxes an interpretation of FASB Statement No. 109, (FIN 48) effective February 4, 2007. FIN 48 prescribes a comprehensive model for the financial statement recognition, measurement, presentation and disclosure of uncertain tax positions taken or expected to be taken in an income tax return. Under FIN 48, the Company may recognize the tax benefit from an uncertain tax position only if it is more likely than not that the tax position will be sustained on examination by the taxing authorities, based on the technical merits of the position. The tax benefits recognized in the financial statements from such a position should be measured based on the largest benefit that has a greater than fifty percent likelihood of being realized upon ultimate settlement. FIN 48 also provides guidance on derecognition, classification, interest and penalties on income taxes, accounting in interim periods and requires increased disclosures.

As a result of the implementation of FIN 48, the Company did not recognize a liability for unrecognized tax benefits or reverse any accruals for uncertain tax positions previously recorded, and accordingly, was not required to record any cumulative effect adjustment to beginning of year retained earnings. As of both the date of adoption and February 2, 2008, there was no liability for income tax associated with unrecognized tax benefits.

The Company classifies interest related to unrecognized tax benefits as well as any related penalties in operating expenses in its consolidated statements of operations, which is consistent with the recognition of these items in prior reporting periods. As of the date of adoption, the Company was not required to have an accrual for the payment of interest and penalties. No accrual for interest and penalties related to uncertain tax positions was required as of February 2, 2008.

The Company operates stores throughout the United States and Puerto Rico, and as a result, files income tax returns in the United States federal jurisdiction and various state and foreign jurisdictions. In the normal course of business, the Company is subject to examination by taxing authorities. The statute of limitations for examinations by the Internal Revenue Service has expired for years ending on and before January 31, 2005.

State and foreign income tax returns are generally subject to examination for a period of three to five years after filing of the respective return. The state impact of any federal changes remains subject to examination by various states for a period of up to one year after formal notification to the states. The Company is not currently under examination in any state or foreign jurisdictions.

BASIC AND DILUTED NET INCOME PER COMMON SHARE

Basic net income per common share is computed by dividing net income by the weighted average number of common shares outstanding during the period. Diluted net income per common share includes, in periods in which they are dilutive, the dilutive effect of those common stock equivalents where the average market price of the common shares exceeds the exercise prices for the respective years.

Basic and diluted net income per common share are computed as follows:

	200	07		AL YEAR 006	2	2005
Net income basic	\$	5,730	\$ 4,3	89,662	\$ 15,	,259,017
Add: interest on convertible note			4	65,798		371,458
Net income diluted	\$	5,730	\$ 4,8	55,460	\$ 15,	,630,475
Denominator:						
Weighted average number of shares for basic net income per share	3,05	8,797	3,0	00,471	2,	,949,146
Options to purchase common stock				60,975		69,890
Convertible note			4	44,444		444,444
Denominator for dilutive net income per share	3,058,797		3,505,890		3,	,463,480
Basic net income per common share	\$		\$	1.46	\$	5.17
Diluted net income per common share	\$		\$	1.38	\$	4.51

All common stock equivalents are excluded from the computation of diluted net income per share in fiscal year 2007 because the result is antidilutive. Excluded from the above computations of weighted-average shares for diluted net income per share were options to purchase 4,504 shares and 5,256 shares of common stock for fiscal years 2006 and 2005 respectively, because the exercise price was greater than the average market price of the Company s common stock during the period and therefore, the effect was antidilutive. The convertible note was issued in December 2004. See further discussion at Note 7.

F-10

Table of Contents

ASSET IMPAIRMENT

The carrying value of long-lived assets is evaluated whenever events or changes in circumstances indicate that the carrying values of such assets may be impaired. An evaluation of recoverability is performed by comparing the carrying values of the assets to projected future cash flows in addition to other quantitative and qualitative analyses. Upon indication that the carrying values of such assets may not be recoverable, the Company recognizes an impairment loss. The impairment loss is determined based on the difference between the net book value and the fair value of the assets. The estimated fair value is based on anticipated discounted future cash flows. Any impairment is charged to operations in the period in which it is identified. Property and equipment assets are grouped at the lowest level for which there are identifiable cash flows when assessing impairment. Cash flows for retail assets are identified at the individual store level.

SHARE BASED COMPENSATION

The Company has two stock option plans which provide for share-based awards to its employees and directors (collectively, the Plans). Under the Plans, the Company has reserved approximately 1,000,000 shares of common stock, of which approximately 590,000 options have been granted and 148,000 options are outstanding. All stock options have an exercise price that is equal to the fair market value of the Company s stock on the date the options were granted. The term of the stock option awards is ten years from the date of grant. All options are fully vested.

Effective January 29, 2006, the beginning of the Company s first fiscal quarter of 2006, the Company adopted Statement of Financial Accounting Standards (SFAS) No. 123(R), Share-Based Payment (SFAS 123R), using the modified prospective transition method and began recording compensation expense associated with stock options. SFAS 123R requires companies to recognize in the statement of operations the cost of employee services received in exchange for awards of equity instruments based on the grant date fair value of those awards (with limited exceptions). Prior to the adoption of SFAS 123R, the Company accounted for stock-based employee compensation using the intrinsic value method prescribed by Accounting Principles Board (APB) Opinion No. 25, Accounting for Stock Issued to Employees. Accordingly, compensation expense had only been recorded in the consolidated financial statements for any stock options granted below fair market value of the underlying stock as of the date of grant.

The modified prospective transition method requires that share-based compensation expense be recorded for (i) all new stock options granted on or after January 29, 2006 based on the grant date fair value determined under the provisions of SFAS 123R and (ii) all unvested stock options granted prior to January 29, 2006 based on the grant date fair value as determined under the provisions of SFAS 123. Results for prior periods have not been restated, as provided for under the modified prospective transition method.

The Company recognized pre-tax compensation expense of approximately \$71,000 and \$64,000 respectively, in the consolidated statements of operations for stock options granted and vested during fiscal year 2007 and 2006, respectively. All of the stock option compensation expense was recorded as a component of selling, general and administrative expenses in the Company s consolidated statements of operations.

F-11

The following table illustrates the pro forma effects on net income and net income per common share in fiscal year 2005 had compensation expense been recognized based upon the estimated fair value on the grant date of stock options in accordance with SFAS 123 (in thousands except per share amounts).

	 AL YEAR 2005
Net income, as reported	\$ 15,259
Add: Total stock-based employee compensation expense included in reported net income, net of tax	
Deduct: Total stock-based employee compensation expense determined under fair market value based method, net of tax	(883)
Proforma net income	\$ 14,376
Net income per common share-basic:	
As reported	\$ 5.17
Stock based compensation	(0.30)
Proforma	\$ 4.87
Net income per common share-diluted:	
As reported	\$ 4.51
Stock based compensation	(0.26)
Proforma	\$ 4.25

Pro-forma disclosure for fiscal year 2007 and 2006 are not presented above because the share-based compensation amounts are recognized in the accompanying consolidated financial statements.

The fair value for these stock options was estimated at the date of grant using a Black-Scholes option pricing model with the following weighted average assumptions:

	FI	FISCAL YEAR			
	2007	2006	2005		
Expected life (years)	5	5	7		
Expected stock price volatility	206%	173%	164%		
Risk-free interest rates	2.82%	4.39%	3.88%		
Expected dividend yield	0%	0%	0%		

The expected life of the options represents the estimated period of time until exercise and is based on historical experience of similar awards, giving consideration to the contractual terms, vesting schedules and expectations of future employee behavior. The expected stock price volatility is estimated using the historical volatility of the Company s stock. The risk-free interest rate is based on the implied yield available on U.S. Treasury zero coupon issues with a term equal to the option s expected life. The Company has not paid dividends in the past and does not intend to in the foreseeable future.

For the purposes of the proforma presentation of employee share-based compensation expense, the Company currently amortizes the expense over the related vesting period. The weighted average estimated fair values of options granted during fiscal year 2007, 2006, and 2005 were \$23.57, \$21.42 and \$12.09 per share, respectively. The fair value of options that vested during fiscal years 2007, 2006 and 2005 was approximately \$71,000, \$36,000 and \$1,338,000 respectively.

PRE-OPENING EXPENSES

Pre-opening expenses related to new stores are expensed as incurred.

F-12

Table of Contents

SHIPPING AND HANDLING FEES AND COSTS

Income generated from shipping and handling fees is classified as revenues. The Company classifies the costs related to shipping and handling as cost of goods sold. The income and cost associated with shipping and handling when combined is immaterial.

ADVERTISING COSTS

Advertising expense for the fiscal years 2007, 2006 and 2005 was approximately \$2,771,000, \$2,679,000 and \$1,650,000, respectively, and is charged to expense when incurred. There were no cooperative advertising amounts received from vendors for fiscal years 2007, 2006 and 2005.

RENT EXPENSE

Perfumania leases retail stores under operating leases. Minimum rental expenses are recognized over the term of the lease on a straight-line basis. For purposes of recognizing minimum rental expenses, the Company uses the date when possession of the leased space is taken from the landlord, which includes a construction period of approximately two months prior to store opening. For tenant improvement allowances and rent holidays, the Company records a deferred rent liability in accrued expenses on the consolidated balance sheets and amortizes the deferred rent over the terms of the leases as reductions to rent expense on the consolidated statements of operations. For scheduled rent escalation clauses during the lease terms or for rental payments commencing at a date other than the date of initial occupancy, the Company records minimum rental expenses on a straight-line basis over the terms of the leases on the consolidated statements of operations.

Certain leases provide for contingent rents, which are primarily determined as a percentage of gross sales in excess of specified levels and are not measurable at inception. The Company records a contingent rent liability in accrued expenses on the consolidated balance sheets and the corresponding rent expense when specified levels have been achieved or when management determines that achieving the specified levels during the fiscal year is probable.

RECENT ACCOUNTING PRONOUNCEMENTS

In March 2008, the FASB issued SFAS 161, Disclosures about Derivative Instruments and Hedging Activities an amendment of FASB Statement No. 133 (SFAS 161). SFAS 161 requires enhanced disclosures regarding derivatives and hedging activities, including: (a) the manner in which an entity uses derivative instruments; (b) the manner in which derivative instruments and related hedged items are accounted for under Statement of Financial Accounting Standards No. 133, Accounting for Derivative Instruments and Hedging Activities; and (c) the effect of derivative instruments and related hedged items on an entity s financial position, financial performance, and cash flows. SFAS 161 is effective for financial statements issued for fiscal years and interim periods beginning after November 15, 2008. As SFAS 161 relates specifically to disclosures, it will have no impact on the Company s results of operations, financial position and cash flows.

In December 2007, the FASB issued SFAS 141(R), *Business Combinations* (SFAS 141(R)). SFAS 141(R) establishes principles and requirements for how an acquirer recognizes and measures in its financial statements the identifiable assets acquired, the liabilities assumed, the goodwill acquired, and any noncontrolling interest in the acquiree. This statement also establishes disclosure requirements to enable the evaluation of the nature and financial effect of the business combination. SFAS 141(R) is effective for fiscal years beginning after December 15, 2008. The Company is in the process of evaluating the effect that the adoption of SFAS 141(R) will have on its results of operations, financial position, and cash flows.

In December 2007, the FASB issued SFAS No. 160, Noncontrolling Interests in Consolidated Financial Statements an amendment of ARB No. 51. (SFAS 160). SFAS 160 amends ARB 51 to establish accounting

F-13

Table of Contents

and reporting standards for the noncontrolling interest in a subsidiary and for the deconsolidation of a subsidiary. It clarifies that a noncontrolling interest in a subsidiary is an ownership interest in the consolidated entity that should be reported as equity in the consolidated financial statements. SFAS 160 is effective for fiscal years beginning on or after December 15, 2008. The Company is in the process of evaluating the effect that the adoption of SFAS 160 will have on its results of operations, financial position, and cash flows.

In February 2007, the FASB issued SFAS 159, *The Fair Value Option for Financial Assets and Financial Liabilities Including an amendment of FASB Statement No. 115.* (SFAS). SFAS 159 permits entities to choose to measure eligible items at fair value at specified election dates and report unrealized gains and losses on items for which the fair value option has been elected in earnings at each subsequent reporting date. SFAS 159 is effective for fiscal 2008. The Company concluded that the adoption of SFAS 159 will not have a material effect on its results of operations, financial position and cash flows.

In September 2006, the FASB issued SFAS No. 157, Fair Value Measurements (SFAS 157). SFAS 157 defines fair value, establishes a framework for measuring fair value and expands disclosure of fair value measurements. SFAS 157 applies under other accounting pronouncements that require or permit fair value measurements and, accordingly, does not require any new fair value measurements. SFAS 157 is effective for the first interim period beginning in fiscal 2008 for financial assets and liabilities, as well as for any other assets and liabilities that are carried at fair value on a recurring basis in financial statements. In November 2007, the FSAB provided a one year deferral for the implementation of SFAS 157 for other nonfinancial assets and liabilities. The Company does not expect the adoption of SFAS 157 to have a material effect on its results of operations, financial position and cash flows.

NOTE 3 RESTATEMENT

Subsequent to the issuance of the Company s 2006 financial statements, management identified errors in the accounting for operating leases and the period of amortization of leasehold improvements.

Under GAAP, rent expense is amortized on a straight-line basis over the term of the lease. In prior periods, the Company had determined that the term of the lease begins on the commencement date of the lease, which generally coincides with the store opening date. Management has re-evaluated FASB Technical Bulletin No. 85-3, Accounting for Operating Leases with Scheduled Rent Increases and other guidance provided by the SEC and determined that the lease term for amortization purposes should commence on the date the Company takes physical possession of the leased space to commence construction of leasehold improvements, which is generally two months prior to a store opening date. The Company has restated its previously reported financial statements in connection with its accounting for rent expense. Rent expense is included in selling, general and administrative expenses on the Company s consolidated statements of operations.

Management has also reviewed its leasehold improvements to ensure amortization over the shorter of their economic lives or their lease term. In prior periods, in determining the lease terms, in addition to the initial term, the Company had included one anticipated renewal term even if the lease did not have a stated renewal option. The Company has adjusted its lease terms for purposes of calculating amortization of leasehold improvements to include a renewal term only if that renewal option is specified in the lease agreement and that exiting the lease after the initial lease term would result in economic loss and therefore normally management anticipates exercising the respective renewal option.

The restatement resulted in a decrease in net income of \$0.1 million in fiscal year 2006, an increase in net income of \$1.0 million in fiscal year 2005 and a decrease in net income of \$2.7 million for all years prior to fiscal year 2005, with a corresponding increase to the accumulated deficit of as of January 29, 2005. The majority of the adjustments relate to periods prior to fiscal year 2005.

F-14

The following is a summary of the effects of the restatement to the Company s Consolidated Financial Statements.

	Consolidated Statements of Operations As previously				S	
	reported Adjustments			justments	As r	estated
Fiscal year ended February 3, 2007	Ī					
Selling general and administrative expenses	\$ 88,6	599,388	\$	304,479	\$ 89,0	003,867
Depreciation and amortization	4,8	363,319		(66,372)	4,	796,947
Total operating expenses	93,5	562,707		238,107	93,	800,814
Income from operations	10,3	366,914		(238,107)	10,	128,807
Income before income taxes	5,8	372,173		(238,107)	5,0	634,066
Income tax (provision) benefit	(1,3	350,243)		105,839	(1,	244,404)
Net income	4,5	521,930		(132,268)	4,	389,662
Basic net income per common share	\$	1.51	\$	(0.05)	\$	1.46
Diluted net income per common share	\$	1.42	\$	(0.04)	\$	1.38
Fiscal year ended January 28, 2006						
Selling general and administrative expenses	\$ 80,8	39,776	\$	165,827	\$ 81,0	005,603
Depreciation and amortization	5,1	55,645		(325,274)	4,	830,371
Total operating expenses	85,9	95,421		(159,447)	85,	835,974
Income from operations	10,505,738			159,447	10,	665,185
Income before income taxes	6,6	528,262		159,447	6,	787,709
Income tax (provision) benefit	7,637,000 834,308		834,308	8,4	471,308	
Net income	14,265,262			993,755	15,	259,017
Basic net income per common share	\$	4.84	\$	0.33	\$	5.17
Diluted net income per common share	\$	4.23	\$	0.28	\$	4.51

	Consolidated Balance Sheet			
	As previously reported	Adjustments	As restated	
February 3, 2007				
Property and equipment, net	\$ 30,213,222	\$ (1,966,789)	\$ 28,246,433	
Deferred tax asset: non-current	6,288,032	940,147	7,228,179	
Total assets	125,748,825	(1,026,642)	124,722,183	
Accrued expenses and other liabilities	7,502,546	801,679	8,304,225	
Total current liabilities	75,625,357	801,679	76,427,036	
Total liabilities	88,178,272	801,679	88,979,951	
Accumulated deficit	(32,961,790)	(1,828,321)	(34,790,111)	
Total shareholders equity	37,570,553	(1,828,321)	35,742,232	
Total liabilities and shareholders equity NOTE 4 NON-CASH TRANSACTIONS	\$ 125,748,825	\$ (1,026,642)	\$ 124,722,183	

Supplemental disclosures of non-cash investing and financing activities:

		FISCAL YEAR	
NON-CASH TRANSACTIONS	2007	2006	2005
Equipment and building acquired under capital leases	\$	\$	\$ 259,430
Compensation cost for President and Chief Executive Officer contributed to capital			406,000
Accounts payable for property and equipment	413,767	135,369	

F-15

NOTE 5 PROPERTY AND EQUIPMENT

Property and equipment consisted of:

					Estimated Useful Lives
	Fe	bruary 2, 2008	Fel	bruary 3, 2007	(In Years)
Furniture, fixtures and equipment	\$	32,388,238	\$	30,146,067	5-7
Leasehold improvements		44,362,343		35,767,650	shorter of 10 years or lease term
Equipment under capital leases		330,293		330,293	shorter of 5 years or lease term
Building under capital lease		8,188,945		8,188,945	15
		85,269,819		74,432,955	
Less:					
Accumulated depreciation and amortization		(48,681,884)		(46,186,522)	
	\$	36,587,935	\$	28,246,433	

Depreciation and amortization expense for fiscal years 2007, 2006, and 2005 was \$6,196,880, \$4,796,947 and \$4,689,123, respectively. Accumulated depreciation for building and equipment under capital leases was \$2,806,766 and \$2,151,657 as of February 2, 2008 and February 3, 2007, respectively.

See Note 12 for further discussion of capital leases.

NOTE 6 RELATED PARTY TRANSACTIONS

Glenn and Stephen Nussdorf (the Nussdorfs) own an aggregate 1,113,144 shares or approximately 36% of the total number of shares of the Company s common stock as of February 2, 2008, excluding shares issuable upon conversion of the Convertible Note discussed below in Note 6. Stephen Nussdorf has served as the Company s Chairman of the Board since February 2004.

The Nussdorfs are officers and principals of Model and its affiliate, Quality King Distributors, Inc. (Quality King) and their subsidiaries, including Quality King Fragrances, Inc. Model is a diversified wholesale and retail fragrance company and Quality King distributes pharmaceuticals and health and beauty care products. The Company s President and Chief Executive Officer, Michael W. Katz is an executive of Model and Quality King and the Company s principal shareholders, Stephen Nussdorf, the Chairman of the Company s Board of Directors and Glenn Nussdorf, his brother, are shareholders and executives of Model and Quality King.

Sales To and Purchases From Affiliated Companies

During fiscal year 2007, the Company purchased approximately \$32,138,000 of merchandise from Model and its subsidiaries, representing approximately 15% of the Company s total purchases, and sold approximately \$57,715,000 of different merchandise to Model and its subsidiaries, which represented substantially all of the Company s wholesale sales. There were approximately \$32,385,000 and \$30,547,000 of purchases from Model and its subsidiaries and approximately \$13,624,000 and \$17,845,000 of merchandise sales to Model and its subsidiaries during fiscal years 2006 and 2005, respectively. The wholesale sales made to Model and its subsidiaries result from the Company s supplier relationships and its ability to obtain certain merchandise at better prices and in greater quantities than Model and its subsidiaries are able to achieve. The amounts due to

Table of Contents

Model and its subsidiaries at February 2, 2008 and February 3, 2007, were approximately \$36,124,000 and \$16,897,000 respectively. Accounts payable due to Model and its subsidiaries are non-interest bearing and are included in the accounts payable-affiliates in the accompanying consolidated balance sheets.

Glenn Nussdorf beneficially owns approximately 12% of the outstanding common stock of Parlux Fragrances, Inc. (Parlux). In February 2007, Mr. Nussdorf reached an agreement with Parlux which called for equal representation in Parlux s Board of Directors by the then current independent directors and Mr. Nussdorf s nominees. Accordingly, Mr. Nussdorf s three nominees were appointed to the Parlux Board, and one of these Board members currently serves as Parlux s Chief Executive Officer.

Purchases of merchandise from Parlux amounted to approximately \$44,743,000, \$17,644,000 and \$23,004,000 in fiscal years 2007, 2006 and 2005, representing approximately 21%, 12% and 18%, respectively, of the Company s total purchases. In fiscal year 2006, Parlux sold all 378,102 shares of the Company s common stock that it had previously owned to an unrelated third party. The amount due to Parlux on February 2, 2008 and February 3, 2007, was approximately \$12,526,000 and \$7,213,000, respectively. Accounts payable due to Parlux are non-interest bearing. Purchases from related parties are generally payable in 90 days, however, due to the seasonality of the Company s business, these terms are generally extended. Related party accounts are historically brought closer to terms at the end of the holiday season, however, we are dependent upon these extended terms for much of our liquidity during the year.

Merger Agreement With Model and Other Arrangements

As reported in the Company s Form 8-K filed with the SEC on December 21, 2007, the Company entered into an Agreement and Plan of Merger (the Merger Agreement) with Model, a New York corporation controlled by the family of Glenn and Stephen Nussdorf, principal shareholders of the Company, on December 21, 2007. Model is a diversified wholesale and retail fragrance company. Stephen Nussdorf is the Chairman of our Board of Directors, and Michael W. Katz, our President and Chief Executive Officer, is an executive of Model. Pursuant to the terms of the Merger Agreement, Model will be merged into a newly formed wholly-owned subsidiary of the Company in exchange for the issuance of 5,900,000 shares of the Company s common stock and warrants to acquire an additional 1,500,000 shares of the Company s common stock at an exercise price of \$23.94. The warrants will have a 10-year term, will not be exercisable for the first three years and will not be transferable, with limited exceptions.

The Merger Agreement was reached after extensive negotiations between Model and a special committee comprised of the independent Board members of the Company, which retained independent counsel and financial advisors. The special committee received an opinion from Financo, Inc. that, as of the date of the Merger Agreement and subject to various assumptions and qualifications set forth therein, the consideration to be paid by the Company in the merger is fair to the shareholders of the Company (other than shareholders of the Company who own or whose affiliates own securities of Model) from a financial point of view.

The Merger Agreement also requires Model to refinance debt owed by Model to its affiliate, Quality King. The refinancing is to be through a \$50 million three-year term loan from trusts for the benefit of Glenn and Stephen Nussdorf and their families and, if needed, a transfer of inventory from Model to Quality King to pay any remaining balance. The new loan will be subordinated to the new secured credit facility described below.

The consummation of the contemplated Merger is subject to certain conditions, including approval of a majority of the votes cast at a meeting of shareholders of the Company and the availability of a new \$280 million secured credit facility to replace the Company s and Model s existing third party credit facilities.

Included in selling, general and administrative expenses on the Company s consolidated statement of operations for fiscal year 2007 are \$1.2 million of expenses related to the Merger. These expenses consist primarily of financial advisory, legal and due diligence fees.

F-17

Table of Contents

Following the contemplated Merger, the Nussdorf s, who currently own 36% of the Company s outstanding common stock, would own approximately 51% of the Company s outstanding common stock. Giving effect to the conversion of our \$5 million subordinated convertible note currently held by them and the exercise of the warrants to be issued to them in the transaction, but not assuming the exercise of any outstanding options held by the Company s officers and directors, they would own approximately 58% of the Company s common stock.

In addition, the Company s Board approved changing the Company s name to Perfumania Holdings, Inc., subject to shareholder approval.

On March 12, 2008, the Company filed a preliminary proxy statement with the SEC discussing the Merger Agreement and the related required shareholder approvals, including the issuance of the additional shares of the Company s common stock pursuant to the Merger Agreement and the related amendments to the Company s articles of incorporation.

On August 2, 2007, the Company entered into an Information Technology Services Agreement (the Services Agreement) with Model and its subsidiaries, whereby among other services, the Company will manage and monitor the IT systems of Model in exchange for a monthly service fee of \$25,000. The Services Agreement terminates 30 days after the effective date of the merger discussed above, or if such merger offer is not consummated on or before December 31, 2008, either party may terminate 30 days after providing written notice of termination. During fiscal year 2007, the Company recorded income of \$150,000 related to this Services Agreement which is included as a reduction to selling, general and administrative expenses on the consolidated statement of operations for fiscal year 2007.

NOTE 7 BANK LINE OF CREDIT AND NOTES PAYABLE

The bank line of credit and notes payable, all of which is classified as current as of February 2, 2008, consist of the following:

	Feb	oruary 2, 2008	Feb	oruary 3, 2007
Bank line of credit, interest payable monthly, secured by a pledge of substantially all of				
Perfumania s assets	\$	32,840,872	\$	26,919,115
Subordinated convertible note payable-affiliate	\$	5,000,000	\$	5,000,000

Perfumania s senior secured credit facility provides for borrowings of up to \$60 million, depending on the Company s levels of eligible inventories. Advances under the line of credit are based on a formula of eligible inventories and bear interest at a floating rate ranging from (a) prime to prime plus 1.25% or (b) LIBOR plus 2.5% to 3.75% depending on a financial ratio test. As of February 2, 2008, \$32.8 million was outstanding under the line of credit and \$18.1 million was available to support normal working capital requirements and other general corporate purposes based upon our eligible borrowing base, net of the restrictions concerning minimum undrawn availability discussed below. Advances are secured by a first lien on all assets of Perfumania. The credit facility contains limitations on additional borrowings, capital expenditures and other items, and contains various covenants including a fixed charge coverage ratio, a leverage ratio and capital expenditure limits as defined. The credit facility also prohibits the Company from paying cash dividends. In June 2007, the minimum undrawn availability as defined was amended so that Perfumania is required to maintain at all times, a minimum undrawn availability of not less than \$5,000,000, and shall maintain a monthly average undrawn availability, as defined, of not less than \$7,500,000. In May 2008, the fixed charge covenant and the covenant concerning the limitation on capital expenditures were amended with an effective date of February 1, 2008. As a result of these modifications, Perfumania was in compliance with all covenant requirements as of February 2, 2008.

The credit facility expired on May 12, 2008. In May 2008, the Company obtained a ninety day extension of the credit facility to August 11, 2008. As discussed in Note 6, the Merger with Model is subject to a number of

terms, one of which includes the requirement that the Company and Model obtain a commitment for asset-based loans and term debt facilities in amounts and on terms that are substantially similar to those contemplated by a previous commitment that was obtained in August 2007. That commitment provided for replacement senior credit facilities for us and Model in an aggregate amount of up to \$280 million, to be established upon the closing of the Merger. That commitment expired by its terms in November 2007. A comparable commitment on substantially similar terms and amounts was obtained from the same lender on May 16, 2008.

Failure of the Merger with Model to close prior to the expiration of the credit facility on August 11, 2008 will result in the Company having to seek an additional extension of this facility, or to obtain an alternative source of financing. Management of the Company believes that alternative financing sources on terms acceptable would be available, however there is no guarantee that such will occur.

In March 2004, the Nussdorfs provided a \$5,000,000 subordinated secured demand loan to Perfumania. The demand loan required quarterly interest payments at the prime rate plus 1%. There were no prepayment penalties and the loan was subordinate to all bank related indebtedness. On December 9, 2004, the Company issued a Subordinated Convertible Note (the Convertible Note) in exchange for the \$5,000,000 subordinated secured demand loan. The Convertible Note bears interest at the prime rate plus 1%, requires quarterly interest payments and is secured by a security interest in the Company s assets pursuant to a Security Agreement, by and among the Company and the Nussdorfs. There are no prepayment penalties and the Convertible Note is subordinate to all bank related indebtedness. The Convertible Note was originally payable in January 2007 however it was modified in January 2006 to extend the due date to January 2009. The Note allows the Nussdorfs to convert the Convertible Note into shares of the Company s common stock at a conversion price of \$11.25, which equaled the closing market price of the Company s common stock on the date of the exchange.

NOTE 8 IMPAIRMENT OF ASSETS

The Company recognized non-cash impairment charges relating to its retail locations that were either closed or identified for closure of approximately \$0.3 million, \$0.2 million and \$0.2 million during fiscal years 2007, 2006 and 2005, respectively. These charges were determined based on the difference between the carrying amounts of the assets, representing primarily fixtures and leasehold improvements, at particular store locations and the fair values of the assets on a store-by-store basis. The estimated fair values are based on anticipated future cash flows discounted at a rate commensurate with the risk involved. These impairment losses are included in selling, general and administrative expenses in the accompanying consolidated statements of operations.

NOTE 9 INCOME TAXES

The income tax (provision) benefit is comprised of the following amounts:

	February 2, 2008	 CAL YEAR ENDED ruary 3, 2007	Janu	ıary 28, 2006
Current:				
Federal	\$ (44,668)	\$ (90,000)	\$	(200,000)
State	(108,010)	(91,000)		(200,000)
	(152,678)	(181,000)		(400,000)
Deferred:				
Federal	(873,558)	(1,710,240)		7,946,273
State	(101,744)	(199,164)		925,035
Foreign	167,913	846,000		
	(807,389)	(1,063,404)		8,871,308
Income tax				
(provision) benefit	\$ (960,067)	\$ (1,244,404)	\$	8,471,308

F-19

The income tax provision differs from the amount obtained by applying the statutory Federal income tax rate to pretax income as follows:

		FISCAL YEAR ENDED	
	February 2, 2008	February 3, 2007	January 28, 2006
Expense at federal statutory rates	\$ (328,371)	\$ (1,915,582)	\$ (2,307,821)
Non-deductible expenses	(332,938)	(23,156)	(18,000)
Change in valuation allowance		2,730,392	10,554,746
Adjustment to net operating loss		(1,763,853)	
Other	(298,758)	(272,205)	242,383
Income tax (provision) benefit	\$ (960,067)	\$ (1,244,404)	\$ 8,471,308

Net deferred tax assets reflect the tax effect of the following differences between financial statement carrying amounts and tax basis of assets and liabilities as follows:

	FISCAL YEAR ENDED		
	February 2, 2008	February 3, 2007	
Assets:			
Net operating loss & tax credit carryforwards	\$ 359,964	\$ 1,882,860	
Puerto Rican net operating loss carryforwards	1,013,472	845,558	
Capital loss carryforward	1,571,773	1,571,773	
Inventories	1,003,441	946,663	
Property and equipment	5,763,844	5,323,697	
Goodwill	40,459	143,643	
Self insured reserves and other	1,061,194	907,342	
Total deferred tax assets	10,814,147	11,621,536	
Valuation allowance	(1,571,773)	(1,571,773)	
Net deferred tax assets	\$ 9,242,374	\$ 10,049,763	

In evaluating the reasonableness of the valuation allowance, management assesses whether it is more likely than not that some portion, or all, of the deferred tax assets will not be realized. Ultimately, the realization of deferred tax assets is dependant upon the generation of future taxable income during those periods in which temporary differences become deductible and/or credits can be utilized. To this end, management considers the level of historical taxable income, the scheduled reversal of deferred tax assets and projected future taxable income. Based on these considerations, and the carryforward availability of the deferred tax assets, management believes it is more likely than not that the Company will realize the benefit of the deferred tax assets, except for capital loss carryfowards. The Company s United States and Puerto Rican net operating loss carryforwards begin to expire in 2019 and 2009, respectively.

The Company previously recorded a full valuation allowance related to the deferred tax assets of its Puerto Rican subsidiary, as realization of these assets was not more likely than not. In fiscal year 2006, management determined that it is more likely than not that these assets will be realized. As a result, the reversal of the valuation allowance on these deferred tax assets of approximately \$2.7 million was reflected as a benefit in the tax provision for fiscal year 2006. In addition, in fiscal year 2006, the Company recorded an increase to the net operating losses of its Puerto Rican subsidiary of approximately \$1.8 million as a result of a Puerto Rico tax audit that was finalized in 2007.

Prior to fiscal year 2005, the Company previously recorded a full valuation allowance related to its deferred tax assets as realization of these assets was not more likely than not. In fiscal year 2005, management determined that it was more likely than not that certain of the Company s deferred tax assets, primarily related to its United

Table of Contents 192

F-20

Table of Contents

States operations, would be realized. As a result, the reversal of the valuation allowance of approximately \$9.7 million was reflected as a benefit in the tax provision for fiscal year 2005. In addition, the reversal of approximately \$2.2 million of valuation allowance related to the exercise of stock options was recorded in the consolidated statement of changes in shareholders—equity in fiscal year 2005.

As a result of the implementation of FIN 48 during the first quarter of fiscal year 2007, the Company did not recognize a liability for unrecognized tax benefits or reverse any accruals for uncertain tax positions previously recorded, and accordingly, was not required to record any cumulative effect adjustment to beginning of year retained earnings. As of both the date of adoption and during fiscal year 2007, there was no liability for income tax associated with unrecognized tax benefits.

NOTE 10 SHAREHOLDERS EQUITY

PREFERRED STOCK

The Company s Articles of Incorporation authorize the issuance of up to 1,000,000 shares of preferred stock. The preferred stock may be issued from time to time at the discretion of the Board of Directors without shareholders approval. The Board of Directors is authorized to issue these shares in different series and, with respect to each series, to determine the dividend rate, and provisions regarding redemption, conversion, liquidation preference and other rights and privileges. As of February 2, 2008, no preferred stock had been issued.

TREASURY STOCK

From time to time, the Company s Board of Directors has approved the repurchase of the Company s common stock. As of February 2, 2008, the Company had repurchased approximately 898,000 shares of common stock for approximately \$8.6 million, all of which are held as treasury shares. There were no repurchases during fiscal years 2007, 2006 or 2005.

STOCK OPTION PLANS

The Company currently has two plans which provide for equity-based awards to its employees and directors. Pursuant to the 2000 Stock Option Plan (the Stock Option Plan) and the 2000 Directors Stock Option Plan (the Directors Plan) (collectively, the Plans), 375,000 shares and 30,000 shares of common stock, respectively, were initially reserved for issuance upon exercise of options under the Plans. Additionally, the number of shares available under the Stock Option Plan automatically increases each year by an amount equal to 3% of the shares of common stock of the Company outstanding at the end of the immediate preceding year. The Company s Board of Directors, or a committee thereof, administers and interprets the Stock Option Plan. The Stock Option Plan provides for the granting of both incentive stock options (as defined in Section 422A of the Internal Revenue Code) and non-statutory stock options. Options can be granted under the Stock Option Plan on such terms and at such prices as determined by the Board, except that the per share exercise price of options will not be less than the fair market value of the common stock on the date of grant. Only non-employee directors are eligible to receive options under the Directors Plan. The Directors Plan provides for an automatic grant of an option to purchase 500 shares of common stock upon election as a director of the Company and an automatic grant of 1,000 shares of common stock upon such person s re-election as a director of the Company, in both instances, at an exercise price equal to the fair value of the common stock on the date of grant.

F-21

Options granted under the Stock Option Plan are exercisable after the period or periods specified in the option agreement, and options granted under the Directors Plan are exercisable immediately. Options granted under the Plans are not exercisable after the expiration of 10 years from the date of grant. A summary of the Company s option activity, and related information for the fiscal year ended February 2, 2008 is as follows:

	2(Shares	007 Weighted Average Exercise Price
Outstanding at beginning of year	149,026	\$ 12.39
Granted	5,296	24.05
Exercised	(6,626)	6.50
Cancelled		3.52
Outstanding at end of year	147,696	\$ 12.80
Options exercisable at end of year	147,696	\$ 12.80

The aggregate intrinsic value of options exercised during fiscal years 2007, 2006 and 2005 was approximately \$86,000, \$909,000 and \$216,000, respectively. The aggregate intrinsic value for fully vested options as of February 2, 2008 was approximately \$1,732,000.

The following table summarizes information about stock options outstanding at February 2, 2008:

		OPTIONS OUTSTANDING Weighted		OPTIONS EXER	RCISABLE
RANGE OF EXERCISE PRICES	NUMBER OUTSTANDING	Weighted Average Exercise Price	Average Remaining Contractual Life in years	NUMBER EXERCISABLE	Weighted Average Remaining Exercise Price
\$2.00 - \$12.52	35,690	\$ 10.01	2.19	35,690	\$ 10.01
\$12.99 - \$12.99	100,000	12.99	7.40	100,000	12.99
\$13.85 - \$22.49	9,006	18.03	6.92	9,006	18.03
\$24.05 - \$24.05	3,000	24.05	9.99	3,000	24.05
	147,696	\$ 12.80	6.17	147,696	\$ 12.80

NOTE 11 EMPLOYEE BENEFIT PLANS

The Company has a 401(k) Savings and Investment Plan (the Plan). Pursuant to such Plan, participants may make contributions to the Plan up to a maximum of 20% of total compensation or the maximum limits allowable under the Internal Revenue Code, whichever is less, and the Company, at its discretion, may match such contributions to the extent of 25% of the first 6% of a participant s contribution. The Company s matching contributions vest over a 4-year period. In addition to matching contributions, the Company may make additional contributions on a discretionary basis in order to comply with certain Internal Revenue Code regulations prohibiting discrimination in favor of highly compensated employees. The Company did not match contributions during fiscal years 2007, 2006 and 2005.

NOTE 12 COMMITMENTS AND CONTINGENCIES

The Company is self-insured for employee medical benefits under the Company s group health plan. The Company maintains stop loss coverage for individual medical claims in excess of \$90,000 and for annual Company medical claims which exceed approximately \$3.2 million in the aggregate. While the ultimate amount of claims incurred are dependent on future developments, in management s opinion, recorded accruals are adequate to cover the future payment of claims incurred as of February 2, 2008. However, it is possible that

F-22

recorded accruals may not be adequate to cover the future payment of claims. Adjustments, if any, to estimates recorded resulting from ultimate claim payments will be reflected in operations in the periods in which such adjustments are determined. The self-insurance accrual at February 2, 2008 and February 3, 2007 was approximately \$219,000 and \$271,000, respectively, which is included in accrued expenses and other liabilities in the accompanying consolidated balance sheets.

The Company leases space for its retail stores. The lease terms vary from month to month leases to ten year leases, in some cases with options to renew for longer periods. Various leases contain clauses, which adjust the base rental rate by the prevailing Consumer Price Index, as well as requiring additional contingent rent based on a percentage of gross sales in excess of a specified amount.

Rent expense is as follows:

		FISCAL YEAR	
	2007	2006	2005
Minimum rentals	\$ 19,255,212	\$ 16,812,743	\$ 14,800,858
Contingent rentals	1,493,294	1,384,501	1,449,216
Total	\$ 20,748,506	\$ 18,197,244	\$ 16,250,074

Future minimum lease commitments under non-cancelable operating leases at February 2, 2008 (excluding future payments for maintenance, insurance and taxes to which the Company is obligated) are as follows:

FISCAL YEAR	
2008	\$ 19,129,081
2009	17,271,125
2010	15,884,482
2011	14,002,190
2012	11,170,328
Thereafter	32,914,134
Total future minimum lease payments	\$ 110,371,340

The Company s capitalized leases consist of a corporate office and distribution facility in Sunrise, Florida, as well as computer hardware and software. The lease for the corporate office and distribution facility is for approximately 14 years with monthly rent ranging from approximately \$81,000 to \$104,000 during the term of the lease. The lease terms for the computer hardware and software vary from one to three years. The following is a schedule of future minimum lease payments under capital leases together with the present value of the net minimum lease payments, at February 2, 2008:

FISCAL YEAR	
2008	\$ 1,209,850
2009	1,231,779
2010	1,239,766
2011	1,326,630
2012	1,326,631
Thereafter	6,522,606
Total future minimum lease payments	12,857,262
Less: Amount representing interest	(5,311,618)
Present value of minimum lease payments	7,545,644

Less: Current portion (355,376)

\$ 7,190,268

F-23

Table of Contents

Depreciation expense relating to capital leases is included in depreciation and amortization expense in the accompanying consolidated statements of operations and totaled approximately \$655,000, \$676,000 and \$687,000 in fiscal years 2007, 2006 and 2005, respectively.

The Company is involved in various legal proceedings in the ordinary course of business. Management cannot presently predict the outcome of these matters, although management believes that the ultimate resolution of these matters will not have a materially adverse effect on the Company s financial position, results of operations or cash flows.

NOTE 13 SEGMENT INFORMATION

Segment information is prepared on the same basis that the Company s chief operating decision maker reviews financial information. The Company operates in two industry segments, specialty retail sales and wholesale distribution of fragrances and related products. Retail sales include sales through our Internet site, perfumania.com. Substantially all wholesale sales are to Model and its subsidiaries. The accounting policies of the segments are the same as those described in the summary of significant accounting policies in Note 2. All of the Company s assets relate to and are owned by the Company s retail segment and the Company does not allocate operating and other expenses to its segments. Accordingly, a reconciliation of segment assets to total assets is not presented. During fiscal years 2007, 2006 and 2005, there were no intersegment revenues. Financial information for these segments is summarized in the following table:

	2007	FISCAL YEAR 2006	2005
Net sales:	2007	2000	2002
Retail	\$ 244,019,763	\$ 229,783,211	\$ 215,841,101
Wholesale	57,814,867	13,825,887	17,852,980
	\$ 301,834,630	\$ 243,609,098	\$ 233,694,081
Gross profit:			
Retail	\$ 108,614,493	\$ 102,975,498	\$ 95,353,919
Wholesale	3,250,669	954,123	1,147,240
	\$ 111,865,162	\$ 103,929,621	\$ 96,501,159

See Note 6 for disclosure of sales to significant customers.

F-24

2007 QUARTER

NOTE 14 QUARTERLY FINANCIAL DATA (UNAUDITED)

Net income (loss) per diluted share (as restated, see Note 3)

Unaudited summarized financial results for fiscal years 2007 and 2006 follows (in thousands, except for per share data) and have been restated to reflect the adjustments discussed in Note 3 to the consolidated financial statements. With the exception of the fourth quarter of 2006 which is comprised of fourteen weeks, all quarters listed below are comprised of thirteen weeks:

FIRST

(0.44)

SECOND

(0.21)

THIRD

(0.46)

FOURTH

2.30

Net sales	\$ 48,139	\$ 73,391	\$ 69,884	\$ 110,421
Gross profit	20,323	24,996	24,329	42,217
Net income (loss) as previously reported	(1,979)	(321)	(1,117)	N/A
Net income (loss) (as restated, see Note 3)	(2,071)	(413)	(1,307)	3,797
Net income (loss) per basic share as previously reported	(0.65)	(0.10)	(0.37)	N/A
Net income (loss) per basic share (as restated, see Note 3)	(0.68)	(0.13)	(0.43)	1.24
Net income (loss) per diluted share as previously reported	(0.65)	(0.10)	(0.37)	N/A
Net income (loss) per diluted share (as restated, see Note 3)	(0.68)	(0.13)	(0.43)	1.09
2006 QUARTER	FIRST	SECOND	THIRD	FOURTH
2006 QUARTER Net sales	FIRST \$ 46,069	SECOND \$ 50,050	THIRD \$ 53,904	FOURTH \$ 93,586
· · ·				
Net sales	\$ 46,069	\$ 50,050	\$ 53,904	\$ 93,586
Net sales Gross profit	\$ 46,069 19,159	\$ 50,050 22,010	\$ 53,904 21,835	\$ 93,586 40,926
Net sales Gross profit Net income (loss) as previously reported	\$ 46,069 19,159 (1,295)	\$ 50,050 22,010 (596)	\$ 53,904 21,835 (1,338)	\$ 93,586 40,926 7,751
Net sales Gross profit Net income (loss) as previously reported Net income (loss) (as restated, see Note 3)	\$ 46,069 19,159 (1,295) (1,313)	\$ 50,050 22,010 (596) (622)	\$ 53,904 21,835 (1,338) (1,387)	\$ 93,586 40,926 7,751 7,712

The Company realizes higher sales, gross profit and net income in the fourth fiscal quarter than the other three fiscal quarters due to increased purchases of fragrances as gift items during the holiday season.

E COM VENTURES, INC. AND SUBSIDIARIES

CONDENSED CONSOLIDATED BALANCE SHEETS

(Unaudited)

	N	1ay 3, 2008	Fe	bruary 2, 2008
ASSETS:				
Current assets:				
Cash	\$	1,279,841	\$	1,035,073
Trade receivables, no allowance required		1,450,620		1,057,499
Deferred tax asset: current		4,040,315		2,261,856
Inventories, net	1	18,856,716		107,479,019
Prepaid expenses and other current assets		4,486,635		2,228,013
Total current assets	1	30,114,127		114,061,460
Property and equipment, net		38,475,547		36,587,935
Goodwill		1,904,448		1,904,448
Deferred tax asset: non-current		6,980,518		6,980,518
Other assets, net		291,857		300,250
Total assets	\$ 1	77,766,497	\$	159,834,611
	Ψ.	77,700,157	Ψ	10,000 1,011
LIABILITIES AND SHAREHOLDERS EQUITY:				
Current liabilities:				
	¢	22,898,921	\$	19,609,065
Accounts payable-non affiliates			Ф	, ,
Accounts payable-affiliates		55,891,254		48,650,294
Accrued expenses and other liabilities		10,974,048		10,327,794
Bank line of credit		42,589,627		32,840,872
Subordinated convertible note payable affiliate		5,000,000		5,000,000
Current portion of obligations under capital leases		363,655		355,376
Total current liabilities	1	37,717,505		116,783,401
T (((11' () 1) () 1		7,000,750		7 100 260
Long-term portion of obligations under capital leases		7,089,759		7,190,268
Total liabilities	1	44,807,264		123,973,669
Commitments and contingencies (see Note 7)				
Shareholders equity:				
Preferred stock, \$.10 par value, 1,000,000 shares authorized, none issued				
Common stock, \$.01 par value, 6,250,000 shares authorized; 3,957,290 shares issued in 2008 and		20.572		20.572
2007		39,573		39,573
Additional paid-in capital		79,182,694		79,182,694
Accumulated deficit	((37,686,090)		(34,784,381)
Treasury stock, at cost, 898,249 shares		(8,576,944)		(8,576,944)
Total shareholders equity		32,959,233		35,860,942
Total liabilities and shareholders equity	\$ 1	77,766,497	\$	159,834,611
1 7		, , ,		, ,-,-

See accompanying notes to condensed consolidated financial statements.

E COM VENTURES, INC. AND SUBSIDIARIES

CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS

(Unaudited)

Net sales to:	Thirteen Week Ended May 3, 2008		Thirteen Weeks Ended May 5, 2007 restated, see Note 2)
Unrelated customers	\$ 46,710,41	\$	42,497,127
Affiliates	9,819,05		5,641,425
Aimates	9,019,03		3,041,423
	56,529,468	3	48,138,552
Cost of goods sold to:			
Unrelated customers	24,843,732		22,614,268
Affiliates	9,178,033	3	5,201,615
	34,021,765		27,815,883
	34,021,70.	,	27,015,005
	22 507 70		20.222.660
Gross profit	22,507,703	3	20,322,669
Operating expenses:			
Selling, general and administrative expenses	24,540,920	5	21,128,301
Depreciation and amortization	1,775,98	7	1,446,701
Total operating expenses	26,316,913	3	22,575,002
Total operating expenses	20,510,51	,	22,373,002
Loss from operations	(3,809,210))	(2,252,333)
Interest expense			
Affiliates	(83,438	3)	(116,910)
Other	(787,520		(970,375)
Other	(707,320	,,	(770,373)
	(870,958	3)	(1,087,285)
Loss before income taxes	(4,680,168	2)	(3,339,618)
Income tax benefit	• • • • • • • • • • • • • • • • • • • •		
income tax benefit	1,778,459)	1,269,000
Net loss	\$ (2,901,709	9) \$	(2,070,618)
Net loss per common share:			
Basic	\$ (0.93	5) \$	(0.68)
	•		, ,
Diluted	\$ (0.95)	5) \$	(0.68)
Diluted	\$ (0.2.	<i>))</i>	(0.00)
Weighted average number of common shares outstanding:			
Basic	3,059,04	l	3,058,170
Diluted	3,059,04	l	3,058,170

See accompanying notes to condensed consolidated financial statements.

F-27

E COM VENTURES, INC. AND SUBSIDIARIES

CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS

(Unaudited)

Net loss \$ (2,901,709) \$ (2,070,6 Adjustments to reconcile net loss to net cash used in operating activities: Deferred income taxes Provision for impairment of assets and store closings 24,849 22,2 Depreciation and amortization 1,775,987 1,446,2 Change in operating assets and liabilities: Trade receivables (393,121) (309,4 Inventories (11,377,697) (17,555,2 Prepaid expenses and other assets (2,250,398) (205,7 Accounts payable-ann affiliates 2,345,747 9,910,2 Accounts payable-ann affiliates 7,240,960 3,111,2 Accrued expenses and other liabilities 925,546 648,6 648,6 Net cash used in operating activities (6,388,295) (6,271,2 Cash flows from investing activities (3,023,462) (2,563,2 Net cash used in investing activities (3,023,462) (2,563,2 Cash flows from financing activities 9,748,755 9,447,2 Net borrowings under bank line of credit 9,748,755 9,447,2 Principal payments under capital lease obligations (92,230) (8		Thirteen Weeks Ended May 3, 2008	hirteen Weeks Ended May 5, 2007 stated, see Note 2)
Adjustments to reconcile net loss to net cash used in operating activities: Deferred income taxes 1,778,459 1,269,469	Cash flows from operating activities:		
Deferred income taxes (1,778,459) (1,269,49) (2,284) 22,2849 22,2849 22,2849 22,2849 22,2849 22,2849 22,2849 22,2849 22,284,2849 22,284,2849 22,284,2849 22,284,2849 22,284,2849 22,284,2849 22,284,2849 23,284,2849 23,284,2849 23,284,2849 23,284,2849 23,284,2849 23,284,2849 23,284,2849 23,284,2849 23,284,2849 23,284,2849 23,284,2849 23,284,2849 23,284,2849 23,284,2849 23,284,2849 23,284,2849 23,284,2849 23,284,2849 23,284,2849 24,284,2849 24,284,2849 23,284,2849 24,284,2849	Net loss	\$ (2,901,709)	\$ (2,070,618)
Deferred income taxes (1,778,459) (1,269,49) 22,2 Provision for impairment of assets and store closings 24,849 22,2 Depreciation and amortization 1,775,987 1,446,7 Change in operating assets and liabilities:	Adjustments to reconcile net loss to net cash used in operating activities:		
Provision for impairment of assets and store closings 24,849 22,2 Depreciation and amortization 1,775,987 1,446,6 Change in operating assets and liabilities: (393,121) (309,4 Inventories (11,377,697) (17,555,6 Prepaid expenses and other assets (2,250,398) (205,7 Accounts payable-non affiliates 2,345,747 9,910,7 Accounts payable-affiliates 7,240,960 3,111,7 Accrued expenses and other liabilities 925,546 648,6 Net cash used in operating activities (6,388,295) (6,271, Cash flows from investing activities: 3,023,462 (2,563,205,205,205,205,205,205,205,205,205,205	· ·	(1.778.459)	(1,269,000)
Depreciation and amortization 1,775,987 1,446, Change in operating assets and liabilities: Trade receivables (393,121) (309,417) Inventories (11,377,697) (17,555,27) Prepaid expenses and other assets (2,250,398) (205,74,205,306) Accounts payable-ann affiliates 2,345,747 9,910,311,34,327 Accounts payable-affiliates 7,240,960 3,111,31,31,32,33,33,33,33,33,33,33,33,33,33,33,33,			22,500
Trade receivables (393,121) (309, Inventories (11,377,697) (17,555, 17,5097) (17,555, 17,5097) (17,555, 17,5097) (17,555, 17,5097) (2,250,398) (205, 17,5097) (205, 17,5097) (205, 17,5097) (205, 17,5097) (205, 17,5097) (200, 17,5097) <td< td=""><td></td><td></td><td>1,446,701</td></td<>			1,446,701
Inventories (11,377,697) (17,555,50) Prepaid expenses and other assets (2,250,398) (205,750) Accounts payable-non affiliates 2,345,747 9,910, Accounts payable-affiliates 7,240,960 3,111, Accrued expenses and other liabilities 925,546 648,0 Net cash used in operating activities (6,388,295) (6,271,10) Cash flows from investing activities: 3,023,462) (2,563,20) Net cash used in investing activities 3,023,462) (2,563,20) Cash flows from financing activities: 3,023,462) (2,563,20) Net borrowings under bank line of credit 9,748,755 9,447,20 Principal payments under capital lease obligations (92,230) (83,10) Proceeds from exercise of stock options 35, 35, Net cash provided by financing activities 9,656,525 9,399,0 Increase in cash and cash equivalents 244,768 564,7 Cash and cash equivalents at beginning of period 1,035,073 1,282,3 Cash and cash equivalents at end of period 1,035,073 1,282,3 Cash p	Change in operating assets and liabilities:		
Prepaid expenses and other assets (2,250,398) (205,5 Accounts payable-non affiliates 2,345,747 9,910,0 3,111,4 Accounts payable-affiliates 7,240,960 3,111,4 Accounts payable-affiliates 3,111,4 Accounts payable-affiliates 648,0 656,3 656,3 656,3 656,3 656,2 9,399,3 656,525 9,399,3 683,0 656,525 9,399,3 656,525 9,399,3 656,525 9,399,3 656,4 656,4 656,4 <t< td=""><td>Trade receivables</td><td>(393,121)</td><td>(309,471)</td></t<>	Trade receivables	(393,121)	(309,471)
Accounts payable-non affiliates 2,345,747 9,910, Accounts payable-affiliates 7,240,960 3,111, Accrued expenses and other liabilities 925,546 648,6 Net cash used in operating activities (6,388,295) (6,271,100,000) (2,563,100,000) (2,5	Inventories	(11,377,697)	(17,555,728)
Accounts payable-non affiliates 2,345,747 9,910, Accounts payable-affiliates 7,240,960 3,111, Accrued expenses and other liabilities 925,546 648,6 Net cash used in operating activities (6,388,295) (6,271,100,000) (2,563,100,000) (2,5	Prepaid expenses and other assets	(2,250,398)	(205,793)
Accrued expenses and other liabilities 925,546 648,6 Net cash used in operating activities (6,388,295) (6,271,1) Cash flows from investing activities: 3,023,462) (2,563,203,462) Net cash used in investing activities (3,023,462) (2,563,203,462) Cash flows from financing activities: 3,023,462) (2,563,203,462) Net borrowings under bank line of credit 9,748,755 9,447,2 Principal payments under capital lease obligations (92,230) (83,023,462) Proceeds from exercise of stock options 35, Net cash provided by financing activities 9,656,525 9,399,2 Increase in cash and cash equivalents 244,768 564,7 Cash and cash equivalents at beginning of period 1,035,073 1,282,3 Cash and cash equivalents at end of period \$1,279,841 \$1,847,3 Supplemental Information: Cash paid during the period for: Interest \$863,463 \$943,4		2,345,747	9,910,139
Net cash used in operating activities (6,388,295) (6,271,171) Cash flows from investing activities: 3,023,462) (2,563,371) Additions to property and equipment (3,023,462) (2,563,372) Net cash used in investing activities: 3,023,462) (2,563,372) Cash flows from financing activities: 9,748,755 9,447,755 Net borrowings under bank line of credit 9,748,755 9,447,755 Principal payments under capital lease obligations (92,230) (83,363) Proceeds from exercise of stock options 35, Net cash provided by financing activities 9,656,525 9,399,30 Increase in cash and cash equivalents 244,768 564,76 Cash and cash equivalents at beginning of period 1,035,073 1,282,50 Cash and cash equivalents at end of period \$1,279,841 \$1,847,30 Supplemental Information: Cash paid during the period for: Interest \$863,463 \$943,40	Accounts payable-affiliates	7,240,960	3,111,447
Cash flows from investing activities: Additions to property and equipment (3,023,462) (2,563,3023,462) (2,563,3023,462) Cash used in investing activities Cash flows from financing activities: Net borrowings under bank line of credit Principal payments under capital lease obligations Proceeds from exercise of stock options Net cash provided by financing activities Net cash provided by financing activities 9,656,525 9,399,302 Increase in cash and cash equivalents 244,768 564,7 Cash and cash equivalents at beginning of period 1,035,073 1,282,303 Supplemental Information: Cash paid during the period for: Interest 863,463 943,4		925,546	648,633
Additions to property and equipment (3,023,462) (2,563,50). Net cash used in investing activities (3,023,462) (2,563,50). Cash flows from financing activities: Net borrowings under bank line of credit 9,748,755 9,447,50. Principal payments under capital lease obligations (92,230) (83,60). Proceeds from exercise of stock options 35,50. Net cash provided by financing activities 9,656,525 9,399,50. Increase in cash and cash equivalents 244,768 564,50. Cash and cash equivalents at beginning of period 11,035,073 1,282,50. Cash and cash equivalents at end of period \$1,279,841 \$1,847,50. Supplemental Information: Cash paid during the period for: Interest \$863,463 \$943,60.	Net cash used in operating activities	(6,388,295)	(6,271,190)
Additions to property and equipment (3,023,462) (2,563,50). Net cash used in investing activities (3,023,462) (2,563,50). Cash flows from financing activities: Net borrowings under bank line of credit 9,748,755 9,447,50. Principal payments under capital lease obligations (92,230) (83,60). Proceeds from exercise of stock options 35,50. Net cash provided by financing activities 9,656,525 9,399,50. Increase in cash and cash equivalents 244,768 564,50. Cash and cash equivalents at beginning of period 11,035,073 1,282,50. Cash and cash equivalents at end of period \$1,279,841 \$1,847,50. Supplemental Information: Cash paid during the period for: Interest \$863,463 \$943,60.	Cash flows from investing activities:		
Cash flows from financing activities: Net borrowings under bank line of credit Principal payments under capital lease obligations Proceeds from exercise of stock options Net cash provided by financing activities Net cash provided by financing activities 9,656,525 9,399,2 Increase in cash and cash equivalents 244,768 564, Cash and cash equivalents at beginning of period 1,035,073 1,282,3 Cash and cash equivalents at end of period \$1,279,841 \$1,847,3 Supplemental Information: Cash paid during the period for: Interest \$863,463 \$943,0	Additions to property and equipment	(3,023,462)	(2,563,336)
Net borrowings under bank line of credit Principal payments under capital lease obligations Proceeds from exercise of stock options Net cash provided by financing activities Net cash provided by financing activities 9,656,525 9,399,2 Increase in cash and cash equivalents Cash and cash equivalents at beginning of period 1,035,073 1,282,3 Cash and cash equivalents at end of period \$1,279,841 \$1,847,3 Supplemental Information: Cash paid during the period for: Interest \$863,463 \$943,6	Net cash used in investing activities	(3,023,462)	(2,563,336)
Net borrowings under bank line of credit Principal payments under capital lease obligations Proceeds from exercise of stock options Net cash provided by financing activities Net cash provided by financing activities 9,656,525 9,399,2 Increase in cash and cash equivalents Cash and cash equivalents at beginning of period 1,035,073 1,282,3 Cash and cash equivalents at end of period \$1,279,841 \$1,847,3 Supplemental Information: Cash paid during the period for: Interest \$863,463 \$943,6	Cash flows from financing activities:		
Principal payments under capital lease obligations (92,230) (83,0) Proceeds from exercise of stock options 35,1 Net cash provided by financing activities 9,656,525 9,399,2 Increase in cash and cash equivalents 244,768 564,7 Cash and cash equivalents at beginning of period 1,035,073 1,282,3 Cash and cash equivalents at end of period \$1,279,841 \$1,847,3 Supplemental Information: Cash paid during the period for: Interest \$863,463 \$943,0		9.748.755	9,447,219
Proceeds from exercise of stock options 35,3 Net cash provided by financing activities 9,656,525 9,399,2 Increase in cash and cash equivalents Cash and cash equivalents at beginning of period 1,035,073 1,282,3 Cash and cash equivalents at end of period \$1,279,841 \$1,847,3 Supplemental Information: Cash paid during the period for: Interest \$863,463 \$943,6			(83,035)
Increase in cash and cash equivalents Cash and cash equivalents at beginning of period 1,035,073 1,282,5 Cash and cash equivalents at end of period \$1,279,841 \$1,847,5 Supplemental Information: Cash paid during the period for: Interest \$863,463 \$943,6		(-,,	35,100
Cash and cash equivalents at beginning of period 1,035,073 1,282,5 Cash and cash equivalents at end of period \$1,279,841 \$1,847,3 Supplemental Information: Cash paid during the period for: Interest \$863,463 \$943,6	Net cash provided by financing activities	9,656,525	9,399,284
Cash and cash equivalents at beginning of period 1,035,073 1,282,5 Cash and cash equivalents at end of period \$1,279,841 \$1,847,3 Supplemental Information: Cash paid during the period for: Interest \$863,463 \$943,6	Increase in cash and cash equivalents	244.768	564,758
Supplemental Information: Cash paid during the period for: Interest \$ 863,463 \$ 943,000 \$ 943,000 \$ 100,			1,282,546
Cash paid during the period for: Interest \$ 863,463 \$ 943,60	Cash and cash equivalents at end of period	\$ 1,279,841	\$ 1,847,304
Cash paid during the period for: Interest \$ 863,463 \$ 943,60	Supplemental Information:		
Interest \$ 863,463 \$ 943,0			
		\$ 863,463	\$ 943,047
Income taxes 40,000	Income taxes	40,000	, .
Accounts payable for property and equipment 664,816 297,4 See accompanying notes to condensed consolidated financial statements.			297,487

E COM VENTURES, INC. AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

(Unaudited)

NOTE 1 OPERATIONS AND BASIS OF PRESENTATION

E Com Ventures, Inc., a Florida corporation (ECOMV), performs all of its operations through two wholly-owned subsidiaries, Perfumania, Inc. (Perfumania), a Florida corporation, which is a specialty retailer and wholesaler of fragrances and related products, and perfumania.com, Inc., (perfumania.com), a Florida corporation, which is an Internet retailer of fragrances and other specialty items.

Perfumania is a leading specialty retailer and wholesale distributor of a wide range of brand name and designer fragrances. As of May 3, 2008, Perfumania operated a chain of 313 retail stores specializing in the sale of fragrances and related products at discounted prices up to 75% below the manufacturers—suggested retail prices. Perfumania—s wholesale division distributes fragrances and related products primarily to an affiliate. Perfumania.com offers a selection of our more popular products for sale over the Internet and serves as an alternative shopping experience to the Perfumania retail stores.

The condensed consolidated financial statements include the accounts of ECOMV and subsidiaries (collectively, the Company). All intercompany balances and transactions have been eliminated in consolidation.

The accompanying unaudited condensed consolidated financial statements have been prepared in accordance with the rules and regulations of the Securities and Exchange Commission (the SEC). Certain information and note disclosures normally included in annual financial statements, prepared in accordance with accounting principles generally accepted in the United States of America (GAAP), have been condensed or omitted pursuant to those rules and regulations. The financial information presented herein, which is not necessarily indicative of results to be expected for the current fiscal year, reflects all adjustments which, in the opinion of management, are necessary for a fair presentation of the interim unaudited condensed consolidated financial statements. It is suggested that these condensed consolidated financial statements be read in conjunction with the financial statements and the notes thereto included in our Annual Report on Form 10-K for the fiscal year ended February 2, 2008.

NOTE 2 RESTATEMENT

Subsequent to the issuance of the Company s 2006 financial statements, management identified errors in the accounting for operating leases and the period of amortization of leasehold improvements.

Under GAAP, rent expense is amortized on a straight-line basis over the term of the lease. In prior periods, the Company had determined that the term of the lease begins on the commencement date of the lease, which generally coincides with the store opening date. Management re-evaluated FASB Technical Bulletin No. 85-3, Accounting for Operating Leases with Scheduled Rent Increases and other guidance provided by the SEC and determined that the lease term for amortization purposes should commence on the date the Company takes physical possession of the leased space to commence construction of leasehold improvements, which is generally two months prior to a store opening date. The Company has restated its previously reported financial statements in connection with its accounting for rent expense. Rent expense is included in selling, general and administrative expenses on the Company s condensed consolidated statements of operations.

Management has also reviewed its leasehold improvements to ensure amortization over the shorter of their economic lives or their lease term. In prior periods, in determining the lease terms, in addition to the initial term, the Company had included one anticipated renewal term even if the lease did not have a stated renewal option. The Company has adjusted its lease terms for purposes of calculating amortization of leasehold improvements to

include a renewal term only if that renewal option is specified in the lease agreement and that exiting the lease after the initial lease term would result in economic loss and therefore normally management anticipates exercising the respective renewal option.

The following is a summary of the effects of the restatement to the Company s Condensed Consolidated Financial Statements.

	Condensed Co As previously	Condensed Consolidated Statements of Operations As previously				
	reported	Adjustments	As restated			
Thirteen weeks ended May 5, 2007						
Selling, general and administrative expenses	\$ 21,094,745	\$ 33,556	\$ 21,128,301			
Depreciation and amortization	1,332,701	114,000	1,446,701			
Total operating expenses	22,427,446	147,556	22,575,002			
Loss from operations	(2,104,777)	(147,556)	(2,252,333)			
Loss before income taxes	(3,192,062)	(147,556)	(3,339,618)			
Income tax benefit	1,213,000	56,000	1,269,000			
Net loss	(1,979,062)	(91,556)	(2,070,618)			
Basic net loss per common share	\$ (0.65)	\$ (0.03)	\$ (0.68)			
Diluted net loss per common share	\$ (0.65)	\$ (0.03)	\$ (0.68)			

	Condensed Consolidated Statements of Cash Flows				
	As previously reported	Adjustments	As restated		
Net loss	\$ (1,979,062)	\$ (91,556)	\$ (2,070,618)		
Deferred income taxes	1,213,000	56,000	1,269,000		
Depreciation and amortization	1,332,701	114,000	1,446,701		
Accrued expenses and other liabilities	615,077	33,556	648,633		

NOTE 3 ACCOUNTING FOR SHARE-BASED PAYMENT

The Company has two stock option plans which provide for equity-based awards to its employees and directors (collectively, the Plans). Under the Plans, the Company has reserved approximately 1,000,000 shares of common stock, of which approximately 590,000 options have been granted and approximately 148,000 options are outstanding. All stock options have an exercise price that is equal to the fair market value of the Company s stock on the date the options were granted. The term of the stock option awards is ten years from the date of grant. All granted and outstanding options were fully vested during the quarter ended May 3, 2008.

The following is a summary of the stock option activity during the thirteen weeks ended May 3, 2008:

	Number of Shares	A	eighted verage cise Price	Weighted Average Remaining Contractual Life	Aggregate Intrinsic Value
Outstanding and exercisable as of February 2, 2008	147,696	\$	12.80	6.17	\$ 1,732,000
Granted					
Exercised					
Forfeited					
Outstanding and exercisable as of May 5, 2008	147,696	\$	12.80	5.92	\$ 1,132,000

The aggregate intrinsic value in the table above is the amount before applicable income taxes which would have been received by the optionees based on the Company s closing stock price as of the last business day of the respective period had all options been exercised on that date.

We compute share based compensation pursuant to Statement of Financial Accounting Standards (SFAS) No. 123(R), Share-Based Payment (SFAS No. 123 (R)) . During the thirteen weeks ended May 3, 2008 and May 5, 2007, the Company did not recognize any share based compensation expense in the condensed consolidated financial statements since no stock options were granted nor were there any modifications of outstanding stock options during such periods.

NOTE 4 BANK LINE OF CREDIT AND NOTES PAYABLE

The bank line of credit and notes payable, all of which is classified as current, consist of the following:

	May 3, 2008	February 2, 2008
Bank line of credit interest payable monthly, secured by a pledge of substantially all of		
Perfumania s assets	\$ 42,589,627	\$ 32,840,872
Subordinated convertible note payable-affiliate	\$ 5,000,000	\$ 5,000,000

Perfumania s senior secured credit facility provides for borrowings of up to \$60 million, depending on the Company s levels of eligible inventories. Advances under the line of credit are based on a formula of eligible inventories and bear interest at a floating rate ranging from (a) prime to prime plus 1.25% or (b) LIBOR plus 2.5% to 3.75% depending on a financial ratio test. As of May 3, 2008, \$42.6 million was outstanding under the line of credit and \$12.0 million was available to support normal working capital requirements and other general corporate purposes based upon our eligible borrowing base, net of the restrictions concerning minimum undrawn availability discussed below. Advances are secured by a first lien on all assets of Perfumania. The credit facility contains limitations on additional borrowings, capital expenditures and other items, and contains various covenants including a fixed charge coverage ratio, a leverage ratio and capital expenditure limits as defined. The credit facility also prohibits the Company from paying cash dividends. In June 2007, the minimum undrawn availability as defined was amended so that Perfumania is required to maintain at all times, a minimum undrawn availability of not less than \$5,000,000, and shall maintain a monthly average undrawn availability, as defined, of not less than \$7,500,000. In May 2008, the fixed charge covenant and the covenant concerning the limitation on capital expenditures were amended with an effective date of February 1, 2008. Perfumania was in compliance with all covenant requirements as of May 3, 2008 and believes it will remain in compliance through the extension period discussed below.

The credit facility expired on May 12, 2008. In May 2008, the Company obtained a ninety day extension of the credit facility to August 11, 2008. In May 2008, the Company received a commitment from the lender to extend the credit facility through May 2009 which it did not execute. As discussed in Note 8, the Merger with Model Reorg, Inc. (Model) is subject to a number of terms, one of which includes the requirement that the Company and Model obtain a commitment for asset-based loans and term debt facilities in amounts and on terms that are substantially similar to those contemplated by a previous commitment that was obtained in August 2007. The August 2007 commitment provided for replacement senior credit facilities for us and Model in an aggregate amount of up to \$280 million, to be established upon the closing of the Merger. That commitment expired by its terms in November 2007. A comparable commitment on substantially similar terms and amounts was obtained from the same lender on May 16, 2008 which is available through August 16, 2008.

Failure of the Merger with Model to close prior to the expiration of the credit facility on August 11, 2008 will result in the Company having to seek an additional extension of its facility, or to obtain an alternative source of financing. Management of the Company believes that alternative financing sources on terms acceptable would be available, however there is no guarantee that such will occur.

In March 2004, Glenn and Stephen Nussdorf (the Nussdorfs) provided a \$5,000,000 subordinated secured demand loan to Perfumania. The demand loan required quarterly interest payments at the prime rate plus 1%. There were no prepayment penalties and the loan was subordinate to all bank related indebtedness. On December 9, 2004, the Company issued a Subordinated Convertible Note (the Convertible Note) in exchange

Table of Contents

for the \$5,000,000 subordinated secured demand loan. The Convertible Note bears interest at the prime rate plus 1%, requires quarterly interest payments and is secured by a security interest in the Company s assets pursuant to a Security Agreement, by and among the Company and the Nussdorfs. There are no prepayment penalties and the Convertible Note is subordinate to all bank related indebtedness. The Convertible Note was originally payable in January 2007 however it was modified in January 2006 to extend the due date to January 2009. The Note allows the Nussdorfs to convert the Convertible Note into shares of the Company s common stock at a conversion price of \$11.25, which equaled the closing market price of the Company s common stock on the date of the exchange. See Note 8 for discussion of transactions between the Company and Model.

NOTE 5 ACCOUNTING FOR INCOME TAXES

The Company adopted the provisions of FIN 48, Accounting for Uncertainty in Income Taxes an interpretation of FASB Statement No. 109, (FIN 48) effective February 4, 2007. FIN 48 prescribes a comprehensive model for the financial statement recognition, measurement, presentation and disclosure of uncertain tax positions taken or expected to be taken in an income tax return. Under FIN 48, the Company may recognize the tax benefit from an uncertain tax position only if it is more likely than not that the tax position will be sustained on examination by the taxing authorities, based on the technical merits of the position. The tax benefits recognized in the financial statements from such a position should be measured based on the largest benefit that has a greater than fifty percent likelihood of being realized upon ultimate settlement. FIN 48 also provides guidance on derecognition, classification, interest and penalties on income taxes, accounting in interim periods and requires increased disclosures.

As a result of the implementation of FIN 48, the Company did not recognize a liability for unrecognized tax benefits or reverse any accruals for uncertain tax positions previously recorded, and accordingly, was not required to record any cumulative effect adjustment to beginning of year retained earnings. As of both the date of adoption and May 3, 2008, there was no liability for income tax associated with uncertain tax positions.

The Company accrues interest related to unrecognized tax benefits as well as any related penalties in operating expenses in its condensed consolidated statements of operations, which is consistent with the recognition of these items in prior reporting periods. As of the date of adoption, the Company was not required to have an accrual for the payment of interest and penalties. No accrual for interest and penalties related to uncertain tax positions was required as of May 3, 2008.

The Company operates stores throughout the United States and Puerto Rico, and as a result, files income tax returns in the United States federal jurisdiction and various state and foreign jurisdictions. In the normal course of business, the Company is subject to examination by taxing authorities. We are no longer subject to income tax examinations by the Internal Revenue Service for fiscal years prior to 2003.

State and foreign income tax returns are generally subject to examination for a period of three to five years after filing of the respective return. The state impact of any federal changes remains subject to examination by various states for a period of up to one year after formal notification to the states. The Company is not currently under examination in any state or foreign jurisdictions.

The Company does not anticipate any material adjustments relating to unrecognized tax benefits within the next twelve months, however the ultimate outcome of tax matters is uncertain and unforeseen results can occur.

NOTE 6 BASIC AND DILUTED LOSS PER COMMON SHARE

Basic loss per common share has been computed by dividing net loss by the weighted average number of common shares outstanding during the period. For all periods presented in the accompanying condensed consolidated statements of operations, all incremental shares attributed to outstanding stock options and convertible notes were not included because the results would be anti-dilutive.

F-32

NOTE 7 CONTINGENCIES

The Company is involved in various legal proceedings in the ordinary course of business. Management cannot presently predict the outcome of these matters, although management believes that the ultimate resolution of these matters will not have a materially adverse effect on the Company s consolidated financial position, results of operations or cash flows.

NOTE 8 RELATED PARTY TRANSACTIONS

Glenn and Stephen Nussdorf (the Nussdorfs) own an aggregate 1,113,144 shares or approximately 36% of the total number of shares of the Company s common stock as of May 3, 2008, excluding shares issuable upon conversion of the Convertible Note discussed in Note 4. Stephen Nussdorf has served as the Company s Chairman of the Board since February 2004.

The Nussdorfs are officers and principals of Model and its affiliate, Quality King Distributors, Inc. (Quality King) and their subsidiaries, including Quality King Fragrances, Inc. Model is a diversified wholesale and retail fragrance company and Quality King distributes pharmaceuticals and health and beauty care products. The Company s President and Chief Executive Officer, Michael W. Katz is an executive of Model and Quality King and the Company s principal shareholders, Stephen Nussdorf, the Chairman of the Company s Board of Directors and Glenn Nussdorf, his brother, are shareholders and executives of Model and Quality King.

Sales To and Purchases From Affiliated Companies

The Company sold approximately \$9.8 million and \$5.6 million of wholesale merchandise to Model and its subsidiaries in the thirteen weeks ended May 3, 2008 and May 5, 2007, respectively. See further discussion at Item 2 Management s Discussion and Analysis of Financial Condition and Results of Operations in this Form 10-Q regarding significant increases in wholesale sales volume. For all periods presented, substantially all of the Company s wholesale sales were to Model and its subsidiaries. The wholesale sales to Model and its subsidiaries result from the Company s supplier relationships and its ability to obtain certain merchandise at better prices and in greater quantities than Model and its subsidiaries are able to achieve. These wholesale sales are included in net sales-affiliates in the accompanying condensed consolidated statements of operations. There were approximately \$0.2 million of accounts receivable due from Model and its subsidiaries as of May 3, 2008.

Glenn Nussdorf (Mr. Nussdorf) beneficially owns approximately 12% of the outstanding common stock of Parlux Fragrances, Inc. (Parlux), a manufacturer and distributor of prestige fragrances and beauty products. In February 2007, Mr. Nussdorf reached an agreement with Parlux which called for equal representation in Parlux s Board of Directors by the then current independent directors and Mr. Nussdorf s nominees. Accordingly, Mr. Nussdorf s three nominees were appointed to the Parlux Board, and one of these nominees currently serves as Parlux s Chief Executive Officer.

Purchases of merchandise from Parlux and Model and its subsidiaries aggregated approximately \$10.5 million and \$7.5 million for the thirteen weeks ended May 3, 2008 and May 5, 2007, representing approximately 23.0% and 17.1% of the Company s total inventory purchases, respectively. The amounts due to Parlux at May 3, 2008 and February 2, 2008 were approximately \$15.4 million and \$12.5 million, respectively. The amounts due to Model and its subsidiaries at May 3, 2008 and February 2, 2008 were approximately \$40.5 million and \$36.1 million, respectively. These amounts are non-interest bearing and are included in accounts payable-affiliates in the accompanying condensed consolidated balance sheets. Purchases from related parties are generally payable in 90 days, however, due to the seasonality of the Company s business, these terms are generally extended. Related party accounts are historically brought closer to terms at the end of the holiday season, however, we are dependent upon these extended terms for much of our liquidity during the year.

F-33

Table of Contents

Merger Agreement with Model and Other Arrangements

As reported in the Company s Form 8-K filed with the SEC on December 21, 2007, the Company entered into an Agreement and Plan of Merger (the Merger Agreement) with Model, a New York corporation controlled by the family of Glenn and Stephen Nussdorf, principal shareholders of the Company, on December 21, 2007. Model is a diversified wholesale and retail fragrance company. Stephen Nussdorf is the Chairman of our Board of Directors, and Michael W. Katz, our President and Chief Executive Officer, is an executive of Model. Pursuant to the terms of the Merger Agreement, Model will be merged into a newly formed wholly-owned subsidiary of the Company in exchange for the issuance of 5,900,000 shares of the Company s common stock and warrants to acquire an additional 1,500,000 shares of the Company s common stock at an exercise price of \$23.94. The warrants will have a 10-year term, will not be exercisable for the first three years and will not be transferable, with limited exceptions.

The Merger Agreement was reached after extensive negotiations between Model and a special committee comprised of the independent Board members of the Company, which retained independent counsel and financial advisors. The special committee received an opinion from Financo, Inc. that, as of the date of the Merger Agreement and subject to various assumptions and qualifications set forth therein, the consideration to be paid by the Company in the merger is fair to the shareholders of the Company (other than shareholders of the Company who own or whose affiliates own securities of Model) from a financial point of view.

The Merger Agreement also required Model to refinance debt owed by Model to its affiliate, Quality King. The refinancing is to be through a \$50 million three-year term loan from trusts for the benefit of Glenn and Stephen Nussdorf and their families and, if needed, a transfer of inventory from Model to Quality King to pay any remaining balance. The new loan will be subordinated to the new secured credit facility described below.

The consummation of the contemplated Merger is subject to certain conditions, including approval by the holders of a majority of the votes cast at a meeting of shareholders of the Company and the availability of a new \$280 million secured credit facility to replace the Company s and Model s existing third party credit facilities for which a commitment was obtained in August 2007, but which expired by its terms in November 2007. A comparable commitment with substantially similar terms was obtained from the same lender in May 2008 which is available through August 16, 2008.

Included in selling, general and administrative expenses on the Company s condensed consolidated statement of operations for the thirteen weeks ended May 3, 2008 are \$0.2 million of expenses related to the Merger. These expenses consist primarily of financial advisory, legal and due diligence fees.

Following the contemplated Merger, the Nussdorf s, who currently own 36% of the Company s outstanding common stock, would own approximately 51% of the Company s outstanding common stock. Giving effect to the conversion of our \$5 million subordinated convertible note currently held by them and the exercise of the warrants to be issued to them in the transaction, but not assuming the exercise of any outstanding options held by the Company s officers and directors, they would own approximately 58% of the Company s common stock.

In addition, the Company s Board approved changing the Company s name to Perfumania Holdings, Inc., subject to shareholder approval.

On March 12, 2008, the Company filed a preliminary proxy statement with the SEC discussing the Merger Agreement and the related required shareholder approvals, including the issuance of the additional shares of the Company s common stock pursuant to the Merger Agreement and the related amendments to the Company s articles of incorporation.

On August 2, 2007, the Company entered into an Information Technology Services Agreement (the Services Agreement) with Model and its subsidiaries, whereby among other services, the Company will

F-34

manage and monitor the IT systems of Model in exchange for a monthly service fee of \$25,000. The Services Agreement terminates 30 days after the effective date of the merger discussed above, or if such merger offer is not consummated on or before December 31, 2008, either party may terminate 30 days after providing written notice of termination. During the thirteen weeks ended May 3, 2008, the Company recorded income of \$75,000 related to this Services Agreement which is included as a reduction to selling, general and administrative expenses on the condensed consolidated statement of operations for the thirteen weeks ended May 3, 2008.

NOTE 9 SEGMENT INFORMATION

Segment information is prepared on the same basis that the Company s chief operating decision maker reviews financial information. The Company operates in two industry segments, specialty retail sales and wholesale distribution of fragrances and related products. Retail sales include sales through our internet site, perfumania.com. Substantially all wholesale sales are to Model and its subsidiaries. The accounting policies of the segments are the same as those described in the summary of significant accounting policies in Note 2 of the Notes to our Consolidated Financial Statements included in our 2007 Annual Report on Form 10-K. All of the Company s assets relate to and are owned by the Company s retail segment and the Company does not allocate operating and other expenses to its segments. Accordingly, a reconciliation of segment assets to total assets is not presented. During the thirteen weeks ended May 3, 2008 and May 5, 2007, there were no intersegment revenues. Financial information for these segments is summarized in the following table.

	Thirteen Weeks Ended May 3, 2008		Thirteen Weeks Ended May 5, 2007	
Net sales:				
Retail	\$ 46,710,411	\$	42,488,370	
Wholesale	9,819,057		5,650,182	
	\$ 56,529,468	\$	48,138,552	
Gross profit:				
Retail	\$ 21,866,679	\$	19,881,608	
Wholesale	641,024		441,061	
	\$ 22,507,703	\$	20,322,669	

See Note 8 for disclosure of sales to significant customers.

NOTE 10 RECENT ACCOUNTING PRONOUNCEMENTS

In March 2008, the FASB issued SFAS 161, Disclosures about Derivative Instruments and Hedging Activities an amendment of FASB Statement No. 133 (SFAS 161). SFAS 161 requires enhanced disclosures regarding derivatives and hedging activities, including: (a) the manner in which an entity uses derivative instruments; (b) the manner in which derivative instruments and related hedged items are accounted for under Statement of Financial Accounting Standards No. 133, Accounting for Derivative Instruments and Hedging Activities; and (c) the effect of derivative instruments and related hedged items on an entity is financial position, financial performance, and cash flows. SFAS 161 is effective for financial statements issued for fiscal years and interim periods beginning after November 15, 2008. As SFAS 161 relates specifically to disclosures, it will have no impact on the Company is results of operations, financial position or cash flows.

In December 2007, the FASB issued SFAS 141(R), Business Combinations (SFAS 141(R)). SFAS 141(R) establishes principles and requirements for how an acquirer recognizes and measures in its financial statements the identifiable assets acquired, the liabilities assumed, the goodwill acquired, and any noncontrolling interest in the acquiree. This statement also establishes disclosure requirements to enable the evaluation of the nature and financial effect of the business combination. SFAS 141(R) is effective for fiscal years beginning after

F-35

Table of Contents

December 15, 2008. The Company is in the process of evaluating the effect that the adoption of SFAS 141(R) will have on its results of operations, financial position, and cash flows.

In December 2007, the FASB issued SFAS 160, Noncontrolling Interests in Consolidated Financial Statements an amendment of ARB No. 51. (SFAS 160). SFAS 160 amends ARB 51 to establish accounting and reporting standards for the noncontrolling interest in a subsidiary and for the deconsolidation of a subsidiary. It clarifies that a noncontrolling interest in a subsidiary is an ownership interest in the consolidated entity that should be reported as equity in the consolidated financial statements. SFAS 160 is effective for fiscal years beginning on or after December 15, 2008. The Company is in the process of evaluating the effect that the adoption of SFAS 160 will have on its results of operations, financial position, and cash flows.

In February 2007, the FASB issued SFAS 159, The Fair Value Option for Financial Assets and Financial Liabilities Including an amendment of FASB Statement 115. (SFAS 159). SFAS 159 permits entities to choose to measure eligible items at fair value at specified election dates and report unrealized gains and losses on items for which the fair value option has been elected in earnings at each subsequent reporting date. SFAS 159 is effective for fiscal 2008. The adoption of SFAS 159 did not have any effect on the Company is results of operations, financial position or cash flows.

In September 2006, the FASB issued SFAS 157, Fair Value Measurements (SFAS 157). SFAS 157 defines fair value, establishes a framework for measuring fair value and expands disclosure of fair value measurements. SFAS 157 applies under other accounting pronouncements that require or permit fair value measurements and, accordingly, does not require any new fair value measurements. SFAS 157 is effective for the first interim period beginning in fiscal 2008 for financial assets and liabilities, as well as for any other assets and liabilities that are carried at fair value on a recurring basis in financial statements. In November 2007, the FASB provided a one year deferral for the implementation of SFAS 157 for other nonfinancial assets and liabilities. The adoption of SFAS 157 did not have a material effect on the Company s results of operations, financial position or cash flows.

F-36

Table of Contents

Independent Auditors Report

Model Reorg, Inc. and Subsidiaries

Bellport, New York

We have audited the accompanying consolidated balance sheets of Model Reorg, Inc. and Subsidiaries as of October 31, 2007 and 2006 and the related consolidated statements of income, stockholders—equity and cash flows for the years ended October 31, 2007, 2006 and 2005. These financial statements are the responsibility of the Company—s management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with auditing standards generally accepted in the United States of America. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes consideration of internal control over financial reporting as a basis for designing audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Company s internal control over financial reporting. Accordingly, we express no such opinion. An audit also includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the consolidated financial statements referred to above present fairly, in all material respects, the financial position of Model Reorg, Inc. and Subsidiaries as of October 31, 2007 and 2006 and the results of its operations and its cash flows for the years ended October 31, 2007, 2006 and 2005 in conformity with accounting principles generally accepted in the United States of America.

As discussed in Note 1 to the consolidated financial statements, the Company has restated its 2007, 2006 and 2005 consolidated financial statements.

/s/ BDO Seidman, LLP

BDO Seidman, LLP

New York, New York

February 12, 2008, except for Note 1, as to which the date is July 24, 2008

F-37

Model Reorg, Inc. and Subsidiaries

Consolidated Balance Sheets

(Dollars In Thousands)

October 31,	2007	2006 (Restated)
Assets		(
Current		
Cash	\$ 2,988	\$ 2,737
Accounts receivable net of allowances of \$269 and \$275	56,960	47,728
Receivables from an affiliate	25,484	16,727
Inventories, net	218,206	209,934
Advances to suppliers for future purchases	1,132	2,133
Prepaids and other	7,554	7,351
	,	,
Total current assets	312,324	286,610
Property and equipment, at cost, less accumulated depreciation	2,496	2,594
Goodwill	20,434	9,392
Other assets, net	9,734	8,248
	•	
	\$ 344,988	\$ 306,844
Liabilities and Stockholders Equity		
Current		
Accounts payable non affiliates	\$ 30,882	\$ 33,303
Accounts payable affiliates	2,152	, ,,,,,,,
Accrued expenses	9,359	7,071
Accrued income taxes	,,,,,	2,169
Other current liabilities	2,548	1,903
Current maturities of long-term debt	209	3,242
		-,
Total current liabilities	45,150	47,688
Long-term debt and other	138,914	124,274
Notes payable former stockholder	100,011	158
Payable to affiliate	76,313	73,089
Tayable to allimate	70,515	75,005
Total liabilities	260,377	245,209
Commitments and Contingencies (Note 10)		
Stockholders equity		
Common stock, no par value 200 shares authorized 111 and 100 shares issued at October 31, 2007 and 2006,		
respectively	1	1
Additional paid-in capital	13,905	2,179
Retained earnings	72,180	60,930
Treasury stock 14 shares at cost	(1,475)	(1,475)
•	· , -,	() /
Total stockholders equity	84,611	61,635
	\$ 344,988	\$ 306,844

See accompanying independent auditors report and notes to consolidated financial statements

F-38

Model Reorg, Inc. and Subsidiaries

Consolidated Statements of Income

(Dollars In Thousands)

Year ended October 31,	2007	2006 (Restated)	2005 (Restated)
Net revenues, unaffiliated	\$ 302,530	\$ 306,089	\$ 278,619
Net revenues, affiliated	25,182	32,272	38,288
Total net revenues	327,712	338,361	316,907
Cost of goods sold, unaffiliated	212,111	216,489	201,674
Cost of goods sold, affiliated	25,092	28,835	33,773
			22,772
Gross Profit	90,509	93,037	81,460
Selling, warehouse, delivery and administrative expenses	61,524	59,269	57,158
(Recovery) provision on vendor advances	(2,367)	2,367	
Income from operations	31,352	31,401	24,302
Interest expense	12,749	14,506	11,179
•			
Income before income taxes	18,603	16,895	13,123
Income taxes	7,353	6,854	5,121
	ŕ	,	ŕ
Net income	\$ 11,250	\$ 10,041	\$ 8,002

See accompanying independent auditors report and notes to consolidated financial statements

Model Reorg, Inc. and Subsidiaries

Consolidated Statements of Stockholders Equity

(Dollars In Thousands)

	Common	stock	Additional Paid-in Capital	Retained Earnings (Restated)	Trea	sury Stock	1	Total ckholders Equity estated)
Balance, November 1, 2004	\$	1	\$ 2,179	\$ 42,887	\$	(1,475)	\$	43,592
Net income				8,002				8,002
Balance, October 31, 2005		1	2,179	50,889		(1,475)		51,594
Net income				10,041				10,041
Balance, October 31, 2006		1	2,179	60,930		(1,475)		61,635
Net income				11,250				11,250
Acquisition of Jacavi			10,800					10,800
Stockholder contribution			926					926
Balance, October 31, 2007	\$	1	\$ 13,905	\$ 72,180	\$	(1,475)	\$	84,611

See accompanying independent auditors report and notes to consolidated financial statements

Model Reorg, Inc. and Subsidiaries

Consolidated Statements of Cash Flows

(Dollars In Thousands)

Year Ended October 31,	2007 (Restated)	2006 (Restated)	2005 (Restated)
Cash flows from operating activities:	(((11
Net income	\$ 11,250	\$ 10,041	\$ 8,002
Adjustments to reconcile net income to net cash (used in) provided by operating activities:			
Deferred tax expense (benefit)	2,615	20	(75)
(Recovery) provision on vendor advances	(2,367)	2,367	
Depreciation and amortization	1,411	1,721	2,469
Provision for losses on accounts receivable	40	77	177
Changes in assets and liabilities:			
(Increase) decrease in:	(0.474)	(4 = 0.4)	
Accounts receivable	(9,272)	(1,781)	11,245
Receivables from affiliate	(8,757)	3,887	(9,211)
Inventories	(8,272)	10,557	(7,118)
Advances to suppliers for future purchases	3,368	1,035	670
Prepaids and other	(2,818)	(970)	568
Increase (decrease) in:			
Accounts payable, non affiliates	(2,421)	9,584	(31,637)
Accounts payable, affiliates	2,152		2051
Accrued expenses and other liabilities	764	2,696	3,051
Total adjustments	(23,557)	29,193	(29,861)
Net cash (used in) provided by operating activities	(12,307)	39,234	(21,859)
Cash flows from investing activities:			
Purchase of property and equipment	(505)	(2,023)	(484)
Acquisition of Northern Group, Inc		(242)	(627)
Acquisition of Jacavi, LLC	(242)		
Other investing activities	(1,368)	(2,345)	(1,312)
Net cash used in investing activities	(2,115)	(4,610)	(2,423)
Cash flows from financing activities:			
Increase (decrease) in revolving credit borrowings	15,459	(21,200)	31,023
Proceeds from (repayment) of affiliated note payable	3,224	(10,643)	(727)
Payment of long-term debt and other	(3,852)	(2,284)	(3,838)
Payment of notes payable former stockholder	(158)	(281)	(432)
Net cash provided by (used in) financing activities	14,673	(34,408)	26,026
Net increase in cash	251	216	1,744
Cash, beginning of year	2,737	2,521	777
Cash, end of year	\$ 2,988	\$ 2,737	\$ 2,521

See accompanying independent auditors report and notes to consolidated financial statements

F-41

Model Reorg, Inc. and Subsidiaries

Notes to Consolidated Financial Statements

(Dollars In Thousands)

1. Summary of Accounting Policies

(a) Basis of Presentation

The consolidated financial statements of Model Reorg, Inc. include the following wholly owned subsidiaries: Five Star Fragrance Co. Inc., Quality King Fragrance, Inc., Scents of Worth, Inc. and Jacavi, LLC (collectively the Company). The Company is an affiliate of Quality King Distributors, Inc. (QKD) and E Com Ventures, Inc. (E Com) through common ownership.

Effective October 2004, Model Reorg acquired all of the stock of Northern Group, Inc., (Northern), a fragrance distributor, for a total purchase price of \$11,998 in a business combination accounted for as a purchase. The results of operations of Northern are included in the accompanying consolidated financial statements as of the date of acquisition. Based on the fair values of the assets acquired and liabilities assumed, the Company recorded goodwill of \$9,392.

On October 5, 2007, the Company acquired all the stock of Jacavi, LLC (Jacavi) a fragrance distributor, in exchange for shares of Model Reorg common stock which was valued at \$10,800. The acquisition was accounted for as a purchase and the results of operations of Jacavi are included in the accompanying consolidated financial statements since the date of acquisition. In connection with the acquisition, the Company obtained a valuation of Jacavi from an independent company. Based on the fair value of the assets acquired and liabilities assumed, and the purchase consideration associated with the acquisition, the Company recorded goodwill of \$11,042, including \$242 of acquisition costs.

Based on the Company s purchase price allocation, the cost of the acquisition was allocated to the fair value of the net assets acquired as follows:

Accounts Receivable and other current assets	\$ 1,566
Inventories	4,990
Goodwill	11,042
Accounts Payable and other liabilities	(1,616)
Payable to Model	(4,940)
Total cost of acquisition	\$ 11,042

All significant intercompany balances and transactions have been eliminated.

(b) Restatements of Previously Issued Consolidated Financial Statements

During 2007, the Company identified incorrectly reported amounts related to inventory reserves, deferred revenue (included in accrued expenses) and federal and state income taxes. As of November 1, 2004, retained earnings was restated by \$5,071 relating to amounts that were previously reported.

F-42

Model Reorg, Inc. and Subsidiaries

Notes to Consolidated Financial Statements

(Dollars In Thousands)

The following is a summary of the effects of the restatements on the 2006 and 2005 financial statements.

	2006	2006	2	2005		2005
	As Repo	rted Restated	As R	eported	Re	estated
Net Income	\$ 9,	744 \$ 10,041	. \$	7,626	\$	8,002
Total Assets	\$ 304,	341 \$ 306,844	\$ 3	318,719	\$3	318,931
Total Liabilities	\$ 248,	450 \$ 245,209	\$ 2	272,572	\$ 2	267,337
Retained Earnings	\$ 55,	184 \$ 60,930	\$	45,440	\$	50,889

The Consolidated Statements of Cash Flows have been restated to reclassify capitalized merger costs from operating to investing activities.

(c) Reclassifications

Certain fiscal 2006 and 2005 amounts in the accompanying consolidated financial statements have been reclassified to conform to the fiscal 2007 presentation.

(d) Organization and Business

The Company is primarily a promotional wholesale distributor of fragrances. Customers include traditional wholesalers, chain stores, mass merchandisers and retail wholesale clubs throughout the United States. In addition, the Company has an arrangement with approximately 3,000 store locations with major retailers and sells designer fragrances on consignment. The Company also manufactures fragrances, which it owns or licenses from others.

(e) Use of Estimates

In preparing financial statements in conformity with accounting principles generally accepted in the United States of America, management is required to make estimates and assumptions that affect the reported amounts of assets and liabilities and the disclosure of contingent assets and liabilities at the date of the financial statements and revenues and expenses during the reporting period. Actual results

could differ from those estimates and assumptions.

In fiscal 2006, the Company recorded an estimated provision for losses of \$2,367 on vendor advances. The advances that were reserved related to the entities involved in the Jacavi acquisition. When the acquisition of Jacavi was certain in 2007, the estimated provision was reversed.

(f) Inventories

Inventories are valued at the lower of cost or market and are costed using the weighted average method, which approximates first-in, first-out (FIFO). Inventory reserves are based on the net realizable value of the inventory and are determined based on historical sales, aging of inventories and estimated marketability.

F-43

Model Reorg, Inc. and Subsidiaries

Notes to Consolidated Financial Statements

(Dollars In Thousands)

(g) Property and Equipment, and Depreciation

Depreciation is computed by the straight-line and accelerated methods over the estimated useful lives of the assets ranging primarily from three to seven years.

(h) Leased Property Under Capital Leases

Property under capital leases is amortized over the lives of the respective leases or useful lives of the assets, whichever is shorter, ranging from five to ten years.

(i) Goodwill

Goodwill and certain indefinite-lived intangibles are not amortized, and are tested for impairment, at least annually. There is a two-step process for impairment testing of goodwill. The first step of this test, used to identify potential impairment, compares the fair value of a reporting unit with its carrying amount, including goodwill. The second step (if necessary) measures the amount of the impairment. Based on the management s impairment review, there was no impairment identified in the recorded goodwill.

(j) Income Taxes

Model Reorg. and subsidiaries are C corporations and provisions have been made for income tax expense at statutory rates. The Company follows the liability method of accounting for income taxes. The primary objectives of accounting for income taxes are to recognize the amount of tax payable for the current year and recognize the amount of deferred tax asset or liability for the future tax consequences of events that have been reflected in the Company s financial statements or tax returns.

(k) Revenue Recognition

Sales are recognized when title passes which occurs either upon shipment of products, delivery of products, or sale to the ultimate customers from consignment inventories. Allowances for estimated returns and pricing adjustments are provided when sales are recognized and are recorded as a reduction of sales. Allowances provided for advertising, marketing and tradeshows are recorded as selling expenses since they are costs for services received from the customer which are separable from the customer s purchase of the Company products. Accruals and allowances are estimated based on available information including third party data.

(1) Concentrations of Credit Risk

The Company is potentially subject to a concentration of credit risk with respect to its trade receivables, the majority of which are due from retailers and wholesale distributors. Credit risks also relate to the seasonal nature of the business. The Company s sales are concentrated in November and December for the holiday season. The Company performs ongoing credit evaluations of its customers and generally does not require collateral. The

F-44

Model Reorg, Inc. and Subsidiaries

Notes to Consolidated Financial Statements

(Dollars In Thousands)

Company maintains allowances to cover potential or anticipated losses for uncollectible accounts. The Company maintains credit insurance on certain receivables, which minimizes the financial impact of uncollectible accounts. (m) Long-Lived Assets

The Company periodically reviews long-lived assets for impairment whenever changes in the circumstances indicate that the carrying amount of the assets may not be fully recoverable. The Company considers cash flow, management s strategic plans, significant decreases in the market value of the asset and other available information in assessing whether the carrying value of the assets can be recovered. No impairment losses have been recognized through October 31, 2007.

(n) New Accounting Pronouncements

In June 2006, the FASB issued FASB Interpretation No. (FIN) 48, Accounting for Uncertainty in Income Taxes-an interpretation of FASB Statement No. 109. FIN 48 clarifies the accounting for uncertainty in income taxes recognized under SFAS No. 109, Accounting for Income Taxes. FIN 48 prescribes a recognition threshold and measurement attribute for financial statement recognition and measurement of a tax position taken or expected to be taken in a tax return and also provides guidance on various related matters such as derecognition, interest and penalties, and disclosure. The effective date of FIN 48 for non-public enterprises is for annual periods beginning after December 15, 2007. The Company has begun evaluating the financial impact of applying the provisions of FIN 48 to all tax positions and it does not believe there will be a material financial impact upon the initial adoption of FIN 48.

2. Inventories

Inventories, which are primarily comprised of finished products, are shown net of reserves for obsolescence and market write downs. Reserves at October 31, 2007 and 2006 were \$2,968 and \$4,092, respectively.

3. Related Party Transactions

(a) The Company receives services from QKD pursuant to a service agreement. The agreement with QKD provides for the allocation of expenses, which are calculated based on various assumptions and methods. The methods employed utilize various allocation bases including the number of transactions processed, estimated delivery miles, warehouse square footage, payroll dollars and sales and inventory ratios. The following table summarizes the expenses, which were allocated to the Company.

Year ended October 31,

	2007		2006		2005				
	Total	All	located	Total	Al	located	Total	A	llocated
Warehouse and delivery	\$ 19,953	\$	1,206	\$ 17,074	\$	1,888	\$ 17,280	\$	2,241
Selling and administrative	41,571		2,884	42,195		3,504	39,878		2,513
Interest expense	12,749		8,031	14,506		7,800	11,179		5,523

\$74,273 \$12,121 \$73,775 \$13,192 \$68,337 \$10,277

F-45

Model Reorg, Inc. and Subsidiaries

Notes to Consolidated Financial Statements

(Dollars In Thousands)

Allocated interest expense represents interest expense on the Company s revolving credit borrowings (see Note 7) and is allocated at the same effective rate as in the QKD credit agreement.

- (b) On August 2, 2007 the Company entered into a service agreement with E Com whereby E Com provides IT Management Support services for all of the Company s computer systems at a rate of \$25 per month, plus out of pocket expenses. This agreement will terminate at consummation of the Company s merger with E Com or January 29, 2009, if the merger has not been finalized.
- (c) The Company had sales and purchase transactions with E Com as follows: for the years ended October 31, 2007, 2006 and 2005 the sales were \$25,182, \$32,272 and \$38,288, respectively, and purchases were \$49,188, \$17,276 and \$17,468, respectively. The Company had a receivable balance from this affiliate of \$25,484 and \$16,727 at October 31, 2007 and 2006, respectively. The Company had a payable balance to this affiliate of \$2,152 at October 31, 2007.
- (d) In December 2007, the Company began subletting new office and warehouse space from QKD (see Note 10).
- (e) During fiscal 2007, additional paid in capital increased \$926 as a result of certain tax benefits relating to the allocation of related party expenses.
- (f) Included in other current liabilities is deferred compensation due to an officer. The Company s liability is calculated by a formula and totals \$1.5 million and \$1.2 million for the years ended October 31, 2007 and 2006, respectively.

4. Property and Equipment

Major classes of property and equipment are stated at cost and consist of the following:

October 31,	2007	2006	
Buildings and improvements	\$ 433	\$ 432	
Machinery and equipment	4,858	6,157	
Construction in progress	1,711	1,728	
Furniture and fixtures	3,470	3,298	
	10,472	11,615	
Less accumulated depreciation	(7,976)	(9,021)	
	\$ 2,496	\$ 2,594	

Model Reorg, Inc. and Subsidiaries

Notes to Consolidated Financial Statements

(Dollars In Thousands)

5. Other Assets

Included in other assets is the net book value of trademarks and licenses of Five Star Fragrance, Inc. Net book values of these intangibles were \$5,574 and \$6,384 at October 31, 2007 and 2006, respectively. Estimated amortization expense for each of the five succeeding fiscal years is \$809 for years 2008 and 2009, \$452 for year 2010, and \$381 for years 2011 and 2012. Based on management s impairment review, there was no impairment identified related to these assets.

6. Payable to affiliate

The Company has a long-term note payable to QKD. This note supplements the Company s borrowings under the Credit Facility, see note 7(a). The Company is charged interest by QKD based on the monthly blended borrowing rates, which are based on the LIBOR and Prime rates. Interest rates on the note for the years ended 2007, 2006, and 2005 averaged 7.6%, 7.4%, and 5.8%, respectively. Interest expense charged to operations relating to the note payable was \$4,571, \$6,474 and \$5,032 for the years ending October 31 2007, 2006 and 2005, respectively. The note is classified as long-term as QKD will not require payment of the loan within one year. (see Note 7).

7. Long-term Debt and other

Long-term debt and other consists of the following:

October 31,	2007	2006
Revolving credit borrowings (a)	\$ 138,278	\$ 122,819
Notes payable to the former stockholders of Northern Group, Inc.		2,999
Capital lease obligations	114	199
Other	731	1,499
	139,123	127,516
Less current maturities	209	3,242
	\$ 138,914	\$ 124,274

(a) At October 31, 2007, the Company along with QKD, was part of a \$500,000 revolving line of credit with a syndicate of lenders (the Credit Agreement), which expires on April 30, 2010. The Credit Agreement provides for borrowings based on eligible accounts receivable and inventories and is secured by the Company s accounts receivable and inventories. The agreement provides the lenders with certain rights available to a secured lender. The interest rate applicable to borrowings under the Credit Agreement is based on, at QKD s option, the LIBOR rate or the prime rate. At October 31, 2007, substantially all of the revolving credit borrowings were at LIBOR, the effective rate of which approximated 6.58%. The borrowings under the Credit Agreement are classified as long-term based on the expiration date of the facility.

In connection with the Credit Agreement, the Company is required to maintain certain financial ratios and comply with certain restrictions. At October 31, 2007, the Company was in compliance with its

financial

F-47

Model Reorg, Inc. and Subsidiaries

Notes to Consolidated Financial Statements

(Dollars In Thousands)

covenants. As of February 1, 2008, the Company was not in compliance with its requirement to file its fiscal 2007 financial statements with the lender by January 31, 2008. Effective February 8, 2008 the Company received a waiver of non-compliance of this requirement from its lenders.

(b) Long term debt (exclusive of the revolving credit borrowings) is payable as follows:

Year ending October 31,	
2008	\$ 209
2009	199
2010	185
2011	159
2012	93

\$ 845

8. Income Taxes

Provisions for federal, state and local income taxes (benefits) consists of the following:

Year ended October 31,	2007	2006 (Restated)	2005 (Restated)	
Current				
Federal	\$ 3,825	\$ 5,440	\$ 4,389	
State and local	913	1,394	807	
	4,738	6,834	5,196	
Deferred:				
Federal	2,232	21	(59)	
State and local	383	(1)	(16)	
	2,615	20	(75)	
	\$ 7,353	\$ 6,854	\$ 5,121	

Net deferred income tax asset, which is included in prepaids and other, consists of the following:

October 31,	2007	2006
Current:		
Inventory valuation	\$ 1,739	\$ 3,777
Accounts receivable allowances	107	106
Depreciable assets	121	420

Other 561	840
-----------	-----

\$ 2,528 \$ 5,143

Actual income tax expense differs from amounts computed by applying the United States federal income tax rate primarily as a result of state income taxes.

F-48

Model Reorg, Inc. and Subsidiaries

Notes to Consolidated Financial Statements

(Dollars In Thousands)

9. Proforma Statements of Income (unaudited)

The Proforma statements of income (unaudited) include the operations of Jacavi for the years ended October 31, 2007 and 2006.

Year ended October 31,	2	2007	2006
Net sales, unaffiliated	\$3	13,369	\$ 325,554
Net sales, affiliated		36,557	53,790
Total net sales	3	49,926	379,344
Cost of goods sold	2	55,486	279,183
Gross Profit		94,440	100,161
Selling, warehouse, delivery and administrative expenses		63,320	61,134
Income from operations		31,120	39,027
Interest expense		13,361	15,540
Income before income taxes		17,759	23,487
Income taxes		7,962	8,544
Net income	\$	9,797	\$ 14,943

The above statements were prepared by combining Jacavi and the Company as if the Company acquired Jacavi as of November 1, 2005. The Proforma statements of income reflect (1) removal of the recovery and provision on vendor advances and (2) tax expense/benefit at the Company s effective tax rate.

10. Commitments and Contingencies

(a) Leases

Total rent expense for warehouse space and equipment charged to operations for the years ended October 31, 2007, 2006, and 2005 was approximately \$4,195, \$2,702, and \$2,642 respectively. This includes allocated warehouse rent from QKD.

In January 2008 the Company began subleasing new office and warehouse facility from QKD at a rate of \$193 per month at an annual escalation of 3%. This sublease expires December 2027. Aggregate future minimum rental payments under operating leases are payable as follows:

Year ending October 31,	
2008	\$ 2,735
2009	3,118
2010	3,138
2011	2,503 2,576
2012	2,576
Thereafter	49,996
	\$ 64,066

Model Reorg, Inc. and Subsidiaries

Notes to Consolidated Financial Statements

(Dollars In Thousands)

(b) Royalties

The Company is party to six license agreements with unaffiliated licensors. Royalty expense was \$2,542, \$2,638 and \$2,967 for the years ended October 31, 2007, 2006 and 2005, respectively. The aggregate future minimum payments under these licensing agreements at October 31, 2007 are payable as follows:

Year ending October 31,	
2008	\$ 2,085
2009	1,975
2010	744
2011	400
2012	400

\$ 5,604

(c) Litigation

The Company is a defendant in various lawsuits and claims which are in various stages of discovery and therefore no conclusions can be made by legal counsel as to their outcomes. It is the opinion of management that the outcome of these pending lawsuits and claims will not materially affect the operations or financial condition of the Company.

11. Statements of Cash Flows

Supplemental disclosure of cash flow information:

Year ended October 31,	2007	2006	2005
Cash paid during the year for:			
Income taxes	\$ 9,811	\$ 5,556	\$ 5,547
Interest (excludes allocated interest from Affiliate)	\$ 8.548	\$ 8.483	\$ 5.941

Non-cash investing activities:

In connection with the acquisition of Jacavi in 2007, the Company issued 10.9 shares of common stock valued at approximately \$10.8 million to acquire Jacavi.

12. Major Customers

For the year ended October 31, 2007 one customer accounted for 15% of net sales. At October 31, 2007 accounts receivable relating to this customer was \$11,076. For the years ended October 31, 2006, and 2005, two customers, one being E Com, accounted for 27%, and 32% of net sales, respectively. At October 31, 2006, and 2005 accounts receivable relating to these two customers accounted for 40%, and 42% of accounts receivable, respectively.

F-50

Model Reorg, Inc. and Subsidiaries

Notes to Consolidated Financial Statements

(Dollars In Thousands)

13. Major Vendors

For the year ended October 31, 2007, two vendors accounted for 44% of net purchases, E Com being one of these vendors. At October 31, 2007 accounts payable relating to these vendors accounted for 40% of accounts payable. For the years ended October 31, 2006 and 2005, two vendors accounted for 42% and 26% of net purchases, respectively. At October 31, 2006 and 2005, accounts payable relating to these two vendors accounted for 63% and 18% of accounts payable, respectively.

14. Acquisition

On October 5, 2007, the Company consummated an Agreement and Plan of Merger with Jacavi Holdings, Inc., a vendor of the Company, whereby the Company acquired 100% of the stock of Jacavi. In connection with the merger, the Company issued 10.9 shares of its common stock to the former shareholders of Jacavi. At the date of the acquisition, the operations of Jacavi were merged into Quality King Fragrance, Inc.

15. Subsequent Event

On December 21, 2007, the Company and its stockholders entered into an Agreement and Plan of Merger with E Com Ventures, Inc. whereby the Company will be merged into a newly formed wholly-owned acquisition subsidiary of E Com in exchange for the issuance of 5,900,000 shares of E Com s common stock and warrants to acquire an additional 1,500,000 common shares of E Com. The acquisition subsidiary will be the surviving entity and for accounting purposes, the Company will be the acquirer. The consummation of the Merger is subject to certain conditions, including shareholder approval, approval by NASDAQ of the listing of shares to be issued, and the availability of a new \$280 million secured credit facility to replace the Company s and E Com s existing third party credit facilities.

16. Segment Information

Model Reorg operates in three industry segments, wholesale distribution, retail and manufacturing. Management reviews segment information by segment and on a consolidated basis each month. Model Reorg distributes fragrances to wholesalers and mass market retailers. Retail sales are made on a consignment basis at leased store locations. The manufacturing of owned and licensed brands is outsourced to third party fillers. The accounting policies for the segments are the same as those described in the Summary of Accounting Policies in the notes to the audited consolidated financial statements. Model Reorg s chief operating decision maker assesses segment performance by reference to gross profit. Each of the segments has its own assets, liabilities, revenues and cost of goods sold. While each segment has certain unallocated operating expenses, these expenses are not reviewed by the chief operating decision maker on a segment basis but rather on a consolidated company basis.

Fiscal Year Ended October 31,			
2007	2006	2005	
245,493	\$ 257,826	\$ 244,128	
76,369	73,990	68,971	
5,850	6,545	3,808	
327,712	\$ 338,361	\$ 316,907	
	2007 245,493 76,369 5,850	2007 2006 245,493 \$ 257,826 76,369 73,990 5,850 6,545	

Model Reorg, Inc. and Subsidiaries

Notes to Consolidated Financial Statements

(Dollars In Thousands)

	Fiscal Y	Fiscal Year Ended October 31,			
	2007	2006	2005		
Gross profit:					
Wholesale	\$ 57,584	\$ 59,886	\$ 51,292		
Retail	30,040	29,945	27,941		
Manufacturing	2,885	3,206	2,227		
	\$ 90,509	\$ 93,037	\$ 81,460		

		Fiscal Year Ended October 31,		
		2007	2006	
Total assets:				
Wholesale		\$ 339,630	\$ 344,872	
Retail		75,887	67,728	
Manufacturing		25,401	20,809	
		\$ 440,918	\$ 433,409	
Eliminations	(a)	(95,930)	(126,565)	
Consolidated assets		\$ 344,988	\$ 306,844	

(a) Adjustment to eliminate intercompany receivables and investment in subsidiaries.

At October 31,	2007	2006	2005
Goodwill:			
Wholesale	\$ 27,176	\$ 16,134	\$ 15,893
Retail	\$ 5,356	\$ 5.356	\$ 5,356

17. Quarterly Financial Data (Unaudited)

Unaudited summarized financial results for fiscal years ended October 31, 2007 and 2006 have been set forth below.

(\$ in thousands)	Janua	ary 31, 2007	Apr	il 30, 2007	Jul	y 31, 2007	Octob	oer 31, 2007
2007 QUARTER			_					
Net Sales	\$	102,481	\$	64,851	\$	67,553	\$	92,827
Gross Profit	\$	27,190	\$	20,745	\$	19,480	\$	23,094
Net Income	\$	3,760	\$	3,140	\$	1,656	\$	2,694
	Ja	nuary 31, 2006	A	pril 30, 2006	J	July 31, 2006	Oc	tober 31, 2006
2006 QUARTER								

\$

63,016

15,838

(430)

\$

\$

\$

100,624

28,989

4,640

67,520

18,962

1,199

Table of Contents 239

\$

107,201

29,248

4,632

Net Sales

Gross Profit

Net Income (loss)

F-52

Model Reorg, Inc. and Subsidiaries

Consolidated Balance Sheets

(Dollars In Thousands)

	ril 30, 2008 naudited)	Octol	ber 31, 2007
Assets			
Current:			
Cash	\$ 5,136	\$	2,988
Accounts receivable net of allowances of \$132 and \$269	28,407		56,960
Receivables from an affiliate	35,844		25,484
Inventories, net	193,620		218,206
Advances to suppliers for future purchases	2,085		1,132
Prepaids and other	5,724		7,554
Total current assets	270,816		312,324
Property and equipment, at cost, less accumulated depreciation	3,141		2,496
Goodwill	20,434		20,434
Other assets, net	12,146		9,734
	\$ 306,537	\$	344,988
Liabilities and Stockholders Equity Current Accounts payable non affiliates Accounts payable affiliates Accrued expenses	\$ 22,965 144 9,276	\$	30,882 2,152 9,359
Other current liabilities	2,935		2,548
Current maturities of long-term debt	208		209
Total current liabilities	35,528		45,150
Long-term debt and other	102,732		138,914
Payable to affiliate	76,912		76,313
Total liabilities	215,172		260,377
Commitments and Contingencies			
Stockholders equity			
Common stock, no par value 200 shares authorized 111 shares issued at October 31, 2007 and April 30, 2008	1		1
Additional paid-in capital	13,905		13,905
Retained earnings	78,934		72,180
Treasury stock 14 shares at cost	(1,475)		(1,475)
Total stockholders equity	91,365		84,611
	\$ 306,537	\$	344,988

See accompanying notes to consolidated financial statements

F-53

Model Reorg, Inc. and Subsidiaries

Consolidated Statements of Income

(unaudited)

(Dollars In Thousands)

	Three Mo	Three Months Ended		ths Ended
	April 30, 2008	April 30, 2007	April 30, 2008	April 30, 2007
Net revenues, unaffiliated	\$ 55,663	\$ 59,144	\$ 156,902	\$ 156,108
Net revenues, affiliated	6,370	5,707	19,685	11,224
Total net revenues	62,033	64,851	176,587	167,332
Cost of goods sold, unaffiliated	40,047	39,564	111,280	111,709
Cost of goods sold, affiliated	4,851	4,542	14,730	7,687
Gross profit	17,135	20,745	50,577	47,936
Selling, warehouse, delivery and administrative expenses	13,790	12,741	33,663	30,331
Income from operations	3,345	8,004	16,914	17,605
Interest expense	2,135	2,770	5,300	6,104
Income before income taxes	1,210	5,234	11,614	11,501
Income taxes	505	2,094	4,860	4,600
Net income	\$ 705	\$ 3,140	\$ 6,754	\$ 6,901

 $See\ accompanying\ notes\ to\ consolidated\ financial\ statements$

Model Reorg, Inc. and Subsidiaries

Consolidated Statements of Cash Flows

(unaudited)

(Dollars In Thousands)

	Six Months En	nded April 30, 2007
Cash flows from operating activities:		
Net income	\$ 6,754	\$ 6,901
Adjustments to reconcile net income to net cash provided by operating activities:		
Depreciation and amortization	713	822
Recovery for losses on accounts receivable	(89)	
Changes in assets and liabilities:		
Decrease (increase) in:		
Accounts receivable	28,642	21,071
Receivables from affiliate	(10,360)	(3,299)
Inventories	24,586	28,742
Advances to suppliers for future purchases	(953)	(3,308)
Prepaids and other	1,830	(1,177)
(Decrease) increase in:		
Accounts payable, non affiliates	(7,917)	(13,155)
Accounts payable, affiliates	(2,008)	, , ,
Accrued expenses and other liabilities	304	1,289
		,
Total adjustments	34,748	30,985
Net cash provided by operating activities	41,502	37,886
Cash flows from investing activities:		
Purchase of property and equipment	(952)	(986)
Other investing activities	(2,819)	(255)
Net cash used in investing activities	(3,771)	(1,241)
Cash flows from financing activities:		
Decrease in revolving credit borrowings	(36,081)	(22,148)
Proceeds from (repayment of) affiliated note payable	599	(13,234)
Payment of long-term debt and other	(101)	(1,635)
Payment of notes payable former stockholder	(=+=)	(158)
Net cash used in financing activities	(35,583)	(37,175)
Net increase (decrease) in cash	2,148	(530)
Cash, beginning of period	2,988	2,737
Cash, end of period	\$ 5,136	\$ 2,207

See accompanying notes to consolidated financial statements

F-55

Model Reorg, Inc. and Subsidiaries

Notes to Consolidated Financial Statements

(unaudited)

(Dollars In Thousands)

1. Summary of Accounting Policies

(a) Basis of Presentation

The consolidated financial statements of Model Reorg, Inc. include the following wholly owned subsidiaries: Five Star Fragrance Co. Inc., Quality King Fragrance, Inc., Scents of Worth, Inc. and Jacavi, LLC (collectively the Company). The Company is an affiliate of Quality King Distributors, Inc. (QKD) and E Com Ventures, Inc. (E Com) through common ownership.

Effective October 2004, Model Reorg acquired all of the stock of Northern Group, Inc., (Northern), a fragrance distributor, for a total purchase price of \$11,998 in a business combination accounted for as a purchase. The results of operations of Northern are included in the accompanying consolidated financial statements as of the date of acquisition. Based on the fair values of the assets acquired and liabilities assumed, the Company recorded goodwill of \$9,392.

On October 5, 2007, the Company acquired all the stock of Jacavi, LLC (Jacavi) a fragrance distributor, in exchange for shares of Model Reorg common stock which was valued at approximately \$10,800. The acquisition was accounted for as a purchase and the results of operations of Jacavi are included in the accompanying consolidated financial statements as of the date of acquisition. In connection with the acquisition, the Company obtained a valuation of Jacavi from an independent company. Based on the fair value of the assets acquired and liabilities assumed, and the purchase consideration associated with the merger, the Company recorded goodwill of \$11,042, including \$242 of acquisition costs.

All significant intercompany balances and transactions have been eliminated.

(b) Organization and Business

The Company is primarily a promotional wholesale distributor of fragrances. Customers include traditional wholesalers, chain stores, mass merchandisers and retail wholesale clubs throughout the United States. In addition, the Company has arrangements with major retailers covering approximately 3,000 store locations and sells designer fragrances on consignment. The Company also manufactures fragrances that it owns or licenses from others.

(c) Use of Estimates

In preparing financial statements in conformity with accounting principles generally accepted in the United States of America, management is required to make estimates and assumptions that affect the reported amounts of assets and liabilities and the disclosure of contingent assets and liabilities at the

date of the financial statements and revenues and expenses during the reporting period. Actual results could differ from those estimates and assumptions.

F-56

Model Reorg, Inc. and Subsidiaries

Notes to Consolidated Financial Statements

(unaudited)

(Dollars In Thousands)

(d) Inventories

Inventories are valued at the lower of cost or market with cost being determined using the weighted average method, which approximates first-in, first-out (FIFO). Inventory reserves are based on the net realizable value of the inventory and are determined based on historical sales, aging of inventories and estimated marketability.

(e) Property and Equipment, and Depreciation

Depreciation is computed by the straight-line and accelerated methods over the estimated useful lives of the assets ranging primarily from three to seven years.

(f) Leased Property Under Capital Leases

Property under capital leases is amortized over the lives of the respective leases or the useful lives of the assets, whichever is shorter, ranging from five to ten years.

(g) Income Taxes

Model Reorg and subsidiaries are C corporations and provisions have been made for income tax expense at statutory rates.

The Company follows the liability method of accounting for income taxes. The primary objectives of accounting for income taxes are to recognize the amount of tax payable for the current year and recognize the amount of deferred tax asset or liability for the future tax consequences of events that have been reflected in the Company s financial statements or tax returns.

(h) Revenue Recognition

Sales are recognized when title passes, which occurs either upon shipment of products or sale to the ultimate customers from consignment inventories. Allowances for estimated returns and pricing adjustments are provided when sales are recognized and are recorded as a reduction of sales. Allowances provided for advertising, marketing and tradeshows are recorded as selling expenses since they are costs for services received from the customer which are separable from the customer s purchase of the Company s products. Accruals and allowances are estimated based on available information including third party data.

(i) Concentrations of Credit Risk

The Company is potentially subject to a concentration of credit risk with respect to its trade receivables, the majority of which are due from retailers and wholesale distributors. Credit risks also relate to the seasonal nature of the business. The Company s sales are concentrated in November and December for the holiday season. The Company performs ongoing credit evaluations of its customers and generally does not require collateral. The Company maintains allowances to cover potential or anticipated losses for

F-57

Model Reorg, Inc. and Subsidiaries

Notes to Consolidated Financial Statements

(unaudited)

(Dollars In Thousands)

uncollectible accounts. The Company maintains credit insurance on certain receivables, which minimizes the financial impact of uncollectible accounts.

2. Inventories

Inventories, which are primarily comprised of finished products, are shown net of reserves for obsolescence and market write downs. Reserves at April 30, 2008 and October 31, 2007 were \$2,976 and \$2,968, respectively.

3. Related Party Transactions

(a) The Company receives services from QKD and E Com pursuant to service agreements. The agreement with QKD provides for the allocation of expenses, which are calculated based on various assumptions and methods. The methods employed utilize various allocation bases including the number of transactions processed, estimated delivery miles, warehouse square footage, payroll dollars and sales and inventory ratios. Allocated operating expenses for the three and six months ended April 30, 2008 were \$1.8 million and \$4.9 million or 13.3% and 35.4% of total operating expenses, as compared to \$3.0 million and \$6.0 million or 24.4% and 46.8% of total operating expense for the three and six months ended April 30, 2007.

Allocated interest expense represents interest expense on the Company s revolving credit borrowings and is allocated at the same effective rate as in the QKD Credit Agreement.

- (b) On August 2, 2007 the Company entered into a service agreement with E Com whereby E Com provides IT Management Support services for all of the Company s computer systems at a rate of \$25 per month, plus out of pocket expenses. This agreement will terminate at consummation of the Company s merger with E Com or January 29, 2009, if the merger has not been finalized.
- (c) The Company had sales and purchase transactions with E Com as follows: sales for the three and six months ended April 30, 2008 were \$6,370 and \$19,685, respectively, and for the three and six months ended April 30, 2007, \$5,707 and \$11,224, respectively. Purchases for the three and six months ended April 30, 2008 were \$9,802 and \$21,088, respectively, and for the three and six months ended April 30, 2007 \$5,707 and \$8,128, respectively. The Company had a receivable balance from this affiliate of \$35,844, and \$25,484 at April 30, 2008 and October 31, 2007, respectively. The Company had a payable balance to this affiliate of \$144 and \$2,152 at April 30, 2008 and October 31, 2007, respectively.
- (d) Effective January 2008, the Company began subletting new office and warehouse space from QKD (see Note 4).

F-58

Model Reorg, Inc. and Subsidiaries

Notes to Consolidated Financial Statements

(unaudited)

(Dollars In Thousands)

- (e) Included in other current liabilities is deferred compensation due to an officer. The Company s liability is calculated by a formula and totals \$1.9 million and \$1.5 million at April 30, 2008 and October 31, 2007, respectively.
- (f) On December 21, 2007, the Company and its stockholders entered into an Agreement and Plan of Merger with E Com Ventures, Inc. whereby the Company will be merged into a newly formed wholly-owned Acquisition Sub of E Com in exchange for the issuance of 5,900,000 shares of E Com s common stock and warrants to acquire an additional 1,500,000 common shares of E Com. The Acquisition Sub will be the surviving entity and for accounting purposes, the Company will be the acquirer. The consummation of the Merger is subject to certain conditions, including shareholder approval, approval by NASDAQ of the listing of shares to be issued, and the availability of a new secured credit facilities to replace the Company s and E Com s existing third party credit facilities. A commitment has been obtained from a lender on May 16, 2008 for a senior credit facility of \$250 million.

4. Commitments and Contingencies

(a) Leases

Total rent expense for warehouse space and equipment charged to operations for the three and six months ended April 30, 2008 was approximately \$1,219 and \$2,700, respectively and for the three and six months ended April 30, 2007 was approximately \$848 and \$1,663, respectively. This includes allocated warehouse rent from QKD.

In January 2008 the Company began subleasing new office and warehouse facility from QKD at a rate of \$193 per month at an annual escalation of 3%. This sublease expires December 2027.

(b) Litigation

The Company is a defendant in various lawsuits and claims which are in various stages of discovery and therefore no conclusions can be made as to their outcomes. Management believes that the outcome of these pending lawsuits and claims will not materially affect the operations, financial condition or cash flow of the Company.

Model Reorg, Inc. and Subsidiaries

Notes to Consolidated Financial Statements

(unaudited)

(Dollars In Thousands)

5. Segment Information

Model Reorg operates in three industry segments, wholesale distribution, retail and manufacturing. Management reviews segment information by segment and on a consolidated basis each month. Model Reorg distributes fragrances to wholesalers and mass market retailers. Retail sales are made on a consignment basis at leased store locations. The manufacturing of owned and licensed brands is outsourced to third party fillers. The accounting policies for the segments are the same as those described in the Summary of Accounting Policies in the notes to the audited Consolidated Financial Statements. Model Reorg s chief operating decision maker assesses segment performance by reference to gross profit. Each of the segments has its own assets, liabilities, revenues and cost of goods sold. While each segment has certain unallocated operating expenses, these expenses are not reviewed by the chief operating decision maker on a segment basis but rather on a consolidated company basis.

	Three Months Ended		Six Months Ended	
	April 30, 2008	April 30, 2007	April 30, 2008	April 30, 2007
Net revenues:				
Wholesale	\$ 45,730	\$ 49,141	\$ 128,297	\$ 120,140
Retail	15,299	14,705	45,661	45,170
Manufacturing	1,004	1,005	2,629	2,022
	\$ 62,033	\$ 64,851	\$ 176,587	\$ 167,332
Gross profit:				
Wholesale	\$ 10,916	\$ 14,640	\$ 32,022	\$ 29,764
Retail	5,884	5,811	17,642	17,341
Manufacturing	335	294	913	831
	\$ 17,135	\$ 20,745	\$ 50,577	\$ 47,936

A 4 A -- -- 21

		At Aprii	0.4.121
		30, 2008	October 31, 2007
Total assets:			
Wholesale		\$ 310,193	\$ 339,630
Retail		76,195	75,887
Manufacturing		25,187	25,401
		\$ 411,575	\$ 440,918
Eliminations	(a)	(105,038)	(95,930)
Consolidated assets		\$ 306,537	\$ 344,988

(a) Adjustment to eliminate intercompany receivables and investment in subsidiaries

Goodwill for the wholesale and retail segments was \$27,176 and \$5,356 respectively, at both April 30, 2008 and October 31, 2007.

F-60

Model Reorg, Inc. and Subsidiaries

Notes to Consolidated Financial Statements

(unaudited)

(Dollars In Thousands)

6. New Accounting Pronouncements

In December 2007, the FASB issued SFAS No, 141(R), Business Combinations (SFAS 141(R)). SFAS 141(R) establishes principles and requirements for how an acquirer recognizes and measures in its financial statements the identifiable assets acquired, the liabilities assumed, the goodwill acquired, and any noncontrolling interest in the acquiree. This statement also establishes disclosure requirements to enable the evaluation of the nature and financial effect of the business combination. SFAS 141(R) is effective for fiscal years beginning after December 15, 2008. We are in the process of evaluating the effect the adoption of SFAS 141(R) will have on our results of operations, financial position and cash flows.

In February 2007, the FASB issued SFAS 159, The Fair Value Option for Financial Assets and Financial Liabilities Including an amendment of FASB Statement No. 115. (SFAS 159). SFAS 159 permits entities to choose to measure eligible items at fair value at specified election dates and report unrealized gains and losses on items for which the fair value option has been elected in earnings at each subsequent reporting date. SFAS 159 is effective for

fiscal year ended October 31, 2009. The adoption of SFAS 159 did not have any effect on our results of operations, financial position and cash flows.

In September 2006, the FASB issued SFAS 157, Fair Value Measurements (SFAS 157). SFAS 157 defines fair value, establishes a framework for measuring fair value and expands disclosure of fair value measurements. SFAS 157 applies under other accounting pronouncements that require or permit fair

value measurements and, accordingly, does not require any new fair value measurements. SFAS 157 is effective for the first interim period beginning in fiscal 2008 for financial assets and liabilities, as well as for any other assets and liabilities that are carried at fair value on a recurring basis in financial statements. In November 2007, the FASB provided a one year deferral for the implementation of SFAS 157 for other nonfinancial assets and liabilities. We do not expect the adoption of SFAS 157 to have a material effect on our results of operations, financial position and cash flows.

In June 2006, the FASB issued FASB Interpretation No. (FIN) 48, Accounting for Uncertainty in Income Taxes-an interpretation of FASB Statement No. 109. FIN 48 clarifies the accounting for uncertainty in income taxes recognized under SFAS No. 109, Accounting for Income Taxes. FIN 48 prescribes a recognition threshold and measurement attribute for financial statement recognition and measurement of a tax position taken or expected to be taken in a tax return and also provides guidance on various related matters such as derecognition, interest and penalties, and disclosure. The effective date of FIN 48 for non-public enterprises is for annual periods beginning after December 15, 2007. The Company has begun evaluating the financial impact of applying the provisions of FIN 48 to all tax positions and it does not believe there will be a material financial impact upon the initial adoption of FIN 48.

F-61

Annex I

AGREEMENT AND PLAN OF MERGER

among

Model Reorg, Inc.,

the stockholders of Model Reorg, Inc.,

E Com Ventures, Inc.

and

Model Reorg Acquisition LLC

Dated as of December 21, 2007

Proxy Statement Exhibit

Notice about Representations and Warranties Herein

This Agreement is included as an exhibit to E Com Ventures, Inc. s Proxy Statement to provide information about the terms of the contractual arrangement between the parties with respect to the Merger described herein, but its filing is not intended to provide factual information about the parties to any other person. The representations and warranties contained in this Agreement were made only for purposes of this Agreement and as of specific dates, were solely for the benefit of the parties hereto, and are subject to limitations agreed to between the contracting parties, including that they are qualified by disclosures between the parties. These representations and warranties may have been made for the purposes of allocating contractual risk between the parties to the Agreement instead of establishing these matters as facts, and may be subject to standards of materiality applicable to the contracting parties that differ from those applicable to disclosures to investors. They should be viewed by investors in this context. Accordingly, investors and third parties should not rely on these representations and warranties as independent characterizations of the actual state of facts at the time they were made or otherwise but should consider them together with the other information in this Proxy Statement or that we have disclosed in other filings with the SEC.

TABLE OF CONTENTS

ART	ICLE 1 PRELIMINARY ACTIONS	1
1.1 1.2	Preliminary Actions Proxy Statement	1 2
ART	ICLE 2 THE MERGER	2
2.1	The Merger	2
2.2	Closing	2
2.3	Effective Time	2
2.4	Effects of the Merger	2
2.5	Certificate of Formation and Limited Liability Company Agreement of the Surviving Entity	2
2.6	Managers and Officers of the Surviving Entity	2
2.7	Conversion of Securities	3
2.8	Status of Certificates	3
2.9	Escrow Fund	3
ART	ICLE 3 REPRESENTATIONS AND WARRANTIES OF E COM	4
3.1	Organization	4
3.2	Capitalization	4
3.3	Authority	5
3.4	Opinion of Financial Advisor	6
3.5	Finders or Brokers	6
ART	ICLE 4 REPRESENTATIONS AND WARRANTIES OF MODEL	6
4.1	Organization	6
4.2	Capitalization	6
4.3	Authority	7
4.4	Consents and Approvals; No Violations	7
4.5	Undisclosed Liabilities, Etc	7
4.6	Absence of Certain Changes or Events	8
4.7	Legal Proceedings	10
4.8	Compliance with Applicable Law; Permits	10
4.9	Tax Matters	10
4.10	Employee Benefits	11
4.11	Labor Matters	13
4.12	Environmental Matters	13
4.13	Insurance	14
4.14	Finder or Brokers	15
4.15	Contracts	15
4.16	Owned Real and Personal Property; Encumbrances	16
4.17	Leased Real and Personal Property	16
4.18	Intellectual Property	16
4.19	Corporate Documents; Books and Records	18
4.20	Bank Accounts	18
4.21	Related Person Transactions	18
4.22	Disclosure	18
ART	ICLE 5 ADDITIONAL COVENANTS AND AGREEMENTS	18
5.1	Conduct of Business	18
5.2	Reasonable Best Efforts	19

1

<u>Table</u>	of Conte	<u>ents</u>	
5.3 5.4 5.5 5.6 5.7 5.8 5.9 5.10 5.11 5.12 ARTI	Public Advices Access Notification Director Delivery Subseque Tax Matt Voting E Filing of ICLE 6 Communication Control of Control of Control of Access Access Notification (Control of Control of Cont	ffers; Fiduciary Obligations nnouncements tion of Certain Matters and Officer Indemnification of Financial Information ent Transactions with Model Shareholders ters C Com Common Stock Listing Application and Blue Sky Forms CONDITIONS TO THE MERGER ons to Each Party s Obligation to Effect the Merger	19 20 20 20 20 21 21 21 21 22 22 23
6.2		ons to Model's Obligations to Effect the Merger	23 24
6.3		ons to E Com s and Acquisition Sub s Obligations to Effect the Merger	
7.1 7.2	Termina	ERMINATION ition Termination	24 24 26
ART	ICLE 8 IN	DEMNIFICATION	26
8.1 8.2 8.3 8.4 8.5	3.2 Indemnification Notice Procedures 3.3 Third-Party Claims 3.4 Payment of Claims		
ART	ICLE 9 MI	ISCELLANEOUS	30
9.1 9.2 9.3 9.4 9.5 9.6 9.7 9.8 9.9 9.10 9.11 9.12 9.13	Amendm Extension Assignme Counterp Entire Ag Governin Notices Severabil Headings Definition Arbitrati	s ons, Construction	30 30 30 31 31 31 33 33 33 33 37
		LIST OF SCHEDULES	
Schedule 2.7		Allocation of Merger Consideration and Escrow Shares	
		LIST OF EXHIBITS	
Exhibit A		Form of Common Stock Purchase Warrant	
Exhibit B		Registration Rights Agreement	
Exhibit C		Terms of Affiliate Debt	

AGREEMENT AND PLAN OF MERGER

This AGREEMENT AND PLAN OF MERGER, dated as of December 21, 2007 (this <u>Agreement</u>), is by and among Model Reorg, Inc., a New York corporation (<u>Model</u>), the stockholders of Model set forth on Schedule 2.7 (the <u>Model Sharehol</u>ders), E Com Ventures, Inc., a Florida corporation (<u>E Com</u>), and Model Reorg Acquisition LLC, a Delaware limited liability company and wholly-owned subsidiary of E Com (<u>Acquisition Sub</u>). Certain terms used in this Agreement are used as defined <u>in Section 9.11</u>.

Recitals

Each of the respective Boards of Directors of E Com and Model deems it advisable that Acquisition Sub acquire Model on the terms and subject to the conditions provided for in this Agreement;

In furtherance thereof, it is proposed that such acquisition be accomplished by the merger of Model with and into Acquisition Sub, with Acquisition Sub being the surviving entity, in accordance with the Business Corporation Law of the State of New York (the <u>BCL</u>) and the Delaware Limited Liability Company Act (the <u>DLLCA</u>), pursuant to which all of the shares of common stock, no par value, of Model (the <u>Model Common Stock</u>) issued and outstanding will be converted into the right to receive an aggregate of 5,900,000 shares of common stock, \$.01 par value of E Com (the <u>E Com Common Stock</u>) and Common Stock Warrants Purchase Warrants to acquire 1,500,000 shares of E Com Common Stock in the form of <u>Exhibit A</u> attached hereto (the <u>Warrants</u> and, together with such shares of E Com Common Stock, the <u>Merger Consideration</u>), on the terms and subject to the conditions provided for in this Agreement, and such surviving entity will be a wholly-owned subsidiary of E Com (the <u>Merger</u>); and

The respective Boards of Directors of E Com and Model have adopted, the sole member of Acquisition Sub has approved and the shareholders of Model have each approved this Agreement and the Merger.

Agreements

NOW, THEREFORE, in consideration of the representations, warranties, covenants and agreements contained in this Agreement, and intending to be legally bound hereby, E Com, Acquisition Sub and Model hereby agree as follows:

ARTICLE 1 PRELIMINARY ACTIONS

- 1.1 <u>Preliminary Actions</u>. As soon as practicable following the date hereof, E Com, acting through its Board of Directors, shall, in accordance with all applicable laws, statutes, ordinances, codes, rules, regulations, decrees and orders of any Governmental Entity (collectively, <u>Law</u>s) and its current certificate of incorporation and by-laws:
- (a) duly call, give notice of, convene and hold an annual or special meeting of E Com s shareholders (the Shareholders Meeting) for the purposes of obtaining the affirmative vote of E Com s shareholders in favor of (i) the issuance of E Com Common Stock and the Warrants pursuant to the Merger and the issuance of E Com Common Stock upon exercise of the Warrants, as required under applicable Law and Nasdaq Market Place Rule 4350(i)(1)(C) and Section 6.1(a) and (ii) the amendment of the certificate of incorporation of E Com (the Charter Amendment) to increase the number of shares of E Com Common Stock that E Com is authorized to issue to 20,000,000 shares (together, the E Com Shareholder Approval) and in favor of the change of the E Com s name to Perfumania Holdings, Inc. ; and
- (b) in consultation with Model, prepare and file with the SEC a preliminary proxy statement relating to the E Com Shareholder Approval and obtain and furnish the information required by the SEC to be included

Table of Contents

therein and, after consultation with Model, respond promptly to any comments made by the staff of the SEC with respect to the preliminary proxy statement and cause a definitive proxy statement (together with all amendments, supplements and exhibits thereto, the Proxy Statement) to be mailed to E Com s shareholders at the earliest practicable date: provide d that no amendments or supplements to the Proxy Statement shall be made by E Com without consultation with Model.

1.2 <u>Proxy Statement</u>. Without limiting any other provision of this Agreement, whenever any party hereto becomes aware of any event or change which is required to be set forth in an amendment or supplement to the Proxy Statement, such party shall promptly inform the other parties thereof and each of the parties shall cooperate in the preparation, filing with the SEC and (as and to the extent required by applicable federal and state securities laws) dissemination to E Com s shareholders of such amendment or supplement. Model shall provide E Com with such information with respect to Model and its affiliates as shall be required to be included in the Proxy Statement.

ARTICLE 2 THE MERGER

- 2.1 <u>The Merger</u>. Upon the terms and subject to the conditions set forth in this Agreement, and in accordance with the BCL and the DLLCA, at the Effective Time Model shall merge with and into Acquisition Sub, and the separate corporate existence of Model shall thereupon cease, and Acquisition Sub shall be the surviving entity in the Merger (the <u>Surviving Entity</u>).
- 2.2 <u>Closing</u>. The closing of the Merger (the <u>Closing</u>) shall take place at 10:00 a.m. (New York City time) at the offices of Edwards Angell Palmer & Dodge LLP, 750 Lexington Avenue, New York, New York, on a date to be specified by the parties, which date shall be no later than the second Business Day after satisfaction or waiver of the conditions set forth in <u>Article 6</u> (other than conditions that by their nature are to be satisfied at the Closing, but subject to the satisfaction or waiver of such conditions), unless another time, place or date, or any or all, are agreed to in writing by the parties hereto. The date on which the Closing is held is herein referred to as the <u>Closing Date</u>.
- 2.3 Effective Time. Subject to the provisions of this Agreement, on the Closing Date the parties shall file certificates of merger with the Department of State of the State of New York and the Secretary of State of the State of Delaware pursuant to the applicable provisions of the BCL and the DLLCA (each, the __Certificate of Merger), executed in accordance with the relevant provisions of the BCL and the DLLCA, and shall make all other filings or recordings required under the BCL and the DLLCA in order to effect the Merger, in each case in forms approved by E Com and Model, which approval shall not be unreasonably withheld. The Merger shall become effective upon the filing of the second of the Certificates of Merger or at such other time as is agreed by the parties hereto and specified in the Certificates of Merger (the time at which the Merger becomes effective is herein referred to as the __Effective Time).
- 2.4 <u>Effects of the Merger</u>. From and after the Effective Time, the Merger shall have the effects set forth in the BCL and the DLLCA. Without limiting the generality of the foregoing, and subject thereto, at the Effective Time, all the properties, rights, privileges, powers and franchises of Model and Acquisition Sub shall vest in the Surviving Entity, and all debts, duties and liabilities of Acquisition Sub and Model shall become the debts, liabilities and duties of the Surviving Entity.
- 2.5 <u>Certificate of Formation and Limited Liability Company Agreement of the Surviving Entity.</u> The certificate of formation and limited liability company agreement of Acquisition Sub shall be applicable to the Surviving Entity until thereafter amended as provided by law and such certificate of formation and limited liability company agreement.
- 2.6 <u>Managers and Officers of the Surviving Entity</u>. The managers and officers of Acquisition Sub immediately prior to the Effective Time shall be the managers and officers of the Surviving Entity until their

2

Table of Contents

respective successors are duly elected or appointed and qualified or their earlier death, resignation or removal in accordance with the limited liability company agreement of the Surviving Entity.

- 2.7 <u>Conversion of Securities</u>. At the Effective Time, by virtue of the Merger and without any action on the part of the holders of any securities of Acquisition Sub or Model:
- (a) The issued and outstanding shares of Model Common Stock at the Effective Time shall be converted into and become an aggregate of 5,900,000 validly issued, fully paid and nonassessable shares of E Com Common Stock and the Warrants. The E Com Common Stock and the Warrants included in the Merger Consideration shall be allocated among the Model Shareholders as set forth on Schedule 2.7 attached hereto.
- (b) Any shares of Model Common Stock that are owned by Model as treasury stock shall be automatically canceled and retired and shall cease to exist and no consideration shall be delivered in exchange therefor.
- (c) Each outstanding membership interest of Acquisition Sub shall remain issued, outstanding and unchanged and owned by E Com.

2.8 Status of Certificates.

- (a) <u>Transfer Books</u>; No Further Ownership Rights in Company Stock. The Merger Consideration to be paid in respect of shares of Model Common Stock shall be paid upon the surrender or exchange of stock certificates representing shares of Model Common Stock (collectively, the <u>Certificates</u>) in accordance with the terms of this <u>Article</u> 2 and shall be deemed to have been paid in full satisfaction of all rights pertaining to the shares of Model Common Stock previously represented by such Certificates. At the Effective Time, the stock transfer books of Model shall be closed and thereafter there shall be no further registration of transfers of shares of Model Common Stock on the records of Model, except for the cancellation of such shares in connection with the Merger. From and after the Effective Time, the holders of Certificates that evidenced ownership of shares of Model Common Stock outstanding immediately prior to the Effective Time shall cease to have any rights with respect to such shares, except as otherwise provided for herein or by applicable Law. If, after the Effective Time, bona fide Certificates are presented to the Surviving Entity for any reason, they shall be canceled and exchanged as provided in this <u>Article 2</u>.
- (b) <u>Lost, Stolen or Destroyed Certificates</u>. If any Certificate shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the Person claiming such Certificate to be lost, stolen or destroyed and, if required by E Com, the posting by such Person of a bond, in such reasonable amount as E Com may direct, as indemnity against any claim that may be made against it with respect to such Certificate, E Com will pay, in exchange for such lost, stolen or destroyed Certificate, the applicable Merger Consideration to be paid in respect of the shares of Model Common Stock formerly represented by such Certificate, as contemplated by this <u>Article 2</u>, in which event such Certificate shall be deemed cancelled.

2.9 Escrow Fund.

At the Closing, E Com shall retain and not deliver to the Model Shareholders certificates representing 295,000 shares of the Merger Consideration (the <u>Escrow Shares</u>, including all dividends, distributions or earnings attributable thereto, collectively, the <u>Escrow Fund</u>) subject to the terms hereof for the purpose of securing the indemnification rights of E Com under this Agreement, such Escrow Shares being allocated among the holders of Model Common Stock as set forth on <u>Schedule 2.7</u>. After the Closing, E Com shall act as escrow agent (the <u>Escrow Agent</u>) with respect to the Escrow Fund. Any dividends, distributions or earnings on the Escrow Shares held in the Escrow Fund shall be deposited by the Escrow Agent in an interest bearing bank account established by the Escrow Agent for such purpose. The Escrow Agent shall comply with the procedures pertaining to the Escrow Fund and any disputes related thereto as set forth in <u>Article 8</u> below. Voting rights to the Escrow Shares shall be exercisable by or on behalf of the Model Shareholders in proportion to the respective allocations of the Escrow Shares. On the thirtieth (30th) day following the filing of E Com s first Annual Report on Form 10-K with the SEC following the Closing that includes audited financial statements that reflect at least

3

Table of Contents

six (6) months of the operations of the Surviving Entity, or if such day shall not be a Business Day on the next succeeding Business Day (the Escrow Release Date), the Escrow Agent will deliver the remaining shares of E Com Common Stock in the Escrow Fund, if any, and any dividends, distributions or earnings with respect to such Escrow Shares then held in the Escrow Fund and accrued interest thereon, to the Model Shareholders pro rata based upon their respective allocations of the Escrow Shares; provided, however, that subject to and in accordance with the terms of Article 8, the Escrow Agent shall withhold from delivery of the Escrow Fund such number of Escrow Shares as shall have a value, based upon the Consideration Per Share Price, equal to any amounts then in dispute relating to indemnification obligations arising under this Agreement, provided further that the withheld Escrow Shares, to the extent not applied in satisfaction of indemnification obligations pursuant to Article 8, shall be delivered to the Model Shareholders pro rata based upon their respective allocations of the Escrow Shares upon resolution of such dispute.

ARTICLE 3 REPRESENTATIONS AND WARRANTIES OF E COM

E Com represents and warrants to Model that, except as set forth in the disclosure letter delivered by E Com to Model simultaneously with the execution of this Agreement (the <u>E Com Disclosure Schedule</u>):

3.1 Organization.

- (a) E Com is a corporation duly organized, validly existing and in good standing under the laws of the State of Florida and has the requisite corporate power and authority necessary to own or lease all of its properties and assets and to carry on its business as it is now being conducted. E Com is duly licensed or qualified to do business and is in good standing in each jurisdiction in which the nature of the business conducted by it or the character or location of the properties and assets owned or leased by it makes such licensing or qualification necessary, except where the failure to be so licensed, qualified or in good standing would not reasonably be expected to have a Material Adverse Effect on E Com and would not reasonably be expected to prevent or delay E Com s performance of its obligations under this Agreement.
- (b) Each of E Com s Subsidiaries, including without limitation Acquisition Sub, listed on Section 3.1(b) of the E Com Disclosure Schedule is duly organized, validly existing and in good standing under the laws of the jurisdiction of its organization and has the requisite power and authority to own or lease all of its properties and assets and to carry on its business as it is now being conducted. Acquisition Sub has not filed an election under Treasury Regulation 301.7701-3 to be classified other than as provided in Treasury Regulation Section 301.7701-3(b) or any similar provision of state, local or foreign law, and shall not make such election prior to the Effective Time. Acquisition Sub has never had greater than one owner at any time during its existence, or had any owner other than its current owner, and agrees that such owner shall not be changed or added to prior to the Effective Time. E Com has made available to Model a complete and correct copy of the certificates of incorporation or formation, and by-laws or limited liability company agreements (or comparable documents), each as amended to date, of each such Subsidiary. Each Subsidiary is duly licensed or qualified to do business and is in good standing in each jurisdiction in which the nature of the business conducted by it or the character or location of the properties and assets owned or leased by it makes such licensing or qualification necessary, except where the failure to be so licensed, qualified or in good standing would not reasonably be expected to have a Material Adverse Effect with respect to E Com. Other than the Subsidiaries listed on Section 3.1(b) of the E Com Disclosure Schedule, E Com does not have any Subsidiaries, and does not own, directly or indirectly, any equity securities or other interests or capital stock of any Person.

3.2 Capitalization.

The authorized capital stock of E Com consists of 6,250,000 shares, \$0.01 par value, of E Com Common Stock and 1,000,000 shares, \$0.10 par value, of preferred stock of E Com. At the close of business on December 20, 2007, there were 3,059,041 shares of E Com Common Stock and no shares of preferred stock of E Com issued and outstanding, and 898,249 shares were held by E Com in its treasury. As of December 20, 2007,

4

Table of Contents

there were 144,696 shares of E Com Common Stock reserved for issuance upon the exercise of outstanding options under the Stock Plans and 444,445 shares of E Com Common Stock reserved for issuance upon the conversion of the Company s \$5,000,000 Subordinated Promissory Note dated December 9, 2004 payable to the order to Stephen Nussdorf and Glenn Nussdorf. For purposes of this Agreement, Stock Plans shall mean the following plans of E Com: the 2000 Stock Option Plan, the Directors 2000 Stock Option Plan, and any other stock option plan, program or arrangement providing for the issuance or grant of any other interest in respect of the capital stock or equity of E Com. All E Com capital stock, voting securities and equity interests have been duly authorized and validly issued and are fully paid, nonassessable and free of preemptive rights. Except as set forth above, E Com does not have outstanding any shares of its capital stock, voting securities or equity interests or any options, warrants, convertible or exchangeable securities or rights that may result in the issuance of any shares of capital stock, voting securities or equity interests of E Com, including, without limitation, any right to purchase E Com Common Stock, phantom stock rights, stock appreciation right or stock based performance units. Section 3.2(a) of the E Com Disclosure Schedule includes a list of all outstanding options and warrants to purchase E Com Common Stock, the holders thereof and the related exercise prices thereof.

3.3 Authority.

- (a) Each of E Com and Acquisition Sub has all necessary corporate, in the case of E Com, and limited liability company, in the case of Acquisition Sub, power and authority to execute and deliver this Agreement, the Warrants and the Registration Rights Agreement to the extent it is a party thereto and, subject, in the case of E Com, to obtaining E Com Shareholder Approval. The execution, delivery and performance by each of E Com and Acquisition Sub of this Agreement, the Warrants and the Registration Rights Agreement to the extent it is a party thereto, and the consummation by each of the Transactions, have been duly authorized and approved by the Board of Directors of E Com, the Board of Managers of Acquisition Sub and the sole member of Acquisition Sub.
- (b) The Special Committee, at a meeting duly called and held, has unanimously approved the Transactions and recommended the adoption of this Agreement to the Board of Directors of E Com, subject to the right of the Special Committee to withdraw, modify or amend such recommendation if the Special Committee determines, in good faith, that failure to take such action would be reasonably likely to result in a breach of its fiduciary duties to E Com s shareholders under applicable Law.
- (c) E Com s Board of Directors, at a meeting duly called and held, has duly adopted resolutions: (i) determining that the terms of the Merger are fair from a financial point of view to the holders of E Com Common Stock (other than shareholders of E Com who own, or whose Affiliates own, securities of Model), (ii) adopting this Agreement and declaring advisable this Agreement and the Transactions and (iii) recommending that the shareholders of E Com approve the issuance of shares of E Com Common Stock and the Warrants pursuant to the Merger and the issuance of E Com Common Stock upon exercise of the Warrants, subject to the right of the Board of Directors of E Com to withdraw, modify or amend such recommendation to the extent that the Board of Directors of E Com determines, in good faith, that failure to take such action would be reasonably likely to result in a breach of its fiduciary duties to E Com s shareholders under applicable Law.
- (d) Except for the E Com Shareholder Approval in the case of E Com, no corporate action not yet taken on the part of E Com or Acquisition Sub is necessary to authorize the execution, delivery and performance by E Com or Acquisition Sub of this Agreement, the Warrants and the Registration Rights Agreement to the extent it is a party thereto and the consummation by each of the Transactions. This Agreement has been duly executed and delivered by each of E Com and Acquisition Sub and constitutes and, following their execution and delivery by the parties thereto, each Warrant and Registration Rights Agreement will each constitute, a valid and binding obligation of E Com and Acquisition Sub, to the extent it is a party thereto, enforceable against E Com or Acquisition Sub, as applicable, in accordance with its terms, except that such enforceability may be limited by applicable bankruptcy, insolvency, fraudulent transfer, moratorium or other similar laws of general application affecting or relating to the enforcement of creditors rights generally (the Bankruptcy Exception).

5

Table of Contents

- 3.4 Opinion of Financial Advisor. The Special Committee has received the opinion of Financo, Inc., dated the date of this Agreement, to the effect that, as of such date, and subject to the various assumptions and qualifications set forth therein, the consideration to be paid by E Com in the Merger is fair to the shareholders of E Com (other than shareholders of E Com who own or whose Affiliates own securities of Model) from a financial point of view.
- 3.5 <u>Finders or Brokers</u>. Except for Financo, Inc., no broker, investment banker, financial advisor or other Person is entitled to any broker s, finder s, financial advisor s or other similar fee or commission in connection with the Transactions based upon arrangements made by or on behalf of E Com or any of its Subsidiaries.

ARTICLE 4 REPRESENTATIONS AND WARRANTIES OF MODEL

Model represents and warrants to E Com and Acquisition Sub that, except as set forth in the disclosure letter delivered by Model to E Com and Acquisition Sub simultaneously with the execution of this Agreement (the <u>Model Disclosure Schedu</u>le):

4.1 Organization.

- (a) Model is a corporation duly organized, validly existing and in good standing under the laws of the State of New York and has the requisite corporate power and authority necessary to own or lease all of its properties and assets and to carry on its business as it is now being conducted. Model has made available to E Com a complete and correct copy of its certificate of incorporation and by-laws as amended to date (the <u>Model Charter Documents</u>). Model is duly licensed or qualified to do business and is in good standing in each jurisdiction in which the nature of the business conducted by it or the character or location of the properties and assets owned or leased by it makes such licensing or qualification necessary, except where the failure to be so licensed, qualified or in good standing would not reasonably be expected to have a Material Adverse Effect with respect to Model and would not reasonably be expected to prevent or delay Model s performance of its obligations under this Agreement.
- (b) <u>Section 4.1(b)</u> of the Model Disclosure Schedule sets forth a true and complete list of all of Model s Subsidiaries. Each of Model s Subsidiaries is duly organized, validly existing and in good standing under the laws of the jurisdiction of its organization and has the requisite power and authority to own or lease all of its properties and assets and to carry on its business as it is now being conducted. Each such Model Subsidiary is duly licensed or qualified to do business and is in good standing in each jurisdiction in which the nature of the business conducted by it or the character or location of the properties and assets owned or leased by it makes such licensing or qualification necessary, except where the failure to be so licensed, qualified or in good standing would not reasonably be expected to have a Material Adverse Effect with respect to Model. Other than the Subsidiaries listed on <u>Section 4.1(b)</u> of the Model Disclosure Schedule, Model does not have any Subsidiaries, and does not own, directly or indirectly, any equity securities or other interests or capital stock of any Person.

4.2 Capitalization.

The authorized capital stock of Model consists of 200 shares of Model Common Stock. There are 96.90 shares of Model Common Stock issued and outstanding, and no shares were held by Model in its treasury. Except therefor, Model does not have any outstanding shares of capital stock, voting securities or equity interests or any options, warrants, convertible or exchangeable securities or rights that may result in the issuance of any shares of capital stock, voting securities or equity interests of Model, including, without limitation, any right to purchase Model Common Stock, phantom stock rights, stock appreciation right or stock based performance units. There are no agreements which would obligate Model (whether by the terms of the creating instrument or upon the demand or request of the holder thereof) to issue or repurchase any of its capital stock.

6

Table of Contents

4.3 Authority.

- (a) Model has all necessary corporate power and authority to execute and deliver this Agreement and to perform its obligations hereunder and to consummate the Transactions. This Agreement has been duly and unanimously adopted by Model s Board of Directors. The execution, delivery and performance by Model of this Agreement, and the consummation by it of the Transactions, have been duly authorized and approved by its Board of Directors.
- (b) Model s Board of Directors, at a meeting duly called and held, has duly adopted resolutions (i) adopting this Agreement and authorizing the Transactions and (iii) recommending that the Model Shareholders approve this Agreement.
- (c) This Agreement has been duly adopted and authorized by the unanimous vote or written consent of the holders of all of the outstanding shares of Model Common Stock, which is the only vote or approval of the holders of any class or series of capital stock of Model or any of its Subsidiaries which is necessary to adopt and authorize this Agreement and approve and authorize the Transactions.
- (d) No other corporate action on the part of Model is necessary to authorize the execution, delivery and performance by Model of this Agreement and the consummation of the Transactions. This Agreement has been duly executed and delivered by Model and constitutes a valid and binding obligation of Model, enforceable against it in accordance with its terms, subject to the Bankruptcy Exception.

4.4 Consents and Approvals; No Violations.

- (a) Except for (i) the filing of the Certificates of Merger with the Department of State of the State of New York and the Secretary of State of the State of Delaware pursuant to the BCL and the DLLCA, respectively, and (ii) applicable requirements of the Hart-Scott-Rodino Act (the <u>HSR Act</u>) and similar applicable competition, antitrust or similar Laws in any relevant foreign country or jurisdiction, no consents or approvals of, or filings, declarations or registrations with, any Governmental Entity are necessary for the consummation by Model of the Transactions, other than such other consents, approvals, filings, declarations or registrations that, if not obtained, made or given, would not reasonably be expected to have a Material Adverse Effect with respect to Model.
- (b) Neither the execution and delivery of this Agreement by Model, nor the consummation by Model of the Transactions, nor compliance by Model with any of the terms or provisions hereof, will (i) conflict with or violate any provision of Model s certificate of incorporation or by-laws, each as amended, or (ii) assuming that the authorizations, consents and approvals referred to in Section 4.4(a) of the Model Disclosure Schedule are obtained and the filings referred to in Section 4.4(a) of the Model Disclosure Schedule are made, violate any Law or any judgment, writ or injunction of any Governmental Entity applicable to Model or any of its Subsidiaries, or (iii) assuming that the notices and consents set forth on Section 4.4(b) of the Model Disclosure Schedule are made or obtained, violate, conflict with or constitute a default (or an event, condition or circumstance which, with notice or lapse of time, would become a default) under, or give any rights of termination or cancellation of, or accelerate the performance required by, or maturity of, or result in the creation of any encumbrance on any assets of Model or its Subsidiaries pursuant to the terms, conditions or provisions of any note, bond, mortgage, indenture, deed of trust, license, lease, contract, Permit, franchise, concession or other agreement to which Model or any of its Subsidiaries is a party, or by which any of their assets are bound, except, with respect to clauses (ii) and (iii), for violations, conflicts or defaults or other occurrences which would not have a Material Adverse Effect on Model.

4.5 Undisclosed Liabilities, Etc.

(a) The audited consolidated financial statements of Model as of and for the years ended October 31, 2004, 2005 and 2006 and unaudited consolidated financial statements of Model as of and for the eleven months ended September 30, 2007 (the <u>Model Financial Statements</u>) have been prepared in accordance with GAAP, and

7

Table of Contents

fairly present in all material respects the consolidated financial position of Model and its consolidated Subsidiaries as of the dates thereof and the consolidated results of their operations and cash flows for the periods then ended (subject, in the case of unaudited quarterly statements, to normal year-end audit adjustments).

- (b) Neither Model nor any of its Subsidiaries has any material liabilities of any nature, whether accrued, absolute, contingent, direct, indirect or otherwise, which, if known, would be required to be reflected or reserved against on a consolidated balance sheet of Model prepared in accordance with GAAP or the notes thereto, except liabilities (i) reflected or reserved against on the balance sheet of Model and its Subsidiaries as of September 30, 2007 (the Model Balance Sheet Date) (including the notes thereto, the Model Balance Sheet), (ii) incurred after the Model Balance Sheet Date in the ordinary course of business consistent with past practice, or (iii) reflected on the Model Disclosure Schedule. The reserves, if any, reflected on the Model Balance Sheet and the Model Financial Statements are adequate, appropriate and reasonable for their purposes, including without limitation, litigation reserves and product warranty reserves, if any.
- (c) The accounts receivable shown on the Model Balance Sheet arose from bona fide transactions in the ordinary course of business. The accounts receivable of Model and its Subsidiaries arising after the Model Balance Sheet Date and prior to the Closing Date arose, or will arise, from bona fide transactions in the ordinary course of business. Except as reserved on the Model Balance Sheet, none of such accounts receivable is subject to any claim of offset or recoupment or counterclaim, and Model has no Knowledge of any specific facts that would be likely to give rise to any such claim. Except as reserved on the Model Balance Sheet, no amount of such accounts receivable is contingent upon the performance by Model or any of its Subsidiaries of any obligation and no agreement for deduction or discount or any type of credit has been made with respect to any such accounts receivable. Except as reserved on the balance sheet, to the Knowledge of Model, no account debtor of Model or any of its Subsidiaries has proposed any discount or reduction with respect to such account debtor s accounts receivable.
- (d) All inventories of Model and its Subsidiaries consist of a quality and quantity usable and salable in the ordinary course of business, except for obsolete items and items of below standard quality which have been written off or written down to net realizable value in the Model Balance Sheet. All inventories (other than as so written off) of Model and it Subsidiaries have been included in the Model Balance Sheet at the lower of cost or market (net realizable value) on an average cost basis. The quantities of inventories maintained by Model and its Subsidiaries are reasonable and warranted, and not excessive, under the present circumstances.
- 4.6 <u>Absence of Certain Changes or Events</u>. Except (i) as disclosed in <u>Section 4.6</u> of the Model Disclosure Schedule, (ii) for the Transactions or (iii) as permitted by this Agreement, since the Model Balance Sheet Date, Model and its Subsidiaries have carried on and operated their respective businesses in all material respects in the ordinary course of business consistent with past practice, and there has not occurred any:
- (a) effect, event or change that would reasonably be expected to have a Material Adverse Effect with respect to Model;
- (b) sale, transfer, distribution, abandonment or other disposal of, or mortgage, pledge or imposition of any encumbrance on any property (including real estate) or other assets of Model and its Subsidiaries that are material to Model and its Subsidiaries taken as a whole, except for (i) sales of inventory in the ordinary course of business consistent with past practice or (ii) dispositions of obsolete or worthless assets in the ordinary course of business consistent with past practice;
- (c) declaration, setting aside or payment of any dividend or other distribution (whether in cash, stock or property) with respect to any class of capital stock of Model or any of its Subsidiaries (other than dividends or other distributions paid to Model by its Subsidiaries), or any repurchase, redemption or other acquisition by Model or any of its Subsidiaries of any capital stock or other securities of, or other ownership interest in, Model or any of its Subsidiaries;

8

Table of Contents

- (d) split, combination or reclassification of any capital stock of Model;
- (e) change in financial or tax accounting methods, principles or practices by Model or its Subsidiaries, except insofar as may have been required by a change in GAAP or applicable Law;
- (f) Tax election or the revocation or change to any Tax election or the settlement or compromise of any Tax liability by Model or any of its Subsidiaries:
- (g) granting by Model or any of its Subsidiaries to any director or employee of any award or bonus or other compensation, except to the extent accrued on the Model Balance Sheet;
- (h) granting by Model or any of its Subsidiaries to any employee of any increase in (or acceleration of vesting or payment of) severance or termination pay, except as required under any employment, severance or termination agreements set forth on Section 4.6 of the Model Disclosure Schedule;
- (i) entry by Model or any of its Subsidiaries into any (or amendment in any material respect of any existing) employment, severance or termination agreement with any employee of Model;
- (j) entry by Model or any of its Subsidiaries into any consulting agreement with any Person;
- (k) acceleration of the vesting of any option issued by Model or any of its Subsidiaries to acquire capital stock of Model or any of its Subsidiaries, except as contemplated in this Agreement;
- (l) capital expenditures by Model or any of its Subsidiaries, taken as a whole, in excess of 1,500,000 individually or \$1,500,000 in the aggregate;
- (m) establishment, amendment or termination of any collective bargaining agreement to which Model or any of its Subsidiaries is a party;
- (n) acceleration of accounts receivable other than in the ordinary course of business consistent with past practice;
- (o) general delay of payments by Model or any of its Subsidiaries to vendors or others to whom Model owes payments (except for disputed payments);
- (p) making of loans to any Persons other than in the ordinary course of business consistent with past practice or loans to any of its Subsidiaries;
- (q) settlement of any action, complaint, claim, petition, investigation, suit or other proceeding, whether civil or criminal, in law or in equity, before any arbitrator or Governmental Entity, to which Model or any of its Subsidiaries are a party or any of their respective properties is subject;
- (r) acceleration of the payment, right to payment or vesting under any benefit, retirement, profit sharing or deferred compensation plan or other compensation arrangement of Model or any of its Subsidiaries;
- (s) incurrence of any Model Indebtedness (other than the Affiliate Debt) in an amount in excess of \$500,000 in the aggregate;
- (t) granting by Model or any of its Subsidiaries of any license or sublicense of any rights under or with respect to any Intellectual Property that is material to Model and its Subsidiaries;
- (u) establishment, amendment or contribution to any pension, retirement, profit sharing, stock bonus, multi-employer or other benefit plan covering any of the current or former employees, officers, directors or consultants of Model or any of its Subsidiaries, except as required by Law or pursuant to such benefit plan;

Table of Contents

267

Table of Contents

- (v) establishment of, or entering into, or financial commitment or contribution to, or amendment to the terms of, or termination of, any joint venture, partnership agreement or similar arrangement or any contract involving a sharing of profits, losses, business or opportunities with any other Person; or
- (w) any agreement or commitment by Model or any of its Subsidiaries to do any of the foregoing.
- 4.7 <u>Legal Proceedings</u>. Except as set forth in <u>Section 4.7</u> of the Model Disclosure Schedule, there is (and in the past five years there has been) neither any pending or, to the Knowledge of Model, threatened, legal or administrative proceeding, claim (including any claim for indemnification under any agreement), suit or action by or against Model or any of its Subsidiaries, any of their respective assets or any of their respective officers or directors in their capacity as such nor any material Order imposed upon Model or any of its Subsidiaries by any Governmental Entity. Model has not received, as of the date hereof, any written notice of any claim that seeks to prevent the Transactions and, to the Knowledge of Model, no such claim has been threatened. No Governmental Entity has challenged in writing the right of Model and its Subsidiaries to conduct their business as currently conducted. Except as set forth in <u>Section 4.7</u> of the Model Disclosure Schedule, all the matters listed on such schedule are covered by insurance, and could not be reasonably expected, individually or in the aggregate, to have a Material Adverse Effect with respect to Model.

4.8 Compliance with Applicable Law; Permits.

- (a) Model and its Subsidiaries are (and during the past five years have been) in compliance in all material respects with all Laws applicable to Model or any of its Subsidiaries. Model and its Subsidiaries hold all material Permits necessary for the lawful conduct of their respective businesses. The operation of the businesses of Model and each of its Subsidiaries and their respective properties and assets are in compliance with all Permits in all material respects. No suspension, cancellation or termination of any such Permit is threatened or imminent other than expirations of Permits requiring renewal in the ordinary course of business and Model knows of no basis for such suspension, cancellation or termination nor basis upon which any expiring Permit will not be renewed. No approval is required under any Permit in connection with the Transactions and the Transactions will not violate any Permit or cause any Permit to terminate or give any Person the right to terminate such Permit.
- (b) Model and its Subsidiaries: (i) are in full compliance with the federal Food Drug and Cosmetic Act and all regulations issued thereunder, (ii) have no reasonable basis to believe that any ingredient used in their products is unsafe when used for its intended purpose, (iii) have conducted appropriate safety testing on each formulation that they produce to assure that such formulations are safe for their intended purpose, (iv) have not received any notification of an adverse event arising from the use of any of their products and/or formulations, (v) have not received any claims or notification of any personal injuries arising from the use of their products or formulations during the past five (5) years and (vi) do not use any ingredients in any of their products or formulations which would be considered unsafe by the Food and Drug Administration.
- (c) To the Knowledge of Model, neither Model nor any of its Subsidiaries has mislabeled any of its products or mischaracterized the natural, organic or other certified nature thereof.
- 4.9 <u>Tax Matters</u>. Except as set forth on Section 4.9 of the Model Disclosure Schedule: (a) each of Model and its Subsidiaries has timely filed, or has caused to be timely filed on its behalf (taking into account any validly obtained extension of time within which to file), all Tax Returns and other Tax Returns required to be filed by it and all such filed Tax Returns are correct and complete in all respects; (b) all Taxes due and payable by Model or any of its Subsidiaries have been fully and timely paid; (c) neither Model nor any of its Subsidiaries currently is the beneficiary of any extension of time within which to file any Tax Return; (d) no claim has ever been made by an authority in a jurisdiction where Model or any of its Subsidiaries does not file a Tax Return that such entity is or may be subject to taxation by that jurisdiction; (e) there are no Liens for Taxes (other than Taxes not yet due and payable) upon any of the assets of Model or its Subsidiaries; (f) each of Model and its Subsidiaries has withheld and paid all Taxes required to have been withheld and paid in connection with any amounts paid or

10

owing to any employee, independent contractor, creditor, stockholder, or other third party; (g) no foreign, federal, state, or local tax audits or administrative or judicial Tax proceedings are pending or being conducted with respect to Model or its Subsidiaries; (h) none of Model or its Subsidiaries has received from any foreign, federal, state, or local taxing authority (including jurisdictions where such entities have not filed Tax Returns) any (i) notice indicating an intent to open an audit or other review, (ii) request for information related to Tax matters, or (iii) notice of deficiency or proposed adjustment for any amount of Tax proposed, asserted, or assessed by any taxing authority against any of Model or its Subsidiaries; (i) Section 4.9 of the Model Disclosure Schedule lists any audit of Model and its Subsidiaries that has been conducted by a Taxing authority within the last three years,; (k) no deficiency or adjustment with respect to Taxes has been proposed, asserted or assessed against Model or any of its Subsidiaries; (j) adequate reserves for Taxes not yet due have been established on the Model Financial Statements in accordance with GAAP; (k) there are no outstanding waivers or agreements extending the statute of limitations for any period with respect to any Tax to which Model or any of its Subsidiaries may be subject; (1) neither Model nor any of its Subsidiaries is party to any agreement, understanding or arrangement (with any Person other than Model and/or any of its Subsidiaries) relating to allocating or sharing any Taxes, and; (m) neither Model nor any of its Subsidiaries is a party to any agreement, contract, arrangement or plan that has resulted or could result, separately or in the aggregate, in the payment of (i) any excess parachute payment within the meaning of Code Section 280G (or any corresponding provision of state, local or foreign Tax law) and (ii) any amount that will not be fully deductible as a result of Code Section 162(m) (or any corresponding provision of state, local or foreign Tax law); (r) neither Model nor any of its Subsidiaries has been a United States real property holding corporation within the meaning of Code Section 897(c)(2) during the applicable period specified in Code Section 897(c)(1)(A)(ii); (s) each of Model and its Subsidiaries has disclosed on its federal income Tax Returns all positions taken therein that could give rise to a substantial understatement of federal income Tax within the meaning of Code Section 6662; (t) none of Model or its Subsidiaries (A) has been a member of an affiliated group filing a consolidated federal income Tax Return (other than a group, the common parent of which was Model) or (B) has any liability for the Taxes of any Person (other than Model and its current Subsidiaries) under Treasury Regulation Section 1.1502-6 (or any similar provision of state, local, or foreign law), as a transferee or successor, by contract, or otherwise; (u) there are currently no limitations on the net operating losses, net capital losses, or credits of any of Model or its Subsidiaries, including under Sections 382, 383, 384 or 269 of the Code; (v) the unpaid Taxes of Model and its Subsidiaries (A) did not, as of the most recent fiscal month end, exceed the reserve for liability for Taxes (rather than any reserve for deferred Taxes established to reflect timing differences between book and Tax income) set forth on the face of the Model Balance Sheet (rather than in any notes thereto) and (B) do not exceed that reserve as adjusted for the passage of time through the Closing Date in accordance with the past custom and practice of the Model and its Subsidiaries in filing their Tax Returns; (w) since the date of the Model Balance Sheet, none of Model or its Subsidiaries has incurred any liability for Taxes arising from unusual gains or losses outside the ordinary course of business consistent with past custom and practice; (x) none of Model or its Subsidiaries has distributed stock of another Person, or has had its stock distributed by another Person, in a transaction that was purported or intended to be governed in whole or in part by Code Section 355.

4.10 Employee Benefits.

(a) Section 4.10(a) of the Model Disclosure Schedule is a true, complete and correct list of (a) each employee benefit plan , as defined in Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended (_ERISA_), (b) each employment, bonus, deferred or incentive compensation, profit sharing, retirement, stock option, stock purchase or other equity compensation, vacation, health, life or other insurance, Internal Revenue Code Section 125 cafeteria plan or flexible benefit arrangement, sick program or fringe benefit plan, arrangement, agreement or program and (c) any other employee plan or agreement, employment agreement, any severance plan, agreement, program or policy, or consulting agreement providing for on-going services that (i) benefits any current or former employee, officer, director, consultant or independent contractor and (ii) is sponsored, maintained or contributed to or required to be contributed to by Model or any ERISA Affiliate (each a _Company Benefit Plan_). Model has furnished or made available to E Com with respect to each Company Benefit Plan, if applicable, (a) each Company Benefit Plan, (b) Form 5500 as filed with the IRS for the three most recent years with any required audited financial statements, (c) the most recent summary plan description

11

Table of Contents

and any subsequent summaries of material modifications, (d) each trust agreement, plan contracts with service providers or insurers and insurance or group annuity contract providing benefits for participants or liability insurance for fiduciaries and other parties in interest or bonding, and (e) the most recent IRS determination letter for all plans qualified under Internal Revenue Code Section 401(a). <u>ERISA Affiliate</u> shall mean Model or any Subsidiary of Model, or any entity, which together with Model would be deemed a single employer within the meaning of Sections 414(b), (c) or (m) of the Internal Revenue Code (the <u>Code</u>) or Section 4001 of ERISA.

- (b) Each Company Benefit Plan has been administered in all material respects in accordance with its terms. Model, its Subsidiaries and each Company Benefit Plan is in compliance in all material respects with the applicable provisions of ERISA, the Code and all other applicable Laws. No individual who has performed services for Model or any Subsidiary of Model has been improperly excluded from participation in any Company Benefit Plan. No Company Benefit Plan or any fiduciary of any such Company Benefit Plan has (i) engaged in any transaction prohibited by ERISA or the Code, (ii) breached any fiduciary duty owed by it with respect to the Plans, or (iii) engaged in any transaction as a result of which Model or any Subsidiary of Model would be subject to any liability pursuant to Sections 406 or 409 of ERISA or to either a civil penalty assessed pursuant to Section 502(i) or Section 502(i) of ERISA or a Tax imposed pursuant to Section 4975 of the Code. Model and each ERISA Affiliate has complied in all material respects with the continuation coverage requirements of Section 4980B of the Code and Sections 601-608 of ERISA applicable to Model or such ERISA Affiliate. No actions, audits, investigations, suits or claims with respect to a Company Benefit Plan (other than routine claims for benefits) are pending or, to the Knowledge of Model, threatened, which could result in or subject Model or any Subsidiary of Model to any liability.
- (c) Each Company Benefit Plan which is intended to be qualified within the meaning of Section 401(a) of the Code has received from the Internal Revenue Service (the <u>IRS</u>) a favorable determination letter that remains in effect on the date hereof and no event has occurred since the date of the most recent determination letter or application therefor relating to any such Company Benefit Plan that is reasonably likely to adversely affect the qualification of such Company Benefit Plan.
- (d) All contributions (including all employer contributions and employee salary reduction contributions), premiums and benefit payments required by and due from Model and any Subsidiary of Model under or in connection with the terms of each Company Benefit Plan have been made within the time periods prescribed by the Company Benefit Plan, ERISA and the Code.
- (e) No Company Benefit Plan is subject to Title IV of ERISA, the minimum funding requirements of Section 302 of ERISA or Section 412 of the Code nor provides for medical or life insurance benefits to retired or former employees of Model or any Subsidiary of Model (other than as required under Code §4980B, ERISA or similar state law). Neither Model nor any ERISA Affiliate is a participating or contributing employer in any multiemployer plan (as defined in Section 3(37) of ERISA) subject to Title IV of ERISA with respect to employees of Model or any Subsidiary of Model and neither Model nor any of its ERISA Affiliates has any outstanding withdrawal liability in connection with a complete or partial withdrawal from any multiemployer plan.
- (f) Neither the execution and delivery of this Agreement nor the consummation of the Transactions will, either alone or in combination with any other event, (i) result in any payment becoming due, or increase the amount of compensation due, to any current or former employee or director of Model or any Subsidiary of Model, (ii) increase any benefits payable under any Company Benefit Plan, or (iii) result in any acceleration of the time of payment or vesting of any such compensation or benefits. Further, neither Model nor any Subsidiary of Model has announced any type of plan or binding commitment to create any additional Company Benefit Plan, to enter into any agreement with any current or former employee or director, or to amend or modify any existing Company Benefit Plan or agreement with any current or former employee or director.

12

Table of Contents

4.11 Labor Matters.

- (a) Except as set forth in Section 4.11(a) of the Model Disclosure Schedule, neither Model nor any of its Subsidiaries is a party to or otherwise bound by any labor or collective bargaining agreement, and there are no labor or collective bargaining agreements that pertain to their respective employees. Except as set forth in Section 4.11(a) of the Model Disclosure Schedule, no labor organization or group of employees of Model or any of its Subsidiaries has made a pending demand for recognition, and, within the preceding six years, there has been no representation or certification proceeding, or petition seeking a representation proceeding, pending or, to the Knowledge of Model, threatened to be brought or filed with the National Labor Relations Board or any other labor relations tribunal or Governmental Entity. Except as set forth in Section 4.11(a) of the Model Disclosure Schedule, within the preceding six years, there has been no organizing activity involving Model or any of its Subsidiaries pending or, to the Knowledge of Model, threatened by any labor organization or group of employees of Model or any of its Subsidiaries.
- (b) There are no pending or, to the Knowledge of Model, threatened investigations of or relating to Model or any of its Subsidiaries by any Governmental Entity responsible for the enforcement of labor or employment Laws.
- (c) There has never been any strike, work stoppage, slowdown, picketing, lockout, walkout, other job action, arbitration, grievance, unfair labor practice charge or complaint or other labor dispute involving Model or any of its Subsidiaries or to the Knowledge of Model, threatened against Model or any of its Subsidiaries, and to the Knowledge of Model, there are no facts or circumstances which could form the basis for any of the foregoing.
- (d) Neither Model nor any of its Subsidiaries has implemented any plant closing or mass layoff of employees that would reasonably be expected to require notification under the Worker Adjustment and Retraining Notification Act (as amended, <u>WARN</u>) or any similar state or local Law, no such plant closing or mass layoff will be implemented before the Closing Date without advance notification to and approval of Model, and there has been no employment loss, as defined by WARN, within the ninety (90) days prior to the Closing Date.
- (e) Model and each of its Subsidiaries are in compliance with all Laws relating to the employment of labor, including all such Laws relating to wages, hours, collective bargaining, discrimination, civil rights, occupational safety and health, workers compensation and the collection and payment of withholding and/or social security Taxes and other Taxes. There are no actions against Model or any of its Subsidiaries pending or, to the Knowledge of Model, threatened to be brought or filed with any public authority, Governmental Entity, arbitrator or court based on, arising out of, in connection with, or otherwise relating to the employment or termination of employment or services by Model or any of its Subsidiaries of any individual, including but not limited to the Civil Rights laws, Americans with Disabilities Act, Age Discrimination in Employment Act (as amended by the Older Workers Benefit Protection Act), Pregnancy Discrimination Act, Equal Pay Act, Fair Labor Standards Act, WARN, and Family and Medical Leave Act, and, to the Knowledge of Model, there are no facts or circumstances which could form the basis for any of the foregoing.

4.12 Environmental Matters.

- (a) For purposes of this Section 4.12, the following definitions will apply:
- (i) <u>Environmental Claims</u> means any complaint, summons, citation, notice, directive, order, claim, litigation, investigation, notice of violation, judicial or administrative proceeding, judgment, letter or other communication from any Governmental Entity, department, bureau, office or other authority, or any third party involving violations of Environmental Laws from (i) any assets, properties or businesses of Model, any of its Subsidiaries or their respective predecessors in interest or (ii) from or onto any facilities which received Hazardous Materials generated by Model, any of its Subsidiaries or their respective predecessors in interest.

13

Table of Contents

- (ii) Environmental Liabilities means any monetary obligations, losses, liabilities (including strict liability), damages, punitive damages, consequential damages, treble damages, costs and expenses (including all reasonable out-of-pocket fees, disbursements and expenses of counsel, out-of-pocket expert and consulting fees and out-of-pocket costs for environmental site assessments, remedial investigation and feasibility studies), fines, penalties, sanctions and interest to the extent any of the foregoing are incurred as a result of any Environmental Claim filed by any Governmental Entity or any third party which relate to any violations of Environmental Laws, Remedial Actions, Releases or threatened Releases of Hazardous Materials from or onto (i) any property presently or formerly owned, leased or operated by Model, any of its Subsidiaries or their respective predecessors in interest, or (ii) any facility which received Hazardous Materials generated by Model, any of its Subsidiaries or their respective predecessors in interest.
- (b) Except as would not likely result in Model or any Subsidiary of Model incurring material Environmental Liabilities:
- (i) The operations of Model and its Subsidiaries are in compliance with Environmental Laws in all material respects;
- (ii) Model and its Subsidiaries have obtained and are in material compliance with all material and necessary permits or authorizations that are required under Environmental Laws to operate the facilities, assets and business of Model;
- (iii) There has been no Release by Model or any of its Subsidiaries or, to the Knowledge of Model, by any other Person at any of the properties currently owned, leased or operated by Model or any of its Subsidiaries, or, to the Knowledge of Model, at any property formerly owned, leased or operated by Model or any of its Subsidiaries or at any disposal or treatment facility which received Hazardous Materials generated by Model, any of its Subsidiaries nor any of their respective predecessors in interest which is reasonably likely to result in material Environmental Liabilities;
- (iv) No pending or unresolved Environmental Claims have been asserted against Model or any of its Subsidiaries or, to the Knowledge of Model, any predecessor in interest of Model or any of its Subsidiaries, nor does Model have notice of any threatened or pending Environmental Claim against Model, any of its Subsidiaries or any such predecessor in interest regarding any facilities that may have received Hazardous Materials generated by Model, any of its Subsidiaries nor any of their respective predecessors in interest which is reasonably likely to result in material Environmental Liabilities;
- (v) Except for Environmental Claims that have been resolved and satisfied solely by the payment of money which has been paid on or before the Model Balance Sheet Date or is reflected as a liability on the Model Balance Sheet, neither Model, any of its Subsidiaries nor any of their respective predecessors in interest, has transported or arranged for the treatment, storage or disposal of any Hazardous Materials to any off-site location that has resulted in an Environmental Claim or is reasonably likely to lead to any Environmental Claim to Model or any of its Subsidiaries under applicable Environmental Laws.
- (c) Model has made available to E Com true and complete copies of all material environmental reports, studies, and investigations regarding any material Environmental Liabilities of Model and its Subsidiaries and of any material environmental conditions at any properties owned or operated by Model, which are in the possession of Model.

4.13 Insurance.

(a) Section 4.13(a) of the Model Disclosure Schedule sets forth a list of all current insurance policies maintained by Model and its Subsidiaries. All such policies are in full force and effect, and neither Model nor any of its Subsidiaries is in default with respect to their material obligations under any such policy so as to cause a loss of coverage. Neither Model nor any of its Subsidiaries has received notice of cancellation or termination in respect of any such policy and knows of no basis for such cancellation, termination or loss of coverage. Such policies provide for insurance of Model and its Subsidiaries which is adequate and customary for businesses of similar type to Model and its Subsidiaries.

14

Table of Contents

- (b) There have been no gaps in coverage under such insurance policies of Model or its Subsidiaries as in effect from time to time during their existence. As of and after the Closing, Model and each of its Subsidiaries will have the right to continued coverage for occurrences prior to the Closing under such policies. The current and historical limits of liability under such policies have not been exhausted or impaired. All losses that have been incurred by Model and its Subsidiaries that are covered by such policies have been properly reported to their respective insurance carriers, and no reservation of rights letters have been issued by such carriers. Except as set forth on Section 4.13(a) of the Model Disclosure Schedule, there are no pending or potential premium audits or adjustments relating to the casualty insurance policies that currently cover Model or any of its Subsidiaries that are not reserved on the Model Balance Sheet.
- 4.14 <u>Finders or Brokers</u>. Except for Wachovia Capital Markets, LLC, no broker, investment banker, financial advisor or other Person is entitled to any broker s, finder s, financial advisor s or other similar fee or commission in connection with the Transactions based upon arrangements made by or on behalf of Model, any of its Subsidiaries or any of its Affiliates.

4.15 Contracts.

- (a) Section 4.15(a) of the Model Disclosure Schedule contains a complete and correct list of the following contracts and agreements (whether written or oral) to which Model or any of its Subsidiaries is a party (collectively, the Model Material Contracts): (i) all contracts (other than relating to purchases and sales of inventory in the ordinary course of business) which involve or are reasonably expected to involve aggregate payments or expenditures by or to Model or any of its Subsidiaries during the fiscal year ending October 31, 2008 in excess of \$250,000; (ii) all Model Indebtedness (other than guarantees by way of endorsement or negotiable instruments in the ordinary course of business); (iii) any contracts containing covenants not to compete (A) binding on Model or any of its Subsidiaries, (B) restricting other Persons for the benefit of Model or any of its Subsidiaries or (C) which otherwise restrict competition granted by Model or any of its Subsidiaries in favor of a third Person; (iv) contracts under which Model or any of its Subsidiaries is obligated to indemnify any Person other than agreements entered into in the ordinary course of business; (v) contracts to loan money or extend credit to any other Person, other than product warranties in the ordinary course of business; and (vi) joint venture, partnership agreements or similar arrangements or contracts involving a sharing of profits, losses, business or opportunities with any other Person.
- (b) With respect to the Model Material Contracts, (i) such Model Material Contracts are in full force and effect and are valid, binding and enforceable against Model or its Subsidiaries, as applicable, and, to the Knowledge of Model, each other party thereto, and (ii) Model and its Subsidiaries, and, to the Knowledge of Model, each other party thereto, have complied in all material respects with all respective covenants and provisions of the Model Material Contracts. Except as set forth on Section 4.15(b) of the Model Disclosure Schedule, none of the rights of Model or any of its Subsidiaries under any Model Material Contract will be subject to termination or modification (nor will any Person have the right to accelerate the performance of Model or such Subsidiary of Model under any Model Material Contract), and no notice to or consent or approval of any other Person is required under any Model Material Contract, as a result of the consummation of the Transactions.
- (c) Except to the extent set forth in Section 4.15(c) of the Model Disclosure Schedule, there are no transactions, agreements, arrangements or understandings between Model or any of its Subsidiaries, on the one hand, and Model s directors, officers, affiliates (other than wholly-owned Subsidiaries of Model) or other Persons, on the other hand.
- (d) Model and its Subsidiaries have good title or valid leasehold interests in all tangible assets that are material to the operation of their respective businesses.
- (e) Except as set forth in Section 4.15(e) of the Model Disclosure Schedule, there are neither any contractual nor, to the Knowledge of Model, any legal, regulatory or other restrictions that prohibit Model or any of its Subsidiaries from transferring cash (or cash equivalents) between accounts outside of and inside the United States or that limit Model s free use of such cash.

15

Table of Contents

(f) Section 4.15(f) of the Model Disclosure Schedule sets forth an itemized list, prepared in good faith, of Model s expenses incurred through October 31, 2007, and the total amount of additional expenses that Model estimates, as of the date hereof, to be incurred by it in connection with the Transactions contemplated by this Agreement, including, without limitation, the fees and expenses of Model s financial advisors and outside counsel and accountants.

4.16 Owned Real and Personal Property; Encumbrances.

- (a) Neither Model nor any of its Subsidiaries owns any real property and neither Model nor any of its Subsidiaries or any of their respective predecessors has ever owned any real property.
- (b) Except for Model Permitted Encumbrances, Model or its Subsidiaries, as the case may be, has good title to all of the material tangible personal property reflected as being owned by it on the Model Balance Sheet (except for personal property sold or otherwise disposed of since the Model Balance Sheet Date in the ordinary course of business consistent with past practices), which personal property, together with personal property leased by Model or its Subsidiaries described on Section 4.16 of the Model Disclosure Schedule, constitutes all personal property necessary for conducting the business of Model and its Subsidiaries as presently conducted. The personal property owned or leased by Model and its Subsidiaries, taken as a whole, is adequate and in a condition sufficient to permit Model and its Subsidiaries to conduct their respective businesses in all material respects in the same manner as it is being conducted as of the date of this Agreement, subject to ordinary wear and tear and routine maintenance. Model Permitted Encumbrances means (i) statutory liens for current Taxes or assessments not yet due or delinquent or the validity of which are being contested in good faith by appropriate proceedings (a list of all such proceedings as of the date of this Agreement being included on Section 4.9 of the Model Disclosure Schedule) and (ii) mechanics , carriers , workers , repairmen s and other similar liens arising or incurred in the ordinary course of business with respect to charges not yet due and payable.
- 4.17 Leased Real and Personal Property. Section 4.17 of the Model Disclosure Schedule sets forth a true and complete list of each lease, sublease or license (each a Model Material Lease) under which Model or any of its Subsidiaries is a lessee, lessor, sublessee, sublessor, licensee or licensor which (a) is a lease of real property or any interest in real property, or (b) is a lease of personal property which provides for payments of more than \$100,000 per year, has a term exceeding one year and may not be canceled upon 90 or fewer days notice without any liability, penalty or premium (other than a nominal cancellation fee or charge). With respect to the Model Material Leases, (i) each Model Material Lease is in full force and effect and is valid, binding and enforceable in accordance with its terms, (ii) all accrued and currently payable rents and other payments required by such Model Material Lease have been paid and (iii) Model and its Subsidiaries, and, to Model s Knowledge, each other party thereto have complied in all material respects with all respective covenants and provisions of the Model Material Lease. Model or one of its Subsidiaries, as applicable, enjoys peaceful and undisturbed possession under each Model Material Lease to which it is a lessee, sublessee or licensee and there is not, with respect to any Model Material Lease, any event of default, or event which with notice or lapse of time or both would constitute an event of default, existing on the part of Model or such Subsidiary or, to the Knowledge of Model, on the part of any other party thereto. Except as set forth on Section 4.19 of the Model Disclosure Schedule, none of the rights of Model or any of its Subsidiaries under any Model Material Lease will be subject to termination or modification (nor will any Person have the right to accelerate the performance of Model or such Subsidiary under any Model Material Lease as a result of the consummation of the Transactions.

4.18 Intellectual Property.

(a) <u>Section 4.18(a)</u> of the Model Disclosure Schedule contains a complete and accurate list of all (i) patented or registered Intellectual Property owned or, to the Knowledge of Model, used by Model and its Subsidiaries, (ii) pending patent applications and applications for other registrations of Intellectual Property filed by or on behalf of Model or any of its Subsidiaries, and (iii) to the Knowledge of Model, material unregistered Intellectual Property owned or used by Model and its Subsidiaries, other than shrink-wrapped or

16

Table of Contents

off-the-shelf software licenses licensed by Model and its Subsidiaries. All patents, patent applications, registered trademarks, trademark applications, trade names, registered service marks, service mark applications, logos, registered copyrights, copyright applications and domain names set forth on Section 4.18(a) of the Model Disclosure Schedule have been duly registered in, filed in, or issued by the United States Patent and Trademark Office, United States Register of Copyrights, Network Solutions, Inc. (or other authorized domain name registry) or the corresponding offices of any other country, state, or other jurisdiction to the extent set forth on Section 4.18(a) of the Model Disclosure Schedule, and have been properly maintained or renewed in accordance with all applicable provisions of Law and administrative regulations in the United States and in each such other country, state, or other jurisdiction.

- (b) <u>Section 4.18(b)</u> of the Model Disclosure Schedule contains a complete and accurate list of all agreements, licenses, royalty agreements, and other rights granted by Model and its Subsidiaries to any other Person with respect to any Intellectual Property set forth on <u>Section 4.18(a)</u> of the Model Disclosure Schedule and all licenses, agreements, and other rights granted by any third Person to Model and its Subsidiaries with respect to any Intellectual Property set forth on <u>Section 4.18(a)</u> of the Model Disclosure Schedule (other than shrink-wrapped or off-the-shelf software licenses licensed by Model and its Subsidiaries), in each case identifying the subject Intellectual Property. Except as set forth on <u>Section 4.18(b)</u> of the Model Disclosure Schedule, Model and its Subsidiaries have the sole and exclusive right to use the Intellectual Property and there are no licenses, sublicenses or agreements relating to the use by any other Person of any of such Intellectual Property now in effect. There is no outstanding and, to the Knowledge of Model, no threatened disputes or disagreements with respect to any such agreement, licenses, royalty agreements or other rights.
- (c) The Intellectual Property set forth on Section 4.18(a) of the Model Disclosure Schedule comprises all Intellectual Property necessary for the operation of the respective businesses of Model and its Subsidiaries as they are currently conducted. Model and its Subsidiaries, as applicable, own all right, title and interest in and to each item of Intellectual Property, free and clear of all encumbrances except for Model Permitted Encumbrances, and have the right to use without payment to any third party all of their respective Intellectual Property. Model and its Subsidiaries are not currently using and it will not be necessary for Model and its Subsidiaries to use any Intellectual Property of any of its present or former employees developed, invented or made prior to their employment by Model and its Subsidiaries except for any such Intellectual Property that have previously been assigned to Model and its Subsidiaries. No present or former employee of Model or any of its Subsidiaries and, except as set forth on Section 4.18(b) of the Model Disclosure Schedule, no other Person owns or has any proprietary, financial or other interest, direct or indirect, in whole or in part, in any Intellectual Property which Model or any of its Subsidiaries owns or uses in the conduct of their respective businesses as now or heretofore conducted. None of the management and salaried employees of Model and its Subsidiaries have signed agreements relating to noncompetition, nonsolicitation, confidentiality and/or assignment of inventions. Model and its Subsidiaries have taken all reasonable and appropriate steps to protect their trade secrets.
- (d) No loss or expiration of any Intellectual Property or related group of Intellectual Property owned or used by Model and its Subsidiaries has occurred which has had or would reasonably be expected to have a Material Adverse Effect with respect to Model, and no loss or expiration of any such Intellectual Property is threatened in writing, or, to Model s Knowledge, pending or reasonably foreseeable.
- (e) (i) There have been no written claims made against Model or any of its Subsidiaries asserting the invalidity, misuse or unenforceability of any of the Intellectual Property owned or used by Model or any of its Subsidiaries, (ii) Model and its Subsidiaries have not received any written notices of any infringement or misappropriation by, or conflict with, any third party with respect to any Intellectual Property (including any demand or request that Model or any of its Subsidiaries licenses any rights from a third party), (iii) the conduct of the business of Model and its Subsidiaries has not infringed, misappropriated or conflicted with and does not infringe, misappropriate or conflict with any Intellectual Property of other Persons and (iv) to the Knowledge of Model, no Person has infringed upon or is currently infringing upon any material Intellectual Property of Model and its Subsidiaries. The transactions contemplated by this Agreement will not materially adversely effect the

17

Table of Contents

right, title or interest in and to the Intellectual Property of Model and its Subsidiaries listed on Section 4.18(a) of the Model Disclosure Schedule and all of such Intellectual Property shall be owned or available for use by Model and its Subsidiaries on substantially identical terms and conditions immediately after the Closing as currently pertains thereto.

- 4.19 <u>Corporate Documents</u>; <u>Books and Records</u>. The books, records and accounts of Model and its Subsidiaries, including all minute books of Model and its Subsidiaries, reflect in all material respects, the actions, meetings, transactions and the assets and liabilities of Model and its Subsidiaries. Model and each of its Subsidiaries have not engaged in any material transaction with respect to their respective businesses, maintained any bank account for their respective businesses, or used any of the funds of Model or any of its Subsidiaries in the conduct of their respective businesses, except for transactions, bank accounts and funds which have been and are reflected in the books and records of Model and each of its Subsidiaries.
- 4.20 <u>Bank Accounts</u>. <u>Section 4.20</u> of the Model Disclosure Schedule sets forth each of the bank accounts and the employees of Model and each of its Subsidiaries that are authorized signatories with respect to such accounts.
- 4.21 <u>Related Person Transactions</u>. Except as set forth in the footnotes to the Model Financial Statements and <u>Section 4.21</u> of the Model Disclosure Schedules and for the Affiliate Consignment and issuance of Affiliate Debt, during the period from November 1, 2003 through the date hereof (or, at the Closing, through the Closing Date), there have been no transactions, and there are no currently proposed transactions, in which Model or any of its Subsidiaries was or is to be a participant in which any of the Persons described in Item 404 of SEC Regulation S-K had or will have a direct or indirect material interest and which would be required to be disclosed in Item 404 of SEC Regulation S-K without giving effect to the amount involved in such transaction.
- 4.22 <u>Disclosure</u>. This Agreement (including the Model Disclosure Schedules) does not contain any untrue statement of a material fact or omit to state any material fact necessary in order to make the statements contained herein and therein, in light of the circumstances under which such statements were made, not misleading.

ARTICLE 5 ADDITIONAL COVENANTS AND AGREEMENTS

5.1 Conduct of Business.

- (a) Except (i) as permitted or contemplated by this Agreement, (ii) as required by applicable Law, or (iii) in the case of Model or its Subsidiaries, tenant improvements to new facilities and costs relating to holding over in existing facilities pending availability of new facilities or the Affiliate Consignment, during the period from the date of this Agreement until the earlier of the Termination Date or the Closing Date, unless E Com or Acquisition Sub, on the one hand, and Model, on the other hand, otherwise consent in writing, each party hereto shall, and shall cause each of its Subsidiaries to, (x) conduct its business in all material respects in the ordinary course consistent with past practice, and (y) preserve its current relationships with its material customers, vendors, suppliers and other Persons with which they have business relationships.
- (b) During the period from the date of this Agreement until the earlier of the Termination Date or the Closing Date, except as expressly contemplated or permitted by this Agreement or as required by applicable Law, and except as may be agreed in writing by E Com or Acquisition Sub, on the one hand, and Model, on the other hand, each party shall not, and shall not permit any of its Subsidiaries to, take any action or agree, in writing or otherwise, to take any action which would cause any of the representations or warranties of such party set forth in this Agreement (A) that are qualified as to materiality or Material Adverse Effect to be untrue and (B) that are not so qualified to be untrue in any material respect. In addition, each party shall not, and shall not permit any of its Subsidiaries to, acquire or agree to acquire by merging or consolidating with, or by purchasing a

18

Table of Contents

substantial portion of the assets of or equity in, or by any other manner, any Person or portion thereof, or otherwise acquire or agree to acquire any assets of any Person outside the ordinary course of business in accordance with past practice, except as part of a transaction permitted by Section 5.3.

- (c) During the period from the date of this Agreement until the earlier of the Termination Date or the Closing Date, each party shall give the other party notice of any effect, event or change that would reasonably be expected to have a Material Adverse Effect with respect to the notifying party.
- 5.2 Reasonable Best Efforts. Subject to the terms and conditions of this Agreement and subject to the provisions of Section 5.3 below, Model and E Com shall each cooperate with the other and use (and shall cause their respective Subsidiaries to use) their respective reasonable best efforts to (a) promptly (i) take or cause to be taken all actions, and do or cause to be done all things, necessary, proper or advisable under this Agreement and applicable Laws, to consummate the Transactions as soon as practicable, including preparing and filing promptly and fully all documentation to effect all necessary filings, notices, petitions, statements, registrations, submissions of information, applications and other documents (except that the Certificates of Merger shall be filed contemporaneously with the Closing), and (ii) obtain all approvals, consents, registrations, permits, authorizations and other confirmations from any Governmental Entity or third Person necessary, proper or advisable to consummate the Transactions and (b) take all action necessary to ensure that no state takeover statute or similar Law is or becomes applicable to any of the Transactions, take all action necessary to ensure that the Transactions may be consummated as promptly as practicable on the terms contemplated by this Agreement and otherwise minimize the effect of such Law on the Transactions.
- 5.3 Other Offers; Fiduciary Obligations. (a) Unless and until this Agreement shall have been terminated in accordance with its terms, neither E Com nor its Subsidiaries shall, through any officer, manager, director, member, employee, representative or agent of E Com or any of its Subsidiaries solicit or take other action to facilitate any inquiries or the making of any proposal that constitutes or may reasonably be expected to lead to an Acquisition Proposal, participate in any way in any discussions or negotiations relating thereto or in furtherance thereof or accept any Acquisition Proposal, or enter into any agreement, arrangement or understanding requiring E Com to abandon, terminate or fail to consummate the Transactions; provided, however, that nothing contained in this Section 5.3 shall prohibit E Com, or its Board of Directors, from making any disclosure to its shareholders that in the judgment of its Board of Directors is in accordance with, and based upon the advice of outside counsel, is required under applicable Law or in connection with SEC filings or filings under the HSR Act or from issuing press releases in accordance with Section 5.4. For purposes of this Agreement, Acquisition Proposal means any offer to acquire all or any substantial part of the business and properties or capital stock or membership interest of E Com or any of its Subsidiaries, whether by merger, consolidation, sale of assets, tender offer, or similar transaction or series of transactions involving E Com or any of its Subsidiaries. E Com hereby represents that neither it nor any of its Subsidiaries is engaged in discussions or negotiations with any Person other than Model with respect to any Acquisition Proposal.
- (b) Notwithstanding Section 5.3(a), the Board of Directors of E Com, in the exercise of and as required by its fiduciary duties, as determined in good faith by the Board of Directors of E Com, may (i) furnish information (including, without limitation, confidential information) concerning E Com to a third Person who makes an unsolicited request for such information for the purpose of making an Acquisition Proposal, and (ii) engage in discussions or negotiations with a third party who submits in writing an interest in making an Acquisition Proposal that the Board of Directors of E Com believes is reasonably capable of being consummated, provided, in the case of clause (i) or (ii) hereof, that Model shall have been promptly notified in writing of such request for information or Acquisition Proposal, including the identity of the offeror or the party making such offer or proposal and the principal financial terms and conditions of such offer or proposal, as the case may be, and shall be kept reasonably informed as to the status of any discussions or negotiations referred to in clause (ii) above, so long as Model shall have agreed to keep such information confidential on terms reasonably acceptable to E Com.

19

Table of Contents

- (c) Upon compliance with the foregoing, (i) the Special Committee and/or the Board shall be entitled to withdraw or modify their respective recommendations under Section 3.3 following receipt of a Superior Proposal and approve and recommend to the shareholders of E Com such Superior Proposal, (ii) the Board of Directors of E Com shall be entitled to postpone the Shareholders Meeting for a total period of not more than thirty (30) days to consider any Acquisition Proposal, notice of which shall have been provided to Model pursuant to Section 5.3(b), that the Special Committee shall reasonably determine may represent a Superior Proposal, and (iii) E Com shall be entitled to enter into an agreement with such third Person concerning such Superior Proposal on terms no less favorable to E Com than those set forth in the notice to Model pursuant to Section 5.3(b). A Superior Proposal is an Acquisition Proposal that the Board of Directors reasonably believes (A) is likely to be superior from a financial point of view to the transactions contemplated by this Agreement and (B) is capable of being consummated within a reasonable period of time.
- 5.4 <u>Public Announcements</u>. The initial press release with respect to the execution of this Agreement shall be the joint press release heretofore agreed upon by Model and E Com. E Com shall also file a Current Report on Form 8-K reporting the entering into of this Agreement and filing a copy of this Agreement and such joint press release as exhibits thereto. Thereafter, neither Model nor E Com shall, without the prior consent of the other party (which consent shall not be unreasonably withheld or delayed), issue or cause the publication of any press release or other public announcement with respect to this Agreement or the Transactions, except for any such press release or public announcement as may be required by Law as determined in the good faith judgment of the party proposing to make such release or announcement, upon advice of legal counsel, provided that no such release or announcement shall be made before reasonable (in the circumstances) advance notice of the same has been given to the other party.
- 5.5 Access. Upon reasonable notice and subject to applicable Laws relating to the exchange of information, Model shall, and shall cause each of its Subsidiaries to, afford to the officers, employees and other representatives of E Com, during the period prior to the Effective Time, access to its properties, books, contracts, commitments and records, and to its officers, employees and other representatives and, during such period, Model shall, and shall cause its Subsidiaries to, make available to E Com all other information concerning its business, properties and personnel as E Com may reasonably request. Until the Effective Time, the information provided will be used solely in connection with the Transactions.
- 5.6 Notification of Certain Matters. Model shall give prompt notice to E Com, and E Com shall give prompt notice to Model, of (i) any notice or other communication received by such party from any Governmental Entity in connection with the Transactions or from any Person alleging that the consent of such Person is or may be required in connection with the Transactions, and (ii) any actions, suits, claims, investigations or proceedings commenced or, to such party s Knowledge, threatened against, relating to or involving or otherwise affecting such party or any of its Subsidiaries which relate to the Transactions.
- 5.7 <u>Director and Officer Indemnification</u>. From and after the Closing Date and for a period of six (6) years thereafter, the Surviving Entity shall assume all obligations of Model, and cause the Surviving Entity s Subsidiaries to maintain in effect all such obligations owed by them, to each individual who at the Effective Time is, or any time prior to the Effective Time was, a director, officer or employee of Model or any of its Subsidiaries (the <u>Indemnitees</u>) in respect of indemnification and exculpation from liabilities for acts or omissions occurring at or prior to the Effective Time to the extent provided in the Model Charter Documents (or predecessor documents) and the organizational documents of such Subsidiaries as currently in effect and the BCL and the DLLCA as in effect as of the Effective Time. Without limiting the foregoing, E Com, from and after the Effective Time, shall cause the certificate of formation and limited liability company agreement of the Surviving Entity and its Subsidiaries to contain provisions no less favorable to the Indemnitees with respect to limitation of liabilities of directors, officers and managers and indemnification than are set forth as of the date of this Agreement in the Model Charter Documents, which provisions shall not be amended, repealed or otherwise modified in a manner that would adversely affect the rights thereunder of the Indemnitees.

20

Table of Contents

- 5.8 <u>Delivery of Financial Information</u>. From the date hereof to the Closing Date, Model shall deliver to E Com: (i) as soon as available, but in no event later than thirty (30) days after the end of each month subsequent to the date hereof consolidated financial statements of Model prepared in accordance with GAAP and consistent with those used in the preparation of the Model Financial Statements; and (ii) such other financial and operating information regarding the business of Model and its Subsidiaries as E Com reasonably requests, including, without limitation, monthly capital expenditure reports. Such financial statements have been and will be prepared on a basis consistent from period to period.
- 5.9 <u>Subsequent Transactions with Model Shareholders</u>. The Model Shareholders agree not to effect a merger, consolidation, combination or other transaction resulting in the acquisition by the Model Shareholders of all of the outstanding shares of E Com Common Stock, unless they collectively shall control ninety percent (90%) or more of the issued and outstanding shares of E Com Common Stock prior to such transaction.

5.10 Tax Matters.

(a) Regardless of the accuracy of the representations and warranties in Section 4.9 of this Agreement, the Model Shareholders shall be responsible for and shall indemnify and hold E Com and its Affiliates harmless against (i) all Taxes (or the non-payment thereof) of Model and its Subsidiaries for all taxable periods ending on or before the Closing Date and the portion through the end of the Closing Date for any taxable period that includes (but does not end on) the Closing Date (Pre-Closing Tax Period), (ii) all Taxes of any member of an affiliated, consolidated, combined or unitary group of which Model or any of its Subsidiaries (or any predecessor of any of the foregoing) is or was a member on or prior to the Closing Date, pursuant to Treasury Regulation section 1.1502-6 or any analogous or similar state, local, or foreign law or regulation, (iii) any and all Taxes of any person imposed on any of Model or its Subsidiaries as a transferee or successor, by contract or pursuant to any law, rule, or regulation, which Taxes relate to an event or transaction occurring before the Closing Date provided, however, that in the case of clauses (i), (ii) and (iii) above, Model Shareholders shall be liable only to the extent that such Taxes are in excess of the amount, if any, reserved for such Taxes (excluding any reserve for deferred Taxes established to reflect timing differences between book and Tax income) on the face of the Model Balance Sheet (rather than in any notes thereto), as such reserve is adjusted for the passage of time through the Closing Date in accordance with past custom and practice of Model and its Subsidiaries in filing their Tax Returns, but excluding any income Taxes resulting from unusual gains or losses outside of the ordinary course of business occurring after the date of this Agreement and on or prior to the Closing Date and, provided further, for purposes of the Tax indemnification provided by this Section 5.10, the Merger contemplated by this Agreement shall be deemed to occur after the Pr

In the case of Taxes that are payable with respect to any taxable period that begins before the Closing Date but ends after the Closing Date (the Straddle Period), the portion of any Taxes based on or measured by income or receipts of Model or its Subsidiaries for the Pre-Closing Tax Period shall be determined based on an interim closing of the books as of the close of business on the Closing Date (and for such purpose, the taxable period for any partnership or other pass-through entity in which any of Model and/or its Subsidiaries holds a beneficial interest shall be deemed to terminate at such time); and the amount of all other Taxes for a Straddle Period that relates to the Pre-Closing Tax Period shall be deemed to be the amount of such Tax for the entire Straddle Period multiplied by a fraction the numerator of which is the number of days in the period ending on the Closing Date and the denominator of which is the number of days in the entire Straddle Period.

Payment by the indemnifying party of any amount due under this <u>Section 5.10(a)</u> shall be made within ten days following written notice by the indemnified party that payment of such amounts to the appropriate taxing authority is due, provided, that the indemnifying party shall not be required to make any payment earlier than two days before it is due to the appropriate taxing authority.

(b) E Com shall prepare and timely file, or cause to be prepared and timely filed, all Pre-Closing Period Returns due after the Closing Date and E Com shall prepare and timely file, or cause to be prepared and timely filed, all Straddle Period Returns required to be filed after the Closing Date; provided, however, that in the case

21

Table of Contents

of any such Tax Returns, the Shareholder Representative shall have the right (prior to filing) to review and comment on any such Tax Return and E Com shall consider such comments in good faith. All such Tax Returns shall be true and correct in all material respects.

- (c) In the case of any Pre-Closing Period Return, E Com shall provide the Shareholder Representative: (i) a copy of such Tax Return as filed or to be filed; (ii) a schedule computing the Tax shown as due on any such Pre-Closing Period Return or the portion of the Tax shown as due on any such Straddle Period Return which relates to the portion of the period covered by such Straddle Return ending on the Closing Date (the Pre-Closing Tax). Within the time period prescribed in Section 5.10(a), Model Shareholders shall indemnify E Com or its Affiliates against any Pre-Closing Tax that is due and payable, to the extent in excess of the amount of Taxes reserved (excluding any reserve for deferred Taxes established to reflect timing differences between book and Tax income) on the face of the Model Balance Sheet (rather than in any notes thereto).
- (d) E Com and the Model Shareholders shall reasonably cooperate, and shall cause their respective Affiliates, agents, auditors, representatives, officers and employees to reasonably cooperate, in preparing and filing all Tax Returns to be filed by E Com and conducting any Tax audits and proceedings relating to Pre-Closing Taxes.
- (e) All tax-sharing agreements or similar agreements with respect to or involving Model and its Subsidiaries shall be terminated as of the Closing Date and, after the Closing Date, Model and its Subsidiaries shall not be bound thereby or have any liability thereunder.
- (f) All transfer, documentary, sales, use, stamp, registration and other such Taxes, and all conveyance fees, recording charges and other fees and charged (including any penalties and interest) incurred in connection with the consummation by this Agreement shall be borne by the Model Shareholders.
- (i) For Tax purposes, the parties agree to treat all payments made under this <u>Section 5.10</u> and under any other indemnity provisions contained in this Agreement, and for any breaches of representations, warranties, covenants or agreements, as adjustments to the Merger Consideration.
- (ii) This Section 5.10 and Section 8.4(b) shall be the sole provisions governing indemnities for Taxes under this Agreement.
- (iii) For purposes of this Section 5.10, all references to Model or its Affiliates shall include successors.
- (iv) Notwithstanding any provision in this Agreement to the contrary, the covenants and agreements of the parties hereto contained in this Section 5.10 shall survive the Closing and shall remain in full force indefinitely.
- (g) For the avoidance of doubt, no limits on indemnification contained in <u>Section 8.1</u> or <u>9.1</u> shall apply to indemnification for Taxes.
- 5.11 <u>Voting E Com Common Stock</u>. Each Model Shareholder, who is an E Com Shareholder, agrees to vote his, her or its shares of E Com Common Stock in favor of the matters for which the E Com Shareholder Approval is being sought.
- 5.12 Filing of Listing Application and Blue Sky Forms. The parties shall use commercially reasonably efforts to file any required listing applications with Nasdaq with respect to the shares of E Com Common Stock and Warrants to be issued pursuant to the Merger and the shares of E Com Common Stock to be issued upon exercise of the Warrants and any federal or state forms, notices or other documents relating to the issuance of such securities required to be filed under applicable state or federal securities laws.

22

Table of Contents

ARTICLE 6 CONDITIONS TO THE MERGER

- 6.1 <u>Conditions to Each Party</u> s <u>Obligation to Effect the Merger</u>. The respective obligations of each party hereto to effect the Merger shall be subject to the satisfaction (or waiver, if permissible under applicable Law) at or prior to the Effective Time of the following conditions:
- (a) E Com Shareholder Approval shall have been obtained;
- (b) Any waiting period applicable to the consummation of the Merger under the HSR Act shall have expired or been terminated, and no action shall have been instituted by the Department of Justice or Federal Trade Commission challenging or seeking to enjoin the consummation of the Merger or the Transactions, which action shall have not been withdrawn or terminated;
- (c) E Com shall have received confirmation that the any listing application referred to in Section 5.12 shall be effective as of the Effective Time;
- (d) Asset based loans and term debt facilities in amounts and on terms substantially similar to those contemplated in the Commitment Letter shall have been obtained by Acquisition Sub and/or E Com and funds shall have been advanced and/or be available thereunder;
- (e) No Law, injunction, judgment or ruling enacted, promulgated, issued, entered, amended or enforced by any Governmental Entity shall be in effect enjoining, restraining, preventing or prohibiting consummation of the Merger or making the consummation of the Merger illegal;
- (f) The Affiliate Consignment and the issuance of the Affiliate Debt shall have been effected on or prior to the Closing.
- 6.2 <u>Conditions to Model s Obligations to Effect the Merger</u>. The obligations of Model to effect the Merger shall be subject to the satisfaction (or waiver, if permissible under applicable Law) at or prior to the Effective Time of the following conditions:
- (a) The representations and warranties of E Com set forth in this Agreement shall be true and correct as of the date of this Agreement and as of the Closing Date as though made on and as of the Closing Date, except to the extent (i) any inaccuracies in such representations and warranties, individually or in the aggregate, would not have a Material Adverse Effect on E Com (provided that, for purposes of this Section 6.2(a), any representation or warranty of E Com that is qualified by materiality (or words of similar import) or Material Adverse Effect on E Com shall be read as if such language were not present), or (ii) such representations and warranties (other than Section 3.2, which shall remain true and correct on the Closing Date, except for issuances of E Com Common Stock as a result of the exercise of outstanding options under Stock Plans) specifically speak as of an earlier date. Model shall have received a certificate executed on behalf of E Com by its Chief Executive Officer or Chief Financial Officer to such effect:
- (b) E Com shall have performed or complied in all material respects with all of its obligations and covenants required to be performed by E Com under this Agreement at or prior to the Closing Date. Model shall have received a certificate executed on behalf of E Com by its Chief Executive Officer or Chief Financial Officer to such effect;
- (c) All consents from third Persons (other than Affiliates of Model or the Model Shareholders), including Governmental Entities, necessary or advisable to be obtained in connection with the Transactions shall have been obtained and be in full force and effect, including, without limitation, consents of lessors of properties leased by E Com and its Subsidiaries (as lessor or lessee), except where the failure to obtain all requisite consents and approvals required hereby and under Section 6.3(c) (considered as a group) is not reasonably expected to have a Material Adverse Effect on E Com and Model on a consolidated basis;

23

Table of Contents

- (d) E Com shall have executed and delivered a Registration Rights Agreement in the form of Exhibit B hereto among it and each of the shareholders of Model providing piggyback registration rights;
- (e) E Com shall have executed and delivered the Warrants; and
- (f) Since the date hereof no Material Adverse Effect shall have occurred as to E Com.
- 6.3 <u>Conditions to E Com</u> s and <u>Acquisition Sub</u> s <u>Obligations to Effect the Merger</u>. The respective obligations of each of E Com and Acquisition Sub to effect the Merger shall be subject to the satisfaction (or waiver, if permissible under applicable Law) at or prior to the Effective Time of the following conditions:
- (a) The representations and warranties of Model set forth in this Agreement shall be true and correct as of the date of this Agreement and as of the Closing Date as though made on and as of the Closing Date, except to the extent (i) any inaccuracies in such representations and warranties, individually or in the aggregate, would not have a Material Adverse Effect on Model (provided that, for purposes of this Section 6.3(a), any representation or warranty of Model that is qualified by materiality (or words of similar import) or Material Adverse Effect on Model shall be read as if such language were not present), or (ii) such representations and warranties (other than Section 4.2) speak as of an earlier date. E Com shall have received a certificate executed on behalf of Model by its Chief Executive Officer or Chief Financial Officer to such effect;
- (b) Model shall have performed or complied in all material respects with all of its obligations and covenants required to be performed by Model under this Agreement at or prior to the Closing Date. E Com shall have received a certificate executed on behalf of Model by its Chief Executive Officer or Chief Financial Officer to such effect;
- (c) All consents from third Persons, including Governmental Entities, necessary or advisable to be obtained in connection with the Transactions shall have been obtained and be in full force and effect, including, without limitation, consents of lessors of material properties leased by Model and its Subsidiaries (as lessor or lessee), except where the failure to obtain all requisite consents and approvals required hereby and under Section 6.2(c) (considered as a group) is not reasonably expected to have a Material Adverse Effect on E Com and Model on a consolidated basis;
- (d) Each of Model and Quality King Distributors, Inc. shall have executed and delivered the Transition Services Agreement dated as of a date prior to the Effective Time in a form to be mutually agreed to by Model and E Com prior to the Effective Time;
- (e) Each of Model and the landlords of real estate leased by Model shall have executed and delivered the Space Agreement dated as of a date prior to the Effective Time in a form to be mutually agreed to by Model and E Com prior to the Effective Time; and
- (g) Model shall cause to be delivered by each Model Shareholder the certificate of representations, including representations as to investment intent, in a form to be mutually agreed to by Model and E Com prior to the Effective Time.

ARTICLE 7 TERMINATION.

- 7.1 <u>Termination</u>. This Agreement may be terminated and the Transactions abandoned at any time prior to the Effective Time, whether before or after E Com Shareholder Approval thereof,:
- (a) By the mutual written consent of Model and E Com duly authorized by the respective Boards of Directors of Model and E Com.

24

Table of Contents

- (b) By either of Model or E Com, if any Governmental Entity shall have enacted, promulgated, issued, entered, amended or enforced (A) a Law prohibiting the Merger or making the Merger illegal, or (B) an injunction, judgment, order, decree or ruling, or taken any other action, in each case, permanently enjoining, restraining, preventing or prohibiting the Merger and such injunction, judgment, order, decree or ruling or other action shall have become final and non-appealable; <u>provided</u>, that the right to terminate this Agreement under this <u>Section 7.1(b)</u> shall not be available to a party if the issuance of such final, non-appealable injunction, judgment, order, decree or ruling was primarily due to the failure of such party to perform any of its obligations under this Agreement.
- (c) By either of Model or E Com:
- (i) if the Merger shall not have been consummated by June 30, 2008 (the <u>Outside Date</u>) <u>provided, howe</u>ver, that that the right to terminate this Agreement under this <u>Section 7.1(c)(i)</u> shall not be available to any party whose failure to fulfill any obligation or condition under this Agreement has been the cause of, or resulted in, the failure of the Merger to occur on or before such Outside Date;
- (ii) if E Com enters into an agreement for a Superior Proposal; or
- (iii) if the shareholders of E Com fail to give the E Com Shareholder Approval at the Shareholders Meeting, including any adjournment thereof; provided, however, that the right to terminate this Agreement under this Section 7.1(c)(iii) shall not be available to any party to this Agreement whose failure to fulfill any obligation or condition under this Agreement (or which has an Affiliate whose action or omission to act) has been the cause of, or resulted in, the failure to obtain the E Com Shareholder Approval (other than such failure, action or omission arising from any action or omission to act by the Board of Directors of E Com, as required by its fiduciary duties, as determined in good faith by the Board of Directors of E Com, in accordance with Section 5.3).
- (d) By E Com, if:
- (i) the representations and warranties of Model set forth in this Agreement shall not be true and correct on and as of the date of such determination as if made on such date such that the conditions set forth in Section 6.3(a) would not be satisfied, or
- (ii) Model shall have breached or failed in any material respect to perform or comply with any obligation, agreement or covenant required by this Agreement to be performed or complied with by it, such that the conditions set forth in <u>Section 6.3(b)</u> would not be satisfied,

which, in each case under clauses (i) and (ii) of this Section 7.1(d), has not been cured within twenty (20) Business Days after Model receives from E Com written notice of such inaccuracy, breach or failure; provided, however, that E Com may only exercise this termination right prior to the Closing Date; or

- (e) By Model:
- (i) if the Board of Directors of E Com or any committee thereof shall have withdrawn or modified or publicly proposed to withdraw or modify in a manner adverse to Model, its approval or recommendation of this Agreement or any of the Transactions; or
- (ii) if (A) the representations and warranties of E Com set forth in this Agreement shall not be true and correct on and as of the date of such determination as if made on such date such that the conditions set forth in Section 6.2(a) would not be satisfied or (B) E Com shall have breached or failed in any material respect to perform or comply with any obligation, agreement or covenant required by this Agreement to be performed or complied with by it, such that the conditions set forth in Section 6.2(b) would not be satisfied, which, in each case under clauses (A) and (B) has not been cured within twenty (20) Business Days after E Com receives from Model written notice of such inaccuracy, breach or failure; provided, however, that Model may only exercise this termination right prior to the Closing Date.

25

Table of Contents

7.2 <u>Effect of Termination</u>. In the event of the termination of this Agreement as provided in <u>Section 7.1</u>, written notice thereof shall be given to the other party or parties, specifying the provision hereof pursuant to which such termination is made, and this Agreement shall forthwith become null and void (other than the provisions of the Confidentiality Agreement most recently executed (the <u>Confidentiality Agreement</u>) by Model and E Com (subject to its terms), <u>Sections 5.4</u> and <u>7.2</u>, and <u>Article 9</u>, all of which shall survive termination of this Agreement for any reason, and there shall be no liability on the part of E Com or Model or their respective directors, officers and affiliates, except nothing shall relieve any party from liability for fraud or any willful breach of this Agreement.

ARTICLE 8 INDEMNIFICATION.

8.1 Obligation to Indemnify. Subject to the limitations set forth in this Article 8, E Com shall be entitled to indemnification from and against all losses, damages and out-of-pocket expenses (including reasonable attorneys fees and expenses of outside counsel, accountants and other professional fees, costs of investigation, court costs and other expenses) incurred by E Com in the investigation, collection, prosecution or defense of any claims, whether or not involving a third Person, suits, actions, claims, deficiencies, liabilities, penalties, and fines (but not including punitive, exemplary, consequential and all other kinds of special damages, including, without limitation, lost profits and opportunity costs, unless such punitive, exemplary, consequential or other kind of special damages are awarded to a Person in a Third-Party Claim) (Losses) sustained or incurred as a result of, relating to or caused by (i) any misrepresentation, breach of the representations and warranties of Model or a Model Shareholder contained in this Agreement or any certificate, schedule, document or other writing delivered pursuant hereto, or (ii) other than as contemplated by Article 7 and the failure to make payments under Sections 5.10 and 8.4(b), any breach of any of the covenants and agreements of Model or a Model Shareholder contained in this Agreement or any certificate, schedule, document or other writing delivered pursuant hereto; provided that E Com shall not be entitled to indemnification under this Agreement unless a claim is timely asserted during the applicable survival period specified in Section 9.1 by means of an Indemnification Notice meeting in all material respects the requirements of Section 8.2(a).

The rights of E Com to indemnification under this Section 8.1 (other than as to Pre-Closing Taxes) shall be limited as follows:

- (A) The amount of any Losses incurred by E Com shall be reduced by the net amount of the Tax Benefits actually realized by E Com or any of its Affiliates by reason of such Losses and shall be increased by the amount of any Tax to be incurred by E Com or any of its Affiliates (and grossed up for such increase) as a result of the receipt of indemnity payments with respect to any such Loss.
- (B) The amount of any Losses incurred by E Com shall be reduced by the net amount E Com or any of its Affiliates recovers (after deducting all reasonable attorneys fees, expenses and other costs of recovery) from any insurer or other party liable for such Losses, and E Com shall use commercially reasonable efforts to effect any such recovery.
- (C) E Com shall be entitled to indemnification under this <u>Section 8.1</u> only to the extent that the aggregate amount of Losses (reduced, without duplication, as provided in paragraphs (A) and (B) above) exceeds on a cumulative basis \$1,000,000 (the <u>Threshol</u>d), and then only for the amount of any such excess.
- (D) In no event will E Com be entitled to indemnification under this <u>Section 8.1</u> for any individual item (or series of related items) of Loss (as reduced in accordance with paragraphs (A) and (B) above) that is less than \$10,000 and each such item will not be taken into account in determining whether the aggregate amount of Losses exceeds the Threshold.
- (E) In any event the maximum amount for which E Com shall be entitled to indemnification in the aggregate under this <u>Section 8.1</u> (as reduced, without duplication, in accordance with paragraphs (A), (B), (C) and (D) above) shall not exceed the Escrow Fund.

26

Table of Contents

- (F) Except as to Pre-Closing Taxes as provided in <u>Sections 5.10</u> and <u>8.4(b)</u> and Pre-Closing Covenant Breaches, the Escrow Shares and the Escrow Fund shall be the sole and exclusive remedy available to E Com for any Losses arising out of any and all claims relating to the subject matter of this Agreement, and the maximum amount that may be recovered from any Model Shareholder shall be limited to such person s pro rata share of the Escrow Shares.
- (G) For purposes of calculating the amount of Losses incurred arising out of or relating to any breach of a representation or warranty by Model or a Model Shareholder, the references to Knowledge, Material Adverse Effect or other materiality qualifications shall be disregarded.

8.2 Indemnification Notice Procedures.

- (a) Promptly after the discovery of any Losses for which it may be entitled to indemnification pursuant to this <u>Article 8.</u> E Com shall deliver to the Shareholder Representative a notice (the <u>Indemnification Notice</u>), which Indemnification Notice shall:
- (i) state that E Com has paid or incurred Losses for which E Com is entitled to indemnification pursuant to this Agreement; and
- (ii) specify in reasonable detail each individual item of Loss, the basis for any anticipated liability and the nature of the misrepresentation, breach of warranty or breach of covenant to which each such item is related and, if then known, the amount to which E Com claims to be entitled hereunder.
- (b) In the event that the Shareholder Representative shall object to the indemnification of E Com in respect of any claim or claims specified in any Indemnification Notice, the Shareholder Representative shall, within forty-five (45) days after receipt by the Shareholder Representative of such Indemnification Notice, deliver to E Com a notice to such effect and the Shareholder Representative and E Com shall, within the sixty (60) day period beginning on the date of receipt by E Com of such objection, attempt in good faith to agree as to the disposition of each of such claims to which the Shareholder Representative shall have so objected. If E Com and the Seller Representative shall succeed in reaching agreement with respect to any of such claims, E Com and the Seller Representative shall promptly prepare and sign a memorandum setting forth such agreement. Should E Com and the Seller Representative be unable to agree as to any particular item or items or amount or amounts within such time period, then either the Indemnified Party or the Indemnifying Party may submit such dispute to arbitration as provided in Section 9.12.
- (c) Claims for Losses specified in any Indemnification Notice to which the Shareholder Representative shall not object in writing within forty-five (45) days after receipt of such Indemnification Notice, claims for Losses covered by a memorandum of agreement of the nature described in Section 8.2(b), and claims for Losses the validity and amount of which have been the subject of judicial determination as described in Section 8.2(b) or shall have been settled with the consent of the Shareholder Representative, as described in Section 8.3, are hereinafter referred to, collectively, as Agreed Claims .

8.3 Third-Party Claims.

(a) E Com agrees to give the Shareholder Representative notice in writing of the assertion of any claim or demand made by, or an action, proceeding or investigation instituted by, any Person not a party to this Agreement (a <u>Third-Party Claim</u>) in respect of which indemnity will be sought under <u>Section 8.1</u> in accordance with the notice procedures set forth in <u>Section 8.2</u> promptly after E Com learns of such Third-Party Claim; <u>provided</u>, <u>however</u>, that failure to give such notice shall not affect the indemnification provided hereunder, except to the extent there shall have been actual prejudice to the defense of such Third-Party Claim by the Shareholder Representative as a result of such failure (except that there shall be no right to indemnification under this <u>Article 8</u> for any expenses incurred during the period in which E Com failed to give such notice). Thereafter, E Com shall deliver to the Shareholder Representative, within five (5) Business Days after E Com s receipt thereof, copies of all notices and documents (including court papers) received by E Com relating to the Third-Party Claim.

27

Table of Contents

- (b) If a Third-Party Claim is made against E Com, the Shareholder Representative will be entitled to participate in the defense thereof. The Shareholder Representative may elect to assume the defense, at Model Shareholders expense and by counsel selected by the Shareholder Representative, of any Third-Party Claim in accordance with the following conditions:
- (i) If such Third-Party Claim involves (and continues to involve) solely monetary damages and the Shareholder Representative agrees in writing with E Com that E Com shall be entitled (subject to any applicable limitations in Section 8.1) to indemnification hereunder as to such claim and the amount of such claim can be satisfied from the Escrow Fund, then the Shareholder Representative may assume the defense of such claim and, subject to Section 8.3(d), may not settle such claim without the consent of E Com, which consent shall not be unreasonably withheld or delayed.
- (ii) If such Third-Party Claim involves both monetary damages and injunctive relief against E Com, then the Shareholder Representative may assume the defense of such claim and may settle that portion of such claim involving solely monetary damages in accordance with clause (i) above, as applicable, mutatis mutandis, but shall not settle any portion of such claim involving injunctive relief without the consent of E Com, which consent shall not be unreasonably withheld or delayed.
- (iii) If such Third-Party Claim involves (and continues to involve) solely injunctive relief against E Com, E Com shall control the defense of such claim and shall not settle any portion of such claim for monetary damages without the consent of the Shareholder Representative, which consent shall not be unreasonably withheld or delayed.
- (iv) In all cases, if the parties in any action shall include both Model and/or a Model Shareholder, on the one hand, and E Com, on the other hand, and E Com shall have reasonably concluded that counsel selected by the Shareholder Representative has a conflict of interest because of the availability of conflicting defenses, E Com shall have the right to select separate counsel to participate in the defense of such action on its behalf.
- (c) Subject to the foregoing, if the Shareholder Representative elects to assume the defense of such Third-Party Claim, it shall notify E Com of the Shareholder Representative s intent to do so within thirty (30) days (or sooner, if the nature of the Third-Party Claim so requires) of E Com s notice of such Third-Party Claim. If the Shareholder Representative is not entitled or elects not to assume the defense of the Third-Party Claim or fails to notify E Com of its election as herein provided, E Com shall have the right to conduct such defense with counsel reasonably acceptable to the Shareholder Representative, but E Com (or any insurance carrier defending such Third-Party Claim on E Com s behalf) shall be prohibited from compromising or settling the Third-Party Claim without the prior written consent of the Shareholder Representative, which consent shall not be unreasonably withheld or delayed.
- (d) Subject to Section 8.3(b)(iv), should the Shareholder Representative so elect to assume the defense of a Third-Party Claim, as long as the Shareholder Representative conducts such defense, E Com shall not be entitled to indemnification under this Article 8 for legal expenses subsequently incurred by E Com in connection with the defense thereof. If the Shareholder Representative assumes such defense, E Com shall have the right to participate in the defense thereof and to employ counsel, at its own expense, separate from the counsel employed by the Shareholder Representative, it being understood that the Shareholder Representative shall control such defense. E Com shall be entitled to indemnification under this Article 8 for the fees and expenses of counsel employed by E Com for any period during which the Shareholder Representative has not assumed the defense thereof (other than during any period in which E Com shall have not yet given notice of the Third-Party Claim as provided above). If the Shareholder Representative chooses to defend or prosecute any Third-Party Claim, the Shareholder Representative and E Com shall cooperate in the defense or prosecution thereof. Such cooperation shall include the retention and (upon the Shareholder Representative s request) the provision to the Shareholder Representative of records and information that are reasonably relevant to such Third-Party Claim, and making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder. Whether or not the Shareholder Representative shall have assumed the defense of a

28

Table of Contents

Third-Party Claim, E Com shall not admit any liability with respect to, or settle, compromise or discharge, such Third-Party Claim without the Shareholder Representative s prior written consent, which consent shall not be unreasonably withheld or delayed. Notwithstanding the foregoing, if a settlement offer is made by a third-party claimant solely for money damages and does not seek equitable or other relief and the settlement is not likely to establish a precedential custom or practice adverse to the continued business interests of E Com, and the Shareholder Representative notifies in writing E Com of the Shareholder Representative s willingness to accept the settlement offer and, subject to the applicable limitations of this Article-8, cause to be paid the amount called for by such offer, and E Com declines to accept such offer, E Com may continue to contest such Third-Party Claim, free of any participation by the Shareholder Representative, and the amount of any ultimate liability with respect to such Third-Party Claim for which E Com shall be entitled to indemnification hereunder shall be limited to the lesser of (i) the amount of the settlement offer that E Com declined to accept plus the other Losses of E Com relating to such Third-Party Claim prior to the date of its rejection of the settlement offer or (ii) the aggregate Losses of E Com with respect to such Third-Party Claim.

8.4 Payment of Claims.

- (a) Each Agreed Claim (other than as to a Pre-Closing Tax or as to a Pre-Closing Covenant Breach) shall be satisfied from the Escrow Fund as follows:
- (i) E Com shall notify the Shareholder Representative in writing of its intention to satisfy the Agreed Claim from the Escrow Fund.
- (ii) Unless the Shareholder Representative shall have paid (or caused to be paid) to E Com by wire transfer an amount equal to the amount of such Agreed Claim within ten (10) Business Days of such written notice (a <u>Cash Indemnity Payment</u>), E Com may cause to be cancelled such whole number of shares of the Escrow Shares as shall equal the amount of such Agreed Claim divided by the Consideration Per Share Price rounded up to the next whole number.
- (iii) If a Cash Indemnity Payment shall be made with respect to an Agreed Claim, E Com shall cause to be delivered to the Shareholder Representative for delivery to the Model Shareholders pro rata based upon their respective allocations of the Escrow Shares certificates in such denominations as the Shareholder Representative may indicate in the name of one or more of the Model Shareholders representing in the aggregate a whole number of shares of the Escrow Shares equal to the amount of such Agreed Claim divided by the Consideration Per Share Price rounded down to the next whole number.
- (b) Each Agreed Claim with respect to Pre-Closing Tax shall be paid in cash by the Model Shareholders in accordance with Section 5.10.
- (c) E Com shall be entitled to seek indemnity for any Agreed Claim with respect to the breach by Model of any of its covenants or agreements hereunder that was to be performed on or prior to the Closing Date (a <u>Pre-Closing Covenant Breach</u>) directly against Model subject to the terms and conditions of this <u>Article 8</u> (other than <u>Section 8.4(a)</u>).

8.5 Shareholder Representative.

- (a) The Shareholder Representative shall act as the representative of the Model Shareholders, and shall be authorized to act on behalf of the Model Shareholders and to take any and all actions required or permitted to be taken by the Shareholder Representative under this Agreement, with respect to any claims (including the settlement thereof) made by E Com for indemnification or to be held harmless pursuant to this Article 8. The Model Shareholders shall be bound by all actions taken by the Shareholder Representative in its capacity thereof.
- (b) The Shareholder Representative shall at all times act in his or her capacity as Shareholder Representative in a manner that the Shareholder Representative believes in good faith to be in the best interest of the Model Shareholders. Neither the Shareholder Representative nor any of its agents shall be liable to any Person for any

29

Table of Contents

error of judgment, or any action taken, suffered or omitted to be taken, under this Agreement, except in the case of its gross negligence, bad faith or willful misconduct. The Shareholder Representative may consult with legal counsel, independent public accountants and other experts selected by it and shall not be liable for any action taken or omitted to be taken in good faith by it in accordance with the advice of such counsel, accountants or experts. The Shareholder Representative shall not have any duty to ascertain or to inquire as to the performance or observance of any of the terms, covenants or conditions of this Agreement. As to any matters not expressly provided for in this Agreement, the Shareholder Representative shall not be required to exercise any discretion or take any action.

(c) Each Model Shareholder severally shall indemnify and hold harmless and reimburse the Shareholder Representative from and against such Model Shareholder s ratable share of any and all Losses suffered or incurred by the Shareholder Representative arising out of or resulting from any action taken, or omitted to be taken, by the Shareholder Representative under this Agreement, other than such Losses arising out of or resulting from the Shareholder Representative s gross negligence, bad faith or willful misconduct.

ARTICLE 9 MISCELLANEOUS.

- 9.1 No Survival of Representations and Warranties; etc. The representations, warranties and agreements in this Agreement shall terminate at the Effective Time except that the representations and warranties set forth in Articles 3 and 4 shall survive until the Escrow Release Date (other than those as to Taxes set forth in Section 4.9, which shall survive until expiration of the applicable statute of limitations as to Taxes), the agreements set forth in Section 5.7 shall survive the Effective Time for a period of six (6) years, the agreements set forth in Section 5.10 shall survive for the applicable statute of limitations as to Taxes and the agreements set forth in Articles 2 and 8 and this Article 9 shall survive the Effective Time indefinitely.
- 9.2 <u>Amendment or Supplement</u>. At any time prior to the Effective Time, this Agreement may be amended or supplemented in any and all respects, whether before or after approval of any of the transactions contemplated hereby by the Model Shareholders, by written agreement of the parties hereto, and by action taken by the respective Boards of Directors of E Com and Model; <u>provided</u>, <u>however</u>, that following the E Com Shareholder Approval there shall be no amendment or change to the provisions hereof which by Law would require shareholder approval without such approval.
- 9.3 Extension of Time, Waiver, Etc. At any time prior to the Effective Time, any party may, subject to applicable Law, (a) waive any inaccuracies in the representations and warranties of any other party hereto, (b) extend the time for the performance of any of the obligations or acts of any other party hereto or (c) waive compliance by the other party with any of the agreements contained herein or, except as otherwise provided herein, waive any of such party s conditions. Notwithstanding the foregoing, no failure or delay by Model or E Com in exercising any right hereunder shall operate as a waiver thereof nor shall any single or partial exercise thereof preclude any other or further exercise thereof or the exercise of any other right hereunder. Any agreement on the part of a party hereto to any such extension or waiver shall be valid only if set forth in an instrument in writing signed on behalf of such party.
- 9.4 <u>Assignment; Binding Effect</u>. Neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned by any of the parties hereto (whether by operation of law or otherwise) without the prior written consent of the other parties, except that E Com may assign any of its rights, interests or obligations hereunder to any of its Affiliates; <u>provided</u>, <u>however</u>, that any such assignment shall not relieve E Com from its obligations hereunder. Subject to the preceding sentence, this Agreement shall be binding upon and shall inure to the benefit of the parties hereto and their respective heirs, executors, administrators, legal representatives, successors and permitted assigns. Any purported assignment not permitted under this <u>Section 9.4</u> shall be null and void.

30

Table of Contents

- 9.5 <u>Counterparts</u>; <u>Effectiveness</u>. This Agreement may be executed in two or more separate counterparts, each of which shall be deemed to be an original but all of which taken together shall constitute one and the same agreement. This Agreement shall become effective when each party hereto shall have received counterparts hereof signed by the other parties hereto. Signatures sent by fax or pdf shall be deemed originals.
- 9.6 Entire Agreement; No Third-Party Beneficiaries. This Agreement, the exhibits and schedules hereto, the Confidentiality Agreement, the E Com Disclosure Schedule and the Model Disclosure Schedule constitute the entire agreement, and supersede all other prior agreements and understandings, both written and oral, among the parties, with respect to the subject matter hereof and thereof. This Agreement is not intended to and shall not confer upon any Person other than the parties hereto any rights hereunder.
- 9.7 Governing Law; Enforcement; Jurisdiction; Waiver of Jury Trial.
- (a) This Agreement shall be governed by and construed in accordance with the laws of the State of New York, without regard to the principles of conflicts of laws thereof that would defer to the Laws of another jurisdiction.
- (b) The parties agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the parties shall be entitled to seek an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement in the United States District Court for the Eastern District of New York or in New York Supreme Court sitting in Suffolk County, New York, without bond or other security being required, this being in addition to any other remedy to which they are entitled at law or in equity.
- (c) For purposes of enforcing Section 9.12 or an award of the arbitrator(s) pursuant thereto, each of the parties hereto (i) consents to submit itself to the personal jurisdiction of the United States District Court for the Eastern District of New York or in New York Supreme Court sitting in Suffolk County, New York in the event any dispute arises out of this Agreement or any of the Transactions, (ii) agrees that it will not attempt to deny or defeat such personal jurisdiction by motion or other request for leave from any such court, (iii) waives any defense of forum non conveniens, and (iv) agrees that it will not bring any action relating to this Agreement or any of the Transactions in any court other than a Federal or State court sitting in the State of New York.
- (d) Each of the parties hereto hereby irrevocably waives any and all rights to trial by jury in any legal proceeding arising out of or relating to this Agreement or the Transactions.
- 9.8 Notices. All demands, notices, requests, consents and other communications required or permitted under this Agreement shall be in writing and shall be personally delivered or sent by facsimile machine (with a confirmation copy sent by one of the other methods authorized in this Section), reputable commercial overnight delivery service (including Federal Express and U.S. Postal Service overnight delivery service) for the next Business Day delivery, or deposited with the U.S. Postal Service mailed first class, registered or certified mail, postage prepaid, as set forth below:

If to Model, to:

Model Reorg, Inc.

2060 Ninth Avenue

Ronkonkoma, New York 11779

Attention: Michael Katz

Facsimile No.: (631) 737-5154

Table of Contents 289

31

with a copy to: Edwards Angell Palmer & Dodge LLP 750 Lexington Avenue New York, NY 10022 Attention: Patricia L. Kantor, Esq. Facsimile No.: (212) 308-4844 If to the Seller Representative to: Stephen Nussdorf c/o Model Reorg, Inc. 2060 Ninth Avenue Ronkonkoma, NY 11779 Facsimile No.: 631-439-2333 with a copy to: Edwards Angell Palmer & Dodge LLP 750 Lexington Avenue New York, NY 10022 Attention: Patricia L. Kantor, Esq. Facsimile No.: (212) 308-4844 If to E Com or Acquisition Sub, to: E Com Ventures, Inc. 251 International Parkway Sunrise, Florida 33325 Attention: Donovan Chin Facsimile No.: (954) 335-9166 with copies to: Akerman Senterfitt

One Southeast Third Avenue

28th Floor

Miami, FL 33131

Attention: Alan Aronson, Esq.

Facsimile No.: (305) 374-5095

and

Special Committee of the Board of Directors of E Com Ventures, Inc.

c/o Troutman Sanders LLP

405 Lexington Avenue

New York, NY 10174

Attention: Richard A. Rubin, Esq.

Facsimile No.: (212) 704-6288

Notices shall be deemed given upon the earlier to occur of (i) receipt by the party to whom such notice is directed; (ii) if sent by facsimile machine, on the Business Day such notice is sent if sent (as evidenced by the facsimile confirmed receipt) prior to 5:00 p.m. Eastern Time and, if sent after 5:00 p.m. Eastern Time, on the first Business Day after which such notice is sent; (iii) on the first Business Day following the day the same is deposited with the commercial courier if sent by commercial overnight delivery service for next Business Day delivery; or (iv) the fifth Business Day following deposit thereof with the U.S. Postal Service as aforesaid. Each party, by notice duly given in accordance herewith, may specify a different address for the giving of any notice hereunder.

Table of Contents

- 9.9 Severability. If any term or other provision of this Agreement is determined by a court of competent jurisdiction to be invalid, illegal or incapable of being enforced by any rule of law or public policy, all other terms, provisions and conditions of this Agreement shall nevertheless remain in full force and effect. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the parties hereto shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible to the fullest extent permitted by applicable Law in an acceptable manner to the end that the transactions contemplated hereby are fulfilled to the extent possible. If the parties are unable to reach an agreement as to such modification, a court shall have the power to so modify such provision.
- 9.10 <u>Headings</u>. Headings of the Articles and Sections of this Agreement are for convenience of the parties only, and shall be given no substantive or interpretive effect whatsoever.
- 9.11 <u>Definitions; Construction</u>. As used in this Agreement, the following terms have the meanings ascribed thereto below:

Affiliate shall mean, as to any Person, any other Person which, directly or indirectly, controls, or is controlled by, or is under common control with, such Person. For this purpose, <u>control</u> (including, with its correlative meanings, <u>controlled</u> by <u>and under common control</u> with) shall mean the possession, directly or indirectly, of the power to direct or cause the direction of management or policies of a Person, whether through the ownership of securities or partnership or other ownership interests, by contract or otherwise.

Affiliate Consignment means the repayment by Model and its Subsidiaries of that portion, if any, of the principal amount of (plus accrued and unpaid interest on) their indebtedness to Quality King Distributors, Inc. as of the Closing Date that is in excess of \$50,000,000 by the transfer of inventory, which inventory shall be mutually designated by Model and E Com, having a book value equal to the repaid principal and accrued and unpaid interest on all of the indebtedness of Model and its Subsidiaries to Quality King Distributors and the consignment by Quality King Distributors, Inc. of such inventory to Model and/or one or more of its Subsidiaries for sale on its behalf pursuant to a consignment agreement on customary commercial terms and conditions that shall be reasonably acceptable to E Com but shall include the agreement of Quality King Distributors, Inc. not to sell such inventory other than through the Surviving Entity without the Surviving Entity s consent.

<u>Affiliate De</u>bt means subordinated debt payable to certain Model Shareholders on the terms and conditions described <u>on Exhibit</u> C attached hereto and such other terms as shall be mutually agreeable to such Model Shareholders and E Com.

<u>Business Day</u> shall mean a day except a Saturday, a Sunday or other day on which the SEC or banks in the States of New York or Florida are authorized or required by Law to be closed.

<u>Commitment Letter</u> shall mean the commitment letter dated August 29, 2007 between General Electric Capital Corporation and E Com, as extended or modified.

Consideration Per Share Price shall mean \$23.94.

<u>E Com Indebtedness</u> shall mean (i) all indebtedness of E Com and its Subsidiaries for borrowed money, (ii) all obligations of E Com and its Subsidiaries for the deferred purchase price of property or assets, (iii) all obligations of E Com and its Subsidiaries evidenced by notes (including promissory notes issued in consideration for the purchase of stock or assets of any business), bonds, debentures, guarantees or other similar instruments and (iv) all capital leases, which, in the case of clauses (i) through (iv) above, shall include all accrued interest thereon and applicable prepayment premiums and any other fees, costs or expenses payable in connection therewith.

<u>Environmental Laws</u> shall mean all federal, foreign, state and local laws, statutes, codes, regulations, rules, ordinances, judgments, decisions and orders of any Governmental Entity (including consent decrees, administrative orders and self-implementing requirements) relating to human health, safety and protection of the

33

Table of Contents

environment, all as amended or reauthorized to the date hereof and through the Closing Date, including any law relating to emissions, discharges, Releases or threatened Releases of Hazardous Materials into ambient air, surface water, groundwater or land.

<u>GAAP</u> shall mean generally accepted accounting principles in the United States, applied on a consistent basis during the periods involved (except as may be indicated in the notes to any applicable financial statements).

<u>Governmental Entity</u> shall mean any federal, state or local, domestic, foreign or multinational government, court, regulatory or administrative agency, commission or authority or other governmental instrumentality.

<u>Hazardous Materials</u> shall mean (a) any hazardous waste as defined in the Resource Conservation and Recovery Act of 1976, 42 U.S.C. Sections 6901 et seq., as amended, to the date hereof and through the Closing Date, and regulations promulgated thereunder; (b) any hazardous substance as defined in the Comprehensive Environmental Response, Compensation and Liability Act of 1980, 42 U.S.C. Section 9601 et seq., as amended through the Closing Date, and regulations promulgated thereunder; (c) petroleum or petroleum products and all derivatives and constituents thereof; (d) polychlorinated biphenyls (PCBs); (e) asbestos and asbestos-containing materials; and (f) any other pollutant, contaminant, constituent or chemical regarding which liability is imposed under any Environmental Laws.

<u>Intellectual Property</u> means all (i) patents, patent applications, patent disclosures and inventions, whether United States or foreign, including all reissues, continuations, divisions, continuations in part and renewals and extensions thereof, (ii) internet domain names, trademarks and service marks whether United States or foreign, trade dress, trade names, logos and corporate names and registrations and applications for registration thereof, together with all of the goodwill associated therewith, (iii) copyrights (registered or unregistered) and copyrightable works and registrations and applications for registration thereof, (iv) mask works and registrations and applications for registration thereof, (v) computer software (including both source and object code), data, data bases and documentation thereof, (vi) trade secrets and other confidential information (including ideas, formulas, compositions, inventions (whether patentable or unpatentable and whether or not reduced to practice), know-how, manufacturing and production processes and techniques, research and development information, drawings, specifications, designs, plans, proposals, technical data, copyrightable works, financial and marketing plans and customer and supplier lists and information), (vii) other intellectual property rights and (viii) copies and tangible embodiments thereof (in whatever form or medium).

Knowledge of E Com whether or not such term is used herein in initial capitalized form, when used in this agreement in the phrase Knowledge of E Com, or similar phrases, means the actual knowledge of Donovan Chin.

Knowledge of Model whether or not such term is used herein in initial capitalized form, when used in this agreement in the phrase Knowledge of Model, or similar phrases, means the actual knowledge of each of Michael Katz and Stephen Nussdorf.

<u>Material Adverse Effect</u> shall mean, with respect to any party, an effect, event or change which has a material adverse effect on the business, operations, assets, liabilities, results of operations, prospects, or financial condition of such party and its Subsidiaries taken as a whole, other than effects, events or changes arising out of or resulting from (i) changes in conditions in the United States or global economy or capital or financial markets generally, including changes in interest or exchange rates, (ii) changes in general legal, regulatory, political, economic or business conditions or changes in GAAP, unless they disproportionately affect one of the parties, or (iii) any decline in the market price, or change in trading volume, of the capital stock of such party or any failure to meet publicly announced revenue or earnings projections.

<u>Model Indebtedness</u> shall mean (i) all indebtedness of Model and its Subsidiaries for borrowed money, (ii) all obligations of Model and its Subsidiaries for the deferred purchase price of property or assets, (iii) all obligations of Model and its Subsidiaries evidenced by notes (including promissory notes issued in consideration

34

Table of Contents

for the purchase of stock or assets of any business), bonds, debentures, guarantees or other similar instruments and (iv) all capital leases, which, in the case of clauses (i) through (iv) above, shall include all accrued interest thereon and applicable prepayment premiums and any other fees, costs or expenses payable in connection therewith.

<u>Permit(s)</u> shall mean any approvals, consents, registrations, permits, licenses and other authorizations from any Governmental Entity.

<u>Person</u> shall mean an individual, a corporation, a limited liability company, a partnership, an association, a trust, or any other entity, including a Governmental Entity.

<u>Pre-Closing Period Return</u> means all Tax Return for periods ending on or prior to the Closing Date.

Release shall mean any spilling, leaking, pumping, pouring, emitting, emptying, discharging, injecting, escaping, dumping or disposing into the environment, including movement of any Hazardous Materials through or in the environment, and the abandonment or discarding of barrels, containers and other receptacles containing any Hazardous Materials, but excludes the normal application of lawn or garden chemicals such as pesticides, herbicides and fertilizers.

<u>Remedial Action</u> shall mean any investigation, abatement, containment, removal or remediation of any Release or threatened Release of Hazardous Materials conducted or required to be conducted pursuant to any Environmental Laws.

<u>SEC</u> shall mean the U.S. Securities and Exchange Commission.

<u>Shareholder Representative</u> shall mean initially, Glenn Nussdorf, or such other person as shall be designated in writing by a majority-in-interest of the Model Shareholders based upon the number of shares of Model held by the Model Shareholders as of the date hereof.

<u>Special Committee</u> means the committee of the Board of Directors of E Com established to consider the proposed Merger.

<u>Straddle Period Return</u> means a Tax Return (other than a Pre-Closing Period Return) required to be filed by or with respect to Model or its Subsidiaries for periods beginning prior to and ending after the Closing Date.

<u>Subsidiary</u> when used with respect to any Person, shall mean any corporation, limited liability company, partnership, association, trust or other entity the accounts of which would be consolidated with those of such Person in such Person s consolidated financial statements if such financial statements were prepared in accordance with GAAP, as well as any other corporation, limited liability company, partnership, association, trust or other entity, the securities or other ownership interests representing 50% or more of the ordinary voting power (or, in the case of a non-corporate entity, 50% or more of the ownership interests) of which are, as of such date, owned by such party or one or more Subsidiaries of such Person.

<u>Tax Benefit(s)</u> For purposes of this Agreement, E Com or any of its Affiliates shall be deemed to recognize a Tax Benefit with respect to a taxable year if, and to the extent that the aggregate liability of E Com or such Affiliate for Taxes through the end of such taxable year, as calculated by the E Com or such Affiliate by excluding any Tax items attributable to any Loss and the applicable indemnification payments with respect thereto from all taxable years, exceeds the actual aggregate liability of E Com or such Affiliate for Taxes through the end of such taxable year, as calculated by E Com or such Affiliate by taking into account any Tax items attributable to such Loss and the applicable indemnification payments (to the extent permitted by relevant Tax Law and treating such Tax items as the last items claimed for any taxable year).

35

<u>Tax(es)</u> means all taxes, charges, fees, levies, penalties, or other assessments imposed by any foreign or United States federal, state, tribal or local Taxing Authority, including income, excise, property, sales, use, gross receipts, windfall profits, environmental (including taxes under Code Section 59A), employment, severance, stamp, capital stock, disability, real property, personal property, sales, use, unemployment, disability, registration, value added or add-on minimum, compensating, transfer, franchise, license, payroll, withholding, social security, estimated or other taxes (including any escheat or unclaimed property obligations), in each case including any interest, penalties, or additions attributable thereto, whether disputed or not, and including any obligations to indemnify or otherwise assume or succeed to the tax liability of any other Person.

<u>Tax Return(s)</u> means any and all returns, declaration, report, claim for refund, or information return or statement relating to Taxes, including any schedule or attachment thereto, and including any amendment thereof.

<u>Transactions</u> refers collectively to this Agreement and the transactions contemplated hereby, including the Merger.

The following terms are defined in the Sections of this Agreement set forth after such term below:

Acquisition Proposal	5.3(a)	Confidentiality Agreement	7.2
Acquisition Sub	Preamble	Control	9.11
Affiliate	9.11	Controlled By	9.11
Agreed Claims	8.2(c)	DLLCA	Recitals
Agreement	Preamble	E Com	Preamble
Bankruptcy Exception	3.3(d)	E Com Common Stock	Recitals
BCL	Recitals	E Com Disclosure Schedule	3
Business Day	9.11	E Com Indebtedness	9.11
Cash Indemnity Payment	8.4(b)	E Com Shareholder Approval	1.1(a)
Certificate of Merger	2.3	Effective Time	2.3
Certificates	2.8(a)	Environmental Claims	4.12(a)(i)
Charter Amendment	1.1(a)	Environmental Laws	9.11
Closing	2.2	Environmental Liabilities	4.12(a)(ii)
Closing Date	2.2	ERISA	4.10(a)
Code	4.10(a)	ERISA Affiliate	4.10(a)
Company Benefit Plan	4.10(a)	Escrow Fund	2.9
Escrow Release Date	2.9	Permits	9.11
Escrow Shares	2.9	Person	9.11
GAAP	9.11	Pre-Closing Covenant Breach	8.4(c)
Governmental Entity	9.11	Pre-Closing Period Return	9.11
Hazardous Materials	9.11	Pre-Closing Tax	5.10(c)
HSR Act	4.4(a)	Pre-Closing Tax Period	5.10(a)
Indemnification Notice	8.2(a)	Proxy Statement	1.1(b)
Indemnitees	5.7	Release	9.11
Intellectual Property	9.11	Remedial Action	9.11
IRS	4.10(c)	SEC	9.11
Knowledge of E Com	9.11	Shareholders Meeting	1.1(a)
Knowledge of Model	9.11	Shareholder Representative	9.11
Laws	1.1	Special Committee	9.11
Losses	8.1	Straddle Period Return	9.11
Merger	Recitals	Stock Plans	3.2
Merger Consideration	Recitals	Subsidiary	9.11
Model	Preamble	Superior Proposal	5.3(c)
Model Balance Sheet	4.5(b)	Surviving Entity	2.1
Model Balance Sheet Date	4.5(b)	Tax Benefit(s)	9.11

Table of Contents

Model Charter Documents	4.1(a)	Tax(es)	9.11
Model Common Stock	Recitals	Tax Returns	9.11
Model Disclosure Schedule	4	Third-Party Claim	8.3(a)
Model Financial Statements	4.5(a)	Threshold	8.1(c)
Model Indebtedness	9.11	Transactions	9.11
Model Material Contracts	4.15(a)	Under Common Control With	9.11
Model Material Lease	4.17	WARN	4.11(d)
Model Permitted Encumbrances	4.16(b)	Warrants	Recitals
Model Shareholders	Preamble		
Outside Date	7.1(c)(i)		

As used in this Agreement, <u>including</u> shall mean including, without limitation.

Any singular term in this Agreement shall be deemed to include the plural, and any plural term the singular.

As used in this Agreement, pronouns of either gender or neuter shall include, as appropriate, the other pronoun forms

The parties hereto have participated jointly in the negotiation and drafting of this Agreement and, in the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as jointly drafted by the parties hereto and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provision of this Agreement.

9.12 Arbitration.

(a) After the Closing, any dispute, controversy or claim arising out of or relating to this Agreement which cannot be resolved by the parties within thirty (30) days shall be finally settled by binding arbitration administered by the American Arbitration Association in accordance with its Commercial Arbitration Rules then in effect, including its Procedures for Large, Complex Commercial Disputes, except as specifically otherwise provided in this Section 9.12. This Section 9.12 shall be construed and enforced in accordance with the Federal Arbitration Act, notwithstanding any other choice of law provision in this Agreement.

(b) Arbitration Procedures.

- (i) The place of arbitration shall be New York, New York before a single neutral arbitrator, unless any party s claim exceeds Two Million Dollars (\$2,000,000), exclusive of interest and attorneys fees, in which case the dispute shall be heard and determined by three neutral arbitrators.
- (ii) The arbitrator(s) shall allow such discovery as the arbitrator(s) determine appropriate under the circumstances, including examination by deposition of witnesses. Depositions shall be held within thirty (30) days of the making of the request, limited to a maximum of three per party and no more than six hours duration each. Any dispute regarding discovery shall be determined by the arbitrator(s) and all discovery shall be completed within one hundred and twenty (120) days after the selection of the arbitrator(s).
- (iii) The decision of the arbitrator(s) shall be made within nine (9) months of the filing of the notice of intention to arbitrate (demand). The award shall be in writing and shall set forth the reasons for the disposition of any claim, and the arbitrator(s) shall have thirty (30) days thereafter to reconsider and modify such decision if any party so requests within ten (10) days after the decision.
- (iv) The decision of the arbitrator(s) shall be final, binding, and non-appealable with respect to all persons, including (without limitation) persons who have failed or refused to participate in the arbitration process.
- (c) The arbitrator(s) shall have authority to award relief under legal or equitable principles, including interim or preliminary relief, and to allocate responsibility for the costs of the arbitration and to award recovery of reasonable attorneys fees and expenses in such manner as is determined to be appropriate by the arbitrator(s).

Table of Contents

- (d) Judgment upon the award rendered by the arbitrator(s) may be entered in any court having personal and subject matter jurisdiction.
- (e) All proceedings under this <u>Section 9.12</u>, and all evidence given or discovered pursuant hereto, shall be maintained in confidence by all parties and by the arbitrator(s) except to the extent required by legal process.
- 9.13 Model Shareholders Consent. By his, her or its execution hereof, each of the Model Shareholders hereby consents to the Merger.

[Remainder of Page Intentionally Left Blank.]

38

Table of Contents

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed and delivered as of the date first above written.

MODEL REORG, INC.

By: /s/ Michael W. Katz Name: Michael W. Katz Title: Executive Vice President

E COM VENTURES, INC.

By: /s/ Donovan Chin Name: Donovan Chin

Title: Chief Financial Officer and Secretary

MODEL REORG ACQUISITION LLC

By its Sole Member E Com Ventures, Inc.

By: /s/ Donovan Chin Name: Donovan Chin

Title: Chief Financial Officer and Secretary

[Model Shareholder Signatures Follow]

Table of Contents 298

39

The undersigned, Model Shareholders, solely for purposes of agreeing to be bound by Section 5.10 and 5.11 and Article IX of the Agreement and acknowledging the other terms and provisions of the Agreement have executed this Agreement as of the date first above written.

/s/ Glenn Nussdorf Glenn Nussdorf

/s/ Stephen Nussdorf Stephen Nussdorf

/s/ Arlene Nussdorf Arlene Nussdorf

/s/ Rene Garcia Rene Garcia

THE JACQUELINE MARIE GARCIA 2006 FAMILY TRUST u/t/a dated October 30, 2006

By: /s/ Rene Garcia

RENE A. GARCIA, Trustee

By: /s/ Rafael Villoldo

RAFAEL VILLOLDO, Trustee

THE CAROLINA MARIE GARCIA 2006

FAMILY TRUST u/t/a dated October 30, 2006

By: /s/ Rene Garcia

RENE A. GARCIA, Trustee

By: /s/ Rafael Villoldo

RAFAEL VILLOLDO, Trustee

IRREVOCABLE TRUST FOR VICTOR GARCIA u/t/a dated October 30, 2006

By: /s/ Rene Garcia

RENE A. GARCIA, Trustee

By: /s/ Rafael Villoldo

RAFAEL VILLOLDO, Trustee

40

Schedule 2.7

Allocation of Merger Consideration and Escrow Shares

The Merger Consideration and Escrow Shares shall be allocated among the holders of Model Common Stock based on the percentages set forth below:

Glenn Nussdorf	29.5838%
Stephen Nussdorf	29.5838%
Arlene Nussdorf	29.5838%
Rene Garcia	5.8492%
Carolina Marie Garcia	1.7998%
2006 Trust u/t/a dated	
October 30, 2006	
Jacqueline Marie Garcia	1.7998%
2006 Trust u/t/a dated	
October 30, 2006	
Irrevocable Trust	1.7998%
for Victor Garcia u/t/a	
dated October 30, 2006	

41

Exhibit A

THE WARRANT REPRESENTED BY THIS CERTIFICATE AND THE SECURITIES

ISSUABLE UPON EXERCISE HEREOF HAVE NOT BEEN REGISTERED UNDER THE

SECURITIES ACT OF 1933 AND MAY NOT BE OFFERED OR SOLD EXCEPT

PURSUANT TO A REGISTRATION STATEMENT UNDER SUCH ACT OR AN OPINION

OF COUNSEL REASONABLY SATISFACTORY TO THE ISSUER THAT AN EXEMPTION

FROM REGISTRATION UNDER SUCH ACT IS AVAILABLE. THE TRANSFER OF THE

E COM VENTURES, INC.

WARRANT IS FURTHER RESTRICTED AS DESCRIBED HEREIN.

Common Stock Purchase Warrant

No. A-[] [] Shares Date of Issuance: [], 2008

This Common Stock Purchase Warrant (this Warrant) certifies that, for value received, [] (the Holder), is entitled, upon the terms and subject to the limitations and exceptions set forth herein, at any time and from time to time during the period beginning at 12:01 a.m. New York time on the third (3rd) anniversary of the Date of Issuance (the Initial Exercise Date) and ending at 5:00 p.m. New York time on the tenth (10th) anniversary of the Date of Issuance (the Termination Date), to subscribe for and purchase from E Com Ventures, Inc., a Florida corporation (the Company), up to an aggregate of [] ([]) shares (subject to adjustment from time to time pursuant to the terms hereof) (the Warrant Shares) of the Company s Common Stock, \$0.01 par value (the Common Stock). The purchase price of one Warrant Share (the Exercise Price) under this Warrant shall be \$23.94 (subject to adjustment from time to time pursuant to the terms hereof). This Warrant is one of a series of warrants issued as of the date hereof (the Warrants) pursuant to the Agreement and Plan of Merger dated as of December 21, 2007, by and among the Company, Model Reorg, Inc., the stockholders of Model Reorg, Inc. and Model Reorg Acquisition LLC (the Merger Agreement).

1. Exercise of Warrant.

- (a) The purchase rights represented by this Warrant may be exercised by delivery of the Notice of Exercise Form annexed hereto duly executed, at the office of the Company specified in Section 13(j) hereto and upon payment of the Exercise Price of the shares thereby purchased by wire transfer or cashier s check drawn on a United States bank. Promptly after the Date of Exercise, subject to (i) compliance with securities laws applicable to the issuance and sale of the Warrant Shares and (ii) the Company s receipt of this Warrant, the Company s transfer agent shall transmit certificates for Warrant Shares purchased hereunder to the Holder at the address specified in the Notice of Exercise Form. This Warrant shall be deemed to have been exercised and the Holder shall be deemed to have become the holder of record of such shares for all purposes as of the Date of Exercise, notwithstanding that the transfer books of the Company shall then be closed or certificates representing such Warrant Shares shall not then have been actually delivered. A Date of Exercise means the date on which the Holder shall have delivered to the Company the Notice of Exercise Form appropriately completed and duly signed and payment of the Exercise Price for the number of Warrant Shares so indicated by the Holder to be purchased.
- (b) If this Warrant shall have been exercised in part, the Company shall, promptly following receipt of this Warrant, deliver to the Holder at the address specified by the Holder in the Notice of Exercise Form a new Warrant evidencing the rights of the Holder to purchase the unpurchased Warrant Shares called for by this Warrant, which new Warrant shall in all other respects be identical with this Warrant.
- 2. <u>No Fractional Shares or Scrip</u>. No fractional shares or scrip representing fractional shares shall be issued upon the exercise of this Warrant. In lieu of any fraction of a share that the Holder would otherwise be entitled to purchase upon such exercise, such fraction shall be rounded up (for fractions of one-half or greater) or down (for fractions of less than one-half) to the nearest whole share.

Table of Contents

- 3. <u>Charges, Taxes and Expenses</u>. Certificates for Warrant Shares shall be issued without charge to the Holder for any issue or transfer tax or other incidental expense in respect of the issuance of such certificate, all of which taxes and expenses shall be paid by the Company.
- 4. Restriction on Transfer of Warrant and Warrant Shares. The Holder may not sell, transfer, assign, or otherwise dispose of this Warrant or any part hereof except (a) by will or by the laws of descent and distribution or (b) during the Holder's lifetime, to any member of the Holder's family or to any trust or other entity established for estate planning purposes, the majority of the beneficial interests in which are held by members of the Holder's family. Other than pursuant to clause (a) or (b), the Holder (including any such subsequent Holder) may not sell, transfer, assign, or otherwise dispose of this Warrant or any part hereof to any other person without the prior, express written consent of the Company. The transfer of the Warrant Shares shall be restricted to the extent required by applicable securities laws, and the certificate or certificates evidencing the Warrant Shares shall bear the following legend:

THE ISSUANCE OF THE SHARES REPRESENTED BY THIS CERTIFICATE HAS NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933 AND SUCH SHARES MAY NOT BE OFFERED OR SOLD EXCEPT PURSUANT TO A REGISTRATION STATEMENT UNDER SUCH ACT OR AN OPINION OF COUNSEL REASONABLY SATISFACTORY TO THE ISSUER THAT AN EXEMPTION FROM REGISTRATION UNDER SUCH ACT IS AVAILABLE.

- 5. No Rights as Shareholder until Exercise. This Warrant does not entitle the Holder to any voting rights or other rights as a shareholder of the Company before the exercise hereof.
- 6. Loss, Theft, Destruction or Mutilation of Warrant. Upon receipt by the Company of evidence reasonably satisfactory to it of the loss, theft, destruction or mutilation of this Warrant or any stock certificate representing Warrant Shares, and in case of loss, theft or destruction, of indemnity or security reasonably satisfactory to it (which, in the case of the Warrant, shall not include the posting of any bond), and upon surrender and cancellation of such Warrant or stock certificate, if mutilated, the Company will make and deliver a new Warrant or stock certificate of like tenor and dated as of such cancellation, in lieu of such Warrant or stock certificate.
- 7. Acceleration of Initial Exercise Date. At any time before the third (3rd) anniversary of the Date of Issuance, the Initial Exercise Date shall be accelerated and shall be deemed to occur immediately before the consummation of any (a) merger, share exchange or consolidation of the Company with or into any other person or entity, (b) sale of all or substantially all of the Company s assets in one or a series of related transactions, or (c) tender offer or exchange offer pursuant to which holders of not less than ninety percent (90%) of the then outstanding shares of Common Stock tender or exchange their shares for other securities, cash or property, other than any such transaction as a result of which holders of a majority of the voting securities of the Company outstanding immediately before such transaction would continue to have beneficial ownership, directly or indirectly, of a majority of the combined voting power of the Company or the surviving entity (including any person that, as a result of such transaction, owns all or substantially all of the Company s assets either directly or through one or more subsidiaries) outstanding immediately after such transaction.
- 8. Adjustments of Exercise Price and Number of Warrant Shares. The number and kind of securities purchasable upon the exercise of this Warrant and the Exercise Price shall be subject to adjustment from time to time. In case the Company shall (i) pay a dividend in shares of Common Stock or make a distribution in shares of Common Stock to holders of its outstanding Common Stock, (ii) subdivide its outstanding shares of Common Stock into a greater number of shares, (iii) combine its outstanding shares of Common Stock into a smaller number of shares of Common Stock, or (iv) issue any shares of its capital stock in a reclassification of the Common Stock or a compulsory share exchange, then the number of Warrant Shares purchasable upon exercise of this Warrant immediately prior thereto shall be adjusted so that the Holder shall be entitled to receive the kind and number of Warrant Shares or other securities of the Company that he or she would have owned or have been entitled to receive had such Warrant been exercised in advance thereof. Upon each such adjustment of the kind

2

Table of Contents

and number of Warrant Shares or other securities of the Company that are purchasable hereunder, the Holder shall thereafter be entitled to purchase the number of Warrant Shares or other securities resulting from such adjustment at an Exercise Price per Warrant Share or other security obtained by multiplying the Exercise Price in effect immediately before such adjustment by the number of Warrant Shares purchasable hereunder immediately before such adjustment, and dividing such product by the number of Warrant Shares or other securities of the Company resulting from such adjustment. An adjustment made pursuant to this paragraph shall become effective immediately after the effective date of such event retroactive to the record date, if any, for such event, subject to the effect of any exercise hereof between such record date and such effective date.

- 9. Fundamental Transactions. If, at any time while this Warrant is outstanding, (a) the Company effects any merger, share exchange or consolidation of the Company with or into any other person or entity, (b) the Company effects any sale of all or substantially all of its assets in one or a series of related transactions, (c) any tender offer or exchange offer (whether by the Company or another person or entity) is completed pursuant to which holders of Common Stock are permitted to tender or exchange their shares for other securities, cash or property, or (d) the Company effects any reclassification of the Common Stock or any compulsory share exchange pursuant to which the Common Stock is effectively converted into or exchanged for securities of any other issuer, cash or property (in any such case, a Fundamental Transaction), then the Holder shall have the right thereafter to receive, upon exercise of this Warrant, the same amount and kind of securities, cash or property as it would have been entitled to receive upon the occurrence of such Fundamental Transaction if it had been, immediately before such Fundamental Transaction, the holder of the number of Warrant Shares then issuable upon exercise in full of this Warrant (the Alternate Consideration). For purposes of any such exercise, the determination of the Exercise Price shall be appropriately adjusted to apply to such Alternate Consideration based on the amount of Alternate Consideration issuable in respect of one share of Common Stock in such Fundamental Transaction, and the Company shall apportion the Exercise Price among the Alternate Consideration in a reasonable manner reflecting the relative value of any different components of the Alternate Consideration. If holders of Common Stock are given any choice as to the securities, cash or property to be received in a Fundamental Transaction, then the Holder shall be given the same choice as to the Alternate Consideration it receives upon any exercise of this Warrant following such Fundamental Transaction. To the extent necessary to effectuate the foregoing provisions, any successor to the Company or surviving entity in such Fundamental Transaction shall issue to the Holder a new warrant consistent with the foregoing provisions and evidencing the Holder s right to exercise such warrant into Alternate Consideration; provided that this Warrant shall have been cancelled or amended to the extent such cancellation or amendment is necessary so that such new warrant does not unjustly or disproportionately enrich the holder of the new warrant relative to a holder of the number of Shares for which this Warrant is exercisable immediately before such event. The terms of any agreement pursuant to which a Fundamental Transaction is effected shall include terms requiring any such successor or surviving entity to comply with the provisions of this Section 9 and ensuring that this Warrant (or any such replacement security) will be similarly adjusted upon any subsequent transaction analogous to a Fundamental Transaction.
- 10. Notice of Adjustment. Whenever the number of Warrant Shares or number or kind of securities or other property purchasable upon the exercise of this Warrant or the Exercise Price is adjusted, as herein provided, the Company shall promptly give notice thereof to the Holder, which notice shall state the number of Warrant Shares (and other securities or property) purchasable upon the exercise of this Warrant and the Exercise Price of such Warrant Shares (and other securities or property) after such adjustment, setting forth a brief statement of the facts requiring such adjustment and setting forth the computation by which such adjustment was made. Irrespective of any adjustments in the Exercise Price or the number or kind of shares purchasable upon the exercise of this Warrant, Warrants theretofore or thereafter issued may continue to express the same price and number and kind of shares as are stated in this Warrant.
- 11. <u>Notice of Corporate Action</u>. If at any time: (a) the Company shall take a record of the holders of its Common Stock for the purpose of entitling them to receive a dividend or other distribution, or any right to subscribe for or purchase any evidences of its indebtedness, any shares of stock of any class or any other

3

Table of Contents

securities or property, or to receive any other right, or (b) there shall be any capital reorganization of the Company, any reclassification or recapitalization of the capital stock of the Company, or any consolidation, share exchange or merger of the Company with, or any sale, transfer or other disposition of all or substantially all the property, assets or business of the Company to, another person or entity, or (c) there shall be a voluntary or involuntary dissolution, liquidation or winding up of the Company, the Company shall give the Holder (i) at least ten (10) days prior written notice of the date on which a record date shall be selected for such dividend, distribution or right or for determining rights to vote in respect of any such reorganization, reclassification, merger, share exchange, consolidation, sale, transfer, disposition, liquidation or winding up, and (ii) in the case of any such reorganization, reclassification, merger, share exchange, consolidation, sale, transfer, disposition, dissolution, liquidation or winding up, at least ten (10) days prior written notice of the date when the same shall take place. Such notice in accordance with the foregoing clause also shall specify (i) the date on which any such record is to be taken for the purpose of such dividend, distribution or right, the date on which the holders of Common Stock shall be entitled to any such dividend, distribution or right, and the amount and character thereof, and (ii) the date on which any such reorganization, reclassification, merger, share exchange, consolidation, sale, transfer, disposition, dissolution, liquidation or winding up is to take place and the time, if any such time is to be fixed, as of which the holders of Common Stock shall be entitled to exchange their shares for securities or other property deliverable upon such disposition, dissolution, liquidation or winding up.

- 12. Covenants. The Company covenants that:
- (a) during the period the Warrant is outstanding, it will reserve from its authorized and unissued Common Stock a sufficient number of shares to provide for the issuance of the Warrant Shares upon the exercise hereof;
- (b) its issuance of this Warrant shall constitute full authority to its officers who are charged with the duty of executing stock certificates to execute and issue the necessary certificates for the Warrant Shares upon the exercise hereof;
- (c) it will take all such reasonable action as may be necessary to assure that such Warrant Shares may be issued as provided herein without violation of any applicable law or regulation, or of any requirements of the Principal Market (as defined in Section 13(k)) upon which the Common Stock may be listed at such time;
- (d) it will not close its stockholder books or records in any manner that prevents the timely exercise of this Warrant pursuant to the terms hereof;
- (e) all Warrant Shares that may be issued upon exercise hereof will, upon such exercise, be duly authorized, validly issued, fully paid and nonassessable and free from all taxes, liens and charges in respect of the issue thereof;
- (f) except as and to the extent waived or consented to by the Holder, the Company shall not by any action, including, without limitation, amending its articles of incorporation or through any reorganization, transfer of assets, consolidation, merger, dissolution, issue or sale of securities or any other voluntary action, avoid or seek to avoid the observance or performance of any of the terms of this Warrant, but will at all times in good faith assist in the carrying out of all such terms and in the taking of all such actions as may be necessary or appropriate to protect the rights of Holder as set forth in this Warrant against impairment (and without limiting the generality of the foregoing, the Company will (i) not increase the par value of any Warrant Shares above the amount payable therefor upon such exercise immediately before such increase in par value, (ii) take all such action as may be necessary or appropriate in order that the Company may validly and legally issue fully paid and nonassessable Warrant Shares upon the exercise of this Warrant, and (iii) use commercially reasonable efforts to obtain all such authorizations, exemptions or consents from any public regulatory body having jurisdiction thereof as may be necessary to enable the Company to perform its obligations under this Warrant); and

4

Table of Contents

(g) before taking any action that would result in an adjustment in the number of Warrant Shares for which this Warrant is exercisable or in the Exercise Price, the Company shall obtain all such authorizations or exemptions thereof, or consents thereto, as may be necessary from any public regulatory body or bodies having jurisdiction thereof.

13. Miscellaneous.

- (a) <u>Saturdays, Sundays, Holidays, etc.</u> If the last or appointed day for the taking of any action or the expiration of any right required or granted herein (including without limitation the Termination Date) is not a Business Day, then such action may be taken or such right may be exercised on the next succeeding Business Day. As used in this Warrant, a Business Day is any day that is not a Saturday, a Sunday or a legal holiday in the States of New York or Florida.
- (b) Governing Law; Dispute Resolution. All questions concerning the construction, validity, enforcement and interpretation of this Warrant shall be governed by the internal laws of the State of New York, without giving effect to any choice of law or conflict of law provision or rule (whether of the State of New York or any other jurisdictions) that would cause the application of the laws of any jurisdictions other than the State of New York. The dispute resolution provisions of the Merger Agreement shall apply to all disputes arising hereunder as if set forth in full herein. EACH PARTY HEREBY IRREVOCABLY WAIVES ANY RIGHT IT MAY HAVE, AND AGREES NOT TO REQUEST, A JURY TRIAL FOR THE ADJUDICATION OF ANY DISPUTE HEREUNDER OR IN CONNECTION HEREWITH OR ARISING OUT OF THIS WARRANT OR ANY TRANSACTION CONTEMPLATED HEREBY.
- (c) <u>Nonwaiver</u>. No course of dealing or any delay or failure to exercise any right hereunder on the part of the Holder shall operate as a waiver of such right or otherwise prejudice the Holder s rights, powers or remedies.
- (d) <u>Limitation of Liability</u>. No provision hereof, in the absence of any affirmative action by the Holder to exercise this Warrant or purchase Warrant Shares, and no enumeration herein of the rights or privileges of the Holder, shall give rise to any liability of the Holder for the purchase price of any Common Stock or as a stockholder of the Company, whether such liability is asserted by the Company or by creditors of the Company.
- (e) <u>Remedies</u>. The Holder, in addition to being entitled to exercise all rights granted by law, including recovery of damages, will be entitled to specific performance of its rights under this Warrant. The Company agrees that monetary damages would not be adequate compensation for any loss incurred by reason of a breach by it of the provisions of this Warrant and hereby agrees to waive the defense in any action for specific performance that a remedy at law would be adequate.
- (f) <u>Successors and Assigns</u>. Subject to applicable securities laws, this Warrant and the rights and obligations evidenced hereby shall inure to the benefit of and be binding upon the successors of the Company and the successors and permitted assigns of the Holder. The provisions of this Warrant are intended to be for the benefit of all Holders from time to time of this Warrant and shall be enforceable by any such Holder or holder of Warrant Shares.
- (g) Amendment. This Warrant may be modified or amended or any provision hereof waived with the written consent of the Company and the holders of Warrants issued under the Merger Agreement representing two-thirds of the Warrant Shares issuable under Warrants then outstanding as of the date such consent is sought; provided, however, that (i) no such amendment shall adversely affect any holder of any such Warrant differently than it affects all other such holders unless such holder consents thereto and (ii) no amendment may increase the Exercise Price, decrease the number of shares or class of shares obtainable upon exercise of this Warrant or decrease the time period in which this Warrant can be exercised without the written consent of the Holder.
- (h) <u>Severability</u>. Wherever possible, each provision of this Warrant shall be interpreted in such manner as to be effective and valid under applicable law, but if any provision of this Warrant shall be prohibited by

5

Table of Contents

or invalid under applicable law, such provision shall be ineffective to the extent of such prohibition or invalidity, without invalidating the remainder of such provisions or the remaining provisions of this Warrant.

- (i) <u>Headings</u>. The headings used in this Warrant are for the convenience of reference only and shall not, for any purpose, be deemed a part of this Warrant.
- (j) Notices. Any and all notices or other communications or deliveries hereunder (including without limitation any Notice of Exercise Form) shall be in writing and shall be deemed given and effective on the earliest of (i) the date of transmission, if such notice or communication is delivered via electronic mail or facsimile at the electronic mail address or facsimile number specified in this section before 5:00 p.m. (New York City time) on a Business Day, (ii) the next Business Day after the date of transmission, if such notice or communication is delivered via electronic mail or facsimile at the electronic mail address or facsimile number specified in this section on a day that is not a Business Day or later than 5:00 p.m. (New York City time) on any Business Day, (iii) the Business Day following the date of mailing, if sent by nationally recognized overnight courier service, or (iv) upon actual receipt by the party to whom such notice is required to be given. The addresses for such communications shall be:
 - (x) if to the Company, to:

E Com Ventures, Inc.
Attn: []
251 International Parkway
Sunrise, Florida 33325
Facsimile: []
Email: []

(y) if to the Holder, to the address, facsimile number, or email address, respectively, set forth in [].

IN WITNESS WHEREOF, the Company has caused this Warrant to be executed by its officer thereunto duly authorized.

Dated: [], 2008

E COM VENTURES, INC.

By: Name: Title:

6

NOTICE OF EXERCISE

To: E Com Ventures, Inc.

The undersigned hereby elects to purchase Warrant Shares of E Com Ventures, Inc. pursuant to the terms of the attached Warrant, and tenders herewith payment of the exercise price of such Warrant Shares in full, in lawful money of the United States, together with all applicable transfer taxes, if any.

Please issue a certificate or certificates representing said Warrant Shares in the name of the undersigned.

The Warrant Shares shall be delivered to the following:

SIGNATURE OF HOLDER

Dated as of:

HOLDER

By:

Print Name:

7

ASSIGNMENT FORM

{Assignment of the foregoing Warrant is restricted as described therein.}

(To assign the foregoing Warrant, execute this form and supply required information.

Do not use this form to exercise the Warrant.)

FOR VALUE RECEIVED, the foregoing Warrant and all rights evidenced thereby are hereby assigned to:

Name of Assignee:
Assignee s Address:
Date:
Holder s Signature:
Holder s Address:

Signature Guaranteed:
NOTE: The signature to this Assignment Form must correspond with the name as it appears on the face of the Warrant, without alteration or enlargement or any change whatsoever, and must be guaranteed by a participant in the Medallion guaranty program. Officers of corporations and those acting in a fiduciary or other representative capacity must file proper evidence of authority to assign the foregoing Warrant.

8

Exhibit B

E COM VENTURES, INC.

REGISTRATION RIGHTS AGREEMENT

THIS REGISTRATION RIGHTS AGREEMENT (this Agreement) is made as of the day of , 2008, by and among E Com Ventures, Inc., a Florida corporation (the Company), and each of the shareholders of Model Reorg, Inc., a New York corporation (Model Reorg), listed on Schedule A hereto.

RECITALS

WHEREAS, the Company, Model Reorg Acquisition LLC, a Delaware limited liability company, Model Reorg and the Shareholders have entered into an Agreement and Plan of Merger dated as of December 21, 2007 (the Merger Agreement) pursuant to which the Shareholders will receive, in exchange for their shares of Model Reorg common stock, shares of the Company s common stock, par value \$0.01 per share (the Common Stock) and Common Stock Purchase Warrants to acquire Common Stock (the Warrants); and

WHEREAS, it is a condition of the Merger Agreement that the parties hereto enter into this Agreement providing for the Company to register the shares of Common Stock issuable to the Shareholders under the Merger Agreement (excluding any Common Stock issued or issuable upon exercise of any of the Warrants) as set forth herein;

NOW, THEREFORE, the parties hereby agree as follows:

- **1. DEFINITIONS**. For purposes of this Agreement:
- 1.1 Common Stock means shares of the Company s common stock, par value \$0.01 per share.
- 1.2 Company means E Com Ventures, Inc., any successor thereto by operation of law, and any other company whose shares are issued upon conversion of, or whose shares are exchanged for, substantially all of the outstanding shares of Common Stock pursuant to a merger, consolidation or other recapitalization transaction.
- 1.3 Damages means any loss, damage, or liability (joint or several) to which a party hereto may become subject under the Securities Act, the Exchange Act, or other federal or state law, insofar as such loss, damage, or liability (or any action in respect thereof) arises out of or is based upon (i) any untrue statement or alleged untrue statement of a material fact contained in any registration statement of the Company filed pursuant hereto, including any preliminary prospectus or final prospectus contained therein or any amendments or supplements thereto; (ii) an omission or alleged omission to state therein a material fact required to be stated therein, or necessary to make the statements therein not misleading; or (iii) any violation or alleged violation by the indemnifying party (or any of its agents or affiliates) of the Securities Act, the Exchange Act, any state securities law, or any rule or regulation promulgated under the Securities Act, the Exchange Act, or any state securities law based upon, or arising out of, any of such party s obligations arising hereunder.
- 1.4 Exchange Act means the Securities Exchange Act of 1934, as amended from time to time, and the rules and regulations promulgated thereunder.
- 1.5 Excluded Registration means (i) a registration of the sale of securities to directors, employees or consultants of the Company or a subsidiary pursuant to a bona fide compensatory arrangement; (ii) a registration relating to an SEC Rule 145 transaction; (iii) a registration on any form that does not include substantially the same information as would be required to be included in a registration statement covering the sale of the Registrable Securities; (iv) a registration solely of debt or nonconvertible preferred stock; or (v) a registration in which the only Common Stock being registered is Common Stock issuable upon conversion of debt securities that are also being registered.

Table of Contents

- 1.6 GAAP means generally accepted accounting principles in the United States.
- 1.7 Immediate Family Member means a child, stepchild, grandchild, parent, stepparent, grandparent, spouse, domestic partner, sibling, mother-in-law, father-in-law, son-in-law, daughter-in-law, brother-in-law, or sister-in-law, including adoptive relationships, of a natural person referred to herein.
- 1.8 Majority in Interest means Shareholders holding a majority of the Registrable Securities as to which Shareholders have requested inclusion in a registration under Section 2.1.
- 1.9 Participating Holder means a holder of Company securities that is described in clause (a) of Section 2.8 and that, in compliance with Section 2.8, has acquired rights to have such securities registered under the Securities Act by having them included in a registration statement filed by the Company.
- 1.10 Person means any individual, corporation, partnership, trust, limited liability company, association or other entity.
- 1.11 Registrable Securities means those shares of Common Stock issued pursuant to the Merger Agreement (other than (i) any that have been transferred by a Shareholder in a transaction in which the applicable rights under this Agreement are not assigned pursuant to Section 3.1, (ii) any for which registration rights have terminated pursuant to Section 2.9, and (iii) any Common Stock issued or issuable upon exercise of any of the Warrants) and all shares of Common Stock or other securities issued upon conversion or exchange or otherwise in respect thereof, including without limitation pursuant to any stock dividend, stock split, merger, consolidation or other recapitalization transaction.
- 1.12 Restated Certificate means the Amended and Restated Certificate of Incorporation of the Company.
- 1.13 SEC means the Securities and Exchange Commission.
- 1.14 SEC Rule 144 means Rule 144 promulgated by the SEC under the Securities Act, as in effect from time to time.
- 1.15 SEC Rule 145 means Rule 145 promulgated by the SEC under the Securities Act, as in effect from time to time.
- 1.16 Securities Act means the Securities Act of 1933, as amended from time to time, and the rules and regulations promulgated thereunder.
- 1.17 Selling Expenses means all underwriting discounts, selling commissions, and stock transfer taxes applicable to the sale of Registrable Securities, and the fees and expenses of counsel to the selling Shareholders.
- 1.18 Shareholder means a shareholder of Model Reorg listed on Schedule A and his, her or its respective successors and other assigns complying with Section 3.1.
- 2. **REGISTRATION RIGHTS**. The Company covenants and agrees as follows:
- 2.1 <u>Company Registration</u>. Promptly after the Company determines to register (including, for this purpose, a registration by the Company for holders of Common Stock other than the Shareholders) any of its securities under the Securities Act in connection with the public offering of such securities solely for cash (other than in an Excluded Registration), the Company shall give each Shareholder written notice of such proposed registration. Upon the request of any Shareholder given within twenty (20) days after such notice, the Company shall, subject to the provisions of Section 2.2, cause to be registered all of the Registrable Securities that each such Shareholder has requested to be included in such registration.
- 2.2 Underwriting Requirements.
- (a) If the Shareholders intend to distribute the Registrable Securities covered by their registration request by means of an underwriting, they shall so advise the Company and will select (by a Majority

Table of Contents

in Interest) such underwriter(s) as shall be reasonably acceptable to the Company. In such event, the right of any Shareholder to include such Shareholder s Registrable Securities in such registration shall be conditioned upon such Shareholder s participation in such underwriting and the inclusion of such Shareholder s Registrable Securities in the underwriting to the extent provided herein. All Shareholders proposing to distribute their securities through such underwriting shall (together with the Company as provided in Section 2.3(e)) enter into an underwriting agreement in customary form with the underwriter(s) selected for such underwriting.

- (b) Notwithstanding any other provision of this Section 2.2, if the underwriter(s) referred to in clause (a) advise(s) such Shareholders in writing that marketing factors require a limitation on the number of shares to be underwritten, the maximum number of Registrable Securities and securities to be registered on behalf of all Participating Holders that may be included in the underwriting shall be allocated among such Shareholders and Participating Holders in proportion (as nearly as practicable) to the number of Registrable Securities for which each Shareholder and Participating Holder requested registration; provided, however, that the number of Registrable Securities held by the Shareholders to be included in such underwriting shall not be reduced unless all other securities (other than securities to be sold by the Company and Participating Holders) are first entirely excluded from the underwriting.
- (c) For purposes of any apportionment under Section 2.2(b), a Shareholder and any of such Shareholder s transferees shall be deemed to be a single Shareholder, and any pro rata reduction with respect to such Shareholder shall be based upon the aggregate number of Registrable Securities owned by all Persons included in such Shareholder, as defined in this sentence.
- 2.3 <u>Obligations of the Company</u>. Whenever required under this Section 2 to effect the registration of any Registrable Securities, the Company shall, as expeditiously as reasonably possible:
- (a) prepare and file with the SEC a registration statement with respect to such Registrable Securities and use its best efforts to cause such registration statement to become effective and, upon the request of a Majority in Interest, keep such registration statement effective for a period of up to one hundred twenty (120) days or, if earlier, until the distribution contemplated in the registration statement has been completed; provided, however, that (i) such 120-day period shall be extended for a period of time equal to the period the Shareholders refrain, at the written request of an underwriter of Common Stock (or other securities) of the Company, from selling any securities included in such registration and (ii) in the case of any registration of Registrable Securities that are intended to be offered on a continuous or delayed basis, such 120-day period shall be extended for up to an additional sixty (60) days, if necessary, to keep the registration statement effective until all such Registrable Securities are sold;
- (b) subject to clause (i), prepare and file with the SEC such amendments and supplements to such registration statement, and the prospectus used in connection with such registration statement, as may be necessary to keep such registration statement effective for the period specified in clause (a) and to comply with the Securities Act in order to enable the disposition of all securities covered by such registration statement;
- (c) furnish to the selling Shareholders such numbers of copies of the prospectus, including a preliminary prospectus, included in the registration statement, and such other documents, as the Shareholders may reasonably request in order to facilitate their disposition of their Registrable Securities;
- (d) use its commercially reasonable efforts to register and qualify the securities covered by such registration statement under such other securities or blue-sky laws of such jurisdictions as shall be reasonably requested by the selling Shareholders; provided that the Company shall not be required to qualify to do business or to file a general consent to service of process in any such states or jurisdictions unless the Company is already subject to service in such jurisdiction and except as may be required by the Securities Act;

3

Table of Contents

- (e) in the event of any underwritten public offering, enter into and perform its obligations under an underwriting agreement, in usual and customary form, with the underwriter(s) of such offering;
- (f) promptly make available (subject to reasonable confidentiality requirements) for inspection by the selling Shareholders, any underwriter(s) participating in any disposition pursuant to such registration statement, and any attorney or accountant or other agent retained by any such underwriter or selected by the selling Shareholders, all financial and other records, pertinent corporate documents, and properties of the Company, and cause the Company s officers, directors, employees, and independent accountants to supply all information reasonably requested by any such seller, underwriter, attorney, accountant, or agent, in each case, as necessary or reasonably advisable to verify the accuracy of the information in such registration statement and to conduct appropriate due diligence in connection therewith;
- (g) notify each selling Shareholder, promptly after the Company receives notice thereof, of the time when such registration statement has been declared effective or a supplement to any prospectus forming a part of such registration statement has been filed;
- (h) after such registration statement becomes effective, notify each selling Shareholder of any request by the SEC that the Company amend or supplement such registration statement or prospectus; and
- (i) immediately notify each seller of Registrable Securities and each underwriter under such registration statement, at any time when a prospectus relating thereto is required to be delivered under the Securities Act, of the happening of any event of which the Company has knowledge as a result of which the prospectus contained in such registration statement, as then in effect, includes an untrue statement of a material fact or omits to state a material fact required to be stated therein or necessary to make the statements therein not misleading in light of the circumstances then existing, and, at the request of a Majority in Interest, the Company shall promptly prepare a supplement or amendment to such prospectus so that, as thereafter delivered to the purchasers of such Registrable Securities, such prospectus shall not include an untrue statement of a material fact or omit to state any material fact necessary to make the statements therein not misleading; provided that the Company may postpone for up to ninety (90) days the delivery of any such supplement or amendment if the Company s Board of Directors determines in good faith that disclosure of the new information to be contained therein would reasonably be expected to have a material adverse effect on (i) any proposal or plan by the Company or any of its affiliates to engage in any acquisition of assets (other than in the ordinary course of business) or any merger, consolidation, tender offer, reorganization or similar transaction or (ii) any pending or threatened litigation to which the Company is, or is threatened to be made, a party; and
- (j) in the case of an underwritten offering, use its best efforts to furnish, at the request of any Shareholder whose Registrable Securities are included in the registration statement, on the date on which such Registrable Securities are sold to the underwriter, (i) an opinion, dated such date, of the counsel representing the Company for the purposes of such registration, in form and substance as is customarily given to underwriters in an underwritten public offering, addressed to the underwriters and such Shareholder and (ii) a comfort letter dated such date, from the independent certified public accountants of the Company, in form and substance as is customarily given by independent certified public accountants to underwriters in an underwritten public offering, addressed to the underwriters, if any.
- 2.4 <u>Furnish Information</u>. It shall be a condition precedent to the obligations of the Company to take any action pursuant to this Section 2 with respect to the Registrable Securities of any selling Shareholder that such Shareholder shall furnish to the Company such information regarding such Shareholder, the Registrable Securities held by such Shareholder, and the intended method of disposition of such securities as is reasonably required to effect the registration of such Shareholder s Registrable Securities.
- 2.5 Expenses of Registration. All expenses (other than Selling Expenses) incurred in connection with registrations, filings, or qualifications pursuant to this Section 2, including all registration, filing, and

4

Table of Contents

qualification fees, printers and accounting fees, and fees and disbursements of counsel for the Company, shall be borne and paid by the Company, whether or not any such registration or qualification becomes effective. All Selling Expenses relating to Registrable Securities registered pursuant to this Section 2 shall be borne and paid by the Shareholders pro rata on the basis of the number of Registrable Securities registered on their behalf.

- 2.6 <u>Withdrawal of Registration</u>. The Company shall have the right to terminate or withdraw any registration initiated by it before the effective date thereof, whether or not any Shareholder has elected to include Registrable Securities in such registration, without liability of the Company to any holder of Registrable Securities except for payment of registration expenses as provided in Section 2.5. Any Shareholder may withdraw any Registrable Securities from any registration before the effective date of such registration without liability to the Company or any other holder of Common Stock except for payment of such Shareholder s portion of any Selling Expenses incurred with respect to the period before such withdrawal.
- 2.7 <u>Indemnification</u>. If any Registrable Securities are included in a registration statement under this Section 2:
- (a) To the extent permitted by law, the Company will indemnify and hold harmless each selling Shareholder, and the partners, members, officers, directors, and shareholders of each such Shareholder, legal counsel for each such Shareholder, any underwriter (as defined in the Securities Act) for each such Shareholder, and each Person, if any, who controls such Shareholder, underwriter or other Person within the meaning of the Securities Act or the Exchange Act, against any Damages, and the Company will pay to each such Shareholder, underwriter, controlling Person, or other aforementioned Person any legal fees and other expenses reasonably incurred thereby in connection with investigating or defending any claim or proceeding from which Damages may result, as such expenses are incurred; provided, however, that the indemnity agreement contained in this Section 2.7(a) shall not apply to amounts paid in settlement of any such claim or proceeding if such settlement is effected without the consent of the Company, which consent shall not be unreasonably withheld, nor shall the Company be liable for any Damages to the extent that they arise out of or are based upon actions or omissions made in reliance upon and in conformity with written information furnished by or on behalf of any such Shareholder, underwriter, controlling Person, or other aforementioned Person expressly for use in connection with such registration.
- (b) To the extent permitted by law, each selling Shareholder, severally and not jointly, will indemnify and hold harmless the Company, and each of its directors, each of its officers who has signed the registration statement, each Person (if any), who controls the Company within the meaning of the Securities Act, legal counsel for the Company, any underwriter (as defined in the Securities Act), any other Shareholder selling securities in such registration statement, and any controlling Person of any such underwriter, other Shareholder or other Person, against any Damages, in each case only to the extent that such Damages arise out of or are based upon actions or omissions made in reliance upon and in conformity with written information furnished by or on behalf of such selling Shareholder expressly for use in connection with such registration; and each such selling Shareholder will pay to the Company and each other aforementioned Person any legal fees and other expenses reasonably incurred thereby in connection with investigating or defending any claim or proceeding from which Damages may result, as such expenses are incurred; provided, however, that the indemnity agreement contained in this Section 2.7(b) shall not apply to amounts paid in settlement of any such claim or proceeding if such settlement is effected without the consent of the Shareholder, which consent shall not be unreasonably withheld; and provided further that in no event shall the aggregate amounts payable by any Shareholder (net of any Selling Expenses paid by such Shareholder).
- (c) Promptly after receipt by an indemnified party under this Section 2.7 of notice of the commencement of any action (including any governmental action) for which a party may be entitled to

5

Table of Contents

indemnification hereunder, such indemnified party will, if a claim in respect thereof is to be made against any indemnifying party under this Section 2.7, give the indemnifying party notice of the commencement thereof. The indemnifying party shall have the right to participate in such action and, to the extent the indemnifying party so desires, participate jointly with any other indemnifying party to which notice has been given, and to assume the defense thereof with counsel reasonably mutually satisfactory to the parties; provided, however, that an indemnified party (together with all other indemnified parties that may be represented without conflict by one counsel) shall have the right to retain one separate counsel, with the reasonable fees and expenses to be paid by the indemnifying party, if representation of such indemnified party by the counsel retained by the indemnifying party would be inappropriate due to actual or potential differing interests between such indemnified party and any other party represented by such counsel in such action. The failure to give notice to the indemnifying party within a reasonable time of the commencement of any such action will not relieve such indemnifying party of any liability to the indemnifying party under this Section 2.7, except to the extent, and only to the extent, that such failure actually and materially prejudices the indemnifying party s ability to defend such action. The failure to give notice to the indemnifying party will not relieve it of any liability that it may have to any indemnified party otherwise than under this Section 2.7.

- (d) To provide for just and equitable contribution to joint liability under the Securities Act in any case in which either (i) any party otherwise entitled to indemnification hereunder makes a claim for indemnification pursuant to this Section 2.7 but it is judicially determined (by the entry of a final judgment or decree by a court of competent jurisdiction and the expiration of time to appeal or the denial of the last right of appeal) that such indemnification may not be enforced in such case, notwithstanding the fact that this Section 2.7 provides for indemnification in such case or (ii) contribution under the Securities Act may be required on the part of any party hereto for which indemnification is provided under this Section 2.7, then, and in each such case, such parties will contribute to the aggregate losses, claims, damages, liabilities, or expenses to which they may be subject (after contribution from others) in such proportion as is appropriate to reflect the relative fault of each of the indemnifying party and the indemnified party in connection with the statements, omissions, or other actions that resulted in such loss, claim, damage, liability, or expense, as well as to reflect any other relevant equitable considerations. The relative fault of the indemnifying party and of the indemnified party shall be determined by reference to, among other things, whether the untrue or allegedly untrue statement of a material fact, or the omission or alleged omission of a material fact, relates to information supplied by the indemnifying party or by the indemnified party and the parties relative intent, knowledge, access to information and opportunity to correct or prevent such statement or omission; provided, however, that, in any such case, no Person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) will be entitled to contribution from any Person who was not guilty of such fraudulent misrepresentation; and provided further that in no event shall the aggregate amounts payable by any Shareholder by way of indemnity or contribution under Sections 2.7(b) and 2.7(d) exceed the proceeds from the offering received by such Shareholder (net of any Selling Expenses) paid by such Shareholder).
- (e) Notwithstanding the foregoing, to the extent that any provision on indemnification and contribution contained in the underwriting agreement entered into in connection with the underwritten public offering is in conflict with any of the foregoing provisions, the provision in the underwriting agreement shall control.
- (f) The obligations of the Company and Shareholders under this Section 2.7 shall survive the completion of any offering of Registrable Securities in a registration under this Section 2, and otherwise shall survive the termination of this Agreement.
- 2.8 <u>Limitations on Subsequent Registration Rights</u>. The Company shall not, without the prior written consent of the holders of a majority of the Registrable Securities then outstanding, enter into any agreement that would (x) allow any holder or prospective holder of any securities of the Company to include such

6

Table of Contents

securities in any registration under the Securities Act or (y) require the Company to comply with any demand or request by or on behalf of such holder or prospective holder for registration of any securities held by or on behalf of such holder or prospective holder unless (a) such holder or prospective holder is (i) an institutional investor or financial institution that, with the approval of the Board of Directors of the Company, has provided bona fide debt or equity financing to the Company and/or (ii) a holder of Company securities issued with the approval of the Board of Directors of the Company as consideration for the acquisition of a business or assets, and such agreement is entered into as a condition of such financing or acquisition and (b) the terms of such agreement allow the Company to comply with Section 2.1 hereof and require such holder or prospective holder to (i) comply with the underwriting requirements in Section 2.2(a) hereof together with the Shareholders and (ii) include such securities in any such registration only pro rata with the Registrable Securities in compliance with Section 2.2(b) hereof. The Company represents and warrants that no such rights exist as of the date of this Agreement, except as pursuant to this Agreement.

2.9 <u>Termination of Registration Rights</u>. The right of any Shareholder to request inclusion of Registrable Securities in any registration pursuant to Section 2.1 shall terminate on the date that such Shareholder is able to sell, during any three-month period without limitation under Rule 144 of the amount of securities to be sold, all of the remaining Registrable Securities held by or then issuable to such Shareholder.

3. MISCELLANEOUS.

- 3.1 Successors and Assigns. The rights of a Shareholder under this Agreement may be assigned (but only subject to assumption of all related obligations) by a Shareholder to a transferee of Registrable Securities; provided that the Company is, within a reasonable time after such transfer, furnished with written notice of the name and address of such transferee and the Registrable Securities with respect to which such rights are being transferred and such transferee agrees in a written instrument delivered to the Company to assume and be bound by the terms and conditions of this Agreement as a Shareholder hereunder. The terms and conditions of this Agreement inure to the benefit of and are binding upon the respective heirs, executors, administrators, legal representatives, successors and permitted assignees of the parties. Nothing in this Agreement, express or implied, is intended to confer upon any party other than the parties hereto or their respective successors and permitted assignees any rights, remedies, obligations or liabilities under or by reason of this Agreement, except as expressly provided herein.
- 3.2 <u>Governing Law</u>. This Agreement (including any claim or controversy arising out of or relating to this Agreement) shall be governed by and construed in accordance with the laws of the State of New York, without regard to conflict of law principles that would result in the application of any law other than the law of the State of New York.
- 3.3 <u>Counterparts</u>. This Agreement may be executed and delivered in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
- 3.4 Notices. All notices and other communications given or made pursuant to this Agreement shall be in writing and shall be deemed effectively given upon the earlier of actual receipt or: (i) when sent, if sent by electronic mail or facsimile during the recipient s normal business hours, and if not sent during normal business hours, then on the recipient s next business day, (ii) five (5) days after having been sent by registered or certified mail, return receipt requested, postage prepaid, or (iii) one (1) business day after the business day of deposit with a nationally recognized overnight courier, freight prepaid, specifying next-day delivery. All communications shall be sent to the respective parties at their addresses as set forth on Schedule A hereto, or in the case of the Company, to the principal office of the Company, or to such email address, facsimile number, or address as subsequently modified by written notice given in accordance with this Section 3.4.
- 3.5 <u>Amendments and Waivers</u>. Any term of this Agreement may be amended and the observance of any term of this Agreement may be waived (either generally or in a particular instance, and either retroactively or prospectively) only with the written consent of the Company and the Shareholders holding a

7

Table of Contents

majority of the Registrable Securities then held by all Shareholders; provided that any provision hereof may be waived by any waiving party on such party s own behalf, without the consent of any other party. Notwithstanding the foregoing, this Agreement may not be amended or terminated and the observance of any term hereof may not be waived with respect to any Shareholder without the written consent of such Shareholder, unless such amendment, termination, or waiver applies to all Shareholders in the same fashion. The Company shall give prompt notice of any amendment or termination hereof or waiver hereunder to any party hereto that did not consent in writing to such amendment, termination, or waiver. Any amendment, termination, or waiver effected in accordance with this Section 3.5 shall be binding on all parties hereto, regardless of whether any such party has consented thereto. No waivers of or exceptions to any term, condition, or provision of this Agreement, in any one or more instances, shall be deemed to be or construed as a further or continuing waiver of any such term, condition, or provision.

- 3.6 Severability. In case any one or more of the provisions contained in this Agreement is for any reason held to be invalid, illegal or unenforceable in any respect, such invalidity, illegality, or unenforceability shall not affect any other provision of this Agreement, and such invalid, illegal, or unenforceable provision shall be reformed and construed so that it will be valid, legal, and enforceable to the maximum extent permitted by law.
- 3.7 Entire Agreement. This Agreement constitutes the full and entire understanding and agreement among the parties with respect to the subject matter hereof, and any other written or oral agreement relating to the subject matter hereof existing between the parties is expressly canceled.
- 3.8 <u>Dispute Resolution</u>. The dispute resolution provisions of the Merger Agreement shall apply to all disputes arising hereunder as if set forth in full herein
- 3.9 Waiver of Jury Trial. EACH PARTY HEREBY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN ANY LEGAL PROCEEDING DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY (WHETHER BASED ON CONTRACT, TORT OR ANY OTHER THEORY). EACH PARTY HEREBY (A) CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER AND (B) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HAVE BEEN INDUCED TO ENTER INTO THIS AGREEMENT AND THE OTHER TRANSACTION DOCUMENTS BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 3.9.
- 3.10 <u>Delays or Omissions</u>. No delay or omission to exercise any right, power, or remedy accruing to any party under this Agreement, upon any breach or default of any other party under this Agreement, shall impair any such right, power, or remedy of such nonbreaching or nondefaulting party, nor shall it be construed to be a waiver of or acquiescence to any such breach or default, or to any similar breach or default thereafter occurring, nor shall any waiver of any single breach or default be deemed a waiver of any other breach or default theretofore or thereafter occurring. All remedies, whether under this Agreement or by law or otherwise afforded to any party, shall be cumulative and not alternative.

8

Table of Contents

IN WITNESS WHEREOF, the parties have executed this Agreement as of the date first written above.

E COM VENTURES, INC.

By: Name: Title:

SHAREHOLDERS:

9

Exhibit B

SCHEDULE A

SHAREHOLDERS

Table of Contents

EXHIBIT C

Affiliate Debt

Maturity date: three years following the Closing Date

Interest rate: Either 4.75% over LIBOR or 3.75% over the prime rate as published in the Eastern Edition of the Wall Street Journal at E Com s election

Subordinated to senior creditor financings, including the financing pursuant to the Commitment Letter

AMENDMENT TO AGREEMENT AND PLAN OF MERGER

This AMENDMENT, dated July 8, 2008 (this <u>Amendment</u>), to the AGREEMENT AND PLAN OF MERGER, dated as of December 21, 2007 (the <u>Original Agreement</u>), is made and entered into by and among Model Reorg, Inc., a New York corporation (<u>Model</u>), E Com Ventures, Inc., a Florida corporation (<u>E Com</u>), Model Reorg Acquisition LLC, a Delaware limited liability company and wholly-owned subsidiary of E Com (<u>Acquisition Sub</u>), and the shareholders of Model who are signatories thereto and hereto. Capitalized terms used in this Amendment and not otherwise defined herein shall have the meanings ascribed to such terms in the Original Agreement.

RECITALS

WHEREAS, Section 9.2 (Amendment or Supplement) of the Original Agreement provides that, at any time prior to the Effective Time, the Original Agreement may be amended in any and all respects by written agreement of the parties thereto, and by action taken by the respective Boards of Directors of E Com and Model;

NOW, THEREFORE, in consideration of the foregoing recital and the agreements herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto hereby agree that the Original Agreement shall be amended as follows:

1. The definition of Agreement in the preamble to the Original Agreement shall be deleted in its entirety and replaced by the following:

(together with the Exhibits and Schedules thereto, as the foregoing may be supplemented, modified, amended, restated or extended from time to time in the manner provided therein, this <u>Agreement</u>)

- 2. The definition of Affiliate Consignment in Section 9.11 (Definitions; Construction) of the Original Agreement shall be deleted in its entirety and all references to Affiliate Consignment throughout the Original Agreement shall be deleted from the Original Agreement as set forth below:
- (a) Section 4.21 (Related Person Transactions) to the Original Agreement shall be amended to remove and for the Affiliate Consignment as an exception from the first sentence of Section 4.21; and
- (b) Section 5.1(a)(iii) (Conduct of Business) to the Original Agreement shall be amended to remove or the Affiliate Consignment as an exception.
- 3. The definition of Affiliate Debt in Section 9.11 (Definitions; Construction) of the Original Agreement shall be deleted in its entirety and replaced by the following:

Affiliate Debt means the subordinated debt payable by Model to (i) certain Model Shareholders or trusts for the benefit of their family members in the amount and on the terms and conditions substantially similar to those described on Exhibit C attached hereto and such other terms as shall be mutually agreeable to such Model Shareholders and E Com and (ii) Quality King Distributors, Inc. that remains outstanding after the consummation of the QKD Debt Restructuring in the amount and on the terms and conditions substantially similar to those described on Exhibit C attached hereto and such other terms as shall be mutually agreeable to Quality King Distributors, Inc., Model and E Com.

- 4. Exhibit C (Affiliate Debt) to the Original Agreement shall be deleted in its entirety and replaced by Exhibit C attached hereto.
- 5. The definition of Commitment Letter in Section 9.11 (Definitions; Construction) of the Original Agreement shall be deleted in its entirety and replaced by the following:

<u>Commitment Let</u>ter shall mean the commitment letter previously provided to the Board of Directors of E Com dated May 16, 2008 between a senior lender and E Com, as the same may be supplemented, modified, amended, restated or extended from time to time.

Table of Contents

- 6. A new definition shall be added to Section 9.11 (Definitions; Construction) of the Original Agreement and shall read as follows:
- OKD Debt Restructuring means the restructuring of the debt payable by Model to Quality King Distributors, Inc. in the amount and on the terms and conditions substantially similar to those described on Exhibit C attached hereto and such other terms as shall be mutually agreeable to Quality King Distributors, Inc., Model and E Com.
- 7. Section 4.5(a) (Undisclosed Liabilities, Etc.) of the Original Agreement shall be deleted in its entirety and replaced by the following:
- (a) The audited consolidated financial statements of Model as of and for the years ended October 31, 2004, 2005, 2006, and 2007, and unaudited consolidated financial statements of Model as of and for the eleven months ended September 30, 2007 (the Model Financial Statements), and all subsequent interim financial statements provided by Model in accordance with Section 5.8(i) before the Closing Date, have been prepared in accordance with GAAP, and fairly present in all material respects the consolidated financial position of Model and its consolidated Subsidiaries as of the dates thereof and the consolidated results of their operations and cash flows for the periods then ended (subject, in the case of unaudited quarterly statements, to normal year-end audit adjustments).
- 8. Section 6.1(d) (Conditions to Each Party s Obligation to Effect the Merger) of the Original Agreement shall be deleted in its entirety and replaced by the following:
- (d) Asset based loans in the amount and on terms substantially similar to those contemplated in the Commitment Letter shall have been obtained by Acquisition Sub and/or E Com and funds shall have been advanced and/or be available thereunder;
- 9. Section 6.1(f) (Conditions to Each Party s Obligation to Effect the Merger) shall be amended to read in its entirety as follows:
- (f) The transactions described in the definition of Affiliate Debt shall have been put in place, including, without limitation, the QKD Debt Restructuring, on or prior to the Closing.
- 10. Section 6.3(d) (Conditions to E Com s and Acquisition Sub s Obligations to Effect the Merger) shall be deleted in its entirety and replaced by the following:
- (d) Each of Model and Quality King Distributors, Inc. shall have executed and delivered a services agreement in a form that is mutually agreeable to Quality King Distributors, Inc., Model and E Com;
- 11. Section 6.3(e) (Conditions to E Com s and Acquisition Sub s Obligations to Effect the Merger) shall be deleted in its entirety.
- 12. Section 6.3(g) (Conditions to E Com s and Acquisition Sub s Obligations to Effect the Merger) shall be renumbered as Section 6.3(e) and all references to Section 6.3(g) in the Original Agreement shall refer to Section 6.3(e);
- 13. The definition of Outside Date in Section 7.1(c)(i) (Termination) shall be amended to be August 31, 2008.
- 14. The address of Model for notice purposes in Section 9.8 (Notices) shall be deleted in its entirety and replaced by the following:

Model Reorg, Inc.

35 Sawgrass Drive, Suite 2

Bellport, NY 11713

Attention: Michael Katz

Facsimile No.: (631) 866-4231

Table of Contents

15. The address of Seller Representative for notice purposes in Section 9.8 (Notices) shall be deleted in its entirety and replaced by the following:

Stephen Nussdorf

c/o Model Reorg, Inc.

35 Sawgrass Drive, Suite 2

Bellport, NY 11713

Facsimile No.: 631-866-4109

- 16. By his, her or its execution hereof, each of the Model Shareholders hereby consents to this Amendment.
- 17. Except as expressly provided in this Amendment, the Original Agreement is, and shall continue to be, in full force and effect in accordance with its terms, without amendment thereto, and is, in all respects, ratified and confirmed.
- 18. This Amendment shall be subject to the provisions set forth in Article 9 (Miscellaneous) of the Original Agreement, which are incorporated herein by this reference with the same force and effect as if set forth herein.

The remainder of this page left blank.

[Signature page follows]

3

IN WITNESS WHEREOF, in accordance with Section 9.2 (Amendment or Supplement) of the Original Agreement, the parties hereto have caused this Amendment to be duly executed, as of the date first above written.

MODEL REORG, INC.

By: /s/ MICHAEL W. KATZ

Michael W. Katz Chief Executive Officer

E COM VENTURES, INC.

By: /s/ Donovan Chin

Donovan Chin

Chief Financial Officer and Secretary

MODEL REORG ACQUISITION LLC

By E Com Ventures, Inc., its sole member

By: /s/ Donovan Chin

Donovan Chin

Chief Financial Officer and Secretary

[Model Shareholder Signatures Follow]

The undersigned, Model Shareholders, acknowledging the terms and provisions of this Amendment, have executed this Amendment as of the date first above written.

- /s/ Glenn Nussdorf Glenn Nussdorf
- /s/ Stephen Nussdorf Stephen Nussdorf
- /s/ Arlene Nussdorf
 Arlene Nussdorf
 - /s/ RENE GARCIA
 Rene Garcia

THE JACQUELINE MARIE GARCIA 2006

FAMILY TRUST u/t/a dated October 30, 2006

By /s/ Rene A. Garcia, Trustee RENE A. GARCIA, Trustee

By: /s/ Rafael Villoldo, Trustee RAFAEL VILLOLDO, Trustee

Table of Contents

THE CAROLINA MARIE GARCIA 2006

FAMILY TRUST u/t/a dated October 30, 2006

By: /s/ Rene A. Garcia, Trustee RENE A. GARCIA, Trustee

By /s/ Rafael Villoldo, Trustee RAFAEL VILLOLDO, Trustee

IRREVOCABLE TRUST FOR VICTOR GARCIA u/t/a dated October 30, 2006

By: /s/ Rene A. Garcia, Trustee RENE A. GARCIA, Trustee

By: /s/ Rafael Villoldo, Trustee
RAFAEL VILLOLDO, Trustee

Table of Contents

EXHIBIT C

Affiliate Debt

Total Affiliate Debt - Amount: Approximately \$87.3 million after giving effect to the transactions described in this Exhibit C

(a) Subordinated Debt Payable by Model to Certain Model Shareholders

Amount: \$52.3 million

Maturity date: At least 6 months beyond the scheduled maturity date of the senior creditor financing described in the Commitment Letter

Amortization: Bullet at maturity

Interest rate: 2% per annum over the rate applicable from time to time to the revolving credit facility described in the Commitment Letter payable quarterly

Security: Unsecured

Priority: Subordinated to institutional creditor financings, including the financing pursuant to the Commitment Letter

Use of Proceeds: (i) Working capital and (ii) payment of all but \$35 million of the subordinated debt owed by Model to Quality King Distributors, Inc.

(b) Restructured Subordinated Debt Payable to Quality King Distributors, Inc. after the Payment described in Clause (a) of this Exhibit C

Amount: \$35.0 million.

Maturity Date: June 30, 2012

Amortization: Quarterly amortization of \$2.5 million, commencing January 2009, and a final balloon payment 6 months following the scheduled maturity of the senior creditor financing described in the Commitment Letter

Interest rate: 1% per annum over the rate applicable from time to time to the revolving credit facility described in the Commitment Letter

Security: Unsecured

Priority: Subordinated to institutional creditor financings, including the financing pursuant to the Commitment Letter

Annex II Financo Opinion

Letterhead of Financo, Inc.

December 21, 2007

PERSONAL & CONFIDENTIAL

Special Committee of the Board of Directors

E Com Ventures, Inc.

251 International Parkway

Sunrise, FL 33325

Members of the Special Committee:

We understand that E Com Ventures, Inc. (the Company), Model Reorg Acquisition LLC (Acquisition Sub), Model Reorg, Inc. (Model) and the holders of the outstanding capital stock of Model propose to enter into an Agreement and Plan of Merger, substantially in the form of the draft dated December 20, 2007 (the Agreement), which provides, among other things, that (i) Model will merge with and into Acquisition Sub (the Merger) and (ii) all issued and outstanding shares of common stock, no par value, of Model (Model Common Stock), other than shares of Model Common Stock held in the treasury of Model, will be converted into the right to receive 5,900,000 shares (the Shares) of common stock, par value \$0.01 per share, of the Company (Company Common Stock), and warrants (the Warrants, and together with the Shares, the Merger Consideration) to purchase 1,500,000 shares of Company Common Stock.

The Special Committee of the Board of Directors of the Company (the Special Committee) has requested our opinion as to the fairness of the Merger Consideration, from a financial point of view, to holders of Company Common Stock (other than those who own, or whose affiliates own, securities of Model).

Financo, Inc. provides merger and acquisition advisory services to public and private companies. In this capacity, we are continually engaged in performing financial analyses with respect to businesses and their securities in connection with mergers and acquisitions as well as for other transactions and corporate purposes.

In connection with rendering our opinion, we have reviewed and considered such financial and other matters as we have deemed relevant, including, among other things:

- a. drafts of the Agreement, the Warrants and a registration rights agreement, each dated December 20, 2007 which, for purposes of this opinion, we have assumed to be in all material respects identical to the forms of such agreements to be executed;
- b. certain publicly available financial and other information for the Company, including its Annual Report on Form 10-K for the fiscal year ended January 31, 2007 and Quarterly Reports on Form 10-Q for the quarters ended April 30 and July 31, 2007, and certain other relevant financial and operating data furnished to us by management of the Company;
- c. historical financial information of Model, prepared and furnished to Financo by management of Model;
- d. certain internal financial analyses, financial forecasts, reports and other information concerning the Company, prepared and furnished to us by management of the Company and management and representatives of Model;

- e. certain internal financial analyses, financial forecasts and other information concerning Model, furnished to us by management and representatives of Model;
- f. discussions we have had with certain members of Company management concerning the historical and current business operations and strategy, financial condition and prospects of the Company and such other matters as we deemed relevant;

Table of Contents

- discussions we have had with certain members of Model management concerning the historical and current business operations and strategy, financial condition and prospects of Model and such other matters as we deemed relevant;
- h. certain operating results, the reported prices and historical trading activity of Company Common Stock, and the operating results, reported prices and historical trading activity of the securities of certain publicly traded companies we deemed relevant;
- i. certain financial terms of the Merger as compared to the financial terms of certain selected business combinations we deemed relevant; and
- such other information, financial studies, analyses and investigations and such other factors that we deemed relevant for the purposes of this opinion.

In conducting our review and arriving at our opinion, we have, with your consent, assumed and relied, without independent investigation, upon the accuracy and completeness of the information publicly available about the Company and Model, and the financial and other information provided to us or otherwise made available to, discussed with, or reviewed by or for us. We have not assumed any responsibility for the accuracy or completeness, or independently verified, any such information and we have relied on the assurance provided by Company management that they are unaware of any facts that would make such information incomplete or misleading. With respect to the transaction process conducted on behalf of the Company, we have assumed and relied upon, without assuming any responsibility for independent verification of, the accuracy and completeness of the information supplied, summarized or otherwise made available to, discussed with, or reviewed by or for us, including as to the completeness of the process. Our analyses were based, among other things, on the financial projections of the Company (the Company Financial Projections) prepared by management of the Company and the financial projections of Model prepared by management of Model (the Model Financial Projections and collectively with the Company Financial Projections, which were furnished to us, discussed with us or reviewed for us by management of the Company or Model, as the case may be, we have assumed that they have been reasonably prepared on bases reflecting the best currently available estimates and good faith judgments of management of the Company or Model, as the case may be, as to the future competitive, operating and regulatory environments and related financial performance of the Company or Model, as the case may be, and such projections and the assumptions derived therefrom provide a reasonable basis for our opinion. We express no view as to such Financial Projections, or the assumptions on which they are based.

For purposes of rendering our opinion we have assumed in all respects material to our analysis that the representations and warranties of each party contained in the Agreement are true and correct, that each party will perform all of the covenants and agreements required to be performed by it under the Agreement and that all conditions to the consummation of the Merger will be satisfied without waiver or modification thereof. We have also assumed that in the course of obtaining the necessary regulatory and third party approvals, consents and releases for the Merger, no modification, delay, restriction or condition will be imposed that will have a material adverse effect on the Merger and that the Merger will be consummated in accordance with applicable laws and regulations and the terms of the Agreement, without delay, waiver, amendment or modification of any material term, condition or agreement. We have relied without independent verification upon the views of management of the Company concerning the business, operational and strategic benefits and implications of the Merger. We do not express any view on, and this opinion does not address, the fairness of the amount or nature of any compensation to be paid or payable to any of the officers, directors or employees of the Company or Model, or any class of such persons, whether relative to the Merger Consideration or otherwise.

Our analyses must be considered as a whole. Considering any portion of such analyses or the factors considered, without considering all analyses and factors, could create a misleading or incomplete view of the process underlying the conclusions expressed herein.

2

Table of Contents

We have not been asked to pass upon, and express no opinion with respect to, any matters, including any agreements between the Company and Model or any of their respective affiliates, other than the fairness from a financial point of view of the Merger Consideration to holders of Company Common Stock (other than those who own, or whose affiliates own, securities of Model). In connection with our engagement, we have not evaluated nor did the Special Committee or the Company request us to evaluate, alternative transaction structures or financial alternatives other than the Merger and we did not otherwise participate in the transaction process. Our opinion does not address the Company s underlying business decision to effect the Merger.

We have not made or obtained any independent evaluations, valuations or appraisals of the assets or liabilities of the Company or Model nor have we been furnished with such materials. In addition, we have not conducted, nor have we assumed any obligation to conduct, any physical inspection of the properties or facilities of the Company or Model. We have assumed with your consent that there are no legal issues with regard to the Company or Model that would affect our opinion, and we have relied on this assumption without undertaking any independent investigation or inquiry. Our opinion is necessarily based upon economic and market conditions and other circumstances as they exist and can be evaluated by us on the date hereof. It should be understood that although subsequent developments may affect our opinion, we do not have any obligation to update, revise or reaffirm our opinion and we expressly disclaim any responsibility to do so. Our opinion does not address the relative merits of the Merger as compared to other business strategies that might be available to the Company, nor does it address the underlying business decision of the Company to proceed with the Merger. We express no view as to the federal, state or local tax consequences of the Merger. This opinion has been approved by a Fairness Committee of Financo.

It is understood that this letter is intended for the benefit and use of the Special Committee in its consideration of the Merger and may not be used for any other purpose or reproduced, disseminated, quoted, communicated (in whole or in part) or referred to at any time, in any manner or for any purpose without our prior written consent, except that this opinion may be included in its entirety, if required, in any filing made by the Company with respect to the Merger with the Securities and Exchange Commission, provided that this opinion is reproduced in such filing in full and any description of or reference to us or summary of this opinion and the related analyses in such filing is in a form acceptable to us and our counsel in our sole discretion. This letter does not constitute a recommendation to any holder of Company Common Stock, or any other person, as to how such person should act with respect to the Merger.

We have been retained to represent the Special Committee of the Board of Directors of the Company to give an opinion as to the fairness of the Merger Consideration, from a financial point of view, to the holders of Company Common Stock (other than those who own, or whose affiliates own, securities of Model). We have received an initial fee in connection with our services and an additional fee upon our advising the Special Committee that we were prepared to deliver this opinion. No portion of our fee is contingent upon the consummation of the Merger.

Based upon and subject to the foregoing, including the various assumptions and limitations set forth herein, it is our opinion that, as of the date hereof, the Merger Consideration is fair, from a financial point of view, to holders of Company Common Stock (other than those who own, or whose affiliates own, securities of Model).

whose armates own, securities of wioder).				
Very truly yours,				
/s/ Financo, Inc.				
FINANCO, INC.				

3

Annex III Certificate of Amendment to Articles of Incorporation to

Increase Number of Authorized Shares

ARTICLES OF AMENDMENT TO THE

AMENDED AND RESTATED ARTICLES OF INCORPORATION

OF

E COM VENTURES, INC.

a Florida corporation

Pursuant to the provisions of Section 607.1006 of the Florida Business Corporation Act, the Amended and Restated Articles of Incorporation of E Com Ventures, Inc. (the Corporation) are hereby amended as follows:

1. Article III of the Amended and Restated Articles of Incorporation is hereby amended to provide the following:

The aggregate number of shares of all classes of capital stock that the Corporation shall have authority to issue is twenty-one million (21,000,000) shares, consisting of (i) twenty million (20,000,000) shares of common stock, par value \$0.01 per share (the Common Stock), and (ii) one million (1,000,000) shares of preferred stock, par value \$0.01 per share (the Preferred Stock).

Except as hereby amended, the Amended and Restated Articles of Incorporation of the Corporation shall remain the same.

2. The foregoing amendment to the Amended and Restated Articles of Incorporation of the Corporation was recommended by the Board of Directors of the Corporation and submitted to the shareholders of the Corporation for approval at the special meeting of the shareholders of the Corporation, held on , 2008. The amendment to the Amended and Restated Articles of Incorporation of the Corporation was approved by the shareholders of the Corporation, with the number of votes cast for the amendment being sufficient for approval in accordance with the applicable provisions of the Florida Business Corporation Act.

IN WITNESS WHEREOF, the undersigned duly authorized officer of the Corporation has executed these Articles of Amendment to the Amended and Restated Articles of Incorporation as of this day of , 2008.

[Name] [Title]

Annex IV Certificate of Amendment to Articles of Incorporation to Effect Name Change

ARTICLES OF AMENDMENT TO THE

AMENDED AND RESTATED ARTICLES OF INCORPORATION

OF

E COM VENTURES, INC.

a Florida corporation

Pursuant to the provisions of Section 607.1006 of the Florida Business Corporation Act, the Amended and Restated Articles of Incorporation of E Com Ventures, Inc. (the Corporation) are hereby amended as follows:

1. Article I of the Amended and Restated Articles of Incorporation of the Corporation is hereby deleted in its entirety and replaced with the following:

The name of the Corporation is Perfumania Holdings, Inc.

2. The foregoing amendment to the Amended and Restated Articles of Incorporation of the Corporation was recommended by the Board of Directors of the Corporation and submitted to the shareholders of the Corporation for approval at the special meeting of the shareholders of the Corporation, held on , 2008. The amendment to the Amended and Restated Articles of Incorporation of the Corporation was approved by the shareholders of the Corporation, with the number of votes cast for the amendment being sufficient for approval in accordance with the applicable provisions of the Florida Business Corporation Act.

IN WITNESS WHEREOF, the undersigned duly authorized officer of the Corporation has executed these Articles of Amendment to the Amended and Restated Articles of Incorporation as of this day of , 2008.

[Name] [Title]

FORM OF E COM VENTURES PROXY CARD

qFOLD AND DETACH HERE AND READ THE REVERSE SIDEQ

E COM VENTURES, INC.

251 International Parkway

Sunrise, Florida 33325

PROXY FOR SPECIAL MEETING OF SHAREHOLDERS

THIS PROXY IS SOLICITED ON BEHALF OF THE COMPANY S BOARD OF DIRECTORS

The undersigned holder of common stock of E Com Ventures, Inc., a Florida corporation (the Company), hereby appoints Michael Katz, Paul Garfinkle, Donovan Chin and Joseph Bouhadana, or any of them acting singly, as proxies for the undersigned, each with full power of substitution, for and in the name of the undersigned to act for the undersigned and to vote, as designated on the reverse side of this proxy card, all of the shares of stock of the Company held of record by the undersigned at the close of business on June 24, 2008 at the Company s Special Meeting of Shareholders to be held on August 8, 2008, at 8:30 a.m. at the offices of Edwards Angell Palmer & Dodge LLP, 750 Lexington Avenue, New York, NY 10022, and at any postponements or adjournments thereof.

(PLEASE SIGN ON REVERSE SIDE)

$\ensuremath{\mathsf{qFOLD}}$ AND DETACH HERE AND READ THE REVERSE SIDE $\ensuremath{\mathsf{q}}$

THE BOARD OF DIRECTORS RECOMMENDS A VOTE FOR PROPOSALS 1-4

 To approve the issuance of shares of common stock and Warrants under the Agreement and Plan of Merger dated as of December 2007, by and among E Com Ventures, Inc., Model Reorg, Inc., the shareholders of Model Reorg, and Model Reorg Acquisition amended, and the issuance of common stock upon exercise of the Warrants. 						
		FOR "	AGAINST "	ABSTAIN "		
2.	To amend E Com Ventures Articles of Incorporation to increase the number of authorized shares of common stock from 6,250,000 sto 20,000,000 shares.					
		FOR "	AGAINST "	ABSTAIN "		
3. To amend E Com Ventures Articles of Incorporation to change the company s name to Perfumania Holdings, Inc.						
		FOR "	AGAINST "	ABSTAIN "		
4.	To postpone or adjourn, if ne	litional proxies in favor of Proposa	ls 1, 2 or 3.			
		FOR "	AGAINST "	ABSTAIN "		
5.	In their discretion, upon such other business as may properly come before the Special Meeting or any postponement or adjournment thereof. THIS PROXY, WHEN PROPERLY EXECUTED, WILL BE VOTED IN THE MANNER DIRECTED					
	HEREIN BY THE UNDERSIGNED SHAREHOLDER. IF NO DIRECTION IS MADE, THIS PROXY					
		WILL I	BE VOTED FOR EACH PROI	POSAL.		
	PLEASE MA	ARK, SIGN AND DAT	E THIS PROXY CARD AND PR	COMPTLY RETURN IT IN THE		
			ENVELOPE PROVIDED.			
DA'	TE					
SIG	GNATURE					
	GNATURE (If held jointly)					
	- (· j · j /					

Note: Please sign exactly as your name appears hereon and mail it promptly even though you may plan to attend the Special Meeting. When shares are held by joint tenants, both should sign. When signing as attorney, executor, administrator, trustee or guardian, please give full title. If a corporation, please sign in full corporate name by president or other authorized officer. If partnership, please sign in the partnership name by authorized person.